

MEASURING INFRASTRUCTURE PROJECTS SUSTAINABLE DEVELOPMENT GOALS IMPACT (MISI)

Paul Robert Mansell MSc, MPhil, MA

https://orcid.org/0000-0003-2236-5070

A thesis submitted in partial fulfilment of the requirements of London South Bank University for the degree of Doctor of Philosophy

This research was carried out in collaboration with a number of partners including the Institution of Civil Engineers, University College London, the Buildings Research Establishment, the Cabinet Office's Infrastructure and Projects Authority, the Environment Agency, Thames Tideway Tunnel Ltd and the UN Global Compact.

May 2021

Acknowledgements

There are so many people who deserve my thanks without whom this thesis would not have been written. I have no doubt that the moment I submit I will realise I have missed so many of you from this list.

First of all, I must acknowledge my constantly supportive wife, Jo Mansell. I could not have done this without you. I would not have even thought this possible without you. You are my rock.

I owe a huge amount to my supervisors, especially my First Supervisor, Professor Simon Philbin (Director, Nathu Puri Institute for Engineering and Enterprise, London South Bank University), who provided the excellence of doctoral support that the study into PhD Supervision by Lee (2008) characterised as the 'right blend of functional, enculturation, critical thinking, emancipation, and the development of a quality relationship' where supervisors use emotional intelligence to enthuse and care for their students. He could not have performed these support roles more effectively and I doubt I would have completed the research without his levelheaded and pragmatic approach to the balance of studies and ongoing government support as well as my teaching commitments at UCL. I especially welcomed the freedom to develop my own design for the MISI research project, harnessing my existing relationships across government and industry. I was also fortunate to have excellent support from my Second Supervisor, Professor Rao Bhamidimarri (President, Institute of Advanced Research, Gujarat, India), who was the reason why I joined the NPI team; and, my Third Supervisor, Dr Efrosyni Konstantinou (Associate Professor, Bartlett School of Construction and Project Management, University College London). You have each looked out for me in different ways and I have benefited from you as a team. I am particularly indebted to Ralf Müller, Professor of Project Management at BI Norwegian Business School, who provided a number of support interventions, including an academic writing course, pro bono.

I am also indebted to the research partners listed on the front page of this document, with special thanks to Professor Tim Broyd (UCL, ICE and BRE), Ian Nicholson (BRE), Deney Van Rooyen (Cape Town University) and Dr Luca Sabini (Hertfordshire University), Nick Baveystock (Director General Institution of Civil

Engineers); Nathan Baker (Director Knowledge, formerly ICE); Elodie Huiban (Knowledge Strategy Manager, formerly of ICE); David Hancock and Alison Baptiste (Directors, Infrastructure and Projects Authority, UK Cabinet Office and HM Treasury); Fergus Harradence (Deputy Director, UK Department of Business Enterprise, Innovation and Skills); John Russon and Kat Ibbotson from the Environment Agency; Steve Kenzie (CEO, UN Global Compact UK); Ines Faden (Group Treasurer, Thames Tideway Tunnel); without whom this research would not have been possible. To the many company leaders and engineers who took time to complete surveys and participate in interviews and workshops – thank you for your patience and support.

I would also like to thank: Professor Andrew Davies as my Viva external examiner who, as a global leader in the field of systems integration and complex project management delivery, provided challenging and insightful commentary at my Viva with Dr Sara Hasani Darabadi as the internal examiner; Dr Juliano Denicol, Associate Professor UCL and Dr Christine Unterhitzenberger, Associate Professor in Project Management at University of Leeds, who acted as the examiners for the mock Viva, that greatly contributed to the success of the Viva; Professor Phillip Burrell and Dr Deborah Andrews for their commitment to my development demonstrated in RES panels; fellow Nathu Puri team members, including Sunita Selvarajan, Thomas Empson and Pavan Kumar Sala; Dr James Smith-Spark who supported my SPSS queries; and finally, fellow co-authors who supported the formal publication of 7 papers in academic journals and 5 conference papers. On all these 12 papers, I can confirm that 95% of the design and analysis was my work alone – a full list of them is at Appendix 24.

Abstract

The research captured in this thesis has led to the development of a range of models, tools and processes for government and industry that provide a forward-looking approach to the measurement of impact on infrastructure projects. This approach enables measurement of United Nations' Sustainable Development Goals (UN SDG) at the project level to ensure investments are made equitably across economic, environment and social objectives. Application of the results from this research are already being actively used by the Environment Agency to manage impact assessment across its £5.2Bn portfolio of projects and by the Thames Tideway Project (£4.9Bn).

Background. Achievement of the United Nations' Sustainable Development Goals (SDGs) by the year 2030 is of paramount importance and the construction industry has a major role in achieving a measurable impact against the SDG targets. However, linking of 'local' infrastructure project success to 'global' SDG targets is problematic because the targets were designed at the national level and not at the project or programme level (Mansell, et al., 2020a). Furthermore, while the so called 'triple bottom line' (i.e. economy, environment and society) approach to understanding sustainability remains important, there is a need to understand how this can be related to the full project lifecycle as well as a need for improved project governance. This is consistent with the findings of a key UN investigation's Fourth Report (Global Task Force, 2020) which calls for localization of SDGs as well as the need for cooperative governance to establish shared priorities.

Research description. The research was based on two main stages. The first stage, informed by a systematic literature review, comprised a mixed method that involved a survey of 325 engineers to derive quantitative data (Mansell et al., 2020b) along with interviews with 40 CEOs and corporate Heads of Sustainability to capture qualitative data (Mansell et al., 2020c). The second stage involved the development of a prototype that was tested through two further exploratory investigations at two levels: (1) Test 1: is there a Golden Thread from global SDGs, through the organisational level, down to project level SDG impact measurement?; (2) Test 2: does the prototype model, the *Impact Value Chain*, have practical coherence when

assessed in a brief case study of a Water Utility Company (Anglian Water) (Mansell et al., 2020d). Subsequently, and not part of this thesis, the research led to a collaborative partnership to test the prototype model and its approach across the Environment Agency's full portfolio of projects and also, the megaproject of the Thames Tideway Tunnel.

Findings. The survey of 325 engineers (Mansell et al., 2020b) indicated four primary shortfalls for measuring SDGs on infrastructure projects, namely leadership, tools and methods, engineers' business skills in measuring SDG impact, and how project success is too narrowly defined as outputs (such as time, cost and scope) and not outcomes (longer-term local impacts and stakeholder value). Moreover, the interviews with 40 senior executives (Mansell et al., 2020c) from the UK identified that SDG measurement practices are currently 'more talk less walk' and indicated a number of contextual and mechanistic opportunities to increase the outcome success.

Therefore, using empirical evidence the researcher identified a 'golden thread' between best practice sustainability-reporting frameworks at the 'local' project level and those at the organisational and supra-national-levels (Mansell et al., 2020a). In doing so, the research identified that there is sufficient linkage to embed SDG impact targets into the design stage of an infrastructure project. Furthermore, the innovative process model, called the 'Infrastructure SDG Impact-Value Chain' (IVC) to link project delivery with strategic SDG impacts, builds on the concept of creating shared value and creates a practical mechanism to turn theory into meaningful impact in project selection and delivery. The utility of the IVC process model was initially investigated as part of the case study investigation of Anglian Water (Mansell et al., 2020d) and its application has been further demonstrated in the MISI Project (not included in this thesis).

Research Impact. The research produced twelve peer-reviewed papers including being published in seven internationally recognised academic journals, such as: *Sustainability* (2 articles), *Administrative Sciences*, and the *Proceedings of the Institution of Civil Engineers–Engineering Sustainability*. The MISI research outputs have been taken forward by the government and industry partners, specifically the Environment Agency and Thames Tideway Project, working together to establish this new approach for measuring sustainability on infrastructure projects.

Table of Contents

Ackn	owledgements	ii
Abstr	act	.iv
List o	f Figures	x
List o	f Tables	xiv
List o	f Abbreviations	xvi
Key I	Definitions	xvi
Stater	nent of Original Authorship	xix
Chap	oter 1: Introduction	21
1.1	Overview of introduction chapter	22
1.2	Background	22
1.3	Context – Problem Statement	23
1.4	Personal perspective – Adopting the Kolb Learning Cycle	24
1.5	Research Gap, Aim, Questions and Objectives	28
1.6	Significance and Scope	30
1.7	Thesis Outline	31
1.8	Summary and Link to the Next Chapter	34
Chap	oter 2: Literature Review	37
2.1	Outline of this chapter	37
2.2	Literature Review Introduction	37
2.3	Grand Challenges and SDG	38
2.4	Sustainability and Sustainable Development in relation to SDG	41
2.5	Sustainable Infrastructure Projects (IP) and Project Success in relation to SDG	43
2.6	Methodology for Literature Review	45
2.7	Results and discussion of the SLR	49
2.8	Level 2 Analysis of the 13 top MISI articles based on their keyword occurrences	55
2.9	SLR Findings for Design of Research Framework	56
2.10	(1) How extensive (broad and deep) has the research into MISI been to-date?	57
2.11 Prope	(2) What issues and sub-issues were identified that might inform the MISI study's sitions?	57
2.12	(3) What are the emerging MISI Propositions?	58
2.13	SLR Summary and Implications	61
Chap	oter 3: Theoretical Framework	67
3.1	Outline of this chapter	67
3.2	Theoretical lens	67
3.3	Layered Model of Organisational Project Management (OPM)	73

3.4 the Tr	Project Success theoretical framework: Core theories of the Theory of Change and riple Bottom Line	
3.5	Theory of change	82
3.6	Triple bottom line	85
3.7	Creating shared value	87
3.8	Synthesising the initial theoretical model (Impact Value Chain) from literature	89
3.9	Summary and Implications	96
Chap	pter 4: Research Design and Methodology	. 99
4.1	Outline of chapter	99
4.2	Methodology	99
4.3	Research Design	101
4.4	Qualitative or Quantitative, or Both?	107
4.5	Ethics and Limitations	112
4.6	Chapter Conclusion	112
Chap	pter 5: Survey Method, Results and Analysis	113
5.1	Outline of chapter	113
5.2	Methodology and Research Design	113
5.3	The survey	115
5.4	Data analysis and results	118
5.5	Survey results	119
5.6	Inferential statistics	126
5.7	Discussion of survey results and development of theoretical framework	126
5.8 Infras	Development of Conceptual Framework for Measuring the SDG Performance of structure Projects	131
5.9	Limitations of Survey	133
5.10	Conclusions and Input to Next Stage of Study	135
Chap	pter 6: Interview Method, Results and Analysis	137
6.1	Outline of chapter	137
6.2	Methodology and Research Design	137
6.3	Data Analysis	141
6.4	Interview Analysis Process	143
6.5	Verification	145
6.6	Results	145
6.7	Discussion	174
6.8	Chapter Conclusions and development of prototype	179
Chap	pter 7: Triangulation of Results and development of Prototype	183
7.1	Outline of chapter	183
7.2	Triangulation of Results	185

7.3	Iteration of the IVC by including a Methodology for Adoption1	92
7.4	Chapter Conclusions	96
Chap	oter 8: Test 1: Is there a 'Golden Thread' from global SDG, through	~-
the O	Organisational Layer (portfolio), to local project level?	97
8.1	Outline of chapter1	97
8.2	Stage Two – Detailed Investigation1	97
8.3	Methodology for Golden Thread Test approach	98
8.4	Results and Discussion	06
8.5	Conclusions	17
Chap	oter 9: Test 2: Case Study of IVC in Water Utility context	19
9.1	Outline of chapter	19
9.2	Methodology for Case Study approach	19
9.3	Results and Findings	21
9.4	Overview Analysis of Anglian Water's Projects Set against the IVC Framework 2	28
9.5	Conclusions From the Test 2 Case Study	33
Chap	oter 10: Conclusions	37
10.1	Outline of chapter2	37
10.2	Summary of the Research Study Results	37
10.3	Limitations	53
10.4	Recommendations	54
10.5	MISI Impact and Implications for Future Use2	59
10.6	Personal Reflections	63
Refe	rences2	65
Appe	endices	83
Appe	endix 1 - Table of SLR data across the MISI Research Thematics	83
Appe	endix 2 – SLR Analysis of Data with Derived Emerging Propositions 2	89
Арре	endix 3 - Survey questions and selection of the type of question and metric	•
to ali	gn with analysis requirements for measuring engineers' views on projects	s'
SDG	impact	94
Appe you b	endix 4 - Data capture from survey for question "Select the six SDGs that believe that engineers have the greatest impact and opportunity"	95
Appe corre	endix 5 - Data capture from the survey's Chi-Square Tests (with continuit ection, likelihood ratio, and linear-by-linear association)	y 96
Appe valid Instit	endix 6 - Blog of ICE workshop (between survey and interview stages to ate results) led by Paul Mansell with practitioners & academics at the tution of Civil Engineers – The 'Control Group'2	97
Appe	endix 7 – List of Interviewees and company background data	00
Appe State	endix 8 – Nodal Incidence of participants Per Node and Frequency of ments Per Node	01

Appendix 9 - NVivo Codebook of Node Descriptions and Occurrences of Codified Data Collection
Appendix 10 - Data analysis using NVivo: nodal-word linkages
Appendix 11 – participants' Data Charts of Frequency of Nodes and References 306
Appendix 12 - Nodal Touch Point Charts of participants (6 examples)
Appendix 13 - Interview semi-structured approach – Outline of Questions 308
Appendix 14 - Participant Information Form
Appendix 15 – Full Matrix of CEEQUAL Mapping to SDGs
Appendix 16 – Anglian Water's mapping of SDG targets to the 27 Projects 313
Appendix 17 – MISI Impact Statement from Government & Industry Partners 316
Appendix 18 – MISI Impact Statement of this research from the Environment Agency
Appendix 19 – MISI Impact Statement of this research from the Thames tideway project
Appendix 20 – MISI Impact Statement of this research from the UN Global Compact Network UK
Appendix 21 – MISI Impact Statement of this research from the Infrastructure and Projects Authority, UK Government
Appendix 22 – MISI Impact Statement of this research from the Institution of Civil Engineers
Appendix 23 – MISI Impact Statement of this research from the Building Research Establishment (BRE)
Appendix 24 – Publications in Advance of Thesis

List of Figures

Figure 1: Thesis Chapter Map and Workflow
Figure 2. Logic Map of Thesis Findings (red line = Proposition & Variables Framework flow)
Figure 3: The Global Goals for Sustainable Development (United Nations 2015—permission to use from Sustainable Development Goal (SDG) logo Guidelines)
Figure 4: The sustainable development goals (SDG) Targets and Indicators' framework designed by the UN IAEG-SDGs (2017)40
Figure 5: Systems map showing the key thematic areas related to the research question
Figure 6: Adaption of the PRISMA Flow Diagram' (Moher et al., 2009) to explain the de-selection process45
Figure 7: The 'hopper' approach to SLR Level 1 and 2 Analysis
Figure 8: Occurrences of articles appearing in selected journals using the keywords
Figure 9: Occurrence of Keywords in abstracts over time
Figure 10: Type of contributions, using a hierarchical tree map chart, across subset
Figure 11: Research design approaches
Figure 12: Sectoral focus of subset articles
Figure 13: Institutional focus of subset articles
Figure 14: SDG mentions across subset55
Figure 15: Research issues and sub-issues based on the 13 most relevant articles
Figure 16: Structure of issues, sub-issues and emerging supra-themes58
Figure 17. The Critical Realism theory of real-actual-empirical adapted from Zachariadis et al. (2013) linking to the Phenomenon of Interest of the study
Figure 18. The Realist Evaluation (Context–Mechanism–Outcome (C–M–O)) model mapped to the Critical Realism theory of real-actual-empirical adapted from Zachariadis et al. (2013)71
Figure 19: Adaption of the OPM onion model (Müller, Drouin, Sankaran, 2019) showing: three SDG measurement domains: portfolio prioritisation (yellow), local implementation (blue) and project measurement governance (green), mapped to Muller's seven levels. 74
Figure 20: Framework for sustainability and project success reporting depicting the two core sustainable development questions at project level

Figure 21: SDG Hierarchy of SDG target reporting using Impact value Chain (IVC) outcomes and impact causal chain77
Figure 22: Adaption of the 'Ends-Ways-Means' model (Lykke, 1982; Sen, 2013)
Figure 23: The TBL view of economy, environment and social translated into the doughnut view alignment of the United Nations' 17 Sustainable Development Goals by Johan Rockström (2016) (graphic usage agreed)
Figure 24: Concept 1 - The Theory of Change's fundamentals for delivering impact (the number '1' in the diagram refers to the Concept number, this is sequentially repeated in the three following Figures)90
Figure 25: Concept 2 - Delivery of projects within an organisational structure; both with sustainability reporting loops embedded92
Figure 26: Global Sustainability Reporting. Source data from GRI (2016)93
Figure 27: Concept 3 – Infrastructure investments as a 'system of systems' – Source: The Future of National Infrastructure: A System of Systems Approach (Hall et al, 2016; UNOPS, 2017)93
Figure 28: Infrastructure System of systems Map showing 'Ends, Ways, Means'. Adapted from papers by UNOPS (2017) and ITRC's (Hall et al, 2016)95
Figure 29: Concept 4 – Delivering impact measured against the TBL/SDGs, and completing the learning-loop for smarter investments, adapted from ICAS/IIRC's 'The Sustainable Development Goals, integrated thinking and the integrated report' (Adams, 2017)
Figure 30. The mixed-method research design adapted from Creswell (2003 and 2017)100
Figure 31. The Research 'Onion Rings' (Saunders et al., 2009)102
Figure 32. The two-step Sequential Explanatory Design adapted from Creswell (2003 and 2017)109
Figure 33: The Context-Mechanism-Outcome Variables Framework's causal pathway110
Figure 34: Simplified conceptual view of analytical model for testing the propositions across three dimensions of: IVC; C-M-O Framework; and Portfolio-Project Levels
Figure 35: Exploratory Investigation (EI) Flow Chart
Figure 36: Detailed Investigation (DI) Flow Chart
Figure 37: The research approach of mixed method Sequential Explanatory Design, adapted from Creswell (2003 and 2017)114
Figure 38: Survey Questions Sequence; adopting Realist Evaluation methodology of Context-Mechanism-Outcomes (Pawson and Tilley, 2004)116
Figure 39: Statistical Descriptive Data from Survey118

Figure 40: Response to Question 1: Should engineering businesses seek ways to measure and report SDG impact?
Figure 41: Responses of participants to their five top SDGs that engineering projects should measure impact against – showing the top seven list (with six and seven being significantly less popular)120
Figure 42: Preference of the SDG 13 and 15; differentiating between millennials
Figure 43. Response to Question 3: Do commercial realities dictate the SDGs you pick?
Figure 44. Response to Question 4: Do you want to know more about measuring SDG impact on your projects?
Figure 45. Response to Question 5: Do we have a 'fit for purpose' measuring approach to track projects' SDG impact?
Figure 46. Responses to Question 6: What are the greatest challenges for measuring SDG impact?
Figure 47. Responses to Question 7: What are the greatest opportunities for measuring SDG impact?
Figure 48. Responses to Question 6 (opportunities) and 7 (challenges) for measuring SDG impact? (See Appendix 5 for full data)126
Figure 49. View 1 – Maturity of C-M-O evidence across variables and study areas
Figure 50: The research design of mixed method Sequential Explanatory Design, adapted from Creswell (2003 and 2017)138
Figure 51. The nodal evaluation framework for sequential explanatory design from which the semi-structured interview questions were derived139
Figure 52. The attributes and values of the forty interviewees142
Figure 53. The twin-track analysis protocols approach: qualitative and quantitative
Figure 54. The nodal framework used for identification of key words aligned to context–mechanism–outcome (C–M–O) (Creswell, 2017)144
Figure 55. Results of the self-assessed level of awareness-application and process maturity (colour representation shown in columns c and d in titles row)
Figure 56. Graphical representation of results from self-assessed level of "awareness-application"
Figure 57. Graphical representation of results of the self-assessed level of "SDG measurement maturity" (colour representation shown in Figure 8 in column h in the titles row)153
Figure 58. View 2 – Maturity of C-M-O evidence across variables post- interviews

Figure 59: The Impact Value Chain (IVC) theoretical model, adapted from ICAS/IIRC's 'The Sustainable Development Goals, integrated thinking and the integrated report' (Adams 2017) 184
Figure 60: The Triangulation process to build the IVC Prototype 185
Figure 61: The proposed 5-step SDGiPro methodology 195
Figure 62: The SDGiPro iterative workshops
Figure 63: Detailed Investigation (DI) Flow Chart 197
Figure 64: The IVC Testing Research Design 199
Figure 65: Concentual man of notential 'golden thread' from local to global
levels
Figure 66: Tree map linking IVC four concepts to the Key Indicator Words via nodes and sub-nodes
Figure 67: High-level Analytical Matrix Mapping showing linkages between CEEQUAL Categories (y-axis) with GRI Materiality Topics (x-axis).
Figure 68: Bar Chart showing the instances of 'hot spots' where alignment is identified208
Figure 69: Data captured from NVivo analytical tool showing strength of connections across the 4 concepts in IVC from project level to organisational level (full data in Annex). For description of the columns, see Table 28 title
Figure 70: CEEQUAL's relative focus on TBL and across SDGs (full matrix analysis shown in Appendices)
Figure 71: Systems mapping of connections between CEEQUAL and SDGs
Figure 72: Anglian Water alignment of purpose-outcomes and SDGs (Anglian Water, 2018a). [Permission to re-use all graphics agreed by AW]221
Figure 73: Anglian Water's Purpose Wheel (Anglian Water, 2018a), aligned to the triple bottom line
Figure 74: Anglian Water has three business priorities that are balanced across the triple bottom line (for illustration only)225
Figure 75: Anglian Water has three business priorities that are balanced across the triple bottom line
Figure 76. Logic Map for Thesis Findings238
Figure 77. View 3 – Maturity of C-M-O evidence across variables post-Tests 1 & 2
Figure 78. Lifecycle view of MISI Recommendations255

List of Tables

Table 1: Items containing keywords, in the database per Journal47
Table 2: Items containing combinations of keywords 47
Table 3: Data analysis of key words tabulated across the two dimensions of SDG and IP.
Table 4: The resultant reduction of dataset based on combination of keywords
Table 5: Derived propositions from the literature review
Table 6: Examples of the use of Critical Realism and Realist Evaluation C- M-O configuration on engineering projects that informs similar use for MISI research
Table 7: Summary of contextual variables across organisational boundariesusing the Muller Onion Model (Müller, et al., 2019).75
Table 8: Development of stability of Propositions' Results (changes since previous Chap in red)
Table 9: Development of methodological design based on Saunders' Research Onion Ring (2009)
Table 10: Critical Realist Research Designs (amended from Vincent and O'Mahoney, 2018; Ackroyd and Karlsson 2014)105
Table 11: Survey analysis method (based on: Creswell and Creswell, 2017; Nardi, 2015)115
Table 12: Summary of conceptual development of a future prototype modelfor SDG measurement, based on the survey results in this study132
Table 13: Development (2 nd) stability of Proposition Results (changes since previous Chap in red) 136
Table 14: Profiles of participants
Table 15: Text analysis (NVivo) on key words' frequency: context of leadership
Table 16: Text analysis (NVivo) on key words' frequency: context of knowledge
Table 17: Text analysis (NVivo) on key words' frequency: mechanism/context of the Theory of Change. 160
Table 18: Text analysis (NVivo) on key words' frequency: tool, processes and systems
Table 19: Text analysis (NVivo) on key words' frequency: context of change management
Table 20: Text analysis (NVivo) on key words' frequency: performance management

Table 21: Text analysis (NVivo) on key words' frequency: projects-to-portfolios
Table 22: Development (3 rd) of stability of Propositions (changes since previous Chap in red)
Table 23: IVC table illustrating golden thread mapping of the TBL in a 'baseline' model.193
Table 24: Development of stability of Propositions' Results (changes since previous Chap in red) 196
Table 25: Summary of leading infrastructure sustainability reporting tools/methods at organisational and project levels
Table 26: Comparative definition of CEEQUAL and GRI
Table 27: Selected Manuals for analysis: CEEQUAL and GRI206
Table 28: Data Analysis using NVivo: Nodal-Word linkages
Table 29: Key data results from the NVivo text analysis (See Appendix 10 for full data)
Table 30: Development (4 th) of stability of Propositions (changes since previous Chap in red)
Table 31: Anglian Water's performance against outcomes
Table 32: Applying Anglian Water's Wisbech project initiative to the IVCgrid with mapping of the TBL with the five stages of the IVC231
Table 33: Development (5 th) of stability of Propositions' Results (changes from this Chap in red)
Table 34: Research Objectives Summary. 239
Table 35: Summary Results of C-M-O Variables Framework242
Table 36: Summary table of the stability of Propositions
Table 37: Critical Realist Research Designs (amended from Vincent and O'Mahoney, 2018; Ackroyd and Karlsson 2014)
Table 38: Consolidated table of data across the seven MISI Thematics283
Table 39: MISI SLR Research Findings (issues & variables) from the top 13 articles, leading to emerging propositions. 289
Table 40: Anglian Water's mapping of SDG targets to the 27 Projects. The y- axis shows the 35 SDG targets selected at Anglian Water corporate level; the x-axis shows the 27 projects that they are allocated to by Anglian Water (AW).314

List of Abbreviations

BRE = Buildings Research Establishment CR = Critical Realism CSV = Creating Shared Value ICE = Institution of Civil Engineers IP = Infrastructure Projects IPA = Infrastructure and Projects Authority (Cabinet Office and HM Treasury) IPCC = Intergovernmental Panel on Climate Change IVC = Impact Value Chain (MISI model designed as part of this research study) OPM = Organisational Project Management RE = Realist Evaluation SDG = Sustainable Development Goals SDGiPro = SDG Impact on Projects – a process model for prioritising SDG goals and targets (developed by the research scholar) TBL = Triple Bottom Line (Economic, Environmental, Society) ToC = Theory of ChangeUN = United Nations UNOPS = UN Office for Public Services

Key Definitions

(grouped in order of thematics)

Sustainability and the Project Environment

Sustainable Development Goals. '*The Sustainable Development Goals (SDGs), also known as the Global Goals, were adopted by all United Nations Member States in 2015 as a universal call to action to end poverty, protect the planet and ensure that all people enjoy peace and prosperity by 2030' (UNDP, 2019).*

Sustainable Development. Unlike the myriad of definitions for sustainability, the definition of sustainable development is generally agreed from the Brundtland Commission: '*development that meets the needs of the present without compromising the ability of future generations to meet their own needs*' (Brundtland, 1987).

Project. 'A project is a temporary endeavour undertaken to create a unique product, service or result. The temporary nature of projects indicates a definite beginning and end. The end is reached when the project's objectives have been achieved or when the project is terminated because its objectives will not or cannot be met, or when the need for the project no longer exists' (PMI 2008, p.5). Thus it could be adapted to suggest that 'projects are temporary organisations to deliver clearly identifiable outcomes within the limits of time and cost budgets' (Davies, 2017; Atkinson, 1999; Lundin and Soderholm, 1995). Uncertainty is an inherent part of a project, 'When projects are complex, unpredictable, and changing, their plans have to be flexible and able to adjust to situations that cannot foreseen at the outset' (Davies, 2017). 'Within

a project-based organisation, the project is a business mechanism for coordinating and integrating the business function of the firm' (Davies and Hobday, 2005).

Programme. 'A group of related projects managed in a coordinating way to obtain benefits and control not available from managing them individually' (PMI, 2008).

Megaproject. 'Large scale, complex investments that typically cost a billion dollars and up, take many years to develop and build, involve multiple public and private stakeholders, are transformational, and impact millions of people' (Flyvbjerg, 2014; Denicol, Davies & Krystallis, 2020).

Sustainable Infrastructure Systems

Infrastructure. The Global Commission on the Economy and Climate defined Infrastructure as: 'Structures and facilities that underpin power and other energy systems (including upstream infrastructure, such as the fuel production sector), transport, telecommunications, water, and waste management. It includes investments in systems that improve resource efficiency and demand-side management, such as energy and water efficiency measures. Infrastructure includes both traditional types of infrastructure (including energy to public transport, buildings, water supply and sanitation) and, critically, also natural infrastructure (such as forest landscapes, wetlands and watershed protection)' (Bhattacharya, Oppenheim, and Stern, 2015; The New Climate Economy, 2014)

Sustainable Infrastructure. The definition by Ainger and Fenner (2014) was recently developed further by the Inter-American Development Bank (IDB) Group as: *'infrastructure projects that are planned, designed, constructed, operated, and decommissioned in a manner to ensure economic and financial, social, environmental (including climate resilience), and institutional sustainability over the entire life cycle of the project' (IDB, 2018).*

Systems Thinking. 'An approach that focuses on the identification of interrelationships between components (i.e. sub-systems) of a system' (Davies, 2004). **Infrastructure System.** 'A system comprised of assets, institutions and knowledge that provides a society its services. Examples of National Economic Infrastructure systems are: Water, Energy, Transport, Waste, and Telecommunications' (UNOPS, 2017; ITRC's Hall et al, 2016).

Assets. Adapted to: '*The physical components of the system. This also extends to the links that exist between the assets in the system. Note that there are man-made assets but also 'Natural Assets', which provide a service'* (UNOPS, 2017; ITRC's Hall et al, 2016). Examples of man-made assets include roads, bridges, power lines, and pipes. Example of natural assets include wetland systems, and mangrove forests.

Services. '*The functions which the infrastructure system enables. Examples include healthcare services, transport services, and education services*' (UNOPS, 2017; ITRC's Hall et al, 2016).

<u>Logframe</u>

Theory of Change. 'A basis for planning intervention in a given policy or project arena that helps to identify processes and preconditions whereby actions can best attain their intended consequences' (Weiss, 1995).

Value. '*The worth of a good or service as determined by the customer's preferences and the trade-offs they choose to make given their scarce resources, or the value the*

marketplaces on an item' (Porter, 1985). (Note: these are economic-led priorities. Wider societal and environmental considerations are covered in Triple Bottom Line)

Value chain. '*The full range of processes and activities that characterize the lifecycle of a product from production, to manufacturing and processing, to distribution, marketing and retail, and finally to consumption (including waste and disposal across all stages)*' (Porter, 1985; Bloom and Hinrichs, 2011).

Project Logframe. *A project document normally contains a framework with a list of objectives, activities and monitoring indicators, typically called a "logframe" or "results framework". This framework serves as a basis of implementation, monitoring and evaluation* OECD (2019).

Impact. Adapted to: 'A positive or negative contribution to one or more SDG targets across the TBL of environmental, economic, or social thematic areas' (World Bank, 2004). Also, 'What difference does the intervention make? The extent to which the intervention has generated or is expected to generate significant positive or negative, intended or unintended, higher-level effects. Note: Impact addresses the ultimate significance and potentially transformative effects of the intervention' (World Bank, 2004). Also, 'The effects, both positive and negative, which the project is expected to produce upon environment, organization, community, people, etc.' (PMI, 2017) 'Positive and negative, primary and secondary long-term effects produced by a development intervention, directly or indirectly, intended or unintended.' (OECD, 2010)

Project Outcomes. Adapted to: 'A change in the extent or condition of the stocks of capital (natural, produced, social and human) from the use of an output, due to valuechain activities that deliver SDG impacts' (APM, 2012). Also, 'The likely or achieved short-term and medium-term effects of an intervention's outputs.' (OECD, 2010)

Outputs. '*The tangible or intangible product typically delivered by a project*' (APM, 2012). Also, '*The products, capital goods and services which result from a development intervention; may also include changes resulting from the intervention which are relevant to the achievement of outcomes*.' (OECD, 2010)

Activities. 'A task, job, operation or process consuming time and other resources in a project to produce specific outputs' (APM, 2012). Also, 'Actions taken or work performed through which inputs, such as funds, technical assistance and other types of resources are mobilised to produce specific outputs.' (OECD, 2010)

Inputs. Adapted to: '*All those items required to undertake work utilising the stocks of capital including financial, natural, produced, social and human resources'* (APM, 2012). Also, '*The financial, human, and material resources used for the development intervention.*' (OECD, 2010)

The CMO Framework

Context (in C-M-O framework). The conditions in a context of action encompass 'material resources and social structures, including the conventions, rules and systems of meaning in terms of which reasons are formulated' (Sayer 1992, p. 112; in Easton, 2010).

Mechanism (in C-M-O framework). '*The underlying entities, processes, or structures which operate in particular contexts to generate outcomes of interest*' (Astbury and Leeuw, 2010, p. 386).

Outcome (in C-M-O framework). '*The practical effects produced by causal mechanisms being triggered in a given context*' (Tilley 2016, p. 145).

Statement of Original Authorship

The work contained in this thesis has not been previously submitted to meet requirements for an award at this or any other higher education institution. To the best of my knowledge and belief, the thesis contains no material previously published or written by another person except where due reference is made.

(The thesis has 80,002 total words, with additional 16,369 words of References & Appendices)

band Mansh

Signature:

Date:

14 April 2021

Chapter 1: Introduction

A burning platform for change

Covid-19 has resulted in a 'new norm' (Economist, 2020). The pandemic global health crisis has significantly disrupted the global economy and societies (Economist, 2020). As a result, we are experiencing a transformation in the way society, government and businesses operate. The ways we work, socialize, produce and consume have changed (Economist, 2020). The current situation also highlights the need to ensure the vulnerable are protected and no-one is left behind – in line with the principles of the Sustainable Development Goals (SDGs) (UN, 2015). The SDGs remind us that, despite the urgency of managing the pandemic, the world needs to keep in mind the long-term nature of the circular economy transition and global sustainability objectives including the global climate targets and meeting the needs of future generations (Sachs et al., 2020):

"The SDGs are needed more than ever. Their bedrock principles of social inclusion, universal access to public services, and global cooperation are the guideposts for fighting Covid-19 as well as for the investment-led recovery the world should adopt to overcome the economic crisis caused by the pandemic".

However, the latest OECD (2019) and UN (Sachs, et al., 2020) SDG dashboards indicate that the world is falling behind on all SDGs, but especially in the race to avert the climate crisis, which is one of the seventeen SDGs. This acts as a warning beacon for the other SDGs, because as Sachs' notes, the SDGs are interconnected and underperformance on one is likely to have negative impacts on others. For example, 2019 was the hottest year on record, concluding the hottest decade on record (UN Global Compact, 2020), with resulting significant impacts on agriculture (IPCC, 2019) and this impacts the health and wealth of societies. The trend is set to continue. In November 2019, 11,000 climate scientists sounded the alarm, saying the Earth is "clearly and unequivocally" facing a climate emergency and warning that we are running out of time to reverse the trend (IPCC, 2019). We are all impacted by climate change that threatens the livelihoods and safety of billions of people. There is some positive news, such as the announcement at the UN General Assembly in September

by Chinese President, Xi Jinping, that China aimed to go carbon neutral by 2060. However, as Antonio Guterres, the UN Secretary General, says, our immediate and concerted global efforts in 2021, and specifically at the UK-hosted COP26, *"is a make or break moment for the issue*" (BBC, 2021).

1.1 OVERVIEW OF INTRODUCTION CHAPTER

This introduction chapter states the objectives of this programme of research into the Measurement of Infrastructure projects' SDG Impact (MISI) and includes definitions of the key concepts and variables, as well as providing a brief outline of the background and research approach.

This chapter outlines: the background (section 1.2); the context (section 1.3) of the research; my Personal perspective – Adopting the Kolb Learning Cycle (section 1.4); the purpose of the research, including the what, why, how, when (section 1.5); the significance, scope and definitions (section 1.6); the thesis outline (section 1.7); and as with all chapters, it will conclude with a summary (section 1.8) and a link to the next chapter (section 1.9) to ensure clear sign-posting for the reader.

1.2 BACKGROUND

In April 2017, as the deputy team leader of an overland ice expedition to the North Pole, I was extracted with the team (with the film crew of the Amazon funded 'True North' film on cancer recovery and global warming) via a temporary Russian scientific Ice Station. On the 10th April 2017, I met Alexander Danilov, a Russian climate researcher from the Arctic and Antarctic Research Institute (AARI) of the Federal Service for Hydrometeorology and Environmental Monitoring. As the AARI Deputy Director for Research, he expertly explained the background of the problem of climate change and shared a 'jaw-dropping' fact: that the NASA satellite images in February 2017 showed that there was 1 million square kilometres less ice around the North polar region, than in February 2016. This was the turning point for me. To stop talking about "the problem we need to resolve", and instead to "take action by leveraging my experience". The full explanation of the rationale of how and why I have come to the point of a doctoral project, based on my experience-led research, aligned to the Kolb Experiential Learning Cycle (1981, 1984), is provided in section 1.4, Personal Background.

1.3 CONTEXT – PROBLEM STATEMENT

In 2015, the international community responded to the sustainable development challenge with the Sustainable Development Goals (SDGs) for 2030 in their report 'Transforming Our World: The 2030 Agenda for Sustainable Development' (United Nations, 2015). The SDGs are the United Nations' blueprint and have been signed by 193 nations. They address the global challenges, such as poverty, inequality, climate change, environmental degradation, prosperity, and peace and justice. At the global level, the SDGs are interconnected, and the overarching ambition is to 'leave no one behind' in the achievement of the 2030 targets. However, the global nature of the SDGs means that they have a performance framework that is well developed at regional and national levels, but this has been difficult to cascade to the sub-national level, including at both the organisational and project levels (Patel et al., 2017; Galli et al., 2018). The problem manifests because most infrastructure investments are made at a local level, and therefore, without having an approach that makes adequate provision for the longer-term impacts across SDGs, there is a likelihood that practitioners will make less-informed decisions (Adshead et al., 2019; Thacker et al., 2019). Conversely, by using an SDG lens to view infrastructure investments, strategic infrastructure interventions can lead to significant SDG progress. This implies that improved decisions at local level are possible through translating global impact down to the project level.

Alarmingly, after only five years, the global commitment to deliver meaningful SDG action is falling behind on ambitions at both the local and global levels (Office of National Statistics, 2019). This is relevant for project managers because much of tomorrow's resilience and development will be delivered by the project management community, across all sectors, but especially infrastructure. More specifically, the IPCC's 2018 Report identifies that '*directing finance towards investment in infrastructure for mitigation and adaptation*' is key to meeting SDG targets. The estimated \$97.5 trillion USD (Global Infrastructure Hub, 2019) of investment in infrastructure projects that is required globally by 2040, is considered by McKinsey Global Management Consultancy (Bielenberg et al., 2016) to represent a unique opportunity to stimulate economic prosperity, reduce poverty and raise standards in health, education and gender equality. At the same time, the challenge of measuring project outcomes against SDG goals, targets and indicators within existing project

business models is not a new phenomenon, since the difficulty of measuring sustainability outcomes is a well-researched area (Proctor et al., 2011; Boswell et al., 2015). The use of SDGs to measure success at project level is important for two reasons: firstly, they can help monitor progress at an international level (Constanza et al., 2016); secondly, they can help with selecting infrastructure projects which aim to address SDGs in the design stage/front-end of projects (Adshead et al., 2019). In turn, investment decisions can be targeted towards the distribution of funding to infrastructure projects that can achieve broader and longer-lasting impact (Thacker et al., 2019). Up until this point, scientists, policymakers and practitioners seem to have captured the greatest challenges that the world is facing. The UN's recent fourth report on Localizing SDGs (Global Task Force, 2020) has amplified more than ever before the need to find new ways to increase the pace and scale of positive change post-COVID-19 pandemic. This has also been noted by others who have highlighted the need for new thinking on governance models which address the UN SDG priorities as well as for new ways to measure and support the delivery of the UN SDGs 2030 targets (Adshead et al., 2019).

1.4 PERSONAL PERSPECTIVE – ADOPTING THE KOLB LEARNING CYCLE

The author is typically central to research, so it is critical to reflect on my¹ own position to place my research in the context of my experience and motivations. My interest in this research subject was shaped by both my practitioner and research backgrounds. In this way, I have used the Kolb's (1981, 1984) Experiential Learning Cycle theory to help structure my approach across the four stages: Concrete Experience; Reflective Observation of the New Experience; Abstract Conceptualization; and Active Experimentation.

Over the course of my professional life, I have become increasingly involved in senior project management roles, in what Kolb describes as 'Concrete Experience'. By completing three master's level degrees at King's College London, Cambridge and Oxford Universities, I have been able to meaningfully apply the four Kolb learning stages with opportunities to reflect on new experiences, seek abstract

¹ The personal pronoun has been used in this section as it provides my personal perspective.

conceptualisation and apply active experimentation to see what effects the new action achieved.

I am not short of opportunities to apply this approach because I am one of UK's most senior project management advisors, as well as having personally led a number of national programmes, such as the Programme Director for Smart Meters. As such, I regularly represent the HM Treasury and the Cabinet Office as their Team Leader in reviewing and advising on UK's top megaprojects, as defined by Flyvbjerg (2009), often costing the taxpayer in excess of £1Bn and longer than 10 years in delivery and also, identified as critical to UK's infrastructure strategy (although I have led the review of many larger projects, with some megaprojects over £100Bn, including High Speed 2 and the UK Nuclear Deterrent). I regularly write reports for Ministers on project assessment reviews and I am responsible for recommending whether these projects should continue or be stopped. For example, I have been the government's senior advisor on full time consultancy roles to High Speed 2 Railway and also for the national post-Grenfell buildings response, for 26 of the 30 months prior to starting the doctoral research. I have also been on the United Nation's list of senior Governance Advisors, since 2012, for their largest and most complex Infrastructure Projects (UNOPS) which complemented my roles in the Ministry of Defence where I had worked extensively abroad, completed an MPhil at Cambridge University in International Relations and served for 3 years in the British Embassy in Washington DC. However, despite having a key part in the advice to Ministers and senior civil servants to drive improved performance, too often the megaprojects exceeded cost, were delivered outside time estimates and fell short on scope and benefits realisation. This underperformance challenge, noted by Flyvbjerg and colleagues (2003) as: 'the past decade has seen a sharp increase in the magnitude and frequency of major infrastructure projects ... the paradox being ... that many projects have strikingly poor performance records in terms of economy, environment and public support'. This failure has also been identified by many others through their examination of megaprojects such as Terminal 5 (Davies et al., 2009; Denicol, Davies & Krystallis, 2020). For this reason, it is one of the Cabinet Offices' highest priorities to resolve. It is repetitively wasteful and needs addressing.

My frustration with the consistent underperformance of megaprojects led me to complete a MSc in Major Programme Management at Oxford University between 2010-2012, under the mentoring of Professor Bent Flyvbjerg, a leading expert in this field. Whilst at the Said Business School, where I achieved the second highest marks on a course of 55 students and awarded a distinction, I set-up a charity (OxAid) with MBA students to provide pro bono support to global charities. In my first year we won two national project management excellence awards, and this led me to choose a research dissertation that sought to understand why environmental megaprojects were often poorly defined and inadequately reported against. I was given full access to all Worldwide Fund (WWF) for Nature's project data, for whom I had worked for 12 years on a pro bono basis and worked with them to identify whether their approach to the definition of outcomes and benefits was sufficient to deal with global challenges such as the impact of climate change. Sadly, my WWF research concluded that the greater the ambition of the project goal, the less likely the project delivered successfully. The research left a number of unanswered questions, including whether there was a better way to measure impact, in essence, becoming more adept at managing outcomes and less project output focused. Whilst the overall conclusion was not ground-breaking, some of the tools and approaches proposed added significant contributions to practitioners.

Since the completion of my MSc at Oxford I have continued my passion for project management by teaching at University College London (UCL) whilst also managing a busy professional advisory company, that won the Association of Project Management's top project management company in UK in two of its first five years in existence. In my final year of this doctoral research, I have completed my seventh year of lecturing at UCL, where I lead the UCL Strategic Management of Projects MSc's module on Governance and Controls at the Bartlett School of Construction and Project Management. This teaching has complemented my practical project management advisory work, fully embedding the benefits of the Kolb Experiential Learning Cycle (1981; 1984) and has further stimulated my desire to continue my research into measuring megaproject impacts.

More recently I have been working for the government as the senior Team Leader reviewing the Foreign and Commonwealth Office's investment of £1.2Bn into the Prosperity Fund² that seeks to address the UN's Sustainable Development Goals (SDGs). This work highlights the importance of investing smartly in projects that deliver the biggest 'bang for the buck' at a local level, but with impact that can be measured at the global level. I have also become closely involved with the Institution of Civil Engineers to consider how engineering companies can increase their infrastructure projects' impact on SDGs.

With my background in project management and an increasing interest in how to measure the local impact using global SDG goals, I was excited at the prospect of focusing on doctoral-level research to delve much deeper into this topical area. My discussions led me to a number of universities, such as UCL and Oxford University, where leading academics were generous in their offers for me to study with them. However, at LSBU I had particularly rich and fruitful discussions with Professor Rao Bhamidimarri, who led the Nathu Puri Institute (NPI) for Engineering and Enterprise, as well as the Knowledge Director at the Institution of Civil Engineers, Nathan Baker, who jointly offered me the opportunity to deliver the primary empirical research into the ICE's 200th celebration at their Global Engineering Congress with over 3,500 delegates. This gave me an 'open-door' to harness empirical evidence by surveying their members and also interviewing 40 CEOs from member organisations. Together, the ICE and LSBU agreed to sponsor me to research 'Measuring Infrastructure projects Sustainable Development Goals Impact' (MISI), by 'Thinking Global – Acting Local'. The research would identify how engineering firms are equipping themselves better to deal with complex infrastructure project delivery that impact global Grand Challenges such as climate change.

In summary, the many megaproject failures that I had witnessed, as well as the growing need to address megaprojects to deal with Grand Challenges such as climate change, provided a rich area of potential research. At the commencement of my doctoral studies, I found strong evidence that this was a research area that mattered to practitioners and many of the global construction companies. There was evidence of a desire and need, but a gap between 'knowing and doing'. As I explored the literature,

 $^{^2}$ In 2018-9, the Prosperity Fund included a portfolio of 27 multi-year programmes totalling £1.2 billion across a range of areas, including investment in infrastructure and aims to demonstrate how programmes contribute to the United Nations' Sustainable Development Goals (SDGs).

I discovered that whilst the appetite for knowing how to deliver local infrastructure projects to achieve global goals was highly valued, very few had mechanisms, processes or tools to do so. Thus, my doctoral studies developed from a growing awareness of the dissonance between what was wanted (a better way to measure local infrastructure investment decisions against global goals) and what was known about how to do it.

1.5 RESEARCH GAP, AIM, QUESTIONS AND OBJECTIVES

The relevance of this research is that finite resources must be invested to produce the 'biggest bang for the buck'. For example, in UK, around £640 billion of gross capital investment (UK HMT, 2020) is planned for infrastructure systems by 2024-25, in line with the National Infrastructure Strategy (2020). The critical point in relation to the challenges of climate change and other existential threats is how these infrastructure investments can deliver environmental, economic and society benefits, whilst contributing to the delivery of the SDG targets.

1.5.1 Research Gap and Purpose

The literature review in chapter two identifies that there is a **research gap** into 'How UN SDG impacts can be measured at project level'. It is an important question to answer because whilst the goals and targets are set at global and national levels, the delivery and impacts are at local level. The research to-date has not addressed this area (Økland, 2015) and therefore this new research provides an opportunity to test emerging theory-driven models with practitioners in the 'front-line' of project delivery. Simply stated, there is evidence that there is a great deal of 'greenwash' talk without confidence that measurement is either meaningful or verified.

Based on the research gap, the **purpose of the study** was to understand the practices of engineering organisations in the construction sector to Measure Infrastructure projects' SDG Impacts (MISI) in order to propose an improved prototype that provided a practical, value-add approach to MISI.

1.5.2 Research Question and Objectives

Based on the preceding discussion, the **research question** is: 'How can global SDG goals be used to define and measure infrastructure projects' SDG impact at organisational and local project levels?'. This question was answered through

empirical research, assessing whether the current outcomes could be improved by the development of a theory-led prototype model that is workable at portfolio and project levels – the intent was that the mechanism would provide a practical way to link the definition of success of local projects, through the organisational construct (at portfolio level), to global SDG goals' impacts. Therefore, the eight **research objectives** cascading from the research question were:

- RO1: To understand the <u>existing knowledge</u> (in theory and practice) on how organisations and projects measure infrastructure projects' SDG Impact.
- RO2: To understand the <u>context</u> (the 'variables') of the current use of mechanisms to measure infrastructure projects' SDG Impact.
- RO3: To assess the <u>current mechanisms</u> used for the Measurement of Infrastructure projects' SDG Impact (MISI) at portfolio and project levels.
- RO4: To understand the <u>perception of individual engineers and</u> <u>organisations'</u> relative perception of the <u>outcomes</u> of the current use of mechanisms to measure infrastructure projects' SDG Impact.
- RO5: To use the theory-led study to inform the <u>development of a prototype</u> model to improve the measurement of infrastructure projects' SDG Impact.
- RO6: To <u>test whether a 'golden thread'</u> of SDG measurement could be identified from global to local levels.
- RO7: To <u>test whether the prototype could be validated</u> with a case study organisation.
- RO8: To <u>build a framework for further development</u>, for researchers and practitioners to utilise, driving improved investment decisions across planet, profit and people outcome criteria, aligned to SDG impacts.

The ROs sequentially support the design of the methodology (as shown in Chapter 4, Table 9), thereby providing findings and insights that iteratively inform the following study stage. This builds RO understanding cumulatively. The thesis will return to the RO in the final chapter when collating the summary findings across the four areas of: Research Objectives (were they achieved?); Variables Framework (the contextual and mechanism areas of study derived from Chapters 2-4); Propositions (derived from Chapter 2); and final Recommendations.

1.6 SIGNIFICANCE AND SCOPE

The contribution of this research is to harness the results of how engineers employ the SDG prototype framework at the project level to examine the 'contextual' strengths and weaknesses of utilizing the SDG measuring 'mechanism'. This provides deep insights for academics and practitioners to improve their understanding of how the SDGs can provide increased impact at the local level. It informs further research into local measurement of SDGs with, for example, the opportunity to assess theoryled investigations that establish a link from the local to global levels, via the portfolio layer. Practitioners can also learn from these developments, seeking new ways to link rhetoric to action (Scheyvens et al., 2016) so that businesses can fully leverage their innovation, responsiveness, and resources to drive SDG success.

The **scientific contribution** of the research is to provide insights for academics to further develop the theory-to-practice understanding of the Theory of Change (Weis, 2003) and its relationship to MISI. It provides an opportunity for researchers to compare their research with the specific context of this study and it adds evidence of a growing coherence to the development of the theories in specific contextual situations.

The **practical implications** are that the research into the specific contextual use of the prototype model by the Environment Agency highlights that it provides a workable approach for practitioner organisations to adopt, to strengthen their understanding of how to identify and measure prioritised SDG targets. In doing so, it enables a more balanced (across the TBL), longer-term, and plausible logic chain to align with existing business case and benefits management approaches. This provides confidence that the use of SDG targets can be used to improve, deepen and be made more relevant to individual projects' definition of success. This is pertinent at the key investment decision points to enhance the strategic clarity and stakeholder alignment by 'thinking global – but acting local'.

The **limitations** of the research are that the very essence of the Realist Evaluation (Pawson and Tilley, 1997; Linsley et al., 2015), the theoretical methodology used for this investigation, is the recognition that it is mostly contextually based and therefore extrapolation for theory and practice is not necessarily linear or binary. It is dependent on the social systems and agents that were

part of this research at this particular time-context. It therefore acts as a basis for knowledge building in this important area.

1.7 THESIS OUTLINE

The structure of the thesis is as follows:

The first chapter provides an overview of the thesis and its component parts.

Chapter 2 provides a literature review which was aimed to ascertain the current understanding of construction sector's approach to Measuring Infrastructure Projects' UN SDG Impacts (MISI). The chapter, based on a detailed Systematic Literature Review (SLR) evaluates three topics, including: Grand Challenges and SDGs; Sustainability and Sustainable Development in relation to SDGs; and, Sustainable infrastructure projects and project success in relation to SDGs. It concludes by identifying the results and emergent themes from the SLR and its implications for shaping the research design.

Chapter 3 discusses the theoretical framework. The chapter describes how the Critical Realism perspective of ideological philosophers such as Bhaskar (1978) informed the Realist Evaluation approach championed by Pawson and Tilley (2004) and this introduces the Context-Mechanism-Outcomes model that provides the overarching research framework. The chapter includes analysis of the primary theories adopted for the research, including the Theory of Change and Triple Bottom Line, which are both embedded in the proposed mechanism, called the Impact Value Chain (IVC). The initial theoretical model of the IVC is developed synthesising the findings from Chapters 2 and 3. It is further developed into a prototype in Chapter 7, based on findings from the survey and interviews. The chapter also introduces the context and mechanism across the project organisational structure developed by Müller, Drouin, and Sankaran (2019a), who proposed a layered model of organisational project management (OPM). This enables the structured evaluation of the concept of SDG, from global to local, at portfolio, mega-project and project levels.

Chapter 4 outlines the research design and methodology for this Sequential Explanatory Design (Creswell, 2017) that uses a mixed methods study of surveys and interviews to address Proposition 1 and 2 (listed at the end of Chapter 2) – gaining insights to individual engineers' and organisations' perceptions of approaches to

MISI. A section on research design describes the philosophical position of the thesis and the research setting that is discussed in more detail in Chapter 5 for the survey, and the interviews in Chapter 6.

Chapter 5 addresses the survey method, its results and the analysis. It describes the sample size, data collection tools, procedure, and data analysis used. Data integration methods, strategies to minimise threats to validity, and ethical issues are described to end this chapter as well as identifying themes that can be utilized to inform the interview stage. It concludes by answering Proposition 1: The current SDG Goals measurement approach (process, tools and governance) at project level are relevant and reliable, as perceived by individual engineers.

Chapter 6 is similar in structure to Chapter 5, but instead, covers the interview method, its results and the analysis. It describes the sample size, data collection tools, procedure, and data analysis used. Interview findings begin with a rich description of the sample which is followed by detailed description of the findings presented in model format and supported by in-depth quotations. The chapter concludes by combining the emergent themes from both the survey and the interviews. It concludes by answering Proposition 2.

Chapter 7 captures the triangulation of results that informs the further (from Chapter 3's initial IVC design) development of the IVC from an initial theoretical model into a more robust prototype, ready for testing. This is done through the iteration of the IVC theoretical model that was initially synthesised from Chapter 2 and 3, described more fully in Section 3.8. In this chapter, the IVC is jointly developed with a methodology, the SDGiPro© (designed by the author alongside the IVC model) that had been identified from the survey and interviews as a necessary extension to the IVC to support its implementation in 'active' projects. This early prototype is developed to a stage that is ready for testing in Test 1 (SDG 'Golden Thread') and Test 2 (in a water utility company context), discussed in Chapters 8 and 9 respectively. In the chapter's conclusion, the approach to testing the IVC prototype model is discussed, leading to Test 1 and Test 2, discussed below in the summary of Chapters 8 and 9 respectively.

In **Chapter 8** the context of the 'Golden Thread' test is discussed. It describes the approach to the identification of the Global Reporting Initiative ('GRI' is the global standard for organisational ESG (Environment, Social and Governance) measurement.

It explains why the Building Research Establishment's CEEQUAL project sustainability measurement approach is jointly used with GRI to test the linkage from SDG, through the GRI, to the project-level assessments. Following this, findings stemming from the analysis are used to answer Proposition 3: A 'golden thread' exists from 'global' level SDG goals and targets, through organisational levels (e.g., portfolio), cascaded to 'local' project levels.

Chapter 9 outlines Test 2, the case study using the Impact Value Chain (IVC) in a water utility organisation. This links the organisation's strategic objectives (corporate goals) with SDGs and maps the pathway to measure the selected SDG goals at project level in the delivery phase. It describes the approach to the identification of the water utility company, Anglian Water, and the use of desk-level analysis of publicly available information and interviews with a few of the leaders, to assess the potential practical use of the IVC. Following this, findings stemming from the analysis are used to answer Proposition 4: A developed prototype, building on empirical findings from testing proposition' 1, 2, and 3 can provide a plausible, testable and achievable logic chain for defining project-SDG impacts. The final sections describe the strengths and limitations of this research.

Chapter 10 concludes the thesis with a summary followed by description of how this research contributes new knowledge. Implications for practice, future research, and policy are outlined. It includes summary findings across the four areas of: Research Objectives (were they achieved?); Variables Framework (the contextual and mechanism areas of study derived from Chapters 2-4); Propositions (derived from Chapter 2); and, final Recommendations. The chapter includes the implications for this research on the Theory of Change, through the IVC, on the MISI approach. It addresses potential value to the scientific and practitioner communities and shares the emergent results from the collaboration with the Environment Agency, Thames Tideway and the UK Government's Infrastructure and Projects Authority (with UCL, ICE, BRE and UN Global Compact – their letters of impact support are included at Appendices 17-23). The dissemination strategy is provided as well as reflections from the researcher using reflexivity, with concluding remarks to close the thesis.

The overall chapter structure for the thesis investigation is shown below:



Figure 1: Thesis Chapter Map and Workflow.

Note: The LSBU Research Degree Code of Practice (2018), states the thesis should contain information on any publications produced as part of the research, "either included as part of the thesis or placed in a pocket at the end of the thesis" (Research Degree Code of Practice, p.24). The full list of peer-reviewed published papers (n=12) are shown at *Appendix 24 – Publications in Advance of Thesis*. Chapters 2, 5, 6, 7, 8, 9 were all developed from published papers in internationally leading journals including Sustainability (two articles), Administrative Sciences, and the Proceedings of the Institution of Civil Engineers–Engineering Sustainability. Copies can be provided on request (weblinks are included in Appendix 24).

1.8 SUMMARY AND LINK TO THE NEXT CHAPTER

This chapter provided an orientation to the thesis. SDGs are of fundamental importance to the global community – the macro effects are global in nature but have local impacts. The corollary is also true, that investing in infrastructure projects that have a meaningful SDG local impact, can contribute to global SDG goals for the benefit of the planet, profit and people.

The author's personal experience across project management and also as an Arctic and Antarctic explorer inform the research approach, using the Kolb Experiential learning Cycle. The thesis organisation across the 10 chapters is explained to orientate readers to its structure and content. The next chapter provides a review of literature relating to MISI.

The Logic Map shown below in Figure 2 illustrates the development of the structure and evidence-trail to respond to the Research Question and its Objectives. The map also depicts the development of the Propositions and the C-M-O Variables Framework. The red boxes and red connecting lines illustrate how this measurement framework was built and how it flows. It highlights the connecting loop, from the design of a comprehensive and evidence-based research structure, that returns to answer the Research Question and Objectives established in Chapter 1.



Figure 2. Logic Map of Thesis Findings (red line = Proposition & Variables Framework flow).
2.1 OUTLINE OF THIS CHAPTER

The central topics of this chapter are: Topic 1, Grand Challenges and SDGs (Section 2.3); Topic 2, Sustainability and sustainable development in relation to SDGs (Section 2.4); Topic 3, Sustainable Infrastructure Projects and project success in relation to the SDGs (Section 2.5). This is followed by a description in Section 2.6 of the methodology employed for the systematic literature review, based on the PRISMA (Liberati et al., 2009) checklist, to investigate the existing UN SDG targets in relation to Infrastructure Projects (IP). In the concluding part of the chapter the preliminary findings from the systematic literature review are listed with a description of how they will inform the construct of the thesis' research framework, thereby providing an organised way to conduct the empirical data collection in Chapters 5, 6, 8 and 9.

2.2 LITERATURE REVIEW INTRODUCTION

This chapter reviews the published knowledge of the research area and provides arguments to support the study focus. The aim of this chapter is to delineate various theoretical positions, and from these, to develop a conceptual framework to underpin the research question. In order to meet the first three Research Objectives (Section 1.5.2)³ there were three sub-research questions chosen to guide the SLR:

(1) How extensive (broad and deep) has the research into MISI been to-date?;

(2) What context and mechanism variables were identified that might inform the building of MISI study's 'Variables Framework'? This framework is developed in Chapters 2 and 4 and is used to help collate the findings in the final Chapter. They also support systematic research in the empirical data gathering stage (Chapter 6,7,8 and 9); and

(3) Subsequent to the doctoral MISI research, how could such a framework be used to provide guidance to a range of stakeholders (including regulators,

³ RO1: To understand the existing MISI knowledge (in theory and practice); RO2: To understand the MISI context (the 'variables'); RO3: To assess the current MISI mechanisms.

policymakers, academia, investors and infrastructure practitioners) on how to align IP and their impacts to the SDGs?

2.3 GRAND CHALLENGES AND SDG

Grand Challenges is a term used, predominantly by the academic community, to qualify and structure responses to so called 'wicked problems' (Head & Alford, 2015) of immense magnitude and impact. 'Grand Challenges' capture ideas that are equally relevant to academics as well as practitioners. They are also, by definition, both ambitious ("capture the peoples' imagination") and achievable ("solve ... problems") (Executive Office of the President, 1989). Additionally, the definition identifies the need for impact and the measurement thereof and impact to demonstrate meaningful progress. The definition of Grand Challenges has evolved since Mertz's (2005) focus on the engineering communities, to a broader group of stakeholders that includes policy shapers, funders, and delivery-to-operations project teams (Omenn, 2006). Consequently, project management professionals have the opportunity to take a leading role in this, especially in providing tangible action that can be implemented by practitioners to effect improved performance against the SDG targets.

More recent research into Grand Challenges (Sakhrani et al., 2017) has identified five characteristics that are helpful in this paper's MISI analysis: Grand Challenges are (a) articulated by stakeholders, (b) specific, (c) ambitious yet feasible, (d) framed in a manner that suggests the use of specific methods or disciplines, and (e) have the potential for broad impact. These characteristics provide a useful reference point for developing a conceptual framework to deepen the research into how the project management community can define, design and measure IP contributions towards the SDGs. In effect, the characteristics of the five grand challenges provide the 'lens' to identify what links SDGs to IP.

The failure of not meeting the 2030 targets of the United Nations Sustainable Development Goals (hereafter, UN SDGs) is amongst the most significant global Grand Challenges threatening our survival today (IPCC, 2018) and there is the potential for the project management community to play a key part (Morris, 2017) in making a positive impact on the 2030 targets, as illustrated in Figure 3.



Figure 3: The Global Goals for Sustainable Development (United Nations 2015—permission to use from Sustainable Development Goal (SDG) logo Guidelines).

Although the SDGs build on the earlier Millennium Development Goals (MDGs) by focusing on similar issues, the SDGs differ from the MDGs because they are for all countries in the world to implement—developed and developing alike. Moreover, unlike the MDGs, the SDGs are focused on monitoring, evaluation and accountability across society, not just at the national level, which is why it is critical that the link is made from the 'bottom-to-top'. This means linking from the delivery at the local project level through to the impacts at the national and global levels. However, there appears to be a gap. The golden thread from the national to the project level seems to be missing. This is key because SDGs have been conceptualised at the global level but actually materialise and are operationalised at the project level (Thacker et al., 2019). This is especially true for infrastructure projects (IP) (Adshead et al., 2019) that often reflect large-scale governmental investments, which are delivered by multiple stakeholders working across boundaries and where the linking of global-to-local impacts at the project level is potentially transformative for business policy and everyday citizens, as discussed in the next section.

The SDG delivery targets are understandably ambitious and needed a reporting framework that would drive meaningful and verifiable progress towards the 2030 targets. In 2017, the UN's Inter-agency Expert Group on Targets and Indicators for Sustainable Development designed a mechanism that linked goals, targets and indicators across the geographic and governance boundaries at national, regional and global levels (IAEG-SDGs, 2017). Within this framework, shown in Figure 4, the

Expert Group designed thematic areas that could also be used at the subnational level but, because the targets and indicators were originally designed to be used at global, regional and national level, they had reduced applicability at organisational or project levels. Simply stated, "one size does not fit all". This provides a significant challenge because most of the investment needed (USD \$94 trillion) to respond to the global goals (Global Infrastructure Hub, 2020) is delivered through the business sector, typically through infrastructure projects, which contribute to the systems and services that can positively impact health, wealth and inequalities.



Figure 4: The sustainable development goals (SDG) Targets and Indicators' framework designed by the UN IAEG-SDGs (2017).

As stated earlier, the SDGs consist of 17 major goals and 169 concrete targets and, because some of the targets are not expressed as concrete numbers, the UN also developed a framework of 232 indicators for monitoring and reviewing the targets. Research into the use of the SDG framework (Mansell et al., 2019a) on infrastructure projects has identified that the targets (N = 169) and indicators (N = 232) are too numerous and complicated and therefore, unfortunately, they are rarely used by engineering and project practitioners. The research concluded that a new way was needed to reduce the scientific and statistical complexity of the SDG measurement framework. The starting point for this approach was to evaluate their usability and applicability at the project level on a sector-by-sector basis. For example, in the infrastructure sector, recent analysis (UNOPS, 2018) indicates that 81% of the SDG targets are influenced by infrastructure investment projects. However, "influence" is a comparatively weak word without specifying "attribution" (i.e., directly impacting with verifiable evidence) or "contribution" (i.e., linkage presumed but without evidence) and, therefore, despite the positive conclusion from the UNOPS's analysis (2018), further research is needed to identify which of the SDG targets can be used at project level.

2.4 SUSTAINABILITY AND SUSTAINABLE DEVELOPMENT IN RELATION TO SDG

Sustainability can be a challenging word. Indeed, fifteen years ago there were up to sixty definitions of sustainability (Hartshorn et al., 2005) with little convergence of how the theory of sustainability could be given meaning in practice. There are those (Zuofa, & Ochieng, 2016; Sverdrup & Rosen, 1998) who suggest that sustainability is essentially the long-term harnessing of an ecosystem to maximise the outcomes whilst ensuring the extraction of the input of resources from the ecosystem do not negatively impact its long-term viability. Alternatively, there are others (Costanza & Patten, 1995) who define sustainability simply as a measure of whether a system can ultimately continue or is self-consuming. Holling (2001) broadened the sustainability systems' definition to include 'development': "Sustainable development ... refers to the goal of fostering adaptive capabilities and creating opportunities". It can thus be shown that 'sustainability' has become mired in value-laden language and often vague in concept (Mebratu, 1998; Ciegis et al., 2009; Emas, 2015) that can cause diffusion of interpretation and confusion in practice (Fenner et al., 2006; Ainger and Fenner, 2014; Moore et al., 2017). These examples explain why the definition remains nebulous and why a practical definition has greater utility (Glavic and Lukman, 2007) for project managers.

Salas-Zapata and Ortiz-Muñoz (2019) consider four uses and meanings of the concept of sustainability: (1) a set of criteria, consisting of social-ecological criteria to guide human actions or their products; (2) a vision or goal, which is the convergence of environmental, social and economic purposes, expectations, aims or goals of a system; (3) an object, which is an empirical entity that can be thought and intervened; and (4) an approach, which is the study of social, economic and ecological dimensions or variables of a human activity, product or system.

For the purposes of this thesis, the definition of sustainability builds on the broader definition of sustainable development as "development that meets the needs of the present without compromising the ability of future generations to meet their own needs" (Brundtland, 1987). Over the past 50 years, the phraseology and

understanding of 'sustainable development' (Sachs, 2016) has become an increasingly central theme of nation states and their citizens. Today, the Planetary Boundaries (Rockström, 2009; De Vries et al.2015) provide a global litmus test for how we are doing. The concept of nine planetary boundaries within which humanity can continue to develop and thrive for generations to come was developed in 2009 by environmental scientists from the Stockholm Resilience Centre. The most significant global response to the Planetary Boundary challenge was in 2015, when all governments ratified the UN's 17 SDGs (United Nations, 2015) to be achieved by 2030. This represented a major step-change in the implementation of the sustainability agenda and effective responses to the Planetary Boundary challenge (Rockström, 2009).

It is opined that a gap exists – Infrastructure Projects (IP) are not included in the SDGs' measurement, and the evidence (Martens, & Carvalho, 2016a and 2016b) illustrates that the golden thread from project level measurement to global-national level, is missing. This echoes research highlighting a gap between theory and practice for incorporating sustainability measurement in project management (Økland, 2015).

The problem of identifying suitable SDG measurement is compounded at the indicator level, where a further 232 measurement metrics reside. For example, the UK's Office for National Statistics (ONS) online portal, responsible for reporting UK's progress against global SDG indicator measurement, shows that, in April 2019, they only had data for 173 of the 232 indicators, with 69 being without data (ONS, 2019). The ONS's challenge of collating reporting evidence for the 232 indicators was further corroborated by recent analysis (Mansell et al., 2019a) of the viability of using each of 232 indicators for infrastructure project-level measurement of success. The analysis, based on inductive reasoning using the project success framework proposed by Morris (2013) and Cooke-Davies (2007) and then analysed against the cost-benefit measurement framework from the HMT Green & Orange Books (HM Treasury, 2013) and the World Bank monitoring, reporting, evaluation and learning methodology (Dudwick et al., 2006), highlighted there were only a small number of indicators (N =28; 12%) relevant to engineering projects. Of these, only 8% (N = 20) have close alignment with the engineering projects and 4% (N = 8) have marginal relevance. This analysis highlighted a "gap" of not having suitable indicators below the SDG target level that could be used on infrastructure projects.

2.5 SUSTAINABLE INFRASTRUCTURE PROJECTS (IP) AND PROJECT SUCCESS IN RELATION TO SDG

The Global Commission on the Economy and Climate defined Infrastructure (New Climate Economy 2016, p. 4) as 'structures and facilities that underpin power and other energy systems (including upstream infrastructure, such as the fuel production sector), transport, telecommunications, water, and waste management'. This definition takes a systems perspective and includes energy, transport, buildings, water supply and sanitation, IT and communications, but does not explicitly include natural infrastructure such as forests, deserts, wetlands and other ecosystems (Bhattacharya et al. 2015; The New Climate Economy 2016).

Most of society's developments in recent times can be connected to infrastructure projects (Thacker and Hall, 2018; Thacker et al., 2019) and the UN recognise that the development of infrastructure represents a massive opportunity to stimulate economic prosperity, reduce poverty and raise standards in health, education and gender equality (UNOPS, 2018).

It is evident that ameliorating many of the risks associated with grand challenges, such as climate change, can be supported through investment in appropriate and resilient infrastructure and engineering (OECD, 2019). For example, greenhouse gas emissions cannot be sufficiently reduced without new forms of energy infrastructure or less polluting transport networks; and water security requires investment in new and more resilient forms of water infrastructure (OECD, 2019; United Nations, 2019). This highlights the importance of IP to link from the local investment level to global goals and perhaps provides recognition of the ability of engineering and infrastructure to reduce strategic business risk.

Contribution to the growing literature on the measurement of IPs on sustainability is provided by Shen, et al. (2010), who focus on the balance needed between benefits to society whilst protecting the environment and still achieving the economic benefits envisaged in the project business case. The linkage across the three areas in the construction industry is further defined by Kibert (2013), who suggests the interrelationship between a project's outputs and the society that is impacted is a central component of defining sustainability success of an infrastructure project. This introduces the concept that project success definition needs to consider success against the Triple Bottom Line (TBL) (Elkington, 1994) of social, environmental (or

ecological) and economic (or financial) effects, otherwise noted as the 'Three Pillars' concept of 'people, profit and the planet' (Elkington, 1994, 2018; Griggs et al, 2013).

Defining IP success is central to the understanding of how to link global-national level SDGs with local IP because it allows stakeholders to align their expectations against shorter-term outputs as well as the longer-term outcomes and SDG impacts. More recent research into project success definition (Thiry, 2004; Lavagnon, 2009; Jenner, 2016) has consistently identified benefits and outcomes as being a critical determinant for the assessment of project success. For example, Michael Thiry (2004) highlights that 'too many critical success factors are related to inputs and management processes and not enough on outcomes'. This is further supported by those (Morris, 2013; Terry Cooke-Davies, 2002, 2007) who identify three levels of success – was the right project done?; and consistent project success – were the projects done right, time after time?

Based on the specific sub-research question of this systematic literature review and the earlier exploration of the key thematic areas, the following systems map at Figure 5 was developed to guide the choice of methodology, based on the 6 core areas that are all linked as a systems-of-systems map. This demonstrates their interconnections and the basis for the chosen research approach.



(2) What issues and context variables were identified that might inform the MISI study's thematic framework to support more systematic research in the empirical data gathering stage (Chapter 6,7,8 and 9)?; and (3) Post this doctoral MISI research, how could such a framework be used to provide guidance to a wider range of stakeholders on how to align IP and their impacts to the SDGs?

Figure 5: Systems map showing the key thematic areas related to the research question.

2.6 METHODOLOGY FOR LITERATURE REVIEW

In order to meet the first three Research Objectives (Section 1.5.2)⁴, it was decided to conduct a systematic literature review (SLR) focused on the leading journals that publish articles across the three thematics of business policy, sustainability and project management.

A literature review can broadly be described as a systematic way of collecting and synthesizing previous research (Baumeister & Leary, 1997; Tranfield, Denyer, & Smart, 2003). The selected approach conformed to established SLR protocols that have been mostly based on the rules and standards proposed by Liberati et al. (2009). In development of the PRISMA statement, the researcher (Liberati et al., 2009) laid out a reporting proposal for systematic reviews that had been derived from the metaanalyses of health care studies. As part of this work, the study developed a 'PRISMA Flow Diagram' (Moher et al., 2009) that explains and elaborates on the PRISMA Statement. An example of how this approach was used is shown in the adapted flow diagram, below:



Figure 6: Adaption of the PRISMA Flow Diagram' (Moher et al., 2009) to explain the deselection process.

⁴ Shortened RO: RO1: To understand the existing MISI knowledge (in theory and practice); RO2: To understand the MISI context (the 'variables'); RO3: To assess the current MISI mechanisms.

This was complemented by a Level 2 Analysis that adopted a semi-systematic review process, advocated by Wong et al. (2013), that evaluated the top 13 articles that had the highest prevalence of keywords within the selected dataset, and this supported the derived nodal map of key variable that provided a evidence-based MISI 'Variables Framework' to be studied. These approaches are shown diagrammatically in Figure 7.



Figure 7: The 'hopper' approach to SLR Level 1 and 2 Analysis

With an aim to ensure sufficiently comprehensive coverage across the three thematics (of business policy, sustainability and project management), the focus of the review was on: project management related journals (International Journal of Project Management, and Project Management Journal); sustainability related journals (Journal of Cleaner Production, and Sustainability); and policy related journals (Journal of Environmental Management, Business Strategy and the Environment, Environmental Science and Policy, Research Policy, and Global Environmental Change). Combined, these journals publish the representative coverage of academic research across the three thematics of project management, sustainability and business policy.

In order to assemble the dataset, Scopus⁵ was used as the search engine, identifying articles by source types (peer-reviewed articles within the selected

⁵ The doctoral researcher had support from an assistant at Cape Town University, operating within the design and framework set by the lead researcher. The specific work carried out by the research assistant included: collection and initial analysis of all selected articles from Scopus; collation of data in spreadsheets; stand-in for lead researcher at EPOC Conference to present findings, October 2020 - https://epossociety.org/epoc-2020-website.

journals). As part of the search, the following keywords were used: 'sustainable development goal', 'sustainable development', or 'sustainability' in conjunction with 'project', 'project management' or 'infrastructure' (from 2015 to 2020). The choice of the six keywords was based on clear differentiation of SDG and IP terminology as well as the need to limit the selection to a manageable data size for analysis. These keywords were searched within the fields of 'title', 'abstract' and 'index key words' as defined by the Scopus search engine. This yielded 1,651 articles, shown in the table below (as oft February 2020).

Table 1. Items containing keywords, in the database per Journa	Table 1: Items	containing	keywords,	in the	e database	per Journal
--	----------------	------------	-----------	--------	------------	-------------

Journal	Total
Journal of Cleaner Production	433
Journal of Environmental Management	366
Sustainability (Switzerland)	376
Business Strategy and the Environment	214
Environmental Science and Policy	162
Research Policy	40
Global Environmental Change	37
International Journal of Project Management	17
Project Management Journal	4
Journal of Social Policy	2
	1,651

The occurrence of searched keywords in relation to one another is displayed in in respect of the title, abstract and index key words (see Table 1). Duplications of items containing a combination of more than two keywords were resolved.

		No. of items	
Search keywords combinations across the title, abstract and index key words	Title	Abstract	Index
"SDG"/"sustainable development goal" + "project"	0	47	0
"SDG"/"sustainable development goal" + "project management"	0	0	0
"SDG"/"sustainable development goal" + "infrastructure"	2	27	0
"sustainable development" + "project"	10	163	153
"sustainable development" + "project management"	1	12	78
"sustainable development" + "infrastructure"	5	89	65
"sustainability" + "project"	53	395	129
"sustainability" + "project management"	10	43	76
"sustainability" + "infrastructure"	17	227	100

Table 2: Items containing combinations of keywords

"sustainable infrastructure"	11	17	6
Total items	109	1,020	607
Total	1,736		

Table 3: Data analysis of key words tabulated across the two dimensions of SDG and IP.

Data capture in Title	Project	Project Management	Infrastructure
SDG/Sustainable Development Goal	0	0	2
Sustainable Development	10	1	5
Sustainability	53	10	17
Data capture in Abstract			
SDG/Sustainable Development Goal	47	0	27
Sustainable Development	163	12	89
Sustainability	395	43	227
	_		
Data capture in Index Key Words			
SDG/Sustainable Development Goal	0	0	0
Sustainable Development	153	78	65
Sustainability	129	76	100

To filter out less relevant articles from the sample, a subset was created based on the occurrence of the combinations of keywords, in both title and abstract, as represented in Table 4. The resultant abstracts were browsed for relevance, which eliminated numerous articles (most of which were focused on pedagogies and training around sustainable development goals, or sustainability projects and initiatives in noninfrastructure sectors).

Table 4: The resultant reduction of dataset based on combination of keywords

Subset Delineators: Combined Keywords	Total Items	Relevant Articles
SDG/Sustainable Development Goal + Project	42	12
SDG/Sustainable Development Goal + Infrastructure	22	16
SDG/Sustainable Development Goal + Project + Infrastructure	5	4
	69	32

The 33 articles represent the final subset analysed. These were included in the final review dataset that followed the SLR approach adopted by Jarvis et al. (2003) that codifies data using relevant thematic frameworks, which was based on the SDG IP thematic structure, codifying key information in relation to seven areas of interest:

• Type of contribution (such as framework proposition, framework testing, exploratory, theoretical, etc.).

- Research design (namely, empirical, single/multiple case study, systematic literature review, etc.).
- Primary geographical focus of study (i.e. where the study was focused).
- Primary infrastructure sector (e.g. water, energy, transport, etc.).
- Primary industry (if applicable; e.g. airports).
- Relevant institutional level (e.g. regional, national, organisational, etc.).
- Sustainable Development Goals mentioned in the article.

The summary data across the 33 articles codified against the seven MISI thematics is included at Appendix 1. The results from this analysis are captured and discussed in the following section. The analysis has been completed at two levels: level 1 examines the dataset of 33 articles across the seven MISI thematics and level 2 provides a deeper analysis of the 13 most relevant articles, based on their prevalence of keyword combinations, across MISI research issues and themes.

2.7 RESULTS AND DISCUSSION OF THE SLR

2.7.1 Quantity of Relevant Articles in Dataset

The analysis of the dataset, illustrated in Figure 8, showed that within the 1600+ sample, there were identifiable differences and similarities. For example, across the four sustainability journals there was a total of 1,426 articles which equates to 285 articles per sustainability journal. This compares with a total of 204 across the policy journals. The least represented were the project management journals that had a total of 21 articles.



Figure 8: Occurrences of articles appearing in selected journals using the keywords

2.7.2 Quantity of Use of Keywords in Abstracts over time

The dataset enabled the capture of the prevalence of keywords used over the selected time period of 2015 – February 2020, as shown in Figure 9. The value for researchers in this field is that it highlights the rapid increase in some keywords, especially in the past two years, and suggests that this is an increasing area of importance and relevance. For example, the prevalence of 'SDG' has increased by a factor of 29 with most of the increase in the last two years. Similarly, the appearance of the keywords of 'sustainable development' have increased by a factor of five and 'sustainability' by a factor of three. The latter two keywords have also had a noticeable inflection point in 2017, that is most likely due to the increasing recognition of the SDG terminology since their introduction in 2015.



Figure 9: Occurrence of Keywords in abstracts over time

2.7.3 Level 1 Analysis - Results across the seven codified MISI thematics

This sub section covers the analysis of the seven SLR research areas identified in the earlier part of this SLR.

2.7.3.1 MISI Related Research Finding 1: Type of contribution (framework proposition, framework testing, exploratory, theoretical, etc.)

Since the aim of this chapter was to evaluate the state of knowledge on the specific MISI research, it is useful to identify what approaches have been used to evaluate this area. The purpose was to create relevant MISI research framework, based on any identified gaps in research, or seek insights that facilitate the empirical data gathering such as through the design of the survey agenda (Chapter 5) and the interviews (Chapter 6). In this way, it has been proposed (Baumeister & Leary, 1997; Torraco, 2005) that literature reviews are useful to develop theory and conceptual

models, as completed in Chapter 3 (Theory) and Chapter 7 (Development of the prototype model).



Figure 10: Type of contributions, using a hierarchical tree map chart, across subset.

The results showed a pre-dominance (60%) of articles were based on Framework Testing (in this context a framework means an analytical tool to support a research study) and Framework Propositions, but there were fewer Theoretical articles and Exploratory methods. This might suggest that there is less confidence in existing theories, and that new frameworks are being developed to harness the theories more effectively for the emerging demands of the MISI topic. While there is not necessarily a MISI research gap in the type of contributions, the findings might highlight the preference for frameworks as a way to engage with practitioners that are seeking tools to effect improved ways of measuring SDG impacts on IP.

2.7.3.2 MISI Related Research Finding 2: Research design (empirical, single/multiple case study, systematic literature review, etc.)

The analysis of MISI SLR research finding 2 on research design used the five headings (shown in Figure 11) to give a high-level quantification of design use. This illustrates that empirical analysis was the most favoured approach, and that two thirds of articles used case studies in some form.



Figure 11: Research design approaches.

The results showed that empirical analysis was the favoured approach followed by multi-case studies and single studies, thus indicating that two thirds of articles used case studies in some form. Drawing conclusions on future research focus from these findings is informed by Tranfield et al., (2003) who contends that many SLRs have researcher bias, lack rigour and have insufficient empirical evidence to underpin insights that could enable intervention into the practitioners' operational roles. This SLR research finding therefore underpins the increasing emphasis towards using empirical evidence and case studies. This suggests that future research should seek design methods that relate to practitioners and thereby help inform the MISI policy formulation and implementation.

2.7.3.3 MISI Related Research Finding 3: Primary geographical focus of study (where the study took place or was focused)

The geographical spread of the articles was significant. There were eleven articles that had a focus on the BRICS (Brazil, Russia, India, China, and South Africa) countries and only two from OECD (Organisation for Economic Co-operation and Development) countries. There were also some outliers that included Mozambique, Curacao, Ethiopia and Somalia. The balance of articles between developed countries (n=26) and developing countries (n=4), with a smaller number having a global focus, shows that there remains a predominance of research in the more accessible databases of developed countries.

The larger data set of articles from, and of, developed countries indicates a prevalence of research due to more accessible databases of developed countries as well as a larger body of research resource capacity. The finding is that there is an opportunity to close the gap by focusing more research resources on developing countries since that is where many of the greatest SDG challenges are situated (UN, 2018; OECD, 2019). Whilst this is a finding of the SLR, it is intended to address this gap by designing a MISI model that has broad applicability across all infrastructure sector areas and have international applicability.

2.7.3.4 MISI Related Research Finding 4: Primary infrastructure sector (e.g. water, energy, transport, etc.)

The representation across the infrastructure sector showed (in Figure 12) a dominance of water-focused articles (n=9), followed by urban development (n=6) and energy (n=4). The remaining eleven areas had a combined total of one less article

(n=18, versus n=19) of the top three areas. In some cases, the article covered more than a single sector, hence the increased total above the 33 articles analysed.



Figure 12: Sectoral focus of subset articles.

The dominance of water-focused, urban development and energy articles suggests key SDG-related IP, such as health (SDG 3 – Good Health and Wellbeing) are under-represented and perhaps, the key enabling IP areas such as roads, housing, sanitation, are also research areas that would benefit from increased focus.

2.7.3.5 MISI Related Research Finding 5: Primary industry (e.g. airports)

The more detailed analysis of each article was carried out to identify their specific subsector focus provided less definitive findings. For example, whilst the prevalence of water sector IP articles illustrated that there were six areas of subcategories used (namely rural water services; transboundary resources; water transfer; river basins and water quality; water supply, solid waste), there was little value gained from further analysis of the other sectors.

The sample set was too narrow to make any significant conclusions from the results but to some extent, the results indicated that the spread of subsector topics in articles is broad, if not deep, across sub-sectors. An opportunity might exist to compare the IP related sub-sector topics with SDG goals and targets, to seek research alignment, for example, analysing SDG targets for SDG 6 (Clean Water and Sanitation) could highlight specific topics related to the targets and their indicators that deserve greater focus. In this case, Target 6.1, 'By 2030, achieve universal and equitable access to safe and affordable drinking water for all', has indicator 6.1.1, measuring the 'Proportion of population using safely managed drinking water services', which could

provide a structured way of narrowing the focus for deeper research into specific subsector topical areas.

2.7.3.6 MISI Related Research Finding 6: Relevant institutional level (e.g. regional, national, organisational, project)

The analysis of the dataset of articles (see Figure 13) indicates that there is less research conducted at the higher and lower ends of the hierarchy. For example, the top levels from global-national spanned 15 articles, the lower level from sector-project spanned 11 articles whilst the middle three levels from national to industry included 20 articles.



Figure 13: Institutional focus of subset articles

The concentration of articles in the central area is not necessarily surprising but it does highlight the paucity of research at the lower levels that this study seeks to address. Some of the critical observations of this analysis is that this suggests that at the organisational level there is better coverage but that a gap exists at the sectorproject level. An existing framework for the analysis across levels is provided by Müller et al. (2019) in their organizational levels in project management (OPM) model (addressed in Chapter 3). In project management terms, this includes the project, programme, and portfolio levels of organisational design and this finding could allow the alignment of the MISI analysis with the Müller-developed theory to help explain the SDG interface and interaction of the OPM elements across the layers within the model.

2.7.3.7 MISI Related Research Finding 7: Sustainable Development Goals mentioned in the articles

The final MISI SLR research findings (#7) was a numeric counting of which specific SDGs (of the 17 goals) were the focus of the articles in the dataset. Unsurprisingly, given the heavy sectorial focus on water in the dataset, SDG 6 on Clean Water and Sanitation, was the most prevalent (n=6). This was followed by SDG 11, Sustainable Cities and Communities, which aligns closely with the search parameters of IP (n=4). Eight articles referred to SDGs in a general reference without a specific focus on individual SDGs and six of the SDGs were specifically mentioned in the articles but with lower occurrence.



Figure 14: SDG mentions across subset.

The results on this SLR research finding suggests that there is an opportunity to broaden the research across the SDGs that were under-represented, specifically into those SDGs identified as being most affected by the development of infrastructure, or, to deepen the research in areas already covered, such as SDG 6 or 11.

2.8 LEVEL 2 ANALYSIS OF THE 13 TOP MISI ARTICLES BASED ON THEIR KEYWORD OCCURRENCES

Whilst the SLR methodology used in the preceding sections was based on the rules and standards proposed by Liberati et al. (2009), the findings have only provided a high-level indication of how to design this studies research framework. Following the initial sieve of relevant article these were further reduced to the most relevant based on their prevalence of keywords. The articles were then analysed to identify MISI issues and sub-issues that have relevance in guiding the design of the MISI research framework, developed in Chapters 3 and 4. The data in Figure 15 highlights the

correlation between the issues and sub-issues and the originating author's reference of where the sub-issues are derived from in the top 13 research articles. This provides the opportunity to use the identified issues to help shape a research 'Variables Framework' (including MISI context and mechanism issues) that can be used for the empirical data gathering in Chapters 5, 6, 8 and 9. A summary of all the relevant extracts from the top 13 articles is included in Appendix 1 and a detailed capture of analysis that indicates emerging MISI propositions is shown at Appendix 2.



Figure 15: Research issues and sub-issues based on the 13 most relevant articles

Analysis of the nodal framework shown in Figure 15 indicates that there is a balanced spread of the top articles that relate to the identified MISI research issues (shown in the first column) although there are three that dominate: 'definition and evaluation of project success' (n=4); 'sustainable construction' (n=3); and 'investment priorities in SDGs' (n=3). There is clearer differentiation of the most common subissues across the research themes: 'policy to implementation' (n=8); 'research approach' (n=4); 'opportunities for delivery of sustainability' (n=4); 'challenges to delivery of sustainability' (n=4); and 'outputs versus outcomes' (n=3).

2.9 SLR FINDINGS FOR DESIGN OF RESEARCH FRAMEWORK

The literature review has sought to explore opportunities for research into the assessment of infrastructure projects through SDG targets. The SLR methodology

was adopted to examine a sub-set of relevant journal articles to understand the quantitative data across six MISI thematic areas (see Figure 5) that emerged from the first part of the SLR. Level 2 analysis was then developed to identify some qualitative insights into the type of issues that have been covered within the sample set.

The following sections consolidates the emerging findings from the SLR and have been captured under the headings of the three questions that were identified at the start of this chapter to respond to Research Objectives 1-3.

2.10 (1) HOW EXTENSIVE (BROAD AND DEEP) HAS THE RESEARCH INTO MISI BEEN TO-DATE?

The Level 1 analysis has indicated that there is a growing body of research into MISI, with a marked increase since 2017. The relative coverage of the MISI, against other business and project management research areas is less easy to quantify since SDGs have only been in existence since 2015 and the SDG targets and indicator framework was only fully agreed by the 193 signatory nations in 2016. However, recent SLR studies (Aarseth et al., 2017; Engert et al., 2016) have shown that sustainability in a project context is still a nascent and fragmented research area, but that there is growing awareness of its criticality to business success, and indeed, also to society and environmental success. This reinforces the contention that MISI research is 'nascent' to an even greater degree because there were approximately 300% more sustainability keyword connections than for SDGs.

2.11 (2) WHAT ISSUES AND SUB-ISSUES WERE IDENTIFIED THAT MIGHT INFORM THE MISI STUDY'S PROPOSITIONS?

The level 2 analysis (see Appendix 2) provided useful insights into the issues and sub-issues that emerged from the qualitative analysis of the highest ranked articles in the dataset. This provided an evidence-based MISI 'Variables Framework' (of contextual and mechanism issues) for structuring the MISI research, specifically the emerging propositions. The diagram below (Figure 16) illustrates the emerging issues and context variable in columns 'a' and 'b'. This forms a linkage to supra-themes in column 'c' that are evaluated further in the next chapter.



Figure 16: Structure of issues, sub-issues and emerging supra-themes.

Whilst the analysis of the individual articles indicated their specific research focus areas, the overview across all the research articles provided a strategic perspective of linkages between the issues, sub-issues and derived supra-themes. In many cases the studies did not overtly cover these areas, and this could be evidence of a research gap. More helpful for achieving the research's objectives is an emerging research framework that provides a strategic context to refine the empirical data gathering carried out in Chapters 5, 6, 8 and 9. To give an indication of how this might be used, the sections below provide an overview of the main points emerging from five of the seven MISI Research Findings shown in Figure 16 in column 'a'.

2.12 (3) WHAT ARE THE EMERGING MISI PROPOSITIONS?

A recurring theme emanating from analysis of the selected articles is the need to make the research accessible and relevant to the practitioner community of stakeholders, whether policy makers, investors, project managers or others. Indeed, there was a specific thematic area that was identified as MISI Research Issue 4: Practical application of Theoretical sustainability models, which is a theme Tranfield et al., (2003), championed. For this reason, the first two propositions, shown in the table below, seek to establish the sector's MISI viewpoints at an individual and organisational level (Propositions 1 and 2). If these propositions can be confirmed, then a prototype could be developed, informed by their views (individual and organisational) and then tested for a 'golden thread' to ensure it is both possible and practical (Proposition 3) and then analysed in a case study context to assess whether the prototype could provide utility (Proposition 4).

The focus on practical application of this research to inform the approach to collect empirical data, is supported by the emerging propositions that this SLR chapter has informed and gives a view of the research 'direction of travel'. The four derived propositions, as evidenced in Table 39 (In Appendix 2, column 'e') and Table 5 (below), drive the subsequent design of the research framework.

Table 5: Derived propositions from the literature review.

SLR Issues	Research derived Propositions	
 MISI Research Issue 1. Cascading from Global to the local level (a reality gap from theory to practice) Hall et al., 2017; Terrapon-Pfaff et al., 2018. 	 The literature review highlighted that a significant challenge is how to operationalise the measurement of SDGs. There was a consistent view that it is considered important, but there was a gap in knowledge as to how to do so. The identified gap from a number of the papers reviewed suggests that there would be benefit in assessing engineers' perspectives on whether they have experience of applying MISI successfully on their projects. The access to engineers, and other stakeholders such as designers, architects and investors, at the project level, would also provide useful insights as to the strengths and weaknesses of MISI. Also, the SLR highlighted the gap in evidence-based research from sector experts such as engineers (to give the individual engineer's view at the coalface) and CEOs (at the organisational level). The SLR highlighted the theoretical importance of this work, but it is not underpinned by evidence. This explicitly seeks acknowledgement that the global SDG goals do have relevance at organisational and project levels, but implicitly seeks to understand the reasons why the current approach is sub optimal and what they believe are ways of improving practical application. The papers analysed highlighted that there are no consistent or evidence-based methods for measuring global goals at project level. Derived Proposition: (for 'proposition' or 'hypothesis' choice, see section below this table) 	
	includes to incasure global SDG goals at project level.	
MISI Research Issue 2. Definition and evaluation of Project Success Dean et al., 2017; Schwanitz et al., 2017; and Diaz-Sarachanga et al., 2016; da Silva et al., 2019.	 The literature suggests that organisations do not have sufficient clarity on the definition of project success and often confuse outputs and outcomes when defining benefits. There is evidence that there is insufficient value placed by organisations in the longer-term and wider criteria across environmental and societal factors to balance their existing bias to economic and commercial drivers. This needs to be tested with CEOs and leaders of sustainability in organisations. There is a shortage of research into how organisations balance their 'strong rhetoric' for SDG measurement, with practical application of workable models. 	

	Derived Proposition:
	Proposition 2: Organisations' leaders champion SDG measurement without having evidence-based methods to verify claimed impact at portfolio and project levels.
<i>MISI Research Issue 3.</i> <i>Sustainable Construction</i> Goel et al., 2018; Munyasya and Chileshe, 2018.	 Is there a 'golden thread' that proves a linkage from the global SDGs, through organisational levels, down to local project level? What research across the engineering and infrastructure sector that has focused on measurement of sustainability success on projects can be used for assessment of MISI impact? What existing tools and processes (e.g. LEED, Envision, CEEQUAL) could be utilised for the assessment of MISI impact? Derived Proposition: Proposition 3: Current sustainability measurement at organisational and project levels can be used to demonstrate a 'golden thread' from 'global' SDG to 'local' portfolio and project levels.
MISI Research Issue 4. Practical application of Theoretical sustainability models Tranfield et al., 2003; Dushenko et al., 2018.	 The evidence from the literature highlights the inconsistent use of a Theory of Change as well as inconsistent use of a more balanced investment decisions across the Triple Bottom Line (TBL). There is a shortage of practical models that can apply MISI successfully. The gap identified in the literature indicates the opportunity for further research on strong theoretical or concept models that have relevance for the business and project management communities for the identification and measurement of benefits. Based on the empirical findings from answering Propositions 1 and 2, there is perceived value in development of a theory-led prototype model that can have practical value in unlocking the MISI challenges, such as being aligned to existing business case and benefits realisation models. Theory of Change research leaders and practitioners have developed a set of quality control standards for assessing the ToC (Connell and Kubisch, 1998, p.3): Plausibility; Feasibility and Testability (discussed in Sect 3.5.2). Derived Proposition: Proposition 4: A MISI prototype can provide a plausible, testable and achievable logic chain for defining project and portfolio SDG impacts.
<i>MISI Research Issue 5.</i> <i>Investment priorities in</i> <i>SDG</i> Aust et al., 2020.	 What are the MISI contextual issues (political, cultural, environmental and social) that might affect the investment decisions of future infrastructure projects? What are the likely outcomes of employing a well-defined theoretical MISI model to inform investment decisions? What investment criteria might inform decisions in sustainable infrastructure and how might the existing frameworks be utilised in future? Consideration for future research beyond this thesis: What are the MISI recommendations from this study that can inform a future research framework to inform improved infrastructure investment decisions?

2.12.1 Use of Propositions for Research Framework

Further structuring of the research investigation raises the question of selecting either hypotheses or propositions to guide the research. This study has used accepted wisdom (Merriam, et al., 2016) that hypotheses are primarily for quantitative analysis. Propositions can be described as a tentative and conjectural relationship between constructs that is stated in a declarative form (Merriam, et al., 2016). Propositions are generally derived based on logic (deduction) or empirical observations (induction). In this study they are derived deductively from the literature review. Merriam et al., (2016) suggest that propositions cannot be tested directly because they are derived from associations between abstract constructs. Instead, they are tested indirectly by examining the relationship between corresponding measures (variables) of those constructs. The identified variables are derived initially from the Literature Review's findings, then iteratively developed from each stage of the investigation, starting with the survey findings (Chapter 5), that informs the interviews (Chapter 6), that further uses its 'snowball' findings to inform the development of the prototype (Chapter 7). The variables for testing the propositions in the final prototype testing stage, are informed from the previous research. All variables are placed within the theoretical construct of the Realist Evaluation's Context-Mechanism-Outcome model that is introduced in Chapter 3.

2.13 SLR SUMMARY AND IMPLICATIONS

In the context of climate change as an existential threat to the human race, alongside the COVID-19-exacerbated threats of growing social and economic inequalities, rising social tensions, and mass migration (IPCC, 2020; UN, 2020), the international community has responded to the grand challenge of sustainable development with the '2030 Agenda for Sustainable Development', culminating in 17 Sustainable Development Goals, linked to 169 targets and 232 (unique) indicators. The IPCC has identified that "directing finance towards investment in infrastructure for mitigation and adaptation" is key to meeting SDG targets (2018) and the estimated USD \$94 trillion infrastructure investment that is required globally between 2018 and 2040 (Global Infrastructure Hub, 2019), represents a significant opportunity to stimulate economic prosperity, reduce poverty and raise standards in living, health, education and gender equality. This is relevant for the project management

community, a critical profession in the delivery of infrastructure through projects across all sectors, and thus in development.

As a result of the problem described above, this literature review has assessed the opportunities for research into the assessment of IP through SDG targets, distilled through 3 sub-research questions that were derived from the first three of the 8 Research Objectives⁶: (1) How extensive (broad and deep) has the research into MISI been to-date? (2) What issues and sub-issues were identified that might inform the MISI study's thematic framework to support the empirical data gathering in Chapters 5, 6, 8 and 9? (3) Post this doctoral MISI research, how could such a framework be used to provide guidance to a wider range of stakeholders? Question 1 was answered through a Level 1 analysis of an SLR, which identified seven MISI research findings. Questions 2 and 3 were answered using the Level 2 analysis of the 13 top articles that identified emerging MISI issues, sub-issues and supra-issues.

The SLR research findings identified the recurring need to make research accessible and relevant to the practitioner community by pursuing studies that will result in practical applications for theoretical sustainability models. This culminated in the identification of several findings to structure the next stage of the study: the importance of localising assessment; defining project success in light of the SDGs; expanding on sustainable construction research; driving practical assessment solutions and benefits for stakeholders; and prioritising investment into SDG assessment where it is most critical; both thematically in terms of specific SDGs, as well as geographically, with a renewed focus on developing countries where sustainability challenges abound. With a focus on IP, this SLR finds that SDGs are seldom linked to projects (either in delivery or in their outputs and outcomes) and it is suggested that increased knowledge in this area may improve both IP investment decisions and performance against SDGs.

This has provided an evidence-based framework of contextual issues and propositions for structuring the MISI research and empirical data gathering. In regard to the meaning for the research approach, it was concluded that, in order to develop an

⁶ Research Objective (RO)1: To understand the existing knowledge (in theory and practice) on how organisations and projects measure infrastructure projects' SDG Impact. RO2: To understand the context (the 'variables') of the current use of mechanisms to measure infrastructure projects' SDG Impact); RO3: To assess the current MISI mechanisms.

integrated framework to link global level SDGs with project level features and outputs, a number of propositions were needed to underpin the overall research question. This framework was used to answer the Research Objectives, through appropriate empirical studies, such as through a large-scale survey of practitioners engaged on IP delivery (Chapter 5) as well as through case study research (Chapter 9) on portfolio and projects that need to be linked to the achievement of SDG outcomes.

2.13.1 Derived Research Propositions

As mentioned earlier, Merriam et al., (2016) suggest that propositions cannot be tested directly because they are derived from associations between abstract constructs. Instead, they are tested indirectly by examining the relationship between corresponding measures (variables) of those constructs. These variables are derived from the next chapter's investigation of the theoretical framework. The five primary conclusions and four propositions, that were derived from the SLR, were:

- There is a MISI research gap. SDGs are seldom linked to projects (either in delivery or in their outputs and outcomes). Increased knowledge in this area may improve both IP investment decisions and performance against SDGs.
- The research gap includes the lack of empirical evidence that engineers do not have appropriate tools and processes to measure SDG impacts at project level. The SLR highlighted shortcomings at two main levels: (1) that the overall quantity of material on SDGs measurement at local project level was low, partly explained by the comparatively recent introduction of SDGs to the international lexicon (agreed in 2015 with a measurement framework not agreed until 2017), but also due to the suspected difficulty of cascading the global measurement to project level; (2) the inclusion of frameworks in the SLR papers highlighted that they were conceptual in nature and had reduced application in practice. In addition to these two reasons, the researcher's own personal experience and interface with leading members of the Institution of Civil Engineers (ICE), highlighted a knowledge gap on the availability of suitable tools and processes. This resulted in the author being commissioned by the ICE (co-sponsoring his doctoral research) to specifically review what tools were available and whether these were sufficient. This provided the primary official empirical paper that input to the ICE's 200th anniversary conference on SDGs in London in October 2018, attended by over

2000 delegates from 120 countries. The structure of the survey questions (shown in Chapter 5) that informed the SDG Conference paper specifically addressed this gap in knowledge and provided empirical evidence that over 2/3 of the engineering community surveyed, confirmed that the tools and processes were inadequate to measure SDG impacts successfully at project level.

- A framework for this MISI research has emerged. The findings from the SLR have provided an evidence-based logic to the formulation of the research design in Chapter 4. The framework has also informed the selection of which empirical data research is required and starts to formulate a strategy to achieve the MISI research's objectives.
- More empirical evidence is required for development of practical MISI approach. It is shown that further empirical evidence is required in order to develop an integrated framework to link global level SDGs with project level features and outputs. e.g. a large-scale survey of practitioners engaged on IP delivery (Chapter 5) that might include interviews with CEOs to access the organisational perspective (Chapter 6).
- A MISI prototype, with practical utility, could help address the gap if it is validated as a value-add approach in terms of time and effort to deploy in construction organisations. Such a framework should then be tested through appropriate case studies (Chapters 8 and 9). This could usefully include investigations at the portfolio, megaproject and project levels, to test whether they can utilise global SDG targets at the local project level.

Based on the research question of 'How can global SDG goals be used to define and measure infrastructure projects' SDG impact at organisational and local project levels?, the underpinning propositions, derived from the SLR, were:

- **Proposition 1**: Individual engineers are 'supportive but frustrated' in methods to measure global SDG goals at project level.
- **Proposition 2**: Organisations' leaders champion SDG measurement without having evidence-based methods to verify claimed impact at portfolio and project levels.

- **Proposition 3**: Current sustainability measurement at organisational and project levels can be used to demonstrate a 'golden thread' from 'global' SDG to 'local' portfolio and project levels.
- **Proposition 4**: A MISI prototype can provide a plausible, testable and achievable logic chain for defining project and portfolio SDG impacts.

These propositions will be used to inform the research design discussed in Chapter 4. The initial testing of the propositions in the survey (Chapter 6) will provide insights that can inform the design of the interviews (Chapter 7) in order to give a richer description of the phenomenon of interest and deepen the understanding into the research question. At the end of each chapter, a table will illustrate each propositions' evolving maturity based on analysis against the defined variables.

3.1 OUTLINE OF THIS CHAPTER

This chapter discusses the theoretical framework by first describing (Section 3.2 – Theoretical Lens) how the Critical Realism perspective of ideological philosophers such as Bhaskar (1978) informed the study's choice of the Realist Evaluation approach, championed by Pawson and Tilley (2004). This introduces the Context-Mechanism-Outcomes model that provides the overarching research framework for the study. The chapter then introduces the four theories that have provided the central core to the thesis. Of these, firstly (Section 3.3) it introduces the project organisational structure developed by Müller, Drouin, and Sankaran (2019a) who proposed a layered model of Organisational Project Management (OPM). This enables the structured evaluation of the concept of SDG, from global to local, at portfolio, mega-project and project levels. The chapter then provides analysis of three further theories adopted for the research, including the Theory of Change (Section 3.7). The four theories are all, directly or indirectly, embedded in the proposed prototype mechanism (developed in Chapter 7), called the Impact Value Chain (IVC).

3.2 THEORETICAL LENS

Saunders' research onion ring (Saunders et al., 2009), originally developed in 2007 (shown in Chapter 4 at Figure 31), provides a recognised framework to explain this study's theoretical and methodological position. This chapter deals primarily with Level 1, the philosophical stance, and the following chapter deals with the other five layers: approaches; strategies; choices; time horizon; and the techniques and procedures.

As regards the philosophical stance, the study recognises the separate characteristics of the key ontological and epistemological assumptions and positions, across a spectrum of positivism, realism and interpretivism, as described by Saunders (2009). However, these are boxes that do not always neatly fit with practical considerations of the theoretical frameworks. Some believe that this leads to a flawed dichotomy (Pawson and Tilley 1997; Vincent and O'Mahoney, 2018) between

objectivist (positivist, deductive, and empiricist) approaches, that have typically been used for quantitative empirical methods, and subjectivist (social constructionist, inductive, and interpretive) approaches, that are typically used with qualitative methods. This is not consistent with the proposed SDG phenomenon of study since no single approach covers the issues adequately. To overcome this identified restriction, the approach has sought a 'balanced dualism' (objectivism and subjectivism) (Vincent and O'Mahoney, 2018). This is helped by understanding the difference between ontology (what is real) and epistemology (what we know). Consistent with the Critical Realist school of thinking, this study accepts the reality of an objective (intransitive) world that has properties that can be studied using scientific means but strongly believes that most knowledge in the MISI area of study is a subjective, discursively bound (transitive) (Saunders et al., 2009) and within a constantly moving social construct. Crucially, for the purposes of this chapter, the methodological focus provides a rationale with which to build the overall research, as a link between the epistemological knowledge and reality of the ontological position. In doing so, it enables the researcher to build a stronger research framework, and thereby better understand the world of the SDG phenomenon. This does raise a quandary of the participants of the research, who themselves are often unaware of the structures and organisations in which they exist. As a result of these tensions, this study has chosen a philosophical position aligned with the Critical Realism (CR) perspective of philosophers such as Bhaskar (2013), since CR applies the earlier discussed 'balanced dualism' of objectivism and subjectivism. In subsequent sections, this will be described further as to how CR informed the choice of the Realist Evaluation (RE) approach, primarily because of its practical utility and its widespread use in social science research into the impacts of programmes (Linsley et al. 2015). It also provides a way to develop theory-led investigations, as described in the next chapter, which is what this research seeks to do on SDG measurement.

3.2.1 Critical Realist Ontology

The origin of critical realism, as a philosophy of science, is attributed to a series of books by Roy Bhaskar (1975, 1979, 1994). He argues that the universe, including the social world, is a 'stratified and open system of emergent entities' (Bhaskar and Lawson, 1998). These entities, shown by the three levels in Figure 17, are "things which 'make a difference' in their own right, rather than as mere sums of their parts"

(O'Mahoney and Vincent, 2014). Unlike 'variables' which only record or register (quantifiable) changes and do not offer causal explanations (Sayer, 1992), entities have causal power and properties which can generate real effects (Easton, 2010; Mingers et al., 2013). As shown in Figure 17, these entities can exist in two dimensions, either intransitive or transitive. For this study, they can be interpreted as potentially being physical, social, human, or conceptual entities (Easton, 2010). These provide the basic building blocks for the proposed MISI theoretical understanding and development of a new prototype, that through theory and observation, develop knowledge and insight about the MISI mechanism, through which action causes an outcome, dependent on the context (as shown in Figure 17). The multi-level nature of entities implies that entities can construct reality at different levels in both transitive and intransitive dimensions. Bhaskar (1978) illustrates this ontological position via his assumption of a stratified model of three interrelated domains of reality: the empirical, the actual and the real. As reality is a complex web of mechanisms and events reacting intra and inter levels, that is 'multiply determined' (Bhaskar 1975), multiple 'variables' must be teased out from detailed investigation of the context. It follows that the critical realist approach to the MISI research is that it seeks to find causation and impact at deeper levels than previously known.

The three levels of CR



Realism's three levels (1975, 2002, 2008)



The three domains that Bhaskar (1978) describes (shown above) as the 'empirical', the 'actual' and the 'real', provide an insight into the qualitative study approach. The empirical is typically what is perceived to be 'fact' such as what we observe and hear. This is distinct from the 'actual' event that happens at a specific

time and space that may be different to what we have perceived to be the case. This is further deepened into the 'real' dimension that is constituted of the mechanisms and structures which generate and explain events. Through understanding these stratified layers, the researcher is better able to accept that perceived reality is not the actual reality. This can shape the methodology so that an understanding is explicit that the observer can only attribute meaning to what empirical evidence is provided and through this interpretation, attempt to plausibly define a deeper meaning that more closely aligns to the 'real'. Bhaskar (1975), suggests that multiple causes must be evolved from detailed explorations of the context. The understanding of the context in which the mechanism operates provides a better causal model to understand the outcomes in the actual and potentially, provide insights into the 'real' world/s. As a result, the critical realist researcher seeks a deeper level of understanding beyond direct observation, that affect the outcomes. Therefore, the task of the study and the approach taken by the researcher, was to identify a causally meaningful, testable, and reliable explanation for observed MISI events, or patterns of events, via the selection of an approach that recognised the three domains and also provided a way of distilling the powers, entities, mechanisms and relations which created them. This led to the selection of the C-M-O model discussed below.

3.2.2 Realist Evaluation's Context–Mechanism–Outcome configuration

The Realist Evaluation's Context–Mechanism–Outcome (C–M–O) configuration (Pawson and Tilley, 1997) is widely used across clinical research (Pawson et al., 2005) and increasingly also across the social sciences (Linsley et al., 2015). Pawson and Tilley specifically recommend the C–M–O strategy so that 'programme theories can be tested for the purposes of refining them' (2005, p. 2). In this regard, the investigation is not about what works but asks instead, 'what works for whom in what circumstances and in what respects, how?' (2005, p. 2). Therefore, this approach gives a multi-layered methodological framework for analysing engineers' perception of the context of SDG measurement as well as its potential outcome on redefining investment decisions to achieve broader SDG impacts. For the purposes of the thesis, the definitions of C–M–O are:

• **Context**: The conditions in a context of action encompass 'material resources and social structures, including the conventions, rules and systems of meaning in terms of which reasons are formulated' (Sayer 1992, p. 112; in Easton, 2010).

- Mechanism: The underlying entities, processes, or structures which operate in particular contexts to generate outcomes of interest (Astbury and Leeuw, 2010, p. 386).
- **Outcome**: The practical effects produced by causal mechanisms being triggered in a given context (Tilley 2016, p. 145).

The conceptual interpretation of the C-M-O model, from a Realist Evaluation interpretation of the Critical Realist approach, is shown below.



Critical Realism

Figure 18. The Realist Evaluation (Context–Mechanism–Outcome (C–M–O)) model mapped to the Critical Realism theory of real-actual-empirical adapted from Zachariadis et al. (2013).

In the engineering research field, the C-M-O configuration is not as widely used as it is in the clinical field, and therefore the choice of this approach has been compared with other relevant applications on engineering projects and specifically, what design features can be drawn from them for the MISI study, as detailed in Table 6.

#	C–M–O Configuration Reference	How does this inform this study's MISI analysis?
1	Terminology . Pawson and Tilley's Realistic Evaluation (Pawson and Tilley, 1997) is widely held as the originators of the Realist Evaluation CMO configuration.	Through understanding the origins of the C–M–O approach and its terminology, research into engineering projects can build on established protocols and use their approach to widen our understanding of its employability outside the clinical and educational sectors, where they are most frequently used.
2	Projects application . Pawson and Tilley's Realistic Evaluation (Pawson and Tilley, 1997)	Pawson and Tilley suggest that the value of the C–M–O strategy is that it enables the researcher to better analyse the nature of programmes and projects and, more importantly, how they work. Thus, the core element of the realist approach is to provide a new perspective on how intervention using a mechanism brings about outcomes that represent change. Engineering-based research can thus adopt Pawson and Tilley's approach to

Table 6: Examples of the use of Critical Realism and Realist Evaluation C–M–O configuration on engineering projects that informs similar use for MISI research.

		better understand and explore the mechanism of change
		in order to evaluate a project or programme.
3	Engineering application . Tilley (2016) developed the C–M–O model to assess how it can be used by engineers to improve their decision making for policy and project decisions.	As a proponent of realism, Tilley argues for a pragmatic approach in engineering to be adopted for its evaluations. The research adapts the C–M–O model to the EMMIEI approach that includes Effects, Mechanism, Moderator (or context), Implementation and Economic Impact. Importantly, the differences between engineering physical worlds and the social world are recognised, but it is suggested that both benefit from a pragmatic research strategy. This is helpful for the measurement of SDGs because it gives confidence of relevance to the engineering domain of using the C–M– O configuration
	Engineering application.	The project evaluation of 'e-government for You' used
4	Horrocks and Budd (2015) used the C–M–O structure to evaluate a European e-services systems engineering project to establish the outcomes and understand the why, for whom, and how?	the theory-driven evaluation approach based on the C– M–O model to enhance the focus and granularity for their study. This supports the usage of the C–M–O model for SDG measurement because it allows for the mixed-method approach and structures the analysis framework in a readily understood causal chain.
	Construction project	
5	application . Peters et al. (2013) examined Critical Realism evaluation models to study business networks. They used a UK construction project to explore the managerial phenomenon, specifically the practice of novation in temporary organisational networks.	The origins of the C–M–O model are from the Critical Realism traditions, and therefore, the Peters et al. (2013) article provides a useful insight into the approach that a strongly theoretical lens can use when applied to the ground level of a local construction project. This helps shape the SDG approach by giving confidence that the 'realistic learning' from the Peters study can be replicated for SDGs.
	Construction project	
6	application . Poirier et al. (2016) evaluated the use of a Critical Realism lens to assess the delivery of building projects in the architecture, engineering and construction (AEC) sector. They evaluated how collaboration can improve performance and value across five core entities, namely process, structures, agents, artefacts and context.	There are strong parallels of the study into the AEC's value derived from collaboration in project delivery. The relevance to the C–M–O approach is that they both derive from the critical realism tradition and seek to understand causal patterns and assess what are the outcomes of employing a specific mechanism within a given context. In particular, the study highlights how the learnings can be structured in a way that is most readily understood by practitioners in an area of great complexity.
	Multi-sector application.	
7	Annough Bergeron and Gaboury (2020) come from a clinical care perspective, their recent study highlights some of the challenges of using the C–M–O framework that has relevance to all disciplines and sectors that use its causal framework. Further challenges were also identified in the article described below, by Crosthwaite.	There are a number of methodological challenges that are identified and should be noted by engineering researchers using this approach. Solutions to the analytic difficulties are shared that can help the identification of patterns and assist the researcher to maintain transparency in the analytical process, thus strengthening the ability to make generalisations.
8	Education application . Crosthwaite et al. (2012) used the C–M–O approach to understand the educational impacts of the	While not focused on an engineering project, this article highlights the use of the model in the education sector to evaluate the causal impact based on the identification of specific issues within the C–M–O framework. In regard to the SDG approach, there is value in combining the
	'Engineers without Borders' in	qualitative and quantitative approaches to improve the
----	-----------------------------------	---
	Australia and New Zealand.	understand the observed outcomes.
	Health system application.	
	Greenhalgh et al. (2009) used the	The broad nature of evaluating organisational systemic
	C–M–O method to evaluate a	transformation has similarities of complexity with
	health system. This wide-ranging	research into the measurement of SDG impacts at project
9	analysis of a systems	and organisational level. The simpler C-M-O approach
	transformation environment	offers a means to help explain causal effects more
	sought to understand the reasons	simply, and this has benefits for both research and
	for how and why the outcomes	practitioners.
	were achieved.	
	Projects application. Berge	
10	(2017) used a realist evaluation	Realist evaluation is used to scrutinize what it is about
	approach on a Norwegian telecare	the telecare system that works for whom, why, how, and
	project. While the study has a	in which circumstances. The study provides a more
	clinical orientation, its project	nuanced approach.
	approach is instructive.	

The analysis provided above in Table 6 confirms that the C-M-O configuration does have relevance in the engineering research field, and therefore it is taken forward as the proposed theoretical research framework, and developed further in the next chapter, where it provides the guidance and structure to apply a mixed methods approach. The next sections provide analysis of the primary theories adopted for the research, including:

- The layered model of Organisational Project Management (OPM) proposed by Müller, Drouin, and Sankaran (2019a) in Section 3.3.
- Theory of Change in Section 3.4 and 3.5.
- Triple Bottom Line in Section 3.6.
- Creating Shared Value in Section 3.7.

3.3 LAYERED MODEL OF ORGANISATIONAL PROJECT MANAGEMENT (OPM)

This section establishes the organisational boundaries that are central to the research area and also builds on the 7 contextual variables from the SLR in Chapter 2, by extending them to nine contextual variables that provide a 'lens' for the research design. Given the challenges of cascading global and national SDG targets to the local level identified in Chapter 2, the research took an organisational-centric perspective to better understand the 'global-to-local' context. To remain consistent with the most recent developments in academic literature, the study used the Onion Model proposed by Müller in his book 'Organizational Project Management' (Müller, Drouin and

Sankaran, 2019). Their work offers a seven-layered model that provides 'logical cohesion' and 'logical adhesion' inter and intra layers. It also provided an adaptable framework and a common terminology to distinguish the structural units to be investigated. In this case, the study focused on the three layers of: Business Integration; Organisational Integration; and Project Governance, shown in Figure 19. Within these layers are located the units of study, both contextually and organisationally. This provided a portfolio perspective that could cascade an agreed organisational approach on SDG measurement to its projects and programmes within its portfolio. Finally, the Project Governance layer is highlighted because much of the measurement of SDG impacts is done at this layer and therefore it was important to consider how the SDG measurement approach would be delivered at the project 'coalface'.

Leveraging the analysis from the systematic literature review, shown in Figure 15 and expanded in Appendix 2, that identified a nodal structure in column 'b' and the context variables in column 'c', these issues and variables have been applied to the Müller OPM Onion Model (Müller et al., 2019). This provided the link to shape the evolving evidence based MISI research framework, shown below.



Figure 19: Adaption of the OPM onion model (Müller, Drouin, Sankaran, 2019) showing: three SDG measurement domains: portfolio prioritisation (yellow), local implementation (blue) and project measurement governance (green), mapped to Muller's seven levels.

There were nine contextual variables identified within the OPM model that were included in the study. Seven of these variables were derived from the SLR's findings (highlighted in red text in Figure 19) with the additional two issues being: level of organisational complexity (the three levels shown in the Organisational Integration onion ring); and the 'Type of Organisation' that sits within the Organisational Philosophy layer. The MISI relevance of these nine variables, is shown in Table 7 below.

Layer	Context Variables (9)	Relevance to the research question: 'How can global SDG goals be used to define and measure infrastructure
Organisational Philosophy	from SLR and OPM Type (complexity) of Organisation (its business nature)	The characteristics of the organisation determines its potential SDG approach. This includes: its presence in the marketplace (local, regional or global); its interaction with its partners, suppliers and customers (do they require SDG measurement, such as in UK on Net Zero 2050 targets); sector – private or public (eg using the updated Green Book (2018) that requires alignment with SDG impacts) and also across infrastructure types such as energy, water, etc. Muller et al (2019) describe this as also including whether the organisation is project based (all work is done in projects), project oriented (almost all work is done in projects). Also see Miterev et al., 2017: Soderlund, 2004.
Business Integration	Portfolio Strategic Alignment	The business opportunities are identified, prioritised and selected. The alignment is both internally and with external stakeholders, such as for the Environment Agency, with local communities that could harness the SDG as a common definition of goals.
	Portfolio Transformation Tempo Optimization	The oversight of the portfolio office also provides the function of monitoring the overall transformation tempo of the organisation. If the SDG measurement is excessive, there is liable to be a reduced performance due to 'change overload'.
	Business Case & Benefits Realisation	The management of the portfolio benefits being aligned to SDG impacts provides the link between organisational strategy and delivery success based on business case assumptions. (Killen and Drouin, 2017; Keeys & Huemann, 2017; Marnewick, 2016)
Organisational Integration	The level of the portfolio- programme- project being investigated	Muller et al (2019) consider this layer to be how it addresses the form of organisational integration of project-related work, which has a strong influence on the way project SDG governance is performed. Eg is it a mega-project which imposes other forms of self-governance since they are often temporary self-governing organisations. As such the megaproject typically identified by its scale (over £1Bn), timeline (often over 10 years), complexity and number of sub-projects and suppliers (Flyvbjerg, 2014), often as their own Special Purpose Vehicle and legal entity (Sainati et al., 2017), such as with Thames Tideway project.
Project Governance	Leadership	Leadership is known to be critical to success, both at the project, portfolio and organisational levels. The success of introducing innovative new methods of SDG measurement, which is complicated, requires clarity of direction and allocation of the right authorities to a high performing team. (Davies and Hobday, 2005; Davies et al., 2009)
	Knowledge of SDG and success definition	Knowledge is a critical contextual issue because it enables informed consideration of how to maximise the SDG measurement opportunity. The ability to overlay the new SDG approach needs to be based on a good understanding of both the SDGs and the traditional definition of project success (see links to Benefits Realisation above) (Hwang and Ng, 2013)

Table 7: Summary of contextual variables across organisational boundaries using the Muller Onion Model (Müller, et al., 2019).

Tools, systems, process maturity	The Muller onion diagram included a strong link between roles, policies, relations and methodology in the project governance layer. The methodology includes the tools, systems and process maturity that are both in existence and need to be adapted if the introduction of CMM-type maturity model for SDG measurement is to be successful. (Sun et al., 2009)
Lifecycle Phase	The stage of the project impacts the use of the new approach. Eg the use of the Thames Tideway Tunnel megaproject and the Boston Barrier project were both midway through their delivery stage. The case study seeks to assess whether it can be used for both new and 'in-flight' projects. (Martens and Carvalho, 2017; Okland, 2015; Silvius et al., 2012)

3.3.1 Identification and proposed usage of the 9 'Variables Framework' issues

The 9 contextual variables identified in Table 7 above, in future referred to as the 'Variables Framework', have been derived jointly from the SLR's nodal structure of context variables (shown in Figure 15) and the Müller OPM Onion Model (Müller et al., 2019). This informed the evidence based MISI research framework, shown below that will be taken forward as the 'Variables Framework' for investigating the propositions, since it is suggested (Merriam et al., 2016) that propositions cannot be tested directly because they are derived from associations between abstract constructs. Instead, they are tested indirectly by examining the relationship between corresponding measures (variables) of those constructs, by gathering empirical evidence from the surveys and the interviews. They will be iterated through derived learnings at each stage of the investigation to inform the proposition analysis, thereby meeting the research objectives and answering the research question.

3.3.2 The hierarchy/levels between project-organisation to the SDG Global Goals

When considering sustainability and SDG reporting at project level (Mansell, 2020a), there are two core 'project lifecycle' questions that have SDG related impacts: (1) In project delivery, how does the design and construction of the project impact on the societal and environmental status quo (e.g. what is the impact on air and water quality during construction)?; and (2) What does the completed project do for the community (e.g. by how much does the waste water treatment plant improve sanitation)? These two parts are core to understanding the measurement of sustainable development at project and SDG levels, as shown in Figure 20 below. The first question is focused on the delivery phases and is tactical in nature, while the second seeks to define the longer-term outcomes and impacts, that are more strategic in orientation.

 _				
 n	C	tι	C	nı
u	c	u	L	uı



Question 1 – Delivery Phase 'Doing projects right'

Question 2 – Post-Delivery 'Doing the right projects'

1) In project delivery, how does the design and construction of the project impact on the economic, societal and environmental status quo eg what is the impact on air and water quality during construction?

2) What does the completed project do for the community eg waste water treatment plant improves sanitation, thereby improving health & hygiene?



Figure 20: Framework for sustainability and project success reporting depicting the two core sustainable development questions at project level.

A further dimension that aids understanding of SDG impacts on infrastructure

projects is the hierarchy, or levels, of SDG reporting as shown in Figure 21.



Figure 21: SDG Hierarchy of SDG target reporting using Impact value Chain (IVC) outcomes and impact causal chain⁷.

⁷ The diagram was designed solely by the author but later used extensively by Buildings Research Establishment to explain how they link project sustainability reporting to SDG impacts. See *The Built Environment and Future Sustainability - The relationship between BREEAM and the Sustainable Development Goals*, P6, Figure (BRE, 2020) accessed at https://files.bregroup.com/breeam/sdg/BREEAM SDB A4%20 BRE 115430 0720 web.pdf.

The diagram in Figure 21 highlights three issues: there are two core perspectives at the project level – during project delivery and post-delivery (the linear relationship shown at the lowest point of the cascade) (as shown in Figure 20); there are different reporting requirements at each of the levels (depicted by the numbers 1-3 in Figure 21) from global/national targets, down to project level targets; and the targets at each level are dependent on the context of the social, political, economic and type of national economic infrastructure category.

3.4 PROJECT SUCCESS THEORETICAL FRAMEWORK: CORE THEORIES OF THE THEORY OF CHANGE AND THE TRIPLE BOTTOM LINE

3.4.1 Theory evolvement and selection

The study provided an opportunity to harness the researcher's cross disciplinary background from his completion of masters-levels research (Strategic Studies at KCL; International Relations at Cambridge University and Major Programmes at University of Oxford). These research opportunities introduced strategic theories that, in separate ways, provided frameworks that all helped define delivery success, both at global, national, organisational and project level. These levels represent the 'global to local' paradigm that the study aimed to explore. The five theoretical areas considered are discussed below.

• Global-to-national Strategic Theories. The consideration of political-military strategic decision making was informed by Sun Tzu's 'The Art of War' (540BC) and Clausewitz's 'On War' (1812). Both of these have heavily influenced more recent political-military strategic thinking, especially in the extracted principles that many authors have used to propose modern principles for strategic decision making (Handel, 1986; McNeilly, 2003 and 2012). The relevance of these to the development of 'the Centre of Gravity' and 'Ends, Ways Means' theories are reflected in modern political-military strategic theories, including the Weinberger Doctrine and its successor, the Powell Doctrine. These latter two theories placed high importance on ensuring that the 'ends' were well understood before the 'means' (human, political, financial and other capitals) were committed, and also before the 'ways' (operational tactics) were designed to achieve the 'ends' with the available 'means'.

- Management Theories. The plethora of management theories that the twentieth and twenty-first centuries have spawned provides a rich resource to identify relevant theories for helping analyse the measurement of SDG success from global to local levels. There are many dimensions to this issue and one of the clearest frameworks is provided by Peter Drucker (1995, 2020) in 'People and performance: The best of Peter Drucker on management'. Drucker also updates this (2020) in 'The essential Drucker' where the interrelation between management, the individual and society are explained. Drucker's 'Management by Objectives' (1995) provides a useful framework to analyse the definition and measurement of the objectives, which is central to the study on SDG measurement. Drucker's work also influenced the subsequent work by Kaplan and Norton (1996) that introduced the balanced scorecard to help structure and focus the measurement of success criteria for modern businesses, in effect, focusing more on future indicators such as innovation, skills and capability and the leverage of technology, and less on the traditional emphasis on past economic and financial indicators.
- **Programme Theory**. There are many programme theories that are discussed later from academics such as Peter Morris and Andrew Davies. Their work has informed more recent work on mega-projects and other mechanisms to consider and address grand challenges that are often at the global level, as well as managing national and regionally important project management activities. The SDGs straddle the global to local levels and seek to address many of the Grand Challenges associated with the 17 Global Goals, such as Climate Change (SDG 13).
- **Transition Pathways.** Transition theories, such as by Frank Geels (2004, 2007), provide relevant insights for sustainability in project management. Geels applies systems thinking to better understand the multi-dimensional nature of agency models and how they impact socio-technical system changes. The transition pathway, whilst not fully linear, does align with evolution theories discussed below, which leads to the selection of ToC as the primary theory of research.
- **Evaluation Theories**. There have been a number of theories that link the management theories of academic leaders such as Drucker, with local

implementation. The most prominent of these researchers has been Rossi and Weis (1995) who developed the 'New Approach to Evaluation' (1995). This heralded a new theory subsequently titled the 'Theory of Change', discussed in more detail later in this chapter.

Having evaluated relevant theories related to the study topic of MISI, the study narrowed its focus to the Theory of Change (ToC) because it provided three separate, but mutually reinforcing, theoretical selection criteria listed by Merriam (Merriam and Grenier, 2019) in her work on Qualitative Research.

- Appropriateness. ToC is widely used across the international development sectors and across levels of organisations from global level (the United Nations and World Bank) down through organisations and directly into the projects that deliver the strategic objectives set at higher levels. This is relevant to the MISI study that seeks to identify and harness a model that can demonstrate a 'golden thread' from global to local levels that has utility for practitioners.
- Ease of application. The author has used the ToC widely during his work for the UK government on mega-projects as well as international projects on behalf of the United Nations. This has given him first-hand knowledge of the ease of application as well as the value it can bring. Conversely, the researcher also recognises the ToC's shortcomings, which provides a useful insight into how existing theories might be improved.
- Explanatory power. The power of the causal logic model is self-evident by the wide usage by global organisations such as the UN and World Bank. One indication of the ToC's explanatory power is evidence that it provides a powerful storyline for inexperienced staff in developing countries who are working on projects and who might otherwise not have the formal project training that includes understanding the benefits theories that define impacts and outcomes. The ToC is a theory that can be 'told as a story' which is visually powerful. This helps the research into MISI since the visual illustration of key concepts is critical to making complex interrelationships more digestible to stakeholders and therefore more helpful to align what success means and how it can be measured. The causal logic model is also embedded in core project management theories and practices, such as benefits management, which is directly linked to MISI research.

3.4.2 Project success criteria and output-outcome success criteria

One of the contextual issues identified in Figure 19 is the need for building knowledge on SDGs and the definition of success. This sub-section examines the study of project success, specifically aligned to the logic chain of the output-outcome pathway. This provides the link to the description of the two core theories that underpin the analytical mechanism that is investigated in the case study.

While project success is a heavily researched field of study within the field of project management (Thiry, 2004; Sward, 2006; Jenner, 2010; Müller and Judgev, 2012; Joslin and Müller, 2016), the quantitative analysis of success criteria and their alignment to outputs or outcomes, is less evident. This becomes more problematic if project success is extended to link to the SDGs because they are by definition, global goals, not designed for the local level. Recent studies (Joslin and Müller, 2016) have shown that forty years of research into project success factors and criteria have indicated that project success is the achievement of a particular combination of objective and subjective measures assessed at the project's end. Other research into project success definition (Thiry, 2004; Jenner, 2010; Bradley, 2010a and 2010b; Lavagnon, 2009) has consistently identified benefits and outcomes as being a critical determinant for the assessment of project success. For example, Michael Thiry (2004) highlights that 'too many critical success factors are related to inputs and management processes and not enough on outcomes'. This is further supported by those (Davies, 2000; Morris, 2013; Terry Cooke-Davies, 2002, 2007) who identify three levels of success criteria: project management success – was the project done right?; project success – was the right project done?; and consistent project success – were the projects done right, time after time? The latter two levels of success are especially relevant at the portfolio level and provides a valuable approach to linking project success with SDG impacts. In other words, did the chosen project deliver longer term impacts that align with SDGs and can these be aligned with the existing benefits realisation approach?

In order to understand the limitations of defining project success in the narrower method (by time and cost), it is necessary to understand the profession of project management that at its core, is a discipline that focuses on the initiation, delivery and completion that often transitions into operations with the initiation, development and delivery of projects (Morris, 2017; Davies, 2017; Atkinson 1999; Lundin and

Soderholm 1995; Davies and Hobday, 2005). Simply stated, this means that there is the delivery of outputs during the implementation stage, but the most important part, the delivery of outcomes, is achieved post project, and it is these outcomes that indicate where most value from the investment are achieved (Morris, 2017; Davies, 2017). As a result, there is a growing need for the project management sector and profession to focus more on 'ends' rather than just the 'means'. The 'ends-ways-means' model from strategic politico-military research (Lykke, 1982; Sen, 2013) is a helpful way to imagine the high-level success definition as shown below in Figure 22. This also relates to projects creating value, which many have argued (Brady, Davies & Gann, 2005), is the need to deliver integrated solutions, especially when considering the delivery of complex capital goods, such as infrastructure. This is important for the research because it provides a simple model to re-vision how SDG measurement can be applied to existing project success approaches that too often had previously focused on the 'means and ways' instead of the 'ends'.





So, given the challenges of defining project success, a new method is needed to align definition of success with the delivery of SDG targets. The Triple Bottom Line and Theory of Change provide a normative and causal theory approach, that combined provide a route map to align with the 'ends-ways-means' model.

3.5 THEORY OF CHANGE

The 'Ends-Ways-Means' model is most commonly identified within the Theory of Change (ToC) (Weiss, 1983; 1995). The ToC is widely used across many academic disciplines, including environmental and organisational psychology, but it has also increasingly been connected to sociology and political science (Rossi, 2018). ToC emerged from the field of programme theory and programme evaluation in the mid1990s as a new way of analysing the theories motivating programmes and initiatives working for social and political change. It is focused not just on generating knowledge about whether a programme is effective but also on explaining what methods it uses to be effective. The original work in the 1980s has been developed further by the work of notable methodologists, such as Huey Chen's work on theory-driven evaluations (1990, 1992), Peter Rossi's systematic approach to theory-driven evaluation in social sciences (2018), Michael Patton's focus on integrating the theory with practice (1986, 2014) and Carol Weiss' seminal work that takes a stakeholder-centric perspective (1972, 1983, 1995, 2005) to find more effective ways of evaluating complex community programmes.

Weiss suggests (1995) that complex community programmes had not sufficiently aligned local stakeholders on the change process and what the outcomes will be. She noted that the logic chains are particularly weak in the midsection of the causal chain, without which the longer-term goals are weakened. Weiss uses the term "Theory of Change" to describe the causal links across the inputs–outputs–outcomes pathway. She also focused attention on what users could claim in terms of impacts, separating claims of "attribution" from a wider, less direct, "contribution". Based on her work (1983, 1995), ToC has been applied extensively across international development, public health and human rights and has since become a central theory that underpins the approach to project benefits management (APM, 2018; PMI 2017).

3.5.1 The Causal Link: Outputs-Outcomes-Impacts

For the purposes of this research, the ToC needs to be unpacked further by better understanding the key components of: outputs, outcomes and impact. These are the core terms used to describe long-term, sustainable change in people's lives. However, although the terms are in common use, there are a number of problematic issues with their use. This is critical to MISI because, as discussed earlier, there is evidence that this causal chain has not been established at the sub-national levels. Most organisations recognise the difference between the 'things they do' (activities) and the intended effects/change they want to achieve (impact). But the distinction is not always helpful. In order to achieve desired long-term changes, there may be many steps between an organisation's activities and the desired impact. The MISI research is attempting to find ways to establish a SDG value creation logic. In the results chain (as shown later Figure 24), inputs are used in order to carry out activities. Activities lead to services or products delivered (outputs). The outputs start to bring about change (outcomes) and eventually this will (hopefully) contribute to the impact. However, there are a number of ToC challenges that impact this research:

- Many users have difficulty differentiating between activities and outputs, outputs and outcomes, and outcomes and impacts. Thus, the gathering of empirical evidence might have misuse and inconsistent interpretation of these key words.
- The reality that most outcomes are delivered after completion of a project and therefore do not get measured. This reduces likelihood of defining SDG goals accurately since they rarely get measured and also, the lessons of underperformance are unlikely to be captured. This links strongly to project's benefits management knowledge and shares the same difficulties (Edmondson and Nembhard, 2009).
- Logic flow of the map design. There are different opinions on the most logical way to build a map. There are three common approaches: from outputs, to outcomes to impact (left to right); outcomes to impact and outcomes to outputs (inside to outside); and, impact to outcomes to outputs (right to left). This study, as discussed above is firmly in the 'right to left' position since it aligns with the 'ends-ways-means' model.
- Even if a project or programme has a good monitoring and evaluation system it will not make up for poor project or programme design (although it may help to show up these weaknesses). If SDG objectives are unclear or poorly designed from the start, then monitoring and evaluation becomes much more complex and difficult. 'A meaningful plan for monitoring and evaluation can only exist in relation to clearly defined objectives and strategies' (Okali et al., 1994).

3.5.2 Testing Quality of ToC

As the uptake of ToC increased, there was a perceived need to formalise its foundation and provide a framework for further development. This led to research leaders and practitioners developing a set of quality control standards for assessing the ToC (Connell and Kubisch, 1998, p.3). These are: Plausibility; Feasibility and Testability. For this study, they are defined as: (1) Plausible. Does evidence and

common sense suggest that the activities, if implemented, will lead to desired outcomes? (2) Achievable/doable. Will the economic, technical, political, institutional, and human resources be available to carry out the initiative? (3) Testable. Is the theory of change specific and complete enough for an evaluator to track its progress in credible and useful ways? A further key ingredient stressed by O'Connor (1995) was the need to better understand the 'context' of both the internal and external environments, in order to predict 'outcomes'. The three ToC quality control criteria and the focus on context and outcomes (as part of the Realist Evaluation mixed method approach) are described more fully in the methodology section.

3.6 TRIPLE BOTTOM LINE

At the project level, the Association of Project Management's Body of Knowledge (APM, 2019) defines sustainability as 'an environmental, social and economically integrated approach to development that meets present needs without compromising the environment for future generations'. The APM's definition has been based on the modern concept of sustainable development as derived from the Brundtland Report (Brundtland, 1987), which suggests that efforts to create improvements in the short-term should be without a negative impact in the longer-term. It also recognises that project strategies need to consider success against the triple bottom line (or otherwise noted as TBL or 3BL) of social, environmental (or ecological) and financial aspects.



Figure 23: The TBL view of economy, environment and social translated into the doughnut view alignment of the United Nations' 17 Sustainable Development Goals by Johan Rockström (2016) (graphic usage agreed).

A contribution to the growing literature on the measurement of infrastructure projects on sustainability is provided by Shen, et al. (2010), who focus on the balance needed between benefits to society whilst protecting the environment and still

achieving the economic benefits envisaged in the project business case. The linkage across the three areas in the construction industry is further defined by Kibert (2013), who suggests that the interrelationship between a project's outputs and the society that is impacted is a central component of defining the sustainability success of an infrastructure project. This introduces the concept that project success definition needs to consider success against the triple bottom line (TBL) (Elkington, 1994) of social, environmental (or ecological) and economic (or financial) effects, otherwise noted as the "three pillars" concept of "people, profit and the planet" (Elkington, 1994, 2013, 2018; Griggs et al, 2013). However, the overemphasis on the last of the TBL criteria, namely finance, brings us to the root of the problem of measuring projects' SDG impact (Martens et al., 2016).

The crux of the problem in sustainability reporting lies with the dominance of accounting tools, which has been the pre-eminent business method of reporting for over 500 years, since Luca Paccioli first published his papers on double-entry bookkeeping (Yamey, 1949). This has largely remained unchanged. In other words, there has been a proliferation of mechanisms and economic models to track different elements of the TBL, including ESG (environmental, social and governance) (Elkington, 1994, 2018) that includes the three core areas used in the business investments measurement of ethical and sustainability impacts of a company; Social Return on Investment (SROI) (Emerson et al., 2000; Millar and Hall, 2013), Net Positive (Forum for the Future, 2018; Rainey et al., 2015), Double and Quadruple Bottom Lines (Sawaf and Gabrielle, 2014), a myriad of capital (human, social, manufactured, financial, natural) analysis models, Environmental Full Cost Accounting (Schaltegger and Burritt, 2000), Boston Consulting Group's Total Societal Impact framework, Integrated Reporting (Eccles and Krzus, 2010), Blended and Shared Value (Bonini and Emerson, 2005) and Impact Investment (Bugg-Levine and Emerson, 2011). Recently this has been extended to new frameworks that focus on specific issues such as Sharing and Circular Economies (Preston, 2012), Carbon Productivity (Malhi et al., 2009; Suess, 1980) and Biomimicry (Elkington, 2018). The contention of this research study is that the proliferation of financially driven sustainability measurement theories, tools and concepts causes confusion and often leads to sub-optimal governance because of the short-term focus on profit instead of wider TBL outcomes.

Current analysis suggests that the TBL framework is in need of 'rethinking' (Elkington, 2018). Indeed, Elkington's contention is that his definition has not been implemented according to its true meaning, and he insists that businesses should now monitor and report economic (not just financial), social, and environmental value added—or they will become negatively impacted. Many contend (Hubbard, 2009; Elkington, 1994, 2018; Joyce and Paquin, 2016) that private sector success is still overly influenced by financial perspectives. This is often restrictively linked to share price value and viewed as an inherent weakness of the system that drives short-termism in governance and decision-making (Elkington, 2018). As a result, and relevant to the assessment of how project managers can measure projects' SDG impacts, there has been a growing demand for knowledge on how sustainability reporting can be broadened.

As a result of the increased knowledge and tempo of the uptake of sustainability language, it has become more mainstream with many academics (Tilt, 2009) and practitioners (Perrini and Tencati, 2006) seeking to further develop the current accounting-centric method towards a broader, or holistic, approach, such as the Balanced Scorecard (Kaplan and Norton, 1996). However, the proliferation of sustainability accounting terminology (sustainability accounting is often referred as social accounting, corporate social reporting corporate social responsibility reporting, social and environmental accounting, and non-financial reporting) also negatively impacts the ability to have a single consistent view, and this contributes to the project world being mired in confusion.

3.7 CREATING SHARED VALUE

The debate on businesses' responsibility to 'give back' to society and the environment have evolved from earlier notions associated with corporate social responsibility (CSR) agenda, and these in turn have been influenced by the notion of creating shared value (CSV), which seeks to change the way in which business creates value for its stakeholders (Porter and Kramer, 2011), i.e., through placing socially aware priorities, such as the SDGs, at the heart of core business' thinking and strategies. In Porter's paper (Porter and Kramer, 2011), he states that 'the concept of shared value can be defined as policies and operating practices that enhance the competitiveness of a company while simultaneously advancing the economic and

social conditions in the communities in which it operates'. CSR is clearly a key part of this idea of 'give back to society' and is of importance to investors (Hayward et al., 2013). Hayward goes further by suggesting that the major problem is with businesses, especially the ones that have remained caught in outdated approaches to value creation, that they view too narrowly. In effect, they are seeking shareholder-driven short-term profits before longer-term broader outcome success. Most of the current sustainability efforts have focused on the identification of harms to society in general and the creation of corporate responses to meet those harms. As a result, many sustainability efforts have been largely divorced from the specific business model of each organisation. In reality, sustainability activities have often functioned as 'surplus', 'add-on', 'nice-to-have' actions for the purpose of deflecting stakeholder criticism, conducted regardless of their actual relevance and impact on the business' strategy, capabilities, suppliers or customers. The net effect is to leave core business activities and risks unchanged. Moreover, sustainability cannot be delivered through CSR because it is both inefficient and ineffective (Porter and Kramer, 2011). Porter argues that CSR is inefficient because it can create irrelevant 'add-on' activities that add to the costs of doing business without either adding to the real value created for any of the business' stakeholders or removing real business risks. Additionally, CSR is ineffective because it continues to pit society and business as opposing forces rather than recognising the opportunities dormant in their interdependence (Porter and Kramer, 2011). However, counter to Porter's view of CSV–CSR differentiation, there is an alternative perspective that CSR has rapidly evolved in recent years and closed the CSV-CSR gap. In a recent paper (Albuquerque et al., 2019), they explore how CSR affects systemic risk and impacts the value proposition (Bénabou and Tirole, 2010). This provides useful indicators that suggest CSR is closing a gap with CSV, especially in its social value insights. More importantly for this paper, it illustrates that both CSR and CSV have an important role in linking organisational and project success to SDG impacts.

The CSV business model, first developed by leading business strategist Michael Porter of Harvard Business School (Porter and Kramer, 2011), enables the potential to provide a link from local projects' objectives to global SDGs by rethinking projects' definition of success by demonstrating impact across the TBL (Elkington, 1994), as discussed in the previous section. It can do this because it: (a) recognises the interdependence between society and business; (b) moves society and business away from zero-sum competition to positive-sum competition; (c) enables new ways for the business to create competitive advantage that are more resilient against sustainability risks and mimicry by other firms; and (d) combines traditional CSR and business operations into new, integrated and company-specific, strategies for creating shared value. Since business and society are interdependent, the best outcomes for each will be obtained when businesses develop strategies that integrate social needs with real commercial opportunities and vice versa.

3.8 SYNTHESISING THE INITIAL THEORETICAL MODEL (IMPACT VALUE CHAIN) FROM LITERATURE

Based on the literature review from the previous chapter and the theoretical analysis in this chapter, an initial theoretical model has been developed, called the Impact Value Chain (IVC). This section provides the initial description of its design and this will be developed further into a prototype model in Chapter 7, when the triangulation from the empirical evidence is used to strengthen the IVC model that will be tested in Chapters 8 (for a 'Golden Thread') and 9 (with a water utility company).

The initial IVC model is based on four underpinning theoretical models, discussed earlier in this chapter, including: (1) the Theory of Change (Weiss, 1995 and Stein and Valters, 2012), (2) Creating Shared Value (Porter, 1985, 2011), (3) infrastructure systems approach (Hall et al., 2016, 2017; Thacker and Hall, 2018 and Thacker, et al., 2019) and (4) the Triple Bottom Line (Elkington, 1994, 2013, 2018 and Griggs et al., 2013). The last of these, the TBL, provided the link to SDGs through a more holistic "systems approach" to address infrastructure sustainability in the SDG context. The IVC potentially provides a new holistic method to improve sustainability on projects and programmes by guiding decision makers in their investment choices through confidence that they link to specific SDG targets.

3.8.1 Concept 1 - The Theory of Change's Impact-Value Chain (IVC)

The overarching IVC model has its roots in the development of the Logical Framework Approach (Baccarini, 1999), also known as the 'LogFrame' or Goal Oriented Project Planning (GOPP) and Objectives Oriented Project Planning (OOPP) methods, in the development sector. Leading authorities such as the World Bank (2004), OECD (2007) and UNDP (2009) have combined with academic research

institutions, such as Oxford University's School of Geography and Environment (2014) to develop the temporal logic model to help investors, and wider stakeholders to design, monitor and evaluate development projects (International Fund for Agricultural Development, 2002). These Value Chain methods all stem from the 'Theories of Change' (ToC) (Weiss, 1995; Stein and Valters, 2012) that was discussed earlier. The ToC value chains have a number of principles, which are core to the proposed Infrastructure SDG model: they define long-term goals and then map backward to identify necessary preconditions; they explain the process of change by outlining causal linkages in an initiative, i.e., its shorter-term, intermediate, and longer-term outcomes; and the identified changes are mapped – as the 'outcomes pathway' – showing each outcome in logical relationship to all the others, as well as chronological flow. Finally, the ultimate success of any ToC value chain lies in its ability to demonstrate progress on the achievement of outcomes. Evidence of success confirms the theory and indicates that the initiative is effective. Figure 24 depicts these causal chain relationships.



Figure 24: Concept 1 - The Theory of Change's fundamentals for delivering impact (the number '1' in the diagram refers to the Concept number, this is sequentially repeated in the three following Figures).

A core part of this research is understanding the SDG measurement at project level. This stems from an understanding of the project lifecycle and how sustainability reporting is critical to starting the project 'right' with alignment across stakeholders on what impacts define their success and how the Value Chain will be measured. The following supporting definitions are used for the value chain:

- **Theory of change** (Weiss, 1995): 'A basis for planning intervention in a given policy or project arena that helps to identify processes and preconditions whereby actions can best attain their intended consequences'.
- Value (Porter, 1985): 'The worth of a good or service as determined by the customer's preferences and the trade-offs they choose to make given their scarce resources, or the value the marketplaces on an item'. (Note: these are economic priorities. The wider societal and environmental considerations are

covered within the second dimension of the emerging IVC approach, in the Triple Bottom Line, In para 3.8.4)

- Value chain (Porter, 1985; Bloom and Hinrichs, 2011): 'The full range of processes and activities that characterize the lifecycle of a product from production, to manufacturing and processing, to distribution, marketing and retail, and finally to consumption (including waste and disposal across all stages)'.
- Impact (World Bank, 2004): Adapted to: a positive or negative contribution to one or more SDG targets across the TBL of environmental, economic, or social thematic areas.
- **Outcome** (APM, 2012): Adapted to: a change in the extent or condition of the stocks of capital (namely natural, produced, social and human) from the use of an output, due to value-chain activities that deliver SDG impacts.
- **Output** (APM, 2012): '*The tangible or intangible product typically delivered by a project*'.
- Activity (APM, 2012): 'A task, job, operation or process consuming time and other resources in a project to produce specific outputs'.
- **Input** (APM, 2012): Adapted to: All those items required to undertake work utilising the stocks of capital including financial, natural, produced, social and human resources.

APM references are used above instead of academic references since the majority of academics refer to PMI and APM definitions because they are the terms used most commonly by practitioners. The definitions above are complementary to the definitions in the 'Key Definitions' on pages xv to xvi.

3.8.2 Concept 2 - Delivery of Projects within an Organisational Structure

Infrastructure project management in the built environment is a discipline that focuses on phased delivery, within the parameters of time, cost and scope (and quality), to deliver a defined output, or if a programme, an outcome (APM, 2012). Project management has a well-recognised development process, referred to as the project life cycle (Morris, 2013), which is typically based on a number of iterative and normative stages, such as: plan, design, deliver, operate/maintain, and decommission.

Therefore, understanding the sustainability reporting requirements within the project lifecycle enables the right stakeholders to be engaged at the right time to design the right approach to the definition, measurement, monitoring and reporting of sustainability outputs and outcomes, as shown in Figure 25 below.



Figure 25: Concept 2 - Delivery of projects within an organisational structure; both with sustainability reporting loops embedded.

The project context is important, especially in its relationship with stakeholders, such as clients, designers, and contractors, because alignment of what success means to different stakeholders is a critical success factor in itself. However, these relationships indicate that the project sits at both inter and intra organisational boundaries, where each organisation has its own sustainability reporting requirements as part of an annual reporting cycle. This is highlighted in Figure 25, showing the proliferation of sustainability reporting instruments; currently in excess of 400, including 170 government and stock exchange / financial market regulations, 129 Codes of Conduct or Guidance, 8 Standards, as well as a myriad of other industry frameworks (KPMG, 2017).



Figure 26: Global Sustainability Reporting. Source data from GRI (2016).

3.8.3 Concept 3 – Infrastructure investments as 'system of systems'.

The layered description of the proposed conceptual model, IVC, is iterated below in Figure 27, with a third concept that relates to infrastructure projects that are based on a 'system-of-systems' approach (Hall et al., 2016; Thacker and Hall, 2018).



Figure 27: Concept 3 – Infrastructure investments as a 'system of systems' – Source: The Future of National Infrastructure: A System of Systems Approach (Hall et al, 2016; UNOPS, 2017).

The concept of a system of systems recognises that infrastructure projects in the built environment are more than the sum of their parts, albeit that the strength of their synergy also has an inherent risk of resilience weakness through both complexities and boundary management issues if they are not fully understood and managed (Davies, 2004; ITRC's Hall et al, 2016). The concept challenges the traditional understanding of infrastructure as stand-alone physical assets. The systems approach recognises a number of linear value chain connections that fit with concepts 1 and 2

described earlier. These are illustrated in Figure 28 below, showing the way that investment of resources on specified infrastructure projects deliver assets that can be transformed through programme change initiatives to provide a service to society, which if successful, has strategic impacts that could be aligned to SDGs. Thus, infrastructure operates as a set of both interrelated and interdependent systems, which enable service systems within a certain context to function. The key definitions for the infrastructure systems are (UNOPS, 2017) as follows:

- Systems Thinking: An approach that focuses on the identification of interrelationships between components (i.e. sub-systems) of a system (Davies, 2004). Davies argues (2004) that the provision of integrated solutions and 'systems integration' is becoming an increasingly important part of construction companies skillset. In addition to an ability to design and integrate systems using internal or external sources of product supply, these firms are developing novel combinations of service capabilities (operations, business consultancy and finance) required to provide complete solutions to each customer's needs.
- Infrastructure System (UNOPS, 2017; ITRC's Hall et al, 2016). 'A system comprised of assets, institutions and knowledge that provides a society its services. Examples of National Economic Infrastructure systems are: Water, Energy, Transport, Waste, and Telecommunications'.
- Assets (UNOPS, 2017; ITRC's Hall et al, 2016). Adapted to: The physical components of the system. This also extends to the links that exist between the assets in the system. Note that there are man-made assets but also 'Natural Assets', which provide a service. Examples of man-made assets include roads, bridges, power lines, and pipes. Example of natural assets include wetland systems, and mangrove forests.
- Services (UNOPS, 2017; ITRC's Hall et al, 2016). 'The functions which the infrastructure system enables. Examples include healthcare services, transport services, and education services'.



Figure 28: Infrastructure System of systems Map showing 'Ends, Ways, Means'. Adapted from papers by UNOPS (2017) and ITRC's (Hall et al, 2016).

Infrastructure projects' SDG impact is best understood through the relationship, shown in Figure 28, between the infrastructure systems' service to society. This provides the causal link to the SDG impacts and a coherent way to assess sustainability across the TBL of economic, social and environmental impact areas.

The applicability of understanding infrastructure projects within a system that delivers outcomes, is becoming better understood (Davies, 1997, 2004, 2005, 2014; ICE, 2018; National Infrastructure Commission, 2018; IPA TIP Report, 2017) as a value-based proposition, as evidenced in the ICE transformative research on Project 13 (ICE, 2018). It focuses more on wider stakeholders' social and environmental impacts as well as the business-leading economic requirements of time, cost and scope.

3.8.4 Concept 4 – Delivering Impact measured against the TBL/SDGs

The connection between the SDG impacts and the five main national economic infrastructure systems of power (i.e. energy), water (including collection, treatment, distribution, and disposal), transport, waste, and information and communications technology (ICT), is described below in the fourth concept, which illustrates the TBL link that connects the entire SDG Infrastructure Value-Impact Chain (IVC).



Figure 29: Concept 4 – Delivering impact measured against the TBL/SDGs, and completing the learning-loop for smarter investments, adapted from ICAS/IIRC's 'The Sustainable Development Goals, integrated thinking and the integrated report' (Adams, 2017).

3.9 SUMMARY AND IMPLICATIONS

This chapter has discussed the theoretical framework. The chapter described how the Critical Realism perspective of ideological philosophers such as Bhaskar (1978) informed the Realist Evaluation approach championed by Pawson and Tilley (2004) and this introduces the Context-Mechanism-Outcomes model that provides the overarching research framework. The chapter includes analysis of the primary theories adopted for the research, including the Theory of Change and Triple Bottom Line, which are both embedded in the proposed mechanism, called the Impact Value Chain (IVC – developed in Chapter 7) – highlighting the need for project value creation as an end product-outcome of a project (Brady, Davies & Gann, 2005). The chapter also introduces the context and mechanism across the project organisational structure developed by Müller, Drouin, and Sankaran (2019a), who proposed a layered model of organisational project management (OPM). This theoretically enables the structured evaluation of the concept of SDG, from global to local, at portfolio, mega-project and project levels.

The preceding discussion on theories has established a number of key considerations to be taken forward in this study:

• The 9 contextual variables identified in Table 7, that were developed from the systematic literature review, shown in Figure 15, identified a nodal structure of issues and variables. These were applied to the Müller OPM Onion Model (Müller

et al., 2019). This provided the link to shape the evolving evidence based MISI research framework. It was used to inform the empirical evidence gathering for the surveys and the interviews. The 9 contextual variables provided a way to structure the questioning and supported the proposition analysis.

- Based on the literature review from the previous chapter and the theoretical analysis in this chapter, an initial theoretical model was developed, called the Impact Value Chain (IVC). The IVC was developed further into a prototype model for testing in Chapter 7, when the triangulation from the empirical evidence was used to strengthen the IVC model that was then tested in Chapters 8 (for a 'Golden Thread') and 9 (with a water utility company).
- The review of theories has highlighted the potential benefits and tensions of linking global goals to local delivery on infrastructure projects. As a result of these findings, the derived sub-research question that underpinned Research Objectives 1-3⁸ are:

How do practitioners in the construction sector rate and use global UN SDGs for infrastructure investment decisions at the local level? The sub-questions that flow from this were:

- What contextual issues influence the successful use of an SDG measurement mechanism to achieve the desired outcomes? (This represents the context).
- What mechanism (for measuring SDG impacts) is in place to achieve the outcomes? (This represents the mechanism).
- What are the expected outcomes of using the SDG measurement mechanism? (This represents the outcome).

The four propositions from Chapter 2, the variables identified in this chapter, and the three sub-questions for proposition 1 and 2, shown above, informed the design of the methodology discussed in the next chapter. As a way of building an evolving picture of how the study built its understanding to answer the research question by

⁸ RO1: To understand the existing MISI knowledge (in theory and practice); RO2: To understand the MISI context (the 'variables'); RO3: To assess the current MISI mechanisms.

addressing the propositions, the table shown below will be iterated at the end of each chapter:

Confidence in	P1:	P2:	P3:	P4:
Evidence for	Engineer's view of	Organisational	'Golden Thread'	Prototype
Propositions	current MISI	view of MISI	global - to - local	Development
Theoretical	Not yet	Not yet	Very Low *1	Very Low *1
Analysis				
Survey Analysis	Not yet	Not yet	Not yet	Not yet
Interviews	Not yet	Not yet	Not yet	Not yet
Analysis		-		-
Test 1: Golden	Not yet	Not yet	Not yet	Not yet
Thread		-		-
Test 2: Case	Not yet	Not yet	Not yet	Not yet
Study				-

Table 8: Development of stability of Propositions' Results (changes since previous Chap in red)

Note *1: Table 8 above illustrates the assessment of the quality of propositions' evidence. This is based on a qualitative formula designed by the Clinical Grading Working Group's British Medical Journal's study (GRADE, 2004) that is widely used to assess the maturity of evidence underpinning propositions, based on a combination of quality and consistency. The high level of innovation needed for MISI, suggests that trialling new forms of evidence building (in this case by using a technique for building proposition confidence) is justified. The GRADE Group (2004) use the following definitions: **High**, Further research is very unlikely to change the confidence in the estimate of effect; **Moderate**, Further research is likely to have an important impact on the confidence in the estimate of effect and may change the estimate of effect and is likely to change the estimate; **Low**, Further research is very uncertain.

Chapter 4: Research Design and Methodology

4.1 OUTLINE OF CHAPTER

The research design needs to be logical, relevant and achievable (Merriam and Grenier, 2019). Research can be defined as something that people undertake in order to find out things in a systematic way, thereby increasing their knowledge (Ghauri and Grønhaug, 2005). This chapter explains the 'systematic way' chosen as being the most appropriate for the needs of the MISI study.

The basis for the choice of research method was whether it will help the study answer the research question, the research objectives and the four propositions (Table 5) established in the previous two chapters, whilst also meeting the key criteria of 'logical, relevant and achievable' (Merriam and Grenier, 2019). This chapter also explains how the derived analysis from the earlier chapters has informed the research design and methodology for this Sequential Explanatory Design (Creswell, 2017) that uses a mixed methods study of surveys and interviews to address Proposition 1 and 2 (Table 5) – gaining insights to individual engineers' and organisations' perceptions of approaches to MISI. A more detailed analysis of the research setting, and the sample population is completed in the following two chapters (surveys in Chapter 5, and the interviews in Chapter 6), covering the design, results and discussion, relevant for each approach.

4.2 METHODOLOGY

Using the C-M-O theoretical lens discussed in the previous chapter, an iterative research framework was developed that combined a mixed-method design, shown below in Figure 30, and discussed at length in Section 4.3. The research design involved a three-way data collection approach. At its core, the research design built on the triangulation of qualitative and quantitative datasets, which is well recognised as a method for informing theory-led research development (Creswell, 2017; Easterby-Smith et al., 2002). In what Creswell (2017) describes as a Sequential Explanatory Design, the literature review informed the survey questions and analysis

that iteratively then informed the structure and approach of the interviews. In this way, Merriam and Grenier (2019) suggest that 'the interviews help the researcher understand the responses to the survey as well as provide additional insights into the phenomenon of interest'.

As shown in Figure 30 below, the development of a prototype SDG measurement model was based on the triangulation of learning from the literature review, the survey of 325 engineers and the subsequent interviews of 40 senior executives. A primary advantage of combining the survey with the semi-structured interview method is that it allows an adaptive-responsive approach to ensure the best improvisation to delve deeper into relative areas of importance, based on the Participant's survey responses (Hardon et al., 2004, Rubin & Rubin, 2005, Polit & Beck, 2010) and it also allows for interview participants' verbal expressions to be captured (Robert Wood Johnson Foundation, 2008).



Figure 30. The mixed-method research design adapted from Creswell (2003 and 2017).

The research methodology shown above combines both inductive (interviews) and deductive (literature review and survey) enquiries. This provides a sequential two stage testing of the propositions, firstly by the conduct of a survey to get individual engineers' views on MISI (Chapter 5), then subsequently, through learning from the survey, to develop a deeper and richer understanding of the research phenomenon by completion of 40 interviews (Chapter 6). Both methods, linked explicitly to the research question, addressed the first two propositions, with the final two propositions being addressed in the Case Study 2-Test stage (Chapters 8 and 9). In particular, Chapter 8 and Chapter 9 provide evidence that a link can be made from global to

organisational level. The two chapters highlight the importance of strategically defining their organisational outcomes and strategic interventions in a structure and language that aligned to the SDGs (as shown in Table 31 and Figure 72). For example, Anglian Water used the SDGs to illustrate the strategic priorities that the organisation set as part of its long-term goals (selecting 10 SDG goals and 35 targets to measure their success). The strategic intervention was managed by Anglia Water's portfolio office that used the defined SDG goals and targets to map the delivery pathway at project level in the delivery phase.

4.3 RESEARCH DESIGN

The research was based on two main stages. The first stage, informed by a systematic literature review, comprised a mixed method that involved a survey of 350 engineers (due to late completion of 25, only 325 were included in the analysis) to derive quantitative data (Mansell et al., 2020b) along with interviews with 40 CEOs and corporate Heads of Sustainability to capture qualitative data (Mansell et al., 2020c). The second stage involved the development of a prototype that was tested through further exploratory investigations at two levels: (1) Test 1: is there a Golden Thread from global SDGs, through the organisational-level, down to project level SDG impact measurement; (2) Test 2: does the prototype model, the Impact Value *Chain*, have coherence in the analysis of a brief case study investigation that, through application of the main findings from the empirical stage, evaluated the scope to measure SDG performance for infrastructure projects at a Water Utility Company (Anglian Water) (Mansell et al., 2020d). Subsequently, the research study led to a collaborative partnership to test the protoype model and its approach across the Environment Agency's full portfolio of projects and also, the megaproject of the Thames Tideway Tunnel (supporting letters describing the direct impact of this follow-on stage are evidenced in Appendices 17-23).

4.3.1 Research 'Onion Ring' Framework

The Saunders 'research onion ring' (Saunders et al., 2009), originally developed in 2007, provides a common framework to explain this study's approach to the theoretical and methodological positions.



Figure 31. The Research 'Onion Rings' (Saunders et al., 2009).

The previous chapter described Level 1 of the 'Onion Ring', the philosophical stance, and this chapter deals with the other five layers: Approaches; Strategies; Choices; Time Horizon; and Techniques and procedures. The detailed design of the Survey Stage is covered in Chapter 5, and for the Interview Stage, in Chapter 6. The high-level summary of the choices of design approach for each of the 6 research strategies shown in column 'a' in Table 9 (Saunders, 2007) is shown below. The columns iteratively illustrate the methodology design, starting with the research design in column 'b', listing whether it was primarily a deductive or inductive method. Column 'c' lists the derivation of the input design, providing clarity on how the research strategy was structured to 'give meaning' to the phenomenon of interest (Merriam and Grenier, 2019). Columns 'd' and 'e' list the propositions and research objectives that are addressed, which is shown as a percentage completion in a table at the end of each of the research strategy chapters (2,3,5,6,7,8, and 9) to enable the reader to follow the evidence-building along a recognisable analytical linear pathway. Column 'f' details the method and tools used, such as SPSS and NVivo, and the final column 'g' lists the outputs from the specific research strategy, for example, the four propositions are an output from the SLR in Chapter 2.

Research (a) Strategy: (Saunders, 2009)	Research (b) Approach: Deductive / Inductive	Input Design (c)	Propositions addressed (d)	Research Objectives addressed (e)	Data Collection Methods & tools (f)	Outputs (g)
Literature Review (Ch 2)	Deductive	PRISMA (Liberati et al., 2009) checklist	n/a: SLR led to derived propositions	RO 1 - 2	Excel and interpretive deduction	4* Propositions;
Theory (Ch 3)	Deductive	Research Question and Res. Objectives	Proposition 4 (partly)	RO 1 - 3	Literature investigation with Excel	9* Context variables; 4* sub- questions for data collection on P 1 & 2; Informs RO 5 & 8
Survey (Ch 5)	Deductive	8 process steps: Creswell, 2017; 9 Context variables	Proposition 1	RO 1 - 4	SPSS	Informs RO 5 & 8
Interview (Ch 6)	Inductive	Updated 9 Context variables	Proposition 2	RO 1 - 4	NVivo	Informs RO 5 & 8
Case Study (Ch 8) Test 1: GRI & CEEQUAL	Inductive	Case Study 5 tests (Flyvbjerg, 2006)	Proposition 3	RO 6	Excel and NVivo	Informs RO 8
Case Study (Ch 9) Test 2: Water utility company	Inductive	Case Study 5 tests (Flyvbjerg, 2006)	Propositions 1, 2, 3 and 4	RO 7	Interpretive induction with Excel	Informs RO 8

Table 9: Development of methodological design based on Saunders' Research Onion Ring (2009)

List of Propositions: Proposition 1: Individual engineers are 'supportive but frustrated' in methods to measure global SDG goals at project level.; Proposition 2: Organisations' leaders champion SDG measurement without having evidence-based methods to verify claimed impact at portfolio and project levels; Proposition 3: Current sustainability measurement at organisational and project levels can be used to demonstrate a 'golden thread' from 'global' SDG to 'local' portfolio and project levels.; Proposition 4: A MISI prototype can provide a plausible, testable and achievable logic chain for defining project and portfolio SDG impacts.

List of Research Objectives (RO): RO1: To understand the <u>existing knowledge</u> on how organisations and projects measure infrastructure projects' SDG Impact; RO2: To understand the <u>context</u> of the current use of mechanisms to measure infrastructure projects' SDG Impact; RO3: To assess the <u>current mechanisms</u> used for the Measurement of Infrastructure projects' SDG Impact (MISI) at portfolio and project levels; RO4: To understand the <u>perception of individual engineers and organisations</u>' relative perception of the <u>outcomes</u> of the current use of mechanisms to measure infrastructure projects' SDG Impact; RO5: To use the theory-led study to inform the development of a prototype model to improve the measurement of infrastructure projects' SDG Impact; RO6: To test whether a 'golden thread' of SDG measurement could be identified from global to local levels; RO7: To test whether the prototype could be validated with a case study organisation; RO8: To build a framework for further development, for researchers and practice to utilise, driving improved investment decisions across planet, profit and people outcome criteria, aligned to SDG impacts.

Note: the compilation of the results against all the propositions and research objectives are included in the final chapter's summary of results.

Based on the research question, the research objectives and the propositions to be tested (as shown above in Table 9), the research uses the Critical Realist Evaluation methodology, introduced in the previous chapter, first developed by Pawson and Tilley (1997). The separate investigations needed to be set-up in a way that could analyse workings of recognised sub-systems within the programme (the measurement of SDG impacts on projects) being investigated, and in doing so, create events that the participants could express an opinion on what they observed and felt. Thus, the realist evaluation approach provides data that indicates how the intervention works (the generative mechanism) and also helps identify the conditions that are needed for a particular mechanism to work (i.e., specification of contexts).

Secondly, the choice of the research strategy was influenced by the need to ground theory in practice, which was highlighted in the previous Systematic Literature Review in Chapter 2 that concluded many studies into SDG at project level were too theoretical, lacked insights for practitioners to apply, and were without tools and models that could be tested.

4.3.2 Critical Realist (CR) Research Designs

A summary of the CR design choices for the MISI study are shown in the table below (Table 10). This design, adapted by Vincent and O'Mahoney (2018), uses the CR structure devised by Ackroyd and Karlsson (2014). The table illustrates two design considerations, the first (shown in columns under 'a'), corresponds with the research focus, scaling from intensive research to extensive research. Intensive research emphasises qualitative research designs, such as case-studies. Extensive research investigates the impact of context variables on a mechanism, such as by using surveys (Sayer, 1997). For example, quantitative data from the survey of the engineers can be used to generate taxonomies for the coding of the interviews and is useful more where mechanisms are known (or at least inferred) but the context varies (Ackroyd and Karlsson, 2014). The second design criteria, shown in rows 'b' and 'c' below (Table 10), reflects the relative detachment of the researcher, either from a distance, or whether they are trying to manipulate the mechanism or context under investigation (Vincent and O'Mahoney, 2018).

	Intensive $\leftarrow \rightarrow$ Extensive (a)				
	What is the mechanism?	How do context & mechanism typically behave x OPM levels?	How do context & mechanism historically interact?	What are the context variables that impact outcomes?	
Detached (b)	Case-study (Chap 9)	Comparative case- study (Not used)	Institutional / historical analysis (Not used)	Surveys (Chap 5)	
Engaged (c)	Action research (Not used)	Intensive realist literature evaluations (Chaps 2,3 & 7)	Barefoot research (Not used)	Extensive realist evaluation (Chap 6)	

Table 10: Critical Realist Research Designs (amended from Vincent and O'Mahoney, 2018; Ackroyd and Karlsson 2014)

Based on this CR framework, the proposed future research design methods chosen, with corresponding literature provenance, were:

- **Case-study**. The most frequent form of CR research that enables MISI to test the exploration of using the IVC prototype to assess whether it is practical, plausible and achievable. Beynon (1979), Flyvbjerg (2006), Ruddin (2006) and Gouldner (1964) suggest this enables the researcher to abduct causal mechanisms from their empirical evidence. Chapter 9 used this design.
- Action research. This is described by Friedman and Rogers (2009) as 'a research investigation to explore the workings of a mechanism by triggering it or changing its context'. Cassell and Johnson (2006) and Morgan and Olsen (2008) also champion this approach. This method of research was discounted due to a lack of time, resources and buy-in from potential organisations to develop the approach with, but is considered for the follow-on research, discussed in Chapter 10.
- **Comparative case-study**. The proponents of this approach such as Kirkpatrick et al. (2005), and, Delbridge (1998) suggest this as a suitable way to explore how similar mechanisms operate in different contexts. This was not chosen as an approach but is considered for the follow-on research to select organisations that can be evaluated at a portfolio and project level, and perhaps one megaproject.

- Intensive realist literature evaluations. This supports the explanation of what mechanisms work in different contexts by reviewing the extant literature, as described by Marchal et al. (2012). It is used in Chapters 2, 3 and 7 to build theoretical underpinning to the prototype model.
- Institutional / historical analysis. Clark (2012) and Mutch (2007, 2013, 2014) propose this as a way to examine causal chains over time to explore how mechanisms and contexts interact over time, and the conditions for such interaction (Smith and Meiksins, 1995). This longitudinal approach has not been used because of the challenge of the researcher's time and resources.
- **Barefoot research**. Lindqvist (1979, 1995), and more recently Vincent and O'Mahoney (2018), describes how organisations can '*develop their own interpretations by use of training and encouraging employees of the case study organisation to do their own research*'. This is the approach that might be used for future research since it requires the training of the collaborative MISI partnership for them to conduct their own review and further developments.
- Surveys. Cully et al. (1999) and others (Rea and Parker, 2014; Vaske, 2019; Groves et al, 2011) characterise this approach for '*when the primary focus is on descriptive statistics*' (such as sampling from the population data of the Institution of Civil Engineers in Chapter 5) to investigate the empirical impacts or conditions of using the MISI mechanism. This study uses it to prompt explanatory investigations, that informed the interviews (Chapter 6).
- Extensive realist evaluation. The mixed methods approach in an 'engaged' way, partly reflects the Interviews approach described in Chapter 6, that is informed by survey's statistical techniques to examine how different contexts affect a mechanism (Kazi et al., 2002, Kazi, 2003).
- Methods not used. There were four methods not used for the research. This was for reasons of either non-applicability or because of time and space limitations (being a single researcher). However, these research methods have been re-considered for future research and these proposals are discussed in the final chapter (see Table 37).

4.4 QUALITATIVE OR QUANTITATIVE, OR BOTH?

Ultimately, the choice of method needs to fit with the research question, which is based on the problem definition (Creswell, 2003). Creswell develops this by stating that mixed methods should only be chosen 'when qualitative or quantitative research is insufficient to fully understand the problem'. Creswell further expands this statement in five ways: 'we need to explore before we administer instruments'; 'we need to explain our statistical results by talking to people'; 'we need to see if our quantitative and qualitative results match each other'; 'we need to enhance our experiments by talking with people'; 'we need to develop new instruments by gathering qualitative data'. The following two sections will answer these five statements.

4.4.1 What type of data and analysis will I need?

Based on the Creswell's definition questions (2003) in the previous paragraph, the data requirements of this study, to meet the research objectives, are:

- Quantitative data collection (closed-ended).
 - To capture individual engineers' perceptions on the following subquestions: How do practitioners in the construction sector rate and use global UN SDGs for infrastructure investment decisions at the local level? The sub-questions (SQ) that flow from this are as follows:
 - SQ1 What contextual issues influence the successful use of an SDG measurement mechanism to achieve the desired outcomes? (This represents the context);
 - SQ2 What mechanism (for measuring SDG impacts) is in place to achieve the outcomes? (This represents the mechanism);
 - SQ3 What are the expected outcomes of using the SDG measurement mechanism? (This represents the outcome).
 - The chosen instrument is a survey, using a Likert scale to collate quantifiable perceptions on relevant criteria.
- Quantitative data analysis

- To use numeric data to provide: description of the phenomenon of study; comparison of groups; and for relating the contextual variables identified in Chapter 2. The SPSS tool was used to support statistical analysis.
- Qualitative data collection (open-ended)
 - Using the same sub-questions discussed above, the chosen instrument was to set-up interviews with senior executives, supported by supplementary reading of their organisation's reports and publications, to get an organisational perspective.
- Qualitative data analysis
 - Use text and image data to allow: coding; theme development and for relating themes to the propositions; and, enabling the MISI contextual variables to be analysed in a richer-deeper analytical way, by building on the survey through semi-structured interviews. The NVivo tool was used to support qualitative analysis.

4.4.2 Rationale for mixed methods

Based on the discussion above, the reasons for using the mixed methods was the need for different, multiple perspectives to achieve a more complete understanding. This enabled the interviews to confirm and expand on the results of the survey. In doing so, the two-step Explanatory Sequential Design (Creswell, 2015; Merriam and Grenier, 2019), illustrated in Figure 32 below, built on the first data analysis by: informing the design of the interviews (by closing data gaps); testing emerging trends, better contextualising the approach to the CEOs, by sharing with them the results of the survey, thus gaining credibility and increased likelihood of being given their 'precious time'; and allowing the researcher to mature his skills as part of Kolb's (1981, 1984) Experiential Learning Cycle theory, in which the researcher has time to 'experience, reflect-observe, conceptualise, and experiment'.


Figure 32. The two-step Sequential Explanatory Design adapted from Creswell (2003 and 2017).

4.4.3 Building the CMO Configuration

The first step (of the two-steps in Figure 32) clarifies the programme of intervention (measuring SDG impacts) to be investigated and identifies the core issues within the three C-M-O elements (Pawson and Tilley, 2004; Zachariadis et al., 2013). Having listed the primary elements of CMO (based on earlier surveys, interviews and literature review), the C-M-O configuration was designed as a routemap to identify a propositioned causal chain that could be interrogated, with the ultimate purpose being to consider whether the IVC was usable by practitioners, and in what circumstances it was likely to be successful. The Figure 33 below illustrates the causal pathway C-M-O and shows in red, the primary areas of the investigation.

The two diagrams below (Figure 33 and Figure 34) provide a way to understand how the three lenses of C-M-O have materialised into: (1) the context variables extracted from Muller et al. (2019); (2) the mechanism from the theoretical analysis of ends-ways-means; and (3) the outcome criteria are from the Theory of Chain evaluation, and investigation into definition of project success.



Figure 33: The Context-Mechanism-Outcome Variables Framework's causal pathway.

This can be simplified as a conceptual model that shows the relationship between the core areas to be investigated: the Realist Evaluation's Context-Mechanism-Outcome (CMO) configuration model championed by Pawson and Tilley (2004) and widely used across clinical research (Pawson, 2005); the Impact Value Chain as the central 'mechanism' being investigated; across the three levels (from Muller's OPM) of the project, mega-project and portfolio.



Figure 34: Simplified conceptual view of analytical model for testing the propositions across three dimensions of: IVC; C-M-O Framework; and Portfolio-Project Levels.

4.4.4 Stage One – Exploratory Investigation

The Exploratory stage of the investigation sought to identify the 'landscape' of the phenomenon of interest. Through the iterative stages of research, shown in Figure 35, the purpose was to address a gap in knowledge through a series of events that enabled the gradual deepening of understanding and thereby, to address the research question as well as the propositions.



Figure 35: Exploratory Investigation (EI) Flow Chart.

4.4.5 Stage Two – Detailed Investigation

The Detailed stage of the investigation sought to build on Stage One by firstly triangulating the data and analysis from the literature review, the survey and the interviews, and secondly, to build a prototype model that could be tested. Through the iterative stages of research, shown in Figure 36, the purpose was to address a gap in knowledge through a series of events that enabled the gradual deepening of understanding and thereby, to address the research question as well as the propositions.



Figure 36: Detailed Investigation (DI) Flow Chart.

4.4.6 Quantitative Research Method – the survey (see Chapter 5)

A survey of 325 engineers on what the individual engineers' perspectives on the research question provided the data source to inform the secondary step of the qualitative research method. A detailed description of this event is in the next chapter.

4.4.7 Qualitative Research Method – the interviews (see Chapter 6)

A series of 40 interviews were conducted to provide an organisational perspective on the research question. The approach was informed by the preceding quantitative research method. A detailed description is in Chapter 6.

4.4.8 Triangulation of Data (see Chapter 7)

Triangulation has been used extensively in qualitative and social research. The thinking behind triangulation was introduced by Norman Denzin (1970) in the 1970s who developed the proposals by Webb et al. (1966) and before them, Campbell and Fiske (1959). The premise is that by adopting the maritime usage of triangulation, which uses the reading of stars through astral navigation, to produce a triangle that identifies the users' location, a similar approach is used for research. Thus, triangulation is a method that can improve the validity (extent to which a study accurately reflects or evaluates the concept or ideas being investigated) and credibility (trustworthiness and how believable a study is). Thus, the method design included an Explanatory Sequential Design (Creswell, 2015; Merriam and Grenier, 2019), followed by a triangulation of data to inform the development of the prototype.

4.5 ETHICS AND LIMITATIONS

The ethical considerations of the research and any problems and limitations, as well as any anticipated or actual threats to the validity of the results are described in detail in Chapters 5 and 6.

4.6 CHAPTER CONCLUSION

This chapter has chosen a research design to be logical, relevant and achievable (Merriam and Grenier, 2019). It has described the 'systematic way' that it intends 'to find out things' (Ghauri and Grønhaug, 2005) in the most appropriate way to meet the needs of the MISI study. This chapter also explained how the derived analysis from the earlier chapters has informed the research design and methodology for this Sequential Explanatory Design (Creswell, 2017) that uses a mixed methods study of surveys and interviews to address the propositions. A more detailed analysis of the research setting, and the sample population is completed in the following two chapters, covering the design, results and discussion of the surveys in Chapter 5, and the interviews in Chapter 6.

Chapter 5: Survey Method, Results and Analysis

5.1 OUTLINE OF CHAPTER

This chapter describes the design adopted by this research to achieve the aims and objectives stated in Chapter 1. Section 5.2 discusses the methodology used for this first part of the Sequential Explanatory Design (Creswell, 2017), the survey of 325 engineers, the stages by which the methodology was implemented, and the research design; Section 5.3 details the participants in the study; Section 5.4 provides the survey results that feed into Section 5.6 inferential statistics. The discussion is included in Section 5.7, with the policy implications outlined in Section 5.8.

5.2 METHODOLOGY AND RESEARCH DESIGN

5.2.1 Methodology - Using the Realist Evaluation Methodology to Structure the Survey

As discussed in Chapter 3, the research study adopted the Critical Realism perspective of philosophers such as Bhaskar (2013) to inform the choice of the Realist Evaluation approach, primarily because of its practical utility and its widespread use in social science research into the impacts of programmes (Linsley et al., 2015). Pawson and Tilley specifically recommend the C–M–O strategy so that 'programme theories can be tested for the purposes of refining them' (2005, p. 2). In this regard, the investigation was not about what works but asks instead, 'what works for whom in what circumstances and in what respects, how?' (2005, p. 2).

The survey captured the first-level 'empirical' observed and experienced views of engineers' use of SDG measurement by seeking to quantify their perspectives on three areas: (a) the value and importance they placed in defining SDG outcomes as a measure of project success, (b) their insights into the use of the mechanism (the 'trigger' being the tools, process, structures and strategy for measurement of a prioritised list of SDG goals and targets at the project level) and (c) the context of issues that affect the likely success of achieving the outcomes from the use of the mechanism.

5.2.2 Research Design of the Survey (expanding from Chapter 4)

Building on the high-level research programme's research design described in the previous chapter, this section describes the research design specifically of the survey.

Given the issues identified in the literature review, the methodology needed to be suitable to address the earlier stated research question: How do engineers in the construction sector rate and use global UN SDG goals for businesses and projects at local level? An online survey was used as the first part of a mixed methods approach that provides a triangulation (Creswell and Cresswell, 2017) of data (i.e., through literature review, survey and interviews) to inform the development of the prototype SDG Measurement Model. In this way, in what Creswell (2017) describes as a Sequential Explanatory Design, the literature review informed the survey questions and analysis, that sequentially informed the structure and questions of the interview stage. Creswell (2017) suggested that this sequential approach has the benefit of being the most straightforward in its design because there are discrete stages that are easy to describe and to report. The main difficulty is the length of time in the data collection phase.



Figure 37: The research approach of mixed method Sequential Explanatory Design, adapted from Creswell (2003 and 2017).

5.3 THE SURVEY

The survey approach to data gathering was selected because it is recognised (Lenth, 2001) as one of the most important methods in applied social science. This approach, shown in Table 11, enabled a rapid and cost-effective way to assess a statistically significant group of engineering project managers' perceptions of the relative importance and current capability for measuring SDG impacts on their projects. The participants were all members of the UK's leading civil engineering professional society, the Institution of Civil Engineers (ICE), who sanctioned the survey and provided access to their members' contact details. A benefit of using this style of survey approach is that it delivered both qualitative research outputs, by using open-ended questions that captured text-based commentary, and quantitative research outputs, by using a Likert scoring mechanism aligned to the questions. The eight-step model used to structure the survey is described below in Table 11.

	Survey Design and Analysis Methods					
1	Problem specification & Research Question	As captured in the introduction paragraph of this chapter; formulation of problem and objectives.				
2 Population Definition With support from the Institution of Civil Engineers (ICE); seeking a representative sample from across the engineering community.						
3 Selection of Use of Survey Monkey [©] software tool—for design and running of survey data collection. It also provided statistical analysis, supported by SPSS.						
4	4 Design Instrument In					
5	Specify and test survey Procedure	Build the logic framework in the tool and run 'check tests' with informed analysts to ensure the technical success and the logic of the procedure.				
6	Data Collection	ICE distributes the survey to 1500 of its members; 325 complete the survey, ca. 20% response rate, providing representative sample.				
7	Analysis	Completed in four stages of diagnostic analysis: Stage 1 : Download all data (quantitative and qualitative) in MS Excel [™] ; remove erroneous and false data, e.g., delete test data from the pilot. Structure data for analysis—e.g., charts and graphs to visualize data. Stage 2 : Use software tool on survey monkey and SPSS to analyse the data's statistical significance; identify patterns and gaps/overlaps against research question's objectives. Stage 3 : Analyse data touch points (using C–M–O coding) and correlate findings to the original research question and the C–M–O model. Complete initial write- up for review. Stage 4 : Share data findings with expert panel (of 12 qualified engineers) organised by the ICE; test the findings; keep integrity of the data but use expert panel to assess the implications and possible next steps. For example, the panel suggested that the low level of organisational responses could be addressed in the				

Table 11: Survey analysis method (based on: Creswell and Creswell, 2017; Nardi, 2015).

interview stage of the research. (Note: The three separate workshops were recorded. An example of a Blog written by the ICE Knowledge Manager about the ICE Kn		
		the Control Group, is shown at Appendix 6)
8	Reporting	 Step 1: Build the data charts that illustrate the findings. Step 2: Write up the findings: test and adjust to ensure recommendations and conclusions are consistent with original research question; identify lessons and insights that inform the next stage of research—the 40 interviews.

The questions (shown in Figure 38 and Appendix 3) were structured to measure attitudes in relation to the research question. The questions focused on three areas: the perceived value and importance of measuring SDGs on their projects (i.e. the outcomes); their current approach and capability (i.e. the mechanism), such as skills, tools, processes, structures and methods (Astbury and Leeuw, 2010: 386); and, their identification of the challenges and opportunities (i.e. the context) of measuring projects' SDG impact. SurveyMonkey© was chosen as the web-enabled survey tool because it was highly customisable and provided a comprehensive set of back-end capabilities, such as data analysis and visual representation tools, that helped present the data in a concise and informative manner.



Figure 38: Survey Questions Sequence; adopting Realist Evaluation methodology of Context-Mechanism-Outcomes (Pawson and Tilley, 2004).

The tool also provided guidance on bias-elimination and sample-selection best practice. This enabled a structured approach for presenting the questions, which were designed to capture the required data. This sought to establish whether this research area was of perceived importance to practising professionals, and if it was of high importance, was there a gap between the import of measuring SDG impact versus their capability to do so? It achieved this by using both open-ended and closed questions within a clear structure that explored firstly broader and secondly, more specific areas and concepts within the research areas. The survey was sent to participants by the UK's leading civil engineering institution, the ICE, thereby providing reassurance to the participants since they would recognise the institution's name and logo, which would be likely to increase the response rate. Data protection methods fulfilled ethical and legal data management requirements, including GDPR (general data protection regulation). For example, by sending the survey from the engineering institution to their members, the approach conformed with the members' original opt-in agreement to receive similar knowledge-sharing initiatives.

5.3.1 Access

The survey aimed to access between 200–300 qualified engineers. In actuality, the Institution of Civil Engineers' communications team selected a random representative distribution of its members, aiming to achieve ca. 20% of a total number of 1500 targeted participants. Since the respondents voluntarily opted in, this was considered a non-probability sample, which Tansey (2007) suggests is preferable because it identified what he terms 'elite interview subjects' in order to avoid the randomness of generic sampling. The response rate of 325 completed surveys was relatively high by the ICE's previous experience of surveys, typically achieving only 5% to 30% responses, the latter higher response rates being due to well-publicised events, such as committee elections. In this case, the ICE only sent a single email without any follow-ups; therefore, the response rate was considered good. The ICE also confirmed that the sub-set of the 325 respondents from the 1500 targeted participants was representative of the wider membership population (of 6500) because it included a sample selection across all experience levels, from student to engineers with over 20 years of experience, and this added to the statistical validity of the sample.

The questionnaire included a question on demographics designed to distinguish between the generations and, more specifically, capture the responses of millennials (i.e., people born between 1983 and 2000) (US Public Interest Research Group 2016; Howe and Strauss 1991) who, according to the U.S. Bureau of Labor Statistics (Labor 2017), within the following two years, reflected 50% of the US workforce, growing to 75% by 2030. The millennials can be viewed as the generation who are rapidly becoming the organisational leaders and already acting as policy shapers (Baird 2015), which is relevant to this study as they will increasingly be owning the selection and reporting of SDG priorities on their projects.

5.4 DATA ANALYSIS AND RESULTS

5.4.1 Descriptive statistics

Using an anonymous data collection approach, 325 survey responses were received during June 2018; 24 of which were corporate responses, with an average completion time of 7 minutes. There was a minimum of 159 answers for each survey response from individuals. From the 301 responses from individuals, 81% (243) were from qualified engineers, of which 45% had over 20 years of experience. When all the years of experience were added together it provided a cumulative total of 3,628 years of professional engineering expertise, not including the non-engineers that covered professions ranging from lawyers, investment specialists and academics. This indicates that although the total number of survey respondents was limited to the 325, it did include a high level of expertise that adds to the weight and power, and thus, validity for the research findings (Diekhoff, 1992). The descriptive statistical data is shown in the tables and charts in Figure 39. Also, the data showed that 40% of the respondents were millennials.

Europe		176 resp	onses		Carl Row Prover	and an and a sector	1. 160	
Africa		0 respon	nses	Rese	arch Survey – En	gineering SDG Impa	ct Response	s
Middle Ea	ast	33 respo	onses					500
Indian Su	b-contine	nt 17 respo	onses	24	176	· Starter	43	25
Asia		43 respo	onses					
Australas	ia	11 respo	onses					and the second s
North Am	nerica	24 respo	onses		Sa ?		· ~ ~	11
Latin America		21 respo	onses	2		and a		~~
			(a). Geog	raphical re	gion.			
Qualified engineer					819	%		
	Engi	neer workir	ng towar	ds chartered status 4%				
Studen		ent enginee	r	10%				
Acaden		demic / educ	cator	2%				
F	Sup	oly chain				0%	,	
F	Rela	ted engineer	ring busi	ness		3%	,	
		0	(b). Ty	oes of respo	ondents.	•		
1–5 years	20%	60%		_		45%		
6–10 years	11%	40%						
11–20 years	10%	20%	20%	1156	10%		14%	
20+ years	45%	075						
N/A	14%		1-5 years	6-10 years	11-20 years	20+ years	N/A	
			(c). Ye	ears of expe	rience.			

Figure 39: Statistical Descriptive Data from Survey.

5.5 SURVEY RESULTS

Responses from the survey's primary C-M-O related questions are shown below, with summary data and supporting analysis.

5.5.1 Question 1 (Outcomes): Should engineering businesses seek ways to measure and report SDG impact?

Data overview: The first set of results show that there was overwhelming agreement that it was important that engineering businesses seek ways to measure and report SDG impact. 87% of respondents either agreed or agreed strongly that this was important. Millennials rated this as more important than non-millennials (94% and 82% respectively) (see Figure 40).

Relation to the research question: the data suggests that engineers rate the use of SDG for measuring impact as important and this provides a starting point for delving deeper into the context, mechanism and outcomes issues that affect its application.

Differences analysis: there were few differences, and this strengthened the findings.



Figure 40: Response to Question 1: Should engineering businesses seek ways to measure and report SDG impact?

5.5.2 Question 2 (Outcomes): What are the top 5 SDG goals most relevant to measuring impact of your infrastructure projects and programmes?

Data overview: The survey results showed that engineers have a strong focus on five priority SDGs (shown in Figure 41), namely: SDG6 (clean water and sanitation), SDG7 (affordable and clean energy), SDG9 (industry, innovation, and infrastructure), SD11 (sustainable cities and communities), and SDG13 (climate change).

Relation to the research question: The data suggests that engineers have a priority for measuring 5 SDGs (5,7,9,11 and 13), followed by SDGs 15 and 12. It strengthens the view that MISI is valued by engineers.



Figure 41: Responses of participants to their five top SDGs that engineering projects should measure impact against – showing the top seven list (with six and seven being significantly less popular).

Differences analysis: The results also showed that there was a marked difference in millennial responses as shown in Figure 42.



Figure 42: Preference of the SDG 13 and 15; differentiating between millennials.

5.5.3 Question 3 (Mechanism): Do commercial realities dictate the SDGs you pick?

The next question probed the way in which commercial realities influence the selection of SDGs to measure (see Figure 43).



Figure 43. Response to Question 3: Do commercial realities dictate the SDGs you pick?

Data overview: Respondents were nearly equally split on this issue; 36% either disagreed or strongly disagreed that SDG choice was influenced by commercial realities, while 39% agreed or strongly agreed, with 21% non-committal.

Relation to the research question: The responses can be linked to the second question that seeks to identify problem areas. There is evidently an issue that many engineers feel that the businesses they work for can be too commercially orientated. This is not a consistent view, but potentially an area for further exploration.

Differences analysis: It is likely that the 'agree' type of responses, as well as the non-committal responses, could reflect the difficulty of interpreting the question. For example, if the respondent identified that the use of SDGs was a secondary consideration after ensuring 'business survival' they might have agreed with the proposition, whereas an alternative position might have been to suggest SDGs are good for business based on the wider 'societal shared values'.

5.5.4 Question 4 (Mechanism): Do you want to know more about measuring SDG impact on your projects?

Data overview: In addition, the overwhelming majority of engineers wanted to know more about how to measure SDG impact on their projects better (83% vs. 17%), especially among the millennial generation (see Figure 44).

Relation to the research question: this supports the rationale for deepening the research and provides a useful way to seek interviews with CEOs in the follow-on stage of this research programme.

Difference analysis: there were few differences which strengthened the findings and provided evidence to further evaluation into this area in future research.



Figure 44. Response to Question 4: Do you want to know more about measuring SDG impact on your projects?

Data overview: 87% of respondents either agreed or agreed strongly that this was important. Again, millennials rated this more importantly, at 94% versus 82% for non-millennials (see Figure 44).

Relation to the research question: The survey respondents gave very strong support to the view that it was important that engineering businesses seek ways to measure and report SDG impact. This consolidated the views expressed in earlier questions that this was a matter of import and that they wanted to be engaged in future

knowledge and learning activities. The responses would help galvanise future engagement in this research project.

Difference analysis: millennials were stronger in their responses but not at a significant level.

5.5.5 Question 5 (Mechanism): What is the engineers' view on current infrastructure projects and their achievement of the SDGs?

Data overview: Only 34% of engineers believed that 'there is strong evidence that we have a 'fit for purpose' SDG measuring approach to track our projects' impact on SDGs. 37% neither disagreed nor agreed, probably due to the fact that it is such a complex and difficult challenge to measure impact and to-date, the industry continues to struggle to find a practical and workable solution to this issue (Merry, 2019; Fukunda-Parr and McNeill, 2019).

Relation to the research question: The results reveal some areas where the current measurement of projects' SDG impact needs improvement (see Figure 45). Despite the strong support for the importance of measuring and reporting SDG impact, and the clear identification of five priority SDGs for the sector.

Difference analysis: there were significant differences but across a balanced response curve. This could be interpreted positively (only 24% did not agree) or negatively (only 34% agreed).



Figure 45. Response to Question 5: Do we have a 'fit for purpose' measuring approach to track projects' SDG impact?

5.5.6 Question 6 (Context): What are the greatest challenges for measuring SDG impact?

Data overview: The respondents to the exploratory survey said that the four greatest challenges were (see Figure 46): success definition (56%), business priorities (55%), leadership (52%), and a focus on outputs rather than outcomes (46%).

Relation to the research question: these responses address the aim to identify contextual issues with strong linkage to the study's C-M-O 'Variables Framework'. This starts to build a nodal framework for further investigation in the interview stage and also provides links to the 'outcomes' issues.

Difference analysis: the four issues with the highest incidence are all within 42-58%. Each of the margins between the issues is 4-6%, which shows a balanced and consistent view without outliers.



Figure 46. Responses to Question 6: What are the greatest challenges for measuring SDG impact?

The four top challenges identified in the results can also be interpreted as reflecting the difficulties of integrating business needs with the SDGs in the absence of shared value business strategies, although 'success definition' could also reflect the lack of KPIs to measure SDG performance on engineering projects.

5.5.7 Question 7 (Context): How could the achievement of the SDGs on future infrastructure projects be improved?

Data overview: This exploratory research shows that in terms of the greatest opportunities within engineering firms, the top four opportunities were leadership

(57%), increased education and training in SDG impact skills (57%), use of a simple and widely used tool (55%), and business skills (48%).

Relation to the research question: The penultimate set of results reveals some initial views about how the performance of future projects against SDG targets can be improved (see Figure 47 below). This responds to elements of the research question (2 and 2.1-2.3) and highlights consistency with the earlier question shown in the previous paragraph. Notable alignment is in three areas: leadership; business understanding of success; and further investment in education.



Figure 47. Responses to Question 7: What are the greatest opportunities for measuring SDG impact?

The challenges were also compared to the 'opportunities' in Figure 48 to assess the respondent's understanding of whether the same themes were noted as both a challenge and an opportunity. There were data linkages between the two results, with 'leadership' and 'business skills/success definition' appearing in both of the top four responses.



Figure 48. Responses to Question 6 (opportunities) and 7 (challenges) for measuring SDG impact? (See Appendix 5 for full data)

5.6 INFERENTIAL STATISTICS

Chi-Square analysis provided a p-value of between 0.001 and 0.132 (see Appendix 5). Therefore, the statistical validation was not found to be consistent and as such the findings should be viewed as an indicator of where further research can be focused. However, the survey, even without feedback from the 40 interviews with CEOs and Heads of Sustainability (discussed in the next chapter), has provided increased confidence that there is a gap between current practices and engineers' strong desire to improve how sustainability is addressed in the context of infrastructure projects.

5.7 DISCUSSION OF SURVEY RESULTS AND DEVELOPMENT OF THEORETICAL FRAMEWORK

The survey captured engineers' views on the measurement of SDG impact on projects. Therefore, it provided an individualistic and rich perspective on the views of engineers on the importance of SDG measurement. The survey also specifically examined the mechanism and context of SDG measurement to identify strengths and weaknesses of employing such an approach. In doing so, the survey identified that the overwhelming majority (87%) of engineers surveyed have a strong appetite for action on the SDGs. From the engineers surveyed, millennial engineers are 15-20% more likely than non-millennials to want to work on projects that deliver according to the SDGs. However, this strong focus and desire, almost commitment, to the SDGs and their materialisation was accompanied by an equally strong frustration with the lack of solutions that are fit for purpose and currently available for use in industry.

5.7.1 Analysis of Results and Further Development of the Research Model

The survey highlighted four **findings** to improve the effectiveness of SDG measurement at the sub-national levels: (1) tools and methods, (2) suitable training geared towards an understanding of the SDGs, (3) definition of business success that differentiates between outputs and outcomes, and finally, (4) the most frequently identified success factor was the leadership and governance tailored to driving change under the SDG framework. This led to eight core themes that emerged from the results, which are discussed within the three categories of Context, Mechanism and Outcomes, consistent with the C–M–O Realist Evaluation approach (Zachariadis et al., 2013). The first category of *Context* had four emergent themes: leadership & governance (C1); business skills for engineers (C2); performance measurement tools (C3); and, millennials (C4). The second category of *Mechanism* had three emergent themes: Prioritisation of SDGs (M1); Adapting to Levels of P3M Complexity (M2). The third category was *Outcomes*, which had two themes: outputs versus outcomes as definition of project success (O1); and, adopting a Creating Shared Value approach using the Theory of Change and the Triple Bottom Line (O2).

5.7.2 Context

5.7.2.1 Leadership—Governance (C1)

Unsurprisingly, leadership was identified as a dominant contextual issue in the measurement of SDG impacts at organisational and project levels. Whilst the results from the engineers' survey are insightful, the data needed triangulation by seeking the senior leaders' attitudes in the follow-on interview stage of this research. This captured views of 40 CEOs and senior executives to gain an organisational perspective on the dominant contextual issues. It was expected that leadership would be a prominent

theme because of its role in transformation, which lies at the core of adopting the SDG lens to measure business and project success.

As Metcalf and Benn noted (2013), leadership is a well-recognised success factor in the implementation of sustainability and CSR in businesses. The SDGs have similar complexity patterns to sustainability, with an equal need for adaptive systems that place an extraordinary demand on leaders (Metcalf and Benn, 2013). These leaders need the skill sets that can balance the complexities of achieving the economic business success of profitability with the increasing demands on co-balancing with environmental and social objectives. The visionary leaders will be the ones that harness the CSV mindset and then can empower and align their organisations with the people, profit and planet thematics of TBL, thinking and acting within 'systems of systems' models that seek innovative solutions to the SDG challenges (Fullan 2005).

5.7.2.2 Business Skills for Engineers (C2)

Following on from the previous theme of defining success through outcomes and not just through the traditional outputs of time, cost and scope (and quality), the survey results also indicated the need to build capability and capacity amongst engineers. This complements the work by Zahra et al. (2006) into entrepreneurship and dynamic capabilities needed of engineers beyond their traditional technical expertise. Others (Armanios et al., 2017) proposed that the broadening of skill sets to include business skills of innovation and definition of broader TBL success will play a more dominant part of the education and learning syllabus in future. In this regard, it was suggested by a number of respondents that embedding business skills learning within core engineering educational programmes would help provide opportunities for meaningful improvements in the measurement of SDG performance on projects.

5.7.2.3 Measurement Tools and Processes (Methodologies) (C3)

There are many authors that recognise the need for using benefits and impact performance measurement to ensure strategic plans are delivered effectively and efficiently (Micheli and Manzoni, 2010; Prasad et al., 2003). The survey responses also showed that there is significant room for improvement on availability of 'fit for purpose' engineering tools and methodologies to measure SDGs. These results highlight the need for a new simple tool (such as a set of KPIs linked to the SDG indicators) developed with a global sustainability body, such as the Global Reporting Initiative (GRI). This might allow the engineering community to align projects' SDG reporting with the growing trend of using global standards to report sustainability, with 93% of the world's largest 250 corporations reporting on their sustainability performance in 2018 and 82% (Global Reporting Initiative, 2016) of these using GRI Sustainability Reporting Guidelines to do so.

5.7.2.4 Millennials (C4)

In addressing the research question on engineers' use and rating of SDG measurement at local levels, a further investigation was carried out to explore potential differences between millennials and others. In a highly encouraging note, the survey results indicated that millennial engineers are 15-20% more likely than nonmillennials to want to work on projects that deliver the SDGs. This is a key insight for engineering business leaders to take note of (such as CEOs and other managers) because in the UK alone, 50,000 engineers (all forms) will have to be recruited per year until 2022 to meet the projected level of demand for qualified engineers (ICE, 2018). These talent management priorities will be important to business resilience in the near future. Indeed, the identified lack of alignment between millennials' perceptions of businesses' motivations balanced against their own imperatives is typically shown in their allegiance to employers (Deloitte, 2018b). Simply stated, if businesses do not make a greater effort to demonstrate their ability to create shared value that achieves financial, but also environmental and social, outcomes, then it will not engender loyalty. In turn, this may result in higher staff turnover, thereby damaging its business interests as well as negatively impacting the projects' delivery of SDGs.

The survey results are further evidenced by a report from Deloitte (2018a) on findings from over 10,000 millennials. The Deloitte (2018b) survey indicated a distinct, negative shift in millennials' attitudes on CEO's and business' motivations and ethics. Today, less than half of millennials believe businesses operate ethically (48% balanced against 65% in 2017) as well as a drop in the number of them that believe that CEOs are committed to supporting society (47% balanced against 62% in 2017). This highlights that there is a significant mismatch between what millennials define as responsible organisations and the people that lead them, in terms of what responsible businesses should aspire to achieve. The message is clear that millennials

want business leaders to be proactive about making a positive impact on society. Measuring projects' SDG impact is a way to do this.

5.7.3 Mechanism

5.7.3.1 Prioritisation of SDGs (M1)

The survey highlighted a clear preference for measuring just five SDG goals (there was a 50% reduction in the preferences for the next two SDG goals). This indicated a long tail of ten further goals that did not appear to resonate with participants. The literature review also noted that there is a growing body of evidence (Allen et al., 2019; Bali Swain and Yang-Wallentin, 2020; Jones and Comfort, 2020) that suggests that the complexity of the 17 Goals and the 169 targets needs to be simplified and a reduced selection prioritised for measurement. Combined, the findings from the survey on the top five SDGs and the complexity noted in the literature review is consistent with the advice given by the UN Global Compact in their proposed methodology (GRI, 2015). This indicated an important area for further research to assess how this simplification can be achieved at organisational and project levels.

5.7.3.2 Organisational, Portfolio, Programme and Project (P3M) Complexity (M2)

The survey questions did not specifically address the separate levels of organisations and projects, which in the project management discipline, is referred to as portfolio–programme–project management (P3M) levels. This becomes important in the development of a prototype since the measurement of SDGs at the enterprise level (the portfolio) will likely be different from that at programme and project levels. This required further exploration in the interview stage (next chapter). The non-response error highlighted that the survey of organisational level had not achieved its objectives and that this should be addressed in the next research stage of the forty interviews, which introduced the wider issues of the Muller OPM (2019) structure.

5.7.4 Outcomes

5.7.4.1 Defining Success—Outputs Versus Outcome (O1)

The survey showed that engineers agreed that too often, projects define success by traditional outputs using the so-called 'iron triangle' of time/cost/scope (and quality) dimensions to deliver on the SDG goals. Instead, the majority of survey respondents agreed that the engineering and project management communities need to place a greater emphasis on the achievement of long-term outcomes and a corresponding broader definition of success.

5.7.4.2 Creating Shared Value using Theory of Change and Triple Bottom Line (O2)

As noted above (Section 5.4.1), the survey showed that many engineers agreed that the choice of SDG goals and targets should be primarily selected on the basis of business profitability. This is counter to CSV and TBL. As a result, the longer-term value of making investment decisions based on broader TBL principles could be weakened. Therefore the next research stage, to inform the development of the prototype, investigated how the TBL could be integrated with the measurement of SDGs.

5.8 DEVELOPMENT OF CONCEPTUAL FRAMEWORK FOR MEASURING THE SDG PERFORMANCE OF INFRASTRUCTURE PROJECTS

The identification of the four context issues discussed above offers insights into the situational effects on the likely success of the mechanism. Stated slightly differently, for the measurement of SDG impacts to deliver the outcomes expected, the contextual situation needs to be appropriate. For example, without the requisite organisational leadership inter- and intra-businesses, from the government as well as executives, the measurement of SDGs will not be successful. This survey therefore informed the next stage of the Sequential Explanatory Design (Creswell and Creswell, 2017) that involved interviews of forty senior executives. It was proposed to deepen the exploration of the four context issues by using a nodal coding system (Creswell and Creswell, 2017) for the interviews. This strengthened the analysis and, combined with the literature review to enable the triangulation of data, enabled the development of a robust prototype to test the approach with practitioners.

Four of the mechanism and outcome C–M–O themes, noted in Sections 5.3 and 5.4, are further explored below in Table 12 to identify themes that could be included in the next study stage. As a deduction, it was posited that the shared value approach aligns individual business priorities of specific firms with sustainable development imperatives. Consequently, adopting an enhanced SDG measurement approach is

capable of releasing the energies of businesses to pursue competitive advantage and the SDGs through integrated business strategies.

As a way of summarising the conceptual development of a prototype SDG measurement model (started in Chapter 3, but further refined in Chapter 7), based on the literature review and survey findings presented in this study, a tabulated compendium is shown below that includes some exploratory questions, with derived findings and supporting literature. This has been used to shape the design of the interview questions, shown later in Chapter 6, Figure 51.

Table 12: Summary of conceptual development of a future prototype model for SDG measurement, based on the survey results in this study.

C–M–O Future Research Focus	Findings Derived from Stage 1 Research to inform Prototype development	Supporting evidence from the Literature Review
Prioritisation of SDGs (M1)	 Only a small proportion of the 1554 SDG indicators are currently being measured at the project level, and consequently, there is a large gap between global definitions of SDG objectives and project-level definitions of action. The evidence of the difficulty to use the existing 169 targets and 232 indicators suggests that the derived model should recognise that a contextual perspective needs to be adopted to keep it simple for practitioners who are already heavily committed to other performance measurement frameworks. 	Klopp and Petretta 2017; Donohue et al. 2016; Nerini et al. 2018; Allen et al. 2016; IPCC 2018; Swain 2018; UN 2018; Hall et al. 2016; Martens and Carvalho 2016a, 2016b
Organisation al P3M Complexity and Sustainability (M2)	 Measurement of SDG performance should accommodate the required different organisational levels, namely portfolio, programme and project levels. That existing sustainability measurement at organisational and project levels is well established, and therefore, SDG measurement should be aligned to existing successful approaches, not created as an 'add-on' (i.e., the organisational level will likely have different SDG imperatives and reporting requirements, such as using the GRI, from the project level, which might have limited capability and capacity to track too many targets and indicators.) 	Morris 2013; Cooke- Davies 2007; Morris 2017; NAO Report Projects (National Audit Office 2005); Silvius and Schipper 2014; Silvius et al. 2017; Martens and Carvalho 2016; Økland 2015; Silvius and Schipper 2014; APM 2019; Sawaf and Gabrielle 2014; Bonini and Emerson (2005); Bugg-Levine and Emerson 2011; Preston 2012; Malhi et al. 2009; Suess 1980; Tilt 2009; Perrini and Tencati 2006; Kaplan and Norton 1996
Defining Success— Outputs	1. Measurement of SDG performance should be viewed from a systemic perspective and thereby move beyond the traditional 'iron	Theory of Change and Logic Model: Stein and

Versus Outcome (01)	2	triangle' view of projects in the short term (i.e., according to schedule, budget, scope and quality performance) and additionally, take account of longer-term project outcomes and impacts.	Valters 2012; Weiss 1995; Project Success : Thiry 2004; Themistocleous and Wearne 2000
	2.	the Theory of Change and the Logic Model, with their focus on outcomes measurement, including the analysis of causal linkages, engagement of stakeholders and strategic design with the 'ends' being the starting point for a right to left causal mapping.	
Creating Shared Value using Triple Bottom Line (O2)	1. 2.	Measurement of SDG performance should accommodate the perspective of Creating Shared Value (CSV) (i.e., seeking solutions that are good for business in the short and longer term through balance of profit–planet– people objectives). Measurement of SDG performance should accommodate the perspective of the Triple Bottom Line (i.e., social, environmental and economic performance). This will drive a broader definition of project sustainability that includes the three pillars (i.e., social, environmental and economic performance). It provides simplicity and structure for the analysis in regard to selecting and measuring SDGs	Creating Shared Value : Porter and Kramer 2011; Elkington 1994, 2018; OECD 2019; UN 2018; Triple Bottom Line : Elkington 1994, 2018, 2018; Griggs et al. 2013

5.9 LIMITATIONS OF SURVEY

A potential limitation of the study may have emerged as originally the survey was intended to capture attitudes of both individual engineers as well as organisations. However, this was not successful because respondents were unable (with only five exceptions) to provide an authoritative organisational perspective. This was a form of non-response error (Singleton and Straits 2010), and the strategy to reduce this error was to firstly diagnose the problem and then find ways to mitigate the error. The reasons given when the researcher followed up with a few known participants who had waivered their anonymity and volunteered their feedback was that no official statement would be given by large organisations on a survey without having secured senior leadership sanction. These organisations were later approached at the interview stage of the research study, which involved CEOs and Heads of Sustainability who had the authority to provide a corporate statement on their organisations' SDG measurement strategy. A further limitation might be considered that a full pilot was not included prior to the full go-live of the on-line survey. This might have mitigated the 'non-response' limitation described above. In actuality, the 'Step 5 Specify and test survey Procedure', detailed in Table 11, included the plan to build the logic framework in the chosen survey tool and once completed, to run 'check tests' with informed analysts to ensure the technical success and the logic of the procedure. This allowed the full process of the respondent interface to be checked. Whilst, this preliminary procedure was completed, the subsequent problem with the lack of organisations responding was not identified and provided a useful learning for the researcher.

Singleton and Straits (2010) highlight the need for self-awareness of bias when using surveys and actively address these from the start. As an example, as shown in Table 11, the survey approach addressed four known biases: (1) asking the wrong question, which was addressed by testing the questions in a pilot stage and getting feedback and adapting where necessary; (2) surveying an inadequate sample, which was addressed by partnering with the ICE to benefit from a defined group of engineers (they were all active members of a global standards body), although 'opting-in' meant that the sample was potentially biased in favour of the survey due to participants being more aware of sustainable development; (3) the single nature of the survey format did not allow for a free-flowing of ideas, and the time restraint of making it relevant and accessible to busy professionals to be part of the survey meant it lacked the depth of separate interviews. This was balanced by having a follow-on phase of 40 interviews with construction company CEOs; (4) misrepresenting the data results, which was addressed by having forums hosted at the ICE to share back the findings in discussions (as described in Stage 4 of the Analysis phase in Table 11). This feedback was used to shape the design requirements of the follow-on interviews.

The resultant view against the C-M-O framework after the survey is shown below in Figure 49, as an illustrative subjective analysis based on the researcher's interpretation of the survey's quantitative results. The qualitative maturity of the C-M-O evidence is shown by using 'Harvey Balls' to aid visual communication of qualitative information. They are commonly used in comparison tables to indicate the degree to which a particular item meets a particular criterion, and users of them, such as Groenland (2016) suggest that they discourage readers of a report from '*attributing quantitative meanings to qualitative data*'.



Figure 49. View 1 - Maturity of C-M-O evidence across variables and study areas.

5.10 CONCLUSIONS AND INPUT TO NEXT STAGE OF STUDY

This chapter started from the premise that there is a problem with linking the SDG global-level goals to the local-level delivery on infrastructure projects. This potentially manifests in poorer investment decisions on infrastructure projects because they are too often based on purely financial return on investment (RoI) instead of a broader set of criteria across the TBL of economic, environment and society. In order to explore the identified gap in the literature, the study adopted a mixed-method approach of a Sequential Explanatory Design (Creswell and Creswell, 2017). This addresses the first stage, a survey of 325 engineers, to answer the research question of 'How do engineers in the construction sector rate and use global UN SDG goals for infrastructure investment decisions at local level? To derive the answer, it used the Realist Evaluation methodology of the Context–Mechanism–Outcomes model (Pawson and Tilley, 1997) to structure and evaluate practitioners' views on using SDGs to measure local success.

Despite limitation of the sample (all 325 ICE members 'opted-in', inferring interest), the results of the survey show that the vast majority (87%) of engineers surveyed have a strong appetite for action on the SDGs. From the engineers surveyed, millennial engineers are 15–20% more likely than non-millennials to want to work on

projects that deliver the SDGs. However, this strong focus and desire, almost commitment, to the SDGs and their materialisation was accompanied by a strong frustration with the lack of solutions that are fit for purpose. This stage of the research identified that there is a gap between their perceived importance of measuring SDG impact, contrasted with their current capabilities (such as skills, knowledge, leadership, tools and approaches) to do so.

The limitations of this exploratory research are that it has not provided definitive findings from the perspective of organisations. However, it helped to narrow the scope of future research by establishing priorities for the next stage and a signpost for further analysis that deepens the research. Through use of the refined research framework developed from this survey, with key assumptions and derived research questions, there was an opportunity to deepen understanding of this important area in the interview stage. This guided the next stage of the forty interviews of engineering organisations' CEOs and Heads of Sustainability, discussed in the next chapter.

The building of the evidence to respond to the study's propositions is shown below:

Confidence in	P1:	P2:	P3:	P4:
Evidence for	Engineer's view of	Organisational	'Golden Thread'	Prototype
Propositions	current MISI	view of MISI	global - to - local	Development
Theoretical	Not yet	Not yet	Very Low	Very Low
Analysis				
Survey Analysis	Moderate	Very Low	Very Low	Very Low
Interviews	Not yet	Not yet	Not yet	Not yet
Analysis				
Test 1: Golden	Not yet	Not yet	Not yet	Not yet
Thread				
Test 2: Case	Not yet	Not yet	Not yet	Not yet
Study				

Table 13: Development (2nd) stability of Proposition Results (changes since previous Chap in red)

(Note 1: The assessment of the quality of propositions' evidence is based on a qualitative formula designed by the Clinical Grading Working Group's British Medical Journal's study (GRADE, 2004). Full description in shown in Table 8.)

Chapter 6: Interview Method, Results and Analysis

6.1 OUTLINE OF CHAPTER

This chapter describes the design adopted for the interviews to achieve the aims and objectives stated at the end of Chapter 1. Section 6.2 discusses the methodology used for this second part of the Sequential Explanatory Design (Creswell, 2017), the interviews of 40 CEOs and Heads of Sustainability, the stages by which the methodology was implemented, and the research design; section 6.3 details the data analysis methodology, followed by section 6.4 that provides more detailed description of the interview analysis process. Verification is discussed in Sect 6.5, with the results and discussion included in Sections 6.6 and 6.7, with the conclusions in Section 6.8.

6.2 METHODOLOGY AND RESEARCH DESIGN

6.2.1 Methodology

The broader research design, outlined in Chapter 4, involved a three-way data collection approach (Figure 37). At its core, the research design built on the triangulation of qualitative and quantitative datasets, which is well recognised as a method for informing theory-led research development (Creswell, 2017; Easterby-Smith et al., 2002). In what Creswell (2017) describes as a Sequential Explanatory Design, the literature review informed the survey questions and analysis that has informed the structure and approach of the interviews discussed in this chapter. In this way Merriam and Grenier (2019) suggest that 'the interviews help the researcher understand the responses to the survey (Mansell et al., 2020) as well as provide additional insights into the phenomenon of interest'.



Figure 50: The research design of mixed method Sequential Explanatory Design, adapted from Creswell (2003 and 2017).

As shown in Figure 50, the development of a prototype SDG measurement model was to be based on the triangulation of learning from the literature review, the survey of 325 engineers and the subsequent interviews of 40 senior executives.

This chapter discusses the interview stage in-depth. A primary advantage of the semi-structured interview method is that it allows an adaptive-responsive approach to ensure the best improvisation to delve deeper into relative areas of importance, based on the Participant's responses (Hardon et al. 2004, Rubin & Rubin 2005, Polit & Beck 2010) and it also allows for participants' verbal expressions to be captured (Robert Wood Johnson Foundation, 2008).

6.2.2 Interview Question Design

The semi-structured interviews were designed to explore attitudes in relation to the research question and its subsidiary three sub-questions (shown in Figure 51). The sub questions focused on three areas: the perceived value and importance of measuring SDGs (i.e. the outcomes); their current approach and capability (i.e. the mechanism); and, their identification of the challenges and opportunities (i.e. the context) such as skills, tools, processes, structures and methods (Astbury and Leeuw, 2010: 386). NVivo© was chosen as the web-enabled data collection tool. The selected organisations all 'opted-in' to the interview process and given the seniority of the participants, they had a very strong understanding of their company's strategic vision and the marketplace that they operate in which strengthened their ability to comment

on the value of SDG measurement. They were all prominent in their field and as a result of agreeing to be interviewed, would have ensured that they understood the nature of SDGs and their relevance to their business. For example, the CEO of a transport mega project admitted that his company had not actively used SDG for measuring success, but that the request for an interview had piqued his interest and as a result he was very interested in finding out what others in his sector were doing in this area.





6.2.3 Derivation of the Questions

The questions that are shown in Figure 51 were derived from a variety of sources, both inductively and deductively. The central research question was informed by the literature review, which highlighted a knowledge gap. The importance of understanding why the gap existed and how to close the gap had also been identified in the survey of 325 engineers (discussed in the previous chapter), in which 88% of responses affirmed that stakeholders wanted to increase their ability to measure SDGs on projects. This was strengthened by a response rate of only 34% stating that they had a "fit-for-purpose" mechanism to measure the SDG impacts (Mansell et al., 2020c). The sub-questions 1–3 shown in Figure 51 were derived from the adoption of the realist evaluation's context–mechanism–outcome (C–M–O) configuration (Pawson and Tilley, 1997), which is widely used across clinical research (Pawson et

al., 2005) and increasingly also across the social sciences (Linsley et al., 2015). As stated earlier, Pawson and Tilley specifically recommend the C–M–O strategy so that "programme theories can be tested for the purposes of refining them" (2005, p. 2). The third level of questions for the interviews (shown in the right column in Figure 51) combines the Pawson and Tilley C–M–O framework (1997, 2005) with the survey results (Mansell et al., 2020c). For example, the four contextual questions that were derived from the SWOT (strength, weakness, opportunity and threat) analysis approach discussed earlier, were all topical responses from the surveys that engineers had identified as either "blockers" or opportunities (Mansell et al., 2020c).

6.2.4 Access

The interviews aimed to gain access to 40 CEOs or heads of sustainability. Given the GDPR issues around accessing the names of the senior executives of global companies, the research partnered with the Institution of Civil Engineers (ICE). The ICE vetted the research scope and agreed to provide the personal data on the basis of the work aligning with GDPR legalities. The author contacted a total of 85 organisations at the level of CEO and heads of sustainability, of which 40 agreed to be interviewed.

6.2.5 Sample Size

As mentioned above, sampling was achieved purposefully by partnering with UK's leading construction standards body, the Institution of Civil Engineers (ICE), to identify and select leaders in construction companies who had demonstrated a willingness to be involved in innovative knowledge development. All the interviewees had significant knowledge of the infrastructure sector, but a few (*ca.* 5-6) did not have the same detailed knowledge of how their organisations proactively engaged with the UN SDGs. The reason for some having lower knowledge of the SDGs was that their SDG strategies and approaches were at an embryonic stage. But most were actively involved, at the Board-Executive level, in the SDG agenda of their organisation and were rallying stakeholder commitment in this regard. Given the few that had less knowledge, the sample included 30% (12 of the 40) Heads of Sustainability, who had the requisite detailed knowledge of measurement methodologies and approaches.

6.3 DATA ANALYSIS

6.3.1 Descriptive Statistics

The 40 interviews were conducted between July and September 2018, although two of the interviews had to be cancelled and the participants submitted their answers in writing. The sample profile is shown in Table 14.

Participant ID	Role in Company	ole in Company employees)		Length of Interview
1	Board	Other	Other	45
2	Senior executive	10–25k	Global	55
3	Head of sustainability	25–50k	Global	61
4	CEO	1–5k	Regional	42
5	CEO	1–5k	Regional	53
6	Senior executive	1–5k	Regional	53
7	CEO	>50k	Global	40
8	CEO	1–5k	National	42
9	Head of sustainability	1–5k	National	36
10	Senior government or UN policy director	1–5k	National	52
11	Senior executive	1–5k	National	36
12	CEO	5–10k	National	35
13	Senior executive	<1k	National	42
14	CEO	<1k	National	52
15	Head of sustainability	5–10k	Global	56
16	Board	5–10k	Global	56
17	Senior executive	>50k	Global	21
18	Senior government or UN policy director	Other	Other	36
19	Head of sustainability	10–25k	Global	75
20	Head of sustainability	10–25k	Global	55
21	Board	5–10k	Regional	45
22	Head of sustainability	1–5k	Regional	45
23	Head of sustainability	10–25k	Global	45
24	Senior executive	1–5k	Global	39
25	Senior executive	10–25k	National	43
26	Senior government or UN policy director	<1k	Global	38
27	Senior government or UN policy director	other	National	47
28	Senior executive	10–25k	Global	36
29	Head of sustainability	10–25k	National	46
30	Senior government or UN policy director	other	Other	65
31	Senior executive	<1k	National	59
32	Senior executive	<1k	National	59
33	Head of sustainability	10–25k	Global	43
34	Head of sustainability	10–25k	National	44
35	Board	10–25k	National	44
36	Senior executive	other	Global	65

Table 14: Profiles of participants.

37	Board	5–10k	Global	57
38	Head of sustainability	5–10k	Global	57
39	CEO	other	National	Written
40	CEO	other	Global	Written
			Total	1820
			Average	48

The interviewees were representative of firms that mostly had a global or regional footprint (57%) and had staff levels mostly from 1–25,000 (62%), and they were mostly at, or above, senior executive level (defined as having "director" in their role title), including nearly a third at CEO or board level who reflected individuals who could represent their firm's views.



Figure 52. The attributes and values of the forty interviewees.

6.3.2 Development of the Twin-Track Analysis Protocols, Balancing Qualitative with Quantitative Data Collection

As discussed earlier, the preferred approach was aligned to Frels and Onwuegbuzie approach (2013), who had proposed that even within a specific method choice, such as interviews that are qualitative-dominant, it is appropriate to collect quantitative data during the qualitative interview process. The practical application of the "qualitative-dominant crossover" is shown below in Figure 53, which illustrates a twin-track analysis method, which complemented the use of quantitative and qualitative data collection.



Figure 53. The twin-track analysis protocols approach: qualitative and quantitative.

6.4 INTERVIEW ANALYSIS PROCESS

All interviews were conducted in person and lasted an average of 48 mins (min = 36; max = 75 mins). With the participants' agreement, interviews were recorded using a digital recorder supplemented with hand-written notes. Later, the transcriptions, using the Trint[©] software tool, were uploaded onto NVivo[©] and were then compared and coded using the qualitative data analysis software.

The data was analysed at two levels. Firstly it was analysed using textual analysis and then secondly it was 'made sense of' by using themes and pattern interpretation. Based on the nodal structure described earlier and using the parent-child branching technique (Figure 51), it provided an efficient and effective mechanism to capture and link themes but did not in itself provide any analysis. The analysis took place using nodal coding, which was aligned to the three research sub-questions, based on the realist evaluation C–M–O thematics (Creswell, 2017), and each transcript was coded at three levels: first, second and third level coding. The frequency of participants' statements that were selected for coding, and also the relative frequency of nodal use, is shown in the bar charts in Appendix 8. These groupings of statements under each node were then analysed for similarities and aligned with emerging themes.

In addition to the primary analysis approach discussed above, the study included text mining analysis. This is a commonly used methodology for social scientists (Mische, 2014) because it enables the researchers to manage and quantify large amounts of data in a very short time.



Figure 54. The nodal framework used for identification of key words aligned to context-mechanism-outcome (C-M-O) (Creswell, 2017).
6.5 VERIFICATION

The verification was completed after the interpretation of the data analysis. This involved presenting the findings in 3 workshops (typically 2 hrs long) hosted by the Institution of Civil Engineers (illustrated by the Blog shown at Annex 6) to leading practitioners and academics, organised by the ICE's Knowledge Team Manager.

6.6 **RESULTS**

The results and discussion are structured in three sections that relate to the three sub questions, as shown in Figure 51, that stem from the primary research question: 'How do senior leaders in the construction sector rate and use global UN SDG goals for infrastructure investment decisions at the local level?'. From this, the derived subsections were as follows.

- Thematic area 1: outcome. What are the expected outcomes of successfully using the SDG measurement mechanism?
- Thematic area 2: mechanism. What design criteria enable the mechanism (for measuring SDG impacts) to achieve the outcomes?
- Thematic area 3: context. What issues influence the successful use of an SDG measurement mechanism to achieve the desired outcomes?

Using the twin-track analysis approach (Figure 53), which includes both the qualitative and quantitative data, results were derived from the combined findings. All participants were asked for their views on the strengths, weaknesses, opportunities and threats (SWOT) of the employment of the MISI mechanism. Given the semi-structured interview approach, their responses did not take a standard route and the interviewer used the funnelling technique (Kallio et al., 2016) to increase subject specificity where depth of answer was required.

6.6.1 Thematic Area 1: Outcome. What Are the Expected Outcomes of Successfully Using the SDG Measurement Mechanism?

The "Outcome" section is the first of three thematic areas that focused on the broader organisational ambitions of sustainability, sustainable development and SDGs. The results were collated under the following headings: the challenge/problem, the opportunities and the imperative for change. This thematic node collected the

second highest (out of 23 nodes and sub nodes) number of references (n = 81) in NVivo for leaders' views on the expected outcomes.

6.6.1.1 The Challenge/Problem

The essence of the problem was articulated by Participant 10: "The weaknesses of the impact measures relate to some of the quantification of it in that there is no standard way of doing it and therefore quantifying impact is very difficult.... The leadership is not fully bought into it. It could be you have not got sufficient good tools for learning and education behind it. There is a lack of consistency in the data of how you measure it and the people measure it in different ways and people will have different perspectives of what good looks like". These views are similar to those of Participant 26, who also noted the level of complexity, especially when positioned in a global context with the inherent cultural variations, which is potentially why so many participants only claimed to measure the SDGs at a high level: "This is so complex and it is so different if we are doing things in different countries with different organisations across different environments".

6.6.1.2 Overarching Opportunity

There were many participants that identified opportunities for improvement, and these are mostly captured under section three on "Contexts". The ambition, noted by many, was summed up by Participant 26, who was from an international organisation and who gave this insight into his global organisation's aim: "In three years' time we would like to be in a position to have enough information based on evidence and frameworks in place so that we can have better conversations earlier on with clients about what the potential benefits are for the project and why we should be doing projects possibly in a different way than given to us by donors and others". He continued by anticipating the broader causal impacts of having this mechanism in place: "So, if we understand the linkages and contribution projects can have across several SDGs, and how that impact could be measured, then we can have better conversations to understand where people should be investing their money and how, and what other aspects to bring into our project to ensure long-term sustainability". This places emphasis on using the SDGs to make better investment decisions, which becomes one of the critical success factors of the employment of this mechanism. The upbeat message was also shared by Participant 40, a Spanish former President of a world engineering institution: "Without any doubt, today is a window of opportunity for future SDG impact measurements just for the sake of the pure nature of the new technologies we are using, such as nanotechnology, Biotechnology, Artificial technologies, new material, advanced manufacturing ... all of them allow easier measurements procedures."

6.6.1.3 Imperative for Change and Commitment to Measure SDGs

Participant 13 explained the key part that SDGs contribute to the company's approach to the broader sustainability agenda: "*The SDGs and our impacts on them are of huge relevance to our industry. We are already fully committed to measuring our impact across the triage of economic, social and environmental sustainability themes. Our leadership is fully committed to owning delivery success against these targets, which we jointly assess with our tier 1 contractors. It is now considered core business to ensure the right levels of scrutiny and governance to manage sustainable development performance. In future, this will include measurement against SDG targets but, for now, we need to find a practical method for doing this well." The final comment in the extract highlights the difficulty of moving from "knowing to doing". Another Participant shared some important progress for measuring sustainability: "Progress on achieving sustainability targets is improving. Clients and stakeholder are now demanding a higher level or sustainability performance reporting with clear metrics to gauge success."*

Many commented on the link between SDG measurement and their company's values. For example, Participant 5: "because our purpose is far more than simply generating revenues for shareholders... for us, it is about influencing those solutions to provide the right long-term infrastructure for society. So, we provide jobs and the right training, and we provide the infrastructure we need to connect life together; everything we do depends on it—to try to capture the way we go about doing that in more modern ways for future societies". Although many were better able to relate progress stories with their sustainability measurement, there were others, such as Participant 28 who aligned with SDGs: "the whole world has decided how it can be rapidly made better, so the 169 SDG targets are a compass for humanity". This global connectiveness for companies was a common theme, especially amongst the senior stakeholders, such as Participant 19: "The global challenges that have led to the SDGs are the same as those that are affecting the viability and commercial success of

businesses ... the SDGs have recognised we are all in it together – and importantly provide a collective vision of the future that every country has signed up to. This is the 'carrot' for new business opportunity. The 'stick' is the risks that businesses face if they do nothing. The way forwards is Creating Shared Value which is more than Corporate Social Responsibility."

The theme of creating shared value (Porter and Kramer, 2019) was commented on by a number of participants (2, 5, 8, 10, 11, 13 and 19), one of whom, a CEO, commented, "Since becoming Responsible Business of the Year, we have been working hard to show others how sustainability makes good business sense." This quote emphasises that the notion of creating shared value (CSV) (Porter, 1985 and 2011), whilst not always using the specific language of CSV, is a growing reason to engage with SDGs and sustainability more generally. It implies that the business priorities can be balanced; a strategy that focuses on the environment and society, which can equally achieve economic success. When in harmony, real growth is delivered to the benefit of all, as shared by Participant 11: "For example, our approach to 'product lifecycle management' was learned from the aeronautical and automotive industry from 2004-5 and this meant that we looked at the whole life costs, which not only ensured we were more outcomes focused, but by the way, improved our productivity by three percent each year, year on year, highlighting that good sustainable development also made good business sense". The same senior executive also noted "so, we thought long and hard about, not just the goals that we created, but how did that fit with a set of longer-term outcomes in our region and what that would look like in terms of implementation. This was our way of meaningfully connecting the strategy with outcomes that our stakeholders recognised."

The global context and the relationship of the global SDG goals to businesses was a common theme, as indicated by Participant 24, head of infrastructure for his company, who said "*in a world where populations are increasing, cities are expanding and the effect on our environment is more apparent than ever before, the need for infrastructure that is affordable, sustainable and effective is vital. Engineers have a pivotal role to play in designing infrastructure that is not only effective but does not harm the environment in which we live*".

The first major finding derived from this analysis is as follows.

Finding #1: to achieve the outcomes of measuring SDG impacts at subnational level, business priorities can be aligned across economic, environment and society ambitions, and it can make good business sense to do this.

6.6.2 Thematic Area 2: Mechanism. What Mechanism (for Measuring SDG Impacts) is in Place to Achieve the Outcomes?

The second area of discussion was for the Participant to self-assess their company's "awareness and application" and also, if they were applying SDGs, what was the level of process maturity of their SDG measurement. The data in Figure 55 shows the feedback from the participants when they were asked to score themselves against a Likert-style scale, as shown in the first row in columns c and d. The aim of this assessment was to establish a baseline to better understand their level of awareness of SDG measurement and their self-assessed level of process maturity.

а	b	C	d	е	f	g	h
Participant ID	Role in Company	Awareness Vs Application (1= aware & doing; 2= aware & not doing; 3= unaware & not doing)	Your Company's Level of Maturity in SDG Measurement (0=process not developed; 1= definition developing; 2= early processes in place; 3= using sustainable processes)	Participant ID	Role in Company	Awareness Vs Application	Level of Maturity
1	Board	n/a	n/a	21	Board	2	1
2	Senior Executive	1	2	22	Head of Sustainability	2	1
3	Head of Sustainability	1	3	23	Head of Sustainability	2	
4	CEO	2	1	24	Senior Executive	2	0
5	CEO	1	2	25	Senior Executive	1	1
6	Senior Executive	1	2	26	Senior Government or UN Policy Director	2	1
7	CEO		0	27	Senior Government or UN Policy Director	1	n/a
8	CEO	2	0	28	Senior Executive	1	2
9	Head of Sustainability	1	2	29	Head of Sustainability	2	1
10	Senior Government or UN Policy Director	2	1	30	Senior Government or UN Policy Director	2	
11	Senior Executive	1	2	31	Senior Executive	n/a	n/a
12	CEO	1	2	32	Senior Executive	n/a	n/a
13	Senior Executive	2	0	33	Head of Sustainability	2	1
14	CEO	1	1	34	Head of Sustainability	2	0
15	Head of Sustainability	2	1	35	Board	2	2
16	Board	1	1	36	Senior Executive	2	n/a
17	Senior Executive	1	2	37	Head of Sustainability	2	1
18	Senior Government or UN Policy Director	n/a	n/a	38	Head of Sustainability	2	1
19	Head of Sustainability	1	3	39	CEO	n/a	n/a
20	Head of Sustainability	1	3	40	CEO	n/a	n/a

Figure 55. Results of the self-assessed level of awareness-application and process maturity (colour representation shown in columns c and d in titles row).

6.6.2.1 Company's "Awareness and Application" of SDG Measurement in Construction Projects

As part of the interviews, all participants were asked to describe their awareness of sustainability, sustainable development and SDGs. They were then asked to describe their current level of SDG measurement maturity. The data on these are shown in Figure 55.

At the lower end of the spectrum (level 3 = unaware and not doing it), Participant 37 admitted that, regarding "the United Nations Sustainable Development Goals, I had never heard of them—a request for an interview came through and [name withheld] only heard of them through a bid we were working on that included an SDG question. The SDGs have no current place in our business". As this was a board member, this was surprising because it was expected, in the opinion of the researcher, that senior management would have some level of SDG knowledge. But there were others that also had low levels of knowledge, such as Participant 16, from a large global construction firm, who stated "I'm not sure people in the industry seem to be doing this - they are trying to do things responsibly but in a way that fits with their agenda and their clients' agenda. For us we don't utilise the SDGs". Participant 7 was equally open: "I would say we are unaware and not doing it". This honesty was common amongst interviewees, such as from Participant 21: "Have the workforce bought into the sustainability approach and especially the SDGs? Well, I have to be honest, and I will answer this as honestly as possible - so our level of understanding is very low".

In the middle range (which was "aware and not doing" e.g. not doing refers to not measuring the SDGs, so they are aware and doing, but not systematically, including measuring), representing 47% of the participants, Participant 4's answer was typical: "Awareness is that we are doing some discrete things but not in any depth". The reasons for this varied, but a common theme was that there was not a requirement from governments or clients, as Participant 21 shares: "We do not have a demand from our clients or from our communities that we work to measure against the SDGs. Like many in our industry, these are not common terms that we use… we do not have as much benefit from embedding them as much as a large global company that perhaps needs to demonstrate SDG impact more visibly. A lot of the things we do implicitly encompass the SDGs, but we are not explicitly measuring against them".

In the higher range, which was "aware and measuring", represented by 38% of the participants, there were some examples of significant progress, such as that shared by Participant 7: "Every single project in the organisation will feed into SDG number 11—'sustainable cities and communities'—and every project in the organisation will address at least 4-5 of the SDGs". Some qualified their answers to suggest that they are not measuring SDGs to complete a 'tick box' exercise, but instead, as shared by Participant 10: "I think we definitely are aware, definitely doing it, but we are doing it anyway, not because of the SDGs, because it is the ethos of our organisation, and we do it because it is the right thing to do - but the SDGs provides a useful framework to structure and report on this work". However, there were others that identified the SDGs as being a good reason to change the way they reported, as commented by Participant 19: "Since 2000 we have been reporting on sustainability KPIs, that we refreshed in 2014; sustainability is well established. SDGs represent a new chapter and we have made a commitment to SDG and Paris Declaration and Sendai Agreement. Our strapline is 'Shape a Better World' thus it makes sense for us to align with SDGs and spell out what better looks like in 2030. However, the issue of reporting against SDG's is a hornets' nest".



(n = 34 completed answers of 40)	<u>#</u>	<u>%</u>
LA1: Aware of SDGs and measuring	15	37.5
LA2: Aware and not measuring	18	45
LA3. Unaware	1	2.5
n/a	6	15
total	40	

Figure 56. Graphical representation of results from self-assessed level of "awareness-application".

6.6.2.2 Company's Level of SDG Measurement Process Maturity.

As part of the interviews, the second quantitative question all participants were asked was to describe their current level of SDG measurement maturity. The data on these are shown in Figure 56. The banding levels for this question were: 0 = no SDG

processes, 1 = currently defining processes, 2 = early processes in place and 3 = sustainable SDG processes. Overall, the quantitative data showed that nearly half (49%) were at level 0 or level 1, which meant that no mature processes were in operational use. Only 23% stated that they were at level 2, the early adoption stage of processes, with a very small group (8%) stating that they had repeatable processes in place.

One of the best, Participant 13, stated: "We are at Level 3, we have managed processes, metrics and quality management", which was similar to Participant 23: "we have some consistent ways we do things that are aligned to SDGs, but we do not look at every SDG and answer how they contribute to the goals. But we do cover a lot of the issues at project level".

In reality, many of the participants only conducted measurement at a high level, such as Participant 34: "In the past we have done a review to see how our strategy fits with the SDGs. We found that the SDGs were impacted by our work, some more than others, in terms of the goals and targets; they are not particularly relevant to the work that we do so our priorities have been elsewhere and therefore our resources have been focused elsewhere". About a third of the participants said that they could, at a high level, link their SDG priorities to the formal sustainability reporting that they did on the Global Reporting Index (GRI), such as Participant 26, who stated: "Well, we are all aware and starting to do it. We started using the Global Reporting Index framework on sustainability three years ago and we started reporting on our corporate results yearly on that but, at the project level, we have been a bit slower pushing up to that". This theme of doing more measurement at a high level, more at the corporate than the project levels, was also shared by Participant 2, who stated: "We at least have already produced an integrated SDG report that shares material commitments from our business - so we are already using the SDGs as connectors to open opportunities". Amongst the lowest performers was Participant 9, who stated: "in terms of SDG reporting processes we are close to 1. Our maturity is still low, although our sustainability reporting is much higher. We have not yet made it integrated to SDGs and have not yet generated a report against them. That is what we are talking about now and what we want to achieve". This was similar to Participant 23, who also accepted their immature status, but balanced this with a strong ambition to do better: "we have some consistent ways we do things that are aligned to SDGs, but we don't look at every SDG and answer how they contribute to the goals. But we do cover a lot of the issues at project level". At the lowest level, there was also some surprisingly open admissions, such as Participant 26: "Very little going on... in terms of SDG impact measuring on engineering products - very little."

One of the insightful links for increasing SDG measurement maturity was what Participant 2 referred to as a value chain and the need for the government and clients to increase the relative importance to measure against SDGs: "*My ambition is that we are all at level 3 maturity. All the players in the value chain, but this means that clients and governments need to be high up on the scale. I can't drive it from my position*".



	—	
LM0 = processes not developed	7	17.5
LM1 = process definition stage	13	32.5
LM2 = early adoption of processes	9	22.5
LM3 = repeatable processes in place	3	7.5
n/a	8	20
total	40	

Figure 57. Graphical representation of results of the self-assessed level of "SDG measurement maturity" (colour representation shown in Figure 8 in column h in the titles row).

The second major finding derived from this analysis is as follows.

Finding #2: only a small percentage of companies have a repeatable process as an operational "mechanism" for measuring SDG impacts at company and project levels. Most have an aspiration to do so but believe that the government and their clients need to mandate its implementation.

6.6.3 Thematic Area 3: Context. What Issues Influence the Successful Use of an SDG Measurement Mechanism to Achieve the Desired Outcomes?

The analysis of the contextual issues that affect companies' ability to measure SDG impacts successfully were captured using a strength, weakness, opportunity and threat (SWOT) approach. The eight themes are shown in the nodal framework in Figure 54 and include: leadership and strategy; knowledge; outputs-to-outcomes; tools, processes and systems; change management; performance management; project-to-portfolio levels; and geographic issues. These were all derived from the preceding survey of 325 engineers. The qualitative analysis shared below is complemented by using the twin-track approach described earlier, which includes the text-analysis software-enabled word-count data. The approach was to identify key words and relate their frequency of use to the qualitative findings to assist the understanding of the emerging issues. For example, in this first context thematic, "leadership and strategy", the key words associated with this thematic are: leadership (and its derivatives. such as leader), strategy, CEO/executive and align/governance/direction/vision, which are all words associated with leadership capabilities and actions.

6.6.3.1 Leadership and Strategy

For the leadership and strategy node, there were high levels of relevant statements coded (n = 63) from the 40 participants (using the NVivo software), reflecting the importance of this thematic. In terms of key word usage, this thematic was the fifth most frequently used (n = 584) across the 40 interviews, which equates to once every 120 words. Within this category, the frequency of use of "align", "governance", "direction" and "vision" were noted since these words are all associated with leadership capabilities. This potentially reflects the seniority of the participants who all had roles and responsibilities that focused on strategy setting and alignment across their stakeholders, internally and externally. This also partly explains why there was more emphasis on strategic discussions and less on the tools and processes of measurement, that had the second lowest (out of 8) word frequency usage.

The most impactful statements collected were the frequent references to a "greater value" beyond profit. This sentiment sits well with creating shared value and the triple bottom line discussed earlier. This viewpoint was personified by Participant 11: "*a key part of leadership is doing the right thing because it is the right thing to do,*

not because of a box-ticking exercise". The same participant also focused on the difficulty of making the change stick: "It is 50% belief and 50% belligerence when you start something like this; that is, holding yourself and others to account. That is what I mean by belligerence. In other words, 'seeing it through' and what we wrote down as a mantra: 'Don't you understand"'. In his view, as a senior executive, he stressed the important role of his CEO and Board: "Leadership is the most important critical success factor, both internally and externally, to align and galvanise our employees, our communities and the supply chain. It was about getting us all to be more collaborative in finding novel, innovative ways of delivering sustainable solutions.... It is about the leaders capturing the hearts and minds of the stakeholders to champion changed behaviours to achieve big, bold strategic outcomes." He also noted the moral values that are implicit in the choice of making sustainable development a core business priority for his company. The reference to the core principles of governance (OECD, 2015) of accountability, responsibility and transparency were also noted: "a key part of the leadership is the ownership of the sustainable development strategy. It is also about accountability and having the resources to deliver the solution. That is why the 'Infrastructure Clients' are the single most important stakeholders in addressing sustainable development. If they 'own' and champion the solution, then the supply chain will follow... hence leadership and procurement are the biggest elements of the recent Green Construction Board's 'Three Years On Report - Reducing Carbon Reduces Cost' report'' (Green Construction Board, 2015).

In terms of strategy, one organisation noted the importance of the "ends, ways, means" logic similar to the Theory of Change concept (Carol Weiss, 1972, 1983, 1995, 2005). Participant 9 stated: "you must start with the end in mind, even if you have not got a detailed route map to deliver at every stage of the journey. Part of the mantra is to set big audacious goals and then adopt an attitude of 'I have started so I will finish' and, by the way, you never actually finish, because the end goal is moving, it is like you achieve one peak but realise it is a false horizon, and so you continue your climb to the next summit". The value of having clarity of the strategic ends is noted, albeit with a caution that the identification of targets for tracking performance must not become a "box-ticking" exercise that distorts clarity of outcomes. Participant 11 stated: "if you actually begin with the end in mind of the outcome you are seeking and

how you wire your DNA to achieve that, you are far more likely to achieve those outcomes and, in so doing, the boxes get ticked. But if you predicate your thinking with thoughts about just filling the boxes, you have constrained yourself". Most of the participants linked leadership with strategy, as described by Participant 11: "a key part of the leadership is the ownership of the strategy. It is also about accountability and having the resources to deliver the solution. That is why the 'Infrastructure Clients' are the single most important stakeholders in addressing sustainable development. If they 'own' and champion the solution, then the supply chain will follow... hence leadership and procurement are the biggest elements of the recent UK Cost of Construction reports."

Category	C-M-O	Sub-Category	Word	Count	f1 Word %	f2 Sub- Cat %	f3 Cat %
			leadership	83	0.16%	_	
		Leadership as	leaders	30	0.06%	0.29%	0.80%
		theme	leading	20	0.04%		
			leads	15	0.03%		
		Stratogy	strategy	75	0.14%	- 0.18%	
		Strategy	strategic	23	0.04%		
Leadership	Context	ontext	CEO	26	0.05%	- 0.12%	
	_	Leadership role	executive	37	0.07%		
			align	25	0.05%		
		Churche ai a mlammin a	governance	25	0.05%		
		Strategic planning	alignment	18	0.03%	0.21%	
		competency	direction	18	0.03%		
			vision	28	0.05%	-	

Table 15: Text analysis (NVivo) on key words' frequency: context of leadership.

Finding #3: strong leadership plays a significant part in inculcating SDG measurement as an ambition and core value into an organisation.

Finding #4: the more advanced businesses in SDG measurement noted the need to have a clearly defined strategy that can guide the prioritisation of SDG goals using the "ends, ways, means" model. This requires clarity of the "ends" prior to defining project success (in-project and post-project).

6.6.3.2 Knowledge

For the "knowledge" node, there was a relatively smaller incidence (n = 19) of relevant statements coded from the 40 participants (using the NVivo software). In

terms of key word usage, this thematic was also one of the least frequently used, with "learning", "education" and "experience" being used only 140 times across the 40 interviews, which equates to once every 400 words.

The qualitative analysis identified a strong preference for using education and training to improve their staff's SDG impact skills and business skills, especially in the wider definition of success, which is related to the later discussion of outputs-to-outcomes. An indication of the importance of this was provided by the CEO of one global engineering company, Participant 7: "So, how do we galvanise our community, how do we tell our story better against the SDGs and how do we galvanise our community to be able to share best practices, and what does that mean for education and training?". Another, Participant 25, a national utility senior executive noted: "How can we share knowledge and understand what we can learn from each other? Could learn a lot from one another." This illustrated that there is a general consensus that we can all learn from each other, as stated by Participant 3: "we are going through a huge amount of learning ourselves but certainly around measuring impact yeah it's quite an interesting stage at the moment". It was evident that many identified this as a key leadership responsibility, such as Participant 3: "place emphasis on the leadership role, the key leaders in talking about learning and education".

Learning lessons from success and failures was mentioned by Participant 15: "how do we share knowledge - how can we be more effective and efficient - so how can we avoid repeating the same old problems". Similar to learning lessons is having a feedback loop, as one Participant said (23): "so we have a range of communication events, such as sustainability weeks, when we shine a spotlight on how we're doing on those issues".

Skills covered a number of areas, including the skills to be able to define success definitions, business skills to be able to build performance frameworks and sustainability/SDG skills that helped understand the SDG framework and how they relate at sub global-national levels and at organisational and project levels. Participant 3 stressed its import: "*I think the skills piece is the second most important area because we cannot expect our people to deliver on these KPIs if they do not know what they mean and if they do not know how to measure them and improve them, so investing in how to calculate social value and improve upon them and investing in training in*

social value RoI is very important; it gives us an opportunity to benchmark and improve on it".

Overall, participants seemed to accept that, despite the current supposed level of SDG measurement awareness, there is also a shortage of trained personnel to support the implementation of SDG measurement on their construction projects. The closing of this gap reflects the views of Reffat (2004) on the insufficient number of human resources with the required skills to perform sustainable development on construction operations.

Category	С-М-О	Sub-Category	Word	Count	f1 Word %	f2 Sub- Cat %	f3 Cat %
		Terrine Q	learning	30	0.06%		
	Context -	Education	training	22	0.04%	0.15%	- 0.26%
Vnouladaa			education	29	0.05%		
Knowledge		Exportionco	maturity	30	0.06%	0 110/	
		Experience	experience	29	0.05%	0.11 /0	
				140	0.26%		

Table 16: Text analysis (NVivo) on key words' frequency: context of knowledge.

Finding #5: learning and education plays a critical role in increasing capability and, specifically, in understanding how to better share lessons on SDG measurement for the good of all.

6.6.3.3 Outputs-to-outcomes

The "outputs-to-outcomes" node had the fifth highest incidence (n = 30) of relevant statements coded from the 40 participants (using the NVivo software). In terms of key word usage, this thematic was also one of the most frequently used (shown in the "Theory of Change" key word table, Table 17), with the first half of the causal chain (input, activities and outputs) being cited as frequently as the second half of the value chain (outcome to impacts). This was significantly less than the general reference to longer-term benefits that were synonymous with key words such as "value", "ends" and "goals", which were used 339 times across the 40 interviews.

Within this subcode, most recognised the challenge of differentiating between outputs and outcomes. Too few knew how to do this well and, as a result, the wrong "targets indicators" were sometimes being used to measure success. Participant 8, a CEO of one of the UK's largest infrastructure programmes, said: "programme and project people are sometimes less aware of how we are doing strategically if you are not careful. So, they can often have a bias for cost and schedule focus and lose focus on other priorities we have set". Another way of expressing the inappropriate focus on outputs came from Participant 3 (repeated quote): "we know that, if we just design to code, we end up with projects that are great for today but absolutely do not meet the future that we are expecting". This was expressed very clearly by Participant 19: "We also need to talk in terms of outcomes and impacts. We need to change the dialogue of engineers so that they talk less about output and more about outcomes. Engineers typically focus on what we build, eg roads, water systems, buildings etc. We need to build a different narrative to help engineers of the future, and today, to see the role of engineering in society – in a way, it goes back to the early roots of engineering".

Some organisations have fully embraced the strategic aim of better aligning with outcomes, such as Participant 11: "So we thought long and hard not just about the goals that we created but about how they fitted with a set of outcomes in our region and what that would look like in terms of implementation. This was our way of meaningfully connecting the strategy with outcomes that our stakeholders recognised." The same person described the need to look at the end first to better understand ambitions: "you must start with the end in mind, even if you have not got a detailed route map to deliver at every stage of the journey". He goes on to say: "part of the mantra is to set big audacious goals and then adopt an attitude of 'I've started so ill finish' and by the way, you never actually finish, because the end goal is moving, its like you achieve one peak, but realise it is a false horizon, and so you continue your climb to the next summit."

One of the most common reasons for the overemphasis on "outputs" was shared by Participant 26: "So, the measurables are very weak in terms of linking the engineering and the infrastructure impacts to the higher programme. It is just about 'have you built the hospital' as an output". This was expanded on by Participant 36, who gave a useful example of what she meant: "One of the key things as we look at the evidence that's provided in the assessment is that actually deciding whether you're delivering the right solution isn't always at a project level. It's often a much bigger level. So, for instance a project to take a relatively simple example, a project might be 10 miles of road by-passing a village. If you just look at that 10 miles of road, doing a bypass might be deemed to be the wrong thing to do. But if you put it in to the context of Highways England route management plan that 10 miles might be a very critical part of 100-150-mile-long route. And that 10 miles might be absolutely critical to economic growth, solving problems somewhere else. So therefore, the downside to building that village bypass, if you just assess it at a village bypass level it's the wrong thing - they shouldn't be doing it. But if you assess it on a bigger boundary and think about the route, then it puts it into a completely different context". One Participant (27) shared a solution to the problem discussed above: "I think we have our theory of change in place now. People are required to produce logframe to explain how they are feeding into the theory of change". This proposal gives an important insight into how the linkage of outputs to outcomes can be improved: "at the moment we have one annual report which sets out the theory of change of what we're doing, this helps align key partners to what the ends are".

The cost of implementing SDG measurement was another key issue. According to Participant 26, the SDGs are complicated and not easily adopted and since few clients are willing to pay for it, it is deemed unnecessary. This aligns with the Zhou and Lowe (2003) study that noted general agreement that 'green' construction is more expensive when compared with regular buildings and that SDGs might also weaken the bottom line profits. These cost/profit realities were noted by Participant 6: "We all understand that we're in a place where everybody in our market and business is working for the best outcome for the lowest cost. The trick is to define what we mean by best outcomes and lowest cost means." – "lowest cost ends up with high maintenance costs, or multiple defects, and best outcome is rarely the lowest cost - so in the context of sustainability goals, then both those definitions need to have much more nuancing around them, so people who understand why they want to do things rather than just for the purposes of the financial elements, which are part of it of course, but not by any means the entirety of it".

Table 17: Text analysis (NVivo) on key words' frequency: mechanism/context of the Theory of Change.

Category	С-М-О	Sub-Category	Word	Count	f1 Word %	f2 Sub- Cat %	f3 Cat %
Theory of Change,	Machanian		resources	20	0.04%	0.220/	1 269/
(causal logic chain	Mechanism		cost	57	0.11%	0.33 /0	1.20 /0

from inputs to		costs	17	0.03%	
impacts)	To much has A stimution	funding	16	0.03%	
	input to Activity	efficiency	16	0.03%	
	fron triangle of	money	18	0.03%	
	cost-time-scope	commercial	17	0.03%	
		spend	17	0.03%	
	A attivity to Output	initiatives	27	0.05%	
	(time_cost_coope)	activities	15	0.03%	0.12%
	(time, cost, scope)	outputs	19	0.04%	
		outcomes	60	0.11%	
	Outcome and	outcome	34	0.06%	
	benefits as result of	benefits	23	0.04%	0 179/
	change derived from	benefit	19	0.04%	0.17 /0
	project's outputs.	impact	219	0.41%	
		impacts	19	0.04%	
	longer term goals-	value	101	0.19%	
	values at end of	end	73	0.14%	0.64%
	project	goals	165	0.31%	

Finding #6: the use of the log-frame and Theory of Change provides a means to link outputs to link outputs to outcomes and better identify SDG impacts.

6.6.3.4 Tools, Processes and Systems

The "tools, processes and systems" node had one of the lowest incidences (n = 18) of relevant statements coded from the 40 participants (using the NVivo software). This suggests that senior executives and CEOs have less interest in, or place lower value on, specific tools or methodologies, which might indicate why this is an underinvested area. In terms of key word usage, this thematic was also one of the least frequently used, shown at Table 18, with "processes" being cited twice as frequently as "tools" and "systems". In total, they were used only 177 times across the 40 interviews, which equates to once every 300 words.

The survey (Mansell et al., 2020) that preceded these interviews had identified a common reference to the lack of tools, systems and methodologies. This was not proven in the interviews, although a number of the heads of sustainability (3, 9, 15, 20 and 29) were more likely to mention this as a factor. On the ability of the sector to galvanise and align with a consistent approach, Participant 18 highlighted that there were bigger issues to deal with prior to designing a tool: "*I think it is essential. I have very little confidence in our ability to do it now. Even if you had a decent methodology*

now, I suspect very few people would use it and you probably have a number of competing methodologies, which is typical in this sector."

However, others, such as Participant 20, said: "for me the tools and processes underpin the delivery because, without them, you cannot possibly know where you are or where you need to go". This was consistent with Participant 18, who shared their organisation's investment in this area: "we're going to have the new management methodology in place soon, which is improving our ability to provide that consistency in this centrally controlled process and then building into the electronic the enterprise system". This need for investment in tools, was also shared by Participant 1: "tools, a framework and methodology are all needed to actually report against and see some quantitively proven success". But a key element of the design of a tool was to get the balance right between being too complex and being at the other end of the scale being too high level and therefore superficial—as noted by Participant 10: "I think, in most cases, a consistent framework or reporting approach would be helpful; that gets the balance right between having something that is consistent but watered down to such a high level that it loses meaning, versus having too much detail that is too granular, loses the users in too much complexity and is difficult to fit with your business model and the way you report things into that".

Category	C-M-O	Sub- Category	Word	Count	f1 Word %	f2 Sub- Cat %	f3 Cat %
		Tools	tools	32	0.06%	0.06%	
Tools, Systems;	Mechanism	Drug agon	processes	26	0.05%	0.23%	0.34%
Processes		Process	process	93	0.18%		
		Systems	system	26	0.05%	0.05%	-

Table 18: Text analysis (NVivo) on key words' frequency: tool, processes and systems.

Finding #7: the use of tools, systems and processes to measure SDGs is not a priority for CEOs and board members but it is for senior executives and heads of sustainability. These tools need to be simple enough to understand but robust enough to capture detailed evidence that leads to improved performance.

This very interesting finding raises a number of issues such as: why are tools/systems/processes not a priority?; What did they prioritise instead, why, and

what does this tell us about the Change (as regards ToC), about the TBL and the IVC? These will be answered and developed in Chapters 8 and 9.

6.6.3.5 Change Management

The "change management" node had an average level of incidence (n = 27) of relevant statements coded from the 40 participants (using the NVivo software). In terms of key word usage, this thematic (shown in Table 19) tracked "change culture", "behaviours", "innovation" and "communications", all of which provided a large number of insights from participants. In total, they were used 410 times across the 40 interviews. However, the quantification of the data does little to indicate that this contextual issue was one of the best sources of insightful knowledge.

There was general recognition from the participants that the single most important area for ensuring SDG measurement success is having a successful change programme that ensures a practical approach is made to work for the "users", with the added value of what they are doing. The starting point for this approach was ensuring the right culture in the organisation, characterised by openness and honesty about the difficulties of measuring SDGs and also closing the gap between superficial statements of intent without having the evidence to back up what they say they do. For example, Participant 15 stated: "*[name of company removed] say that they measure against SDGs, but there is a gap between what they say they do and what they actually do*". The same concern in the honesty of many organisations was noted by Participant 5: "*In Terms of our customer base the honest answer is at the moment our customers pay lip service to this stuff - they talk about it. But in reality, they're not really doing things that make a difference. They don't take decisions or anything like a difficult decision that you might say has been heavily influenced by something on the sustainability agenda. Disappointingly - I hold out hope but that is the facts".*

The transparency of data is noted by Participant 12: "trying to get something that is practical, meaningful and simple enough to be understood by your team. We are very open and transparent, what we want to achieve, why and how we are doing against the targets". Participant 11 makes a similar point: "It's obvious that you have to make loads of tough decisions rather them duck them, and then recording your progress in an open, honest and visible way, helps keep you honest in that process." Many identify the key issue of the change management culture is to capture the hearts and minds of their staff, as mentioned by Participant 2: "our change program, as much as anything, is getting the hearts and minds of engineers in the field to understand the potential benefit of doing it and that it's not a waste their time". As one CEO (12) commented: "we can only change people's outlook once people are engaged as a workforce - so its core to having a sustainable business. This is a big part of my ethos of the company that they want to work at" and he also refers to the same theme, "and I guess there's winning the wider hearts and minds to show them that the global agenda really matters as well".

Innovation was a frequently referenced benefit of securing the right change culture and, in doing so, having the means to address the SDG targets more effectively. For example, Participant 11 noted the effect of building long-term supplier relationships that enabled more innovative solutions to be developed: "We wanted to establish meaningful change across the supply chain, and we recognised that, to do this, we had to develop long-term relationships; hence, we contracted on a five-, plus five-, plus five-year basis. This built longevity into our thinking and allowed true innovation to develop solutions to the bigger sustainable development issues across the environment, driving efficiency and effectiveness." The same benefits of a motivated workforce were shared by Participant 14: "We've seen teams come back with new solutions, new ways of working, ideas which for us is incredibly innovative. When you boil them down, they are not radical they have just gone and found ways of doing things differently". One way that a global Head of Sustainability has increased involvement and excitement was shared by Participant 3: "I particularly liked the term of the 'engineers as the clever experts', saying that if the future is going to be like this, how would you, as a clever innovative engineer, design differently than what you did before. If you approach it as an innovation function rather than just a sustainability function and you don't get hung-up by trying to persuade everybody that climate change is the most important thing, you depoliticize it and you play it to areas that engineers are particularly up for - which is doing very clever stuff. That is what engineers are there for".

Communication was also a dominant theme of culture change. Participant 1 noted: "you do not communicate it once, you communicate it nearly every day through many, many different vehicles. You bring people in". Participant 24, a leader of a North

American national civil engineers institution, highlighted the value of leaders who can tell stories that resonated with stakeholders: "people with success stories become your spokespeople and they start to influence others, saying 'hey, you know this works for us' rather than just trying to sell the methodology. It is more, you know, encouraging peers, e.g., peer-to-peer". This highlights the importance of stakeholders and the communication plans used to share SDG progress. The word analysis identified key stakeholders, including: governments; communities; investors; users; suppliers; and most commonly referenced, employees/staff. This was captured by Participant 26: "the SDGs requires us to look at partnerships all across private sector, academia etcetera in achievement of the SDGs - it is obvious that we all need to come together". The main focus for this stakeholder engagement for Participant 11 was: "Our starting point is understanding what is important to our clients, who want to see us make improvements, and where our staff and employees want to make a difference". Participant 32 continued this theme across other specialisations, such as lawyers, designers: "The importance of collaboration with other disciplines still that's great but also apply in areas where landscape architects, accountants and I'm sure others have described to you these goals are not going to be achieved by just one or two groups. They need many people working together with diverse perspectives and diverse backgrounds around the table". The final cautionary note about communication was that the messaging should be kept simple and accessible, as suggested by Participant 11: "We found that our campaign and collaborative working with partners had created a different conversation with different language. Ultimately, accessible language on meaningful outcomes are what people can buy into and this is what creates the momentum of changed behaviours...Through engagement, innovative solutions address the big problems, Wisbech is an example of working with the community to achieve meaningful long-term changes."

Learning from the statements shared above, a good solution proposed by Participant 25 was: "So individual projects would have their own stakeholders and customer satisfaction and community engagement. So, it varies on the size of project - some are very small and some are spending a billion pounds on a major rewiring in London. But each project has outcomes on all the projects, more than cost, time and quality." The theme of collaboration was noted by many of the participants, such as a Board member (#1), who linked the collaboration with strong leadership: "How do you seek to build a sort of coalition of the willing to move in that direction? I think you do it through visionary leaders - leaders that recognise the big picture and do the part they can play in achieving that. So, I think you win over the leaders and the leaders then lead their companies through it. And they then lead that change".

Another key theme of the interviews was the issue of millennials. In total, 33% of participants discussed millennials. The preceding survey (Mansell et al., 2020) included a number of specific questions relating to the interests-perceptions-values of millennials and this survey report had been read by some of the participants prior to the interviews. There were some consistent and interesting considerations, especially in senior executives recognising the demands of millennials in this area and the need to adjust their approach in consideration of the recruitment and retention HR issues. For example, Participant 13, a CEO of one of UK's largest transport programmes, stated: "Young people keen to make a difference, keen to do things sustainably, problem is with more senior people making sure they 'get it' to give young people space to do what they want to do". Others also noted the role of young people as leaders to galvanise businesses to measure SDGs, as shared by Participant 10: "I think millennials have a role here as new project leaders where often they are the people who are most energized. I got into engineering to make a positive difference to people's lives". Many participants recognised the potential power of millennials to change the course of their businesses, such as was noted by Participant 20: "our biggest influencers are the millennials who are more aware, and they see it affecting them more directly - it becomes a challenge since cultural change of the older generation is more difficult to achieve if they are required to change their routine". Although another view was postulated by a CEO (#7) who didn't agree: "No I am a bit sceptical of those Millennial badges of being much more demanding because when I was 20-30 years younger, I had the same passions, I think they might be more of a groundswell, but I was interested in these issues - so I don't like the simple analysis that suggests that millennials do care whereas their predecessors didn't." But there were 6 of the participants that mentioned the positive impact on recruitment and retention, as noted by Participant 23: "being good at these things impacts on the skills shortage by attracting and retaining staff which in a market of shortage is important. It can help us attract retain and motivate good people by presenting good strong ethical company particularly important given our industry faces."

An unexpected but often-quoted issue was on the context of gender influence on SDG measurement. Eight participants (1, 5, 10, 17, 21, 24, 31 and 37) made specific reference to gender impact: "the younger generation really do want to change the world. Interestingly, particularly the female part of that [company name removed] has more than 50% of its membership as female and I pondered why that should be, and I think it is because it appeals to the values of certainly the younger, but actually to the female, side of our institution, who really want to make a difference to the world that they live in. Probably, they are more driven by that than they are by financial reward". Participant 10 suggested that the reason for the female intuitive importance attached to SDG measurement was due to a natural inclination to be more passionate in this area: "Quite often some male engineers just love problem solving but are perhaps less passionate about making a positive impact on people - in other words its less about the technical side that motivates me. It is merely the means to the ends". One senior executive of a global construction company, Participant 24, was open about the value to their business: "SDGs language is great for attracting the future engineers and get the gender balance right". A wider value was linked to the diversity agenda by Participant 31: "The young people coming up are aware of these goals and creates an excitement and there is much to be done - there are things to be figured out and I think it's appealing also to the diversity issue", which was also noted by another CEO (#5) who stated: "SDGs support my interest in promoting gender diversity in our industry and about promoting training and development and apprentices recruitment into our industry."

There were nearly half the participants that promoted the positive effects of harnessing the power of the millennial generation to promote change and thereby help champion the uptake of SDG measurement, which was shared by Participant 1: "So, *if we can find a way of linking into the power of the younger generation*". This attitude was further explored by Participant 10, who noted the obvious fact that millennials are tomorrow's leaders: "*I think millennials have a role here as new project leaders where often they are the people who are most energised*". The beneficial impact on recruitment and retention, mentioned earlier, was described by a CEO, Participant 12: "*… attracting and retaining top talent into the industry, specifically around [location name removed]. It's something that captures the heart of millennials"… "getting the right talent here is critical to us. So, to attract the right people across that age group*

we would want to respond in the right ways." This self-serving reason for using the SDG measurement as a business tool, was also shared by Participant 22: "We are very aware of our millennial global population and making sure that they are engaged and moving the business that continues to attract and retain that talent". Approaches to achieve this were given by Participant 25: "... alignment to your business success to do the right thing, from the sustainable agenda perspective, and I think that feeds into that cultural point of sort of the hearts and the minds of individuals certainly for the millennials, they are less concerned about pension schemes etcetera, they are much more environmentally focused". He continued later in the discussion: "that comes back to how we treat people and our environmental credentials. There is a clear business reason for recruiting and retaining staff". Strong evidence of this view was also provided by Participant 3: "attracting and retaining staff - typically three or four people per week who tell me our sustainable development focus is why they joined our company".

Category	С-М-О	Sub-Category	Word	Count	f1 Word %	f2 Sub- Cat %	f3 Cat %
		Change Culture	change	129	0.24%	0.28%	
		Change Culture	culture	23	0.04%	0.20 /0	
			honest	22	0.04%		
		Behaviours	collaborative	14	0.03%	0.10%	
Change			collective	14	0.03%		
Managamant	Context		opportunity	54	0.10%		0.78%
Management		Innovative	opportunities	20	0.04%	_	
		Commitment to	commitment	42	0.08%	0.34%	
		change	innovation	32	0.06%	_	
			investment	32	0.06%	_	
		Communication	Communication /s	28	0.06%	0.06%	

Table 19: Text analysis (NVivo) on key words' frequency: context of change management.

Finding #8: change management. One of the largest positive impacts for SDG measurement is about engaging, communicating and energising the delivery teams. This involves the internal teams and suppliers. The millennials have a key role to help build and sustain this change momentum.

6.6.3.6 Performance Management

The "performance management" node had the highest level of incidence (n = 82) of relevant statements coded from the 40 participants (using the NVivo software).

In terms of key word usage, this thematic, shown in Table 20, tracked "targets", "measuring", "performance management", "quantitative", "metrics", "qualitative" and "contribution". In total, they were used 1003 times across the 40 interviews, which equates to once every 50 words and represents the most referenced thematic.

The highest frequency of coding on NVivo was using the node for "what to measure", reflecting the importance of this thematic. There were many references to what is measured, and the general theme was that the selection of targets becomes critical in a business environment that is already awash with data collection. Many asked whether they should collect quantitative data or qualitative and also asked what the balance between too little data collection and too much is. Almost all participants accepted that this was an extremely difficult area to resolve and that there were no easy answers. For example, Participant 34 stated: "*I think we are quite confused. It sounds like we are much more advanced than we are in the way we monitor, report and evaluate. Most of our work is about getting the basics right and ensuring we are complying with legal requirements—getting stuff done. We know we need to do more work on understanding sustainability outcomes and how we can develop detailed KPIs that feed into that for measuring our impact. We do not have outcome frameworks in place yet".*

There was a consistent recognition amongst those that had more advanced levels of SDG measurement process maturity (participants 3, 19 and 20) that you had to start by selecting a manageable number of goals (from 17) and targets (from 169). This was explained by Participant 31, who said: "*It is an enormous challenge. I think, out of those 232, the fact that you found 20 that can be measured is actually pretty good if I think about the magnitude of the problem*". Amongst the nine participants that were at the "early processes in place" stage, most were trying to establish hard metrics that could be quantified, such as Participant 15: "*We want hard targets to test our performance. Generally, as a business, qualitative is not very compelling. When we set up our strategy, we did some serious baselining to get some better referenced data.*"

Others were very open that they couldn't see any relevance or value in developing an SDG performance management framework for SDGs, such as Participant 34: "We found that the SDGs were impacted by our work, some more than others, in terms of the goals and targets, but they are not particularly relevant to the

work that we do so our priorities have been elsewhere". There were a few of those that were in the 'early process maturity stage (2, 5, 6, 9, 35) who had similar views to Participant 16: "Its fine for the senior leaders to talk about sustainability, responsible business and the metrics we want to measure - but getting that filtered down as a priority is challenging because at the project level, people are very busy delivering to time and cost and often the extra measurement targets appear to distract from this core focus of being profitable" and later added "all I'm saying is that it is great to have targets with great ambition and objectives and things like that you drive towards but in internally there is this ongoing challenge of making sure that it makes sense and adds value". But in a contrary view, Participant 19 represented those that believed there would be value by measuring impact at project level: "Being able to measure the contribution of individual projects to the SDGs – might be helpful but it's far from straightforward and what you measure depends on why you measure."

The theme of adapting the SDG targets to have relevance to your own organisation, was also noted by Participant 5 (CEO), who also links the approach to millennials: "it is best to set targets that mean something to your company and your people individually, rather than having generic targets which are enforced on you - actually setting a limited number means something to you as a business is the right way forward and of course the trouble is with people who set things they tend to want to measure and really don't buy into the longer game, so everyone should say - so we want to measure over a year or two and see a significant change so that we can all say "We've done that". And that's the problem with the target-based world. This is about actually you know what's the future generation - young people who are the new generation are a bit more wired up to it than those currently in leadership positions today. Maybe it will be quite different in 20 years' time because the emergent leaders coming through will get all this and want to do things differently.

Some provided valuable challenge as to the important differentiation between contribution, versus attribution, such as Participant 19: "*It is about us as engineers understanding the contribution our projects are making to achieving the aspirations of the SDGs (contributing to the SDGs) - not just the goals but the targets. There will be no one-size fits all, and a check list to say a project relates to a goal/target is far removed from a anything that meaningfully assesses impact.*" This is further explored by Participant 27: "*using attribution-contribution seems to be a very helpful way of* not claiming too much and therefore we evidence collection to substantiate whichever one you're going down is different levels of granularity. I think it will be wise to be able to talk about to what extent we are contributing to the wide impact and outcome captured by the SDGs. But we can talk more specifically about the attribution around the outputs and activities towards the SDGs". At a strategic organisational level, Participant 27 placed emphasis on contribution: "Our Strategy is called 'Our contribution". "We have identified 5 SDGs that we feel we have an opportunity to have an impact and where we are already doing a lot in. These coincide with our business model and provide the best opportunity to demonstrate benefit and what we are doing to measure performance against those SDGs". Participant 19, a global sustainable development leader, said: "Being able to measure the contribution of individual projects to the SDGs – might be helpful but it's far from straightforward and what you measure depends on why you measure." She also said, "It is about us as engineers understanding the contribution our projects are making to achieving the aspirations of the SDGs (contributing to the SDGs) - not just the goals but the targets. There will be no one-size fits all, and a check list to say a project relates to a goal/target is far removed from a anything that meaningfully assesses impact."

One of the key problems, mentioned earlier, is the level of complexity in measuring 169 SDG targets. It was frequently explained that this was too complicated for the construction sector, as stated by Participant 2: "*But the indicators are far too detailed and big and sometimes not applicable as well. Therefore, it is better to work at a higher level for the projects. I have more interest in the goals and not the indicators"*.

The emphasis on quantifiable targets was countered by Participant 25: "telling the story of the success against the sustainable development goals, as an example; a lot of the time, it cannot be quantified very easily and therefore telling the story around an outcome perhaps provides more impact and value than just putting a meaningless quantitative score against something". This viewpoint was backed by Participant 2: "In the beginning, I wanted quantification to have numbers that I can use to understand the measurement data. This created a big pushback because engineers tend to want perfect solutions. The assessment was causing some culture issues, so the qualitative aspects have been preserved but not the quantitative. So, we still look for the holy grail but, at this stage, we are going to produce stories. In future we would *like more quantitative that can be assessed at corporate level.*" This was also noted by Participant 27 who stated: "we need to *be able to tell a very compelling story but that's qualitative more than quantitative.*

A further area of research interest was on planning horizons. There was frequent reference to the need to think longer term. SDGs have targets and goals for 2030, but some of the participants discussed moving away from short term RoI (return on investment decisions) that were driven by economics only, to a longer term more balanced approach across environment and society issues. Hence the SDG were viewed as a good way to achieve this ambition, as discussed by Participant 14: "a danger that when you look at those longer-term goals that we allow them to be longer term aims that sit off to one side somewhere and that we want to contribute to, but it's in addition to or on top of our day job". Participant 3 put the same point slightly differently: "it's not just about delivering a building, it's not just about delivering a renewable energy project, that is about every project that we do - It's how can we design it for the future and through that, that is supporting the SDGs as much as possible in a meaningful way".

Category	C-M-O	Sub-Category	Word	Count	f1 Word %	f2 Sub- Cat %	f3 Cat %
			target	31	0.06%		
		Targets and measuring	targets	208	0.39%		
D. (objectives	26	0.05%		
			indicators	76	0.14%		
			measure	142	0.27%	1.36%	1.90%
	Mechanism		measuring	72	0.14%		
			measuring	72	0.14%		
			measurement	54	0.10%		
Management			measured	17	0.03%		
Management			measures	21	0.04%	-	
			management	83	0.16%	-	
		Performance	performance	54	0.10%		
		management -	metrics	46	0.09%	0.43%	
		quantitative	objectives	26	0.05%		
			quantitative	18	0.03%		
		Qualitative contribution	contribution	40	0.08%	0.11%	

Table 20: Text analysis (NVivo) on key words' frequency: performance management.

Finding #9: select a few targets relevant to the construction organisation or project. Keep it simple and build knowledge progressively.

6.6.3.7 Project-to-Portfolio Levels

The "project-to-portfolio" node had the eighth highest level of incidence (n = 21) of relevant statements coded from the 40 participants (using the NVivo software). In terms of key word usage, this thematic (shown in Table 21) tracked "projects", "programmes" and "portfolios". In total, they were used 677 times across the 40 interviews, which equates to once every 80 words. There was wide recognition that the approach needed to be adapted but linked across the project, programme and portfolio levels, as noted by Participant 27: "*I think there is no 'one size fits all'. So, it will vary from programme to programme and be dependent on the country as well*".

Special interest and importance were aligned with the node on "starting projects". The preceding survey (Mansell et al., 2020) had not highlighted the importance of "starting projects well". This node was added during the interviews stage because it was often referred to as the need to use the SDG lens at the "key investment decision point", as noted by Participant 26: "based on evidence frameworks, you can frame your project in a much better way to make sure the impact you get is maximized." The emphasis of getting stakeholder alignment was also mentioned by Participant 19: "They want to demonstrate that their projects contribute to sustainability development goals and develop tools that make sure projects embed sustainability development at the outset, e.g., at their project inception phase". This was also extended to the importance of getting the project staff involved early, as noted by Participant 3: "trying to get the delivery team involved in that as early as possible to think about what future Ready really means and how that can be played back to the client in a powerful way".

There were some, such as Participant 9, the head of sustainability for a utility company, who suggested that the SDG measurement had more relevance at the larger scale of programmes and at the organisational strategic level, represented by the portfolio office: "*Thus we do it more at programme and portfolio level and less at project level. So, we have a mapping process at the portfolio level and align across project and programme SDG targets*".

Category	C-M-O	Sub- Category	Word	Count	f1 Word %	f2 Sub- Cat %	f3 Cat %
		Project Level	project	278	0.52%	0.87%	1.27%
			projects	185	0.35%		
		Programme Level	program	65	0.12%	0.35%	
Project;			program	65	0.12%		
Programme;	Context		programme	31	0.06%		
Fortiono			programs	27	0.05%		
		Portfolio Level	portfolio	26	0.05%		

Table 21: Text analysis (NVivo) on key words' frequency: projects-to-portfolios.

Finding #10: there was evidence that SDGs can be measured at all three levels: projects, programmes and portfolios. There was special value in using the SDG lens at the start of the project to help align stakeholders around the longer-term outcomes and impacts.

6.7 DISCUSSION

This section builds on the 10 core findings and culminates with generalisations across the three sub questions that guided the design of this research into SDG measurements. The three sub questions, as shown in Figure 51, stem from the primary research question: how do senior leaders in the construction sector rate and use global UN SDG goals for infrastructure investment decisions at the local level? The empirical research study, including aforementioned qualitative findings and supporting quantitative data, also allows an evaluation of the theory-driven propositions to be undertaken, which is provided according to the following areas of outcome, mechanism and context.

6.7.1 Outcome Discussion: What Are the Expected Outcomes of Successfully Using the SDG Measurement Mechanism?

The results showed that participants have the appetite and resolve to employ SDG measurement at business and project levels (Finding #2) in order to achieve outcomes that benefit people, the planet and profit. At the same time, they were frustrated by their inability to do so for reasons discussed in the following sections.

Most participants were optimistic that their organisation would achieve the broader outcomes by making SDG measurement more usable, consistent and verifiable across the construction sector, with increasing balance to their investment decisions across environment, economic and societal factors (Finding #1). There was almost unanimous conviction that the "ends" of achieving the desired "outcomes" was good for business (Finding #4). Some viewed this from a realist interpretation of the practical benefits of using the approach and language of SDG measurement to attract and retain millennial staff. Others made the additional connection with Porter and Kramer's Creating Shared Value (2019) theory, that suggests the outcomes from embedding SDG measurement in the delivery of their projects is good for business because, for example, the drive to address the carbon issues typically achieves improvements in efficiently and effectiveness.

Although the results emerged from a different thematic, some of the participants (2, 3, 17, 19, 20, 26 and 27) recognised the value of using Carol Weiss' seminal work (1972, 1983, 1995 and 2005) that uses the LogFrame and Theory of Change approach to take a stakeholder-centric perspective to assist the definition of longer-term impacts and outcomes. They acknowledged that this helps rebalance from an overemphasis on output definition, which is typically used in project management and too often judges success in terms of delivering the infrastructure asset to time, cost and scope (Finding #6).

6.7.2 Mechanism Discussion: What Design Criteria Enable the Mechanism (for Measuring SDG Impacts) to Achieve the Outcomes?

The views were consistent in stating that this was an important area for the construction sector to get right but that there was no best practice established for how to deliver an effective mechanism. Therefore, despite the strong support for its adoption, the depth of knowledge on SDGs was mostly superficial, and only 8% of the organisations interviewed self-assessed their SDG measurement processes as repeatable (Finding #2), with only a further 23% having processes at an "early adoption stage". The majority had not yet defined the SDG measurement processes. Unsurprisingly, there were many, especially at board and CEO level (with notable exceptions, such as 5, 7, 8 and 12), who showed some confusion in their knowledge of SDGs, sustainability and sustainable development. This was reflected in having relatively consistent and well-informed views on specialist areas, such as carbon management, but this was less evident in the details of what the SDGs represented.

The low level of uptake of the SDG measurements at the project level was attributed to the following reasons: (a) The complexity of the SDG framework, with the scale of ambition understandable at a high level but made excessively complicated when examining the 17 goals, 169 targets and 232 indicators; and, (b) The lack of adoption of SDGs by clients did not mandate SDG measurement (Finding #2).

In most participants' view, this had to be significantly simplified if the value of the SDGs were to be achieved (Finding #9). Otherwise, another reason for most organisations only having a very high level, somewhat superficial, view of SDG impacts on their projects, was because their clients did not mandate SDG measurement (Findings #2). There was therefore no incentive to dedicate finite resources on a complicated task that might not deliver them any value – indeed, it might even identify their organisation's weaknesses, which only a few explicitly opined was a good way of learning and developing. The current reality is that a high level approach could carry on. Some would view this as providing most of the reputational benefits of being able to tell a good story, but without similarly adopting – preparing and upholding - the business risk of being held to account for not setting sufficiently demanding targets, or worse, failing to achieve them.

A further design criterion that emerged, to enable the mechanism for measuring SDG impacts to achieve the outcomes, was the ability to find a golden thread from enterprise portfolio level to project level (Finding #10). This was most clearly explained by the participants that were most developed in their SDG measurement processes (2, 3, 11 and 20) but also included others who were actively developing SDG processes (8, 9, 14, 19, 27, 28 and 36). Whilst there was confidence in their self-assessed ability to achieve the golden thread from project to portfolio level (Finding #10), this was mostly not substantiated by any evidence (except 2, 3 and 11).

6.7.3 Context Discussion: What Issues Influence the Successful Use of an SDG Measurement Mechanism to Achieve the Desired Outcomes?

As part of the discussions on strengths and weaknesses, the participants identified a number of contextual issues that affected the likely success of the mechanism achieving the desired outcomes. These "context" issues included leadership (Finding #3), outcome-output definition (Finding #4), knowledge (Finding #5) and change management (Finding #8) capabilities. There were more optimistic discussions than pessimistic ones about the ways they could improve the contextual

issues identified. However, a few had little incentive for, or perceived little value in, adding what they considered a burdensome task onto the shoulders of busy project managers.

As mentioned above, one of the most common explanations for the slow uptake of the SDGs, was related to their apparent complexity. The respondents insisted that the size of the task seemed unmanageable to some of the participants if the full 17 goals and 169 targets were all needing measurement. Some participants (12, 23, 25) linked the perceived complexity to the resulting 'voluntary' high resource costs of implementing a new performance framework. This relates to the earlier mentioned lack of client requirements that have not yet made SDG measurement mandatory, as noted above.

Given the seniority of the participants, it was not surprising that leadership and strategy was a dominant theme in discussions. This led to Finding #3, which states that strong leadership plays a significant part in inculcating SDG measurement as an ambition and core value into an organisation. This was most clearly stated by a senior executive (#11): "Leadership is the most important critical success factor, both internally and externally, to align and galvanise our employees, our communities and the supply chain". Others (2, 10, 17, 19 and 29), none of whom were CEOs or board members, stated that the strategic nature of organisational change had to be driven from the top, consistent with Kotter's change model (2012). There was recognition that, in reality, this meant that leaders at all levels were required as champions, which, for SDG measurement, needed to be aligned with success stories that would make sense to the target audience, expressed in their language and justifying "why" followed by explaining clearly "how".

A further 'Context' (of C-M-O) issue of unanimous agreement across the participants, was the role of learning and education (L&D) in increasing capability and specifically on understanding how to better share lessons on SDG measurement, for the good of all (Finding #5). It was positive to note that most participants shared examples of their organisation's efforts to increase knowledge and skills of employees, such as by learning how to define success in outcome terms (Finding #5). This appears to be aligned with other national studies on challenges of measuring sustainability, such as Hakkinen and Belloni (2011) who identified a shortfall in the knowledge levels of achieving 'green' construction and that these shortcomings limited Finnish

construction practitioners from achieving the desired outcomes. As a CEO (#8) stated, "one of my strategies in creating SDG awareness is through the increased emphasis on learning and development". As discussed in the literature review, Haugh and Talwar (2010) have also indicated from their research into sustainability capability, that the use of action-learning on projects and the benefits from adoption of thorough 'lessons learning' as part of a comprehensive knowledge management system, could provide learning strategies for embedding SDG measurement in organisations. It was also noted by many (1, 3, 7, 12, 15, 23, 25, 27, 32) that investing in L&D, both in engineering courses as well as in on-the-job training, could better support the future generation of construction professionals to apply SDG measurement strategies. It was noted by a few, that this was especially important at the start of projects (Finding #10), when the investment decisions could be broadened by engaging the wider stakeholders on discussions on the longer-term SDG impacts desired.

Linking to the models developed by Kotter (2012) on leading change, the eighth finding was related to the contextual issue of change management (Finding #8). One of the most significant ways to influence the take-up of SDG measurement across organisations is engaging, communicating and energising the delivery teams. Research has shown that this is critical to achieving the right organisational cultures (Garavan & McGuire, 2010). This relates closely to the role of leaders as discussed above but also covers a number of other areas that participants shared. The issues raised included: the need for the 'talk' to be matched by the 'walk', with demonstrative and meaningful action; the need for honesty, to set targets that include the positive as well as the negative impacts, and the confidence to share the shortfall in performance, and learn from it; the essential need to capture the 'hearts and minds', especially of the millennials and across the genders; the positive use of innovation as a strategy to deliver the SDG goals, seeking new solutions to grand challenges; and, the clarity and consistency of communications across the stakeholders, to help galvanize and inspire. It was suggested that verifiable evidence of 'good news stories' would also help with communications and increase the likelihood of the change management succeeding.

The contextual issues identified above provide some indicative insights into broadening our understanding of factors that influence construction companies' decisions on whether to use SDGs as a lens for defining success and, if so, how they might use them effectively. Other studies delve deeper into construction sustainability benefits (Švajlenka et al., 2018) or, for example, the evaluation of modern methods of construction based on wood (as aligned to SDG 12 on responsible consumption and production) (Švajlenka, Kozlovská, 2018). Equally important areas that are not addressed in the thematics discussed above relate to green financing; some authors (Sergi et al., 2019) have provided insights into public–private partnerships as a mechanism for financing sustainable development. This highlights the breadth of relevant thematics and keeps the focus of this paper on just the restricted areas considered most important to the executives interviewed.

The resultant view, after applying the insights from the interviews, against the C-M-O 'Variables Framework' is shown below, as an illustrative subjective analysis based on the researcher's interpretation of the interviews' qualitative results:



Figure 58. View 2 – Maturity of C-M-O evidence across variables post-interviews.

6.8 CHAPTER CONCLUSIONS AND DEVELOPMENT OF PROTOTYPE

This stage of the research study provided empirically grounded insights from the 40 senior leaders on their perceptions of how their organisations rated and used SDGs as a measurement lens. Building on the earlier results from the survey, discussed in the previous chapter, the 10 findings provided a rich and deep insight into answering

the question of how to measure SDG performance on infrastructure projects. The empirical research also validated the theory-driven propositions that were synthesised from the literature. Furthermore, this stage of the research study identified that, whilst SDG measurement practices on infrastructure projects are embraced in theory, they are problematic in practice: rarely does action match rhetoric.

Although the 40 interviews described in the study specifically identified the primary stakeholder group as the senior executives of construction firms, there were a number of other stakeholders interviewed. This included: two senior government experts in the infrastructure sector; one financial advisor; one from the United Nations; and three from standards bodies (these 7 were part of the 40 participants). Consequently, the design of the study sought to include the considerations of wider stakeholders involved in project decision-making and this approach helped to ensure the broader relevance and applicability of the study to the infrastructure sector. The researcher had also consulted with the UK's Institution of Civil Engineers to ensure this broader perspective was adequately captured through the verification workshops with the MISI control group as illustrated in the Blog by the ICE Knowledge Manager from one of the workshops, (see Appendix 6).

One of the primary characteristics of the qualitative research is that the researcher "is the primary instrument for data collection and data analysis" (Merriam and Grenier 2019, p,13). This introduces a researcher's bias, that Singleton and Straits (2010) can be mitigated by becoming more self-aware when designing and implementing an interview data collection approach. If this can be actively addressed from the start by, for example: (1) asking the clear question, but allowing for the interviewee to develop their own line of thinking without shoe-horning them into specific areas; (2) interviewing an adequate sample, which was addressed by partnering with the ICE to benefit from accessing a senior level of executive that might not usually be available to a researcher; (3) sharing the proposed approach and questions, as well as the empirical evidence from the preceding survey (as published by the ICE for their 200th anniversary); (4) not misrepresenting the data results, by deploying best practice use of NVivo and ensuring logical and verifiable analysis. However, there is still a danger that, despite the strength of applying these 4 techniques, there is a remaining weakness since, unlike a survey or scientific experiment, the "human instrument" with all its shortcomings is not able to adjust to
evolving changes. For example, the researcher allowed the interview questions to evolve in a free-flowing discussion (and followed the participants' train of thought when appropriate) when he detected a different line of enquiry. There is thus a need to apply caution to the potential hazard of bringing the researchers' own bias (Alvesson, 2003; Merriam and Tisdell, 2015; Merriam and Grenier, 2019). The researcher protected against this since "it is important to identify them [bias and subjectivity] and monitor them as to how they may be shaping the collection and interpretation of data" (Merriam and Grenier, 2019, p.13).

In regard to Stage 2 of this study, there was a lack of evidence given by participants on their ability to achieve the golden thread of SDG measurement from project to portfolio level (Finding #10) because, often, it was not available at any credible depth or backed up by verifiable evidence. This was an area to be tested in Stage Two to test whether aspirations to achieve this linkage are realistic. There was also the need for further research outside the UK since, while the findings from this study have broad global application due to the regional and global footprint of the participants' organisations, the complexities and challenges in some areas require further SDG measurement research.

The building of the evidence to respond to the study's propositions is shown below:

Confidence in	P1:	P2:	P3:	P4:
Evidence for	Engineer's view of	Organisational	'Golden Thread'	Prototype
Propositions	current MISI	view of MISI	global - to - local	Development
Theoretical	Not yet	Not yet	Very Low	Very Low
Analysis				
Survey Analysis	Moderate	Very Low	Very Low	Very Low
Interviews	Moderate	Moderate	Very Low	Very Low
Analysis				
Test 1: Golden	Not yet	Not yet	Not yet	Not yet
Thread	_	-		-
Test 2: Case	Not yet	Not yet	Not yet	Not yet
Study	-	-	-	-

Table 22: Development (3rd) of stability of Propositions (changes since previous Chap in red)

(Note 1: The assessment of the quality of propositions' evidence is based on a qualitative formula designed by the Clinical Grading Working Group's British Medical Journal's study (GRADE, 2004). Full description and justification for this approach is shown in Table 8.)

Chapter 7: Triangulation of Results and development of Prototype

7.1 OUTLINE OF CHAPTER

This chapter captures the triangulation of results that informs the further development of the *SDG Impact Value Chain* (IVC) from an initial theoretical model into a more robust prototype, ready for testing. This is done through the iteration of the IVC theoretical model that was initially synthesised from Chapters 2 and 3, described more fully in Section 3.8. In this chapter, the IVC is jointly developed with a methodology, the SDGiPro©, that had been identified from the survey and interviews as a necessary extension to the IVC to support its implementation in 'active' projects. This early prototype is developed to a stage that is ready for testing in Test 1 (SDG 'Golden Thread') and Test 2 (in a water utility company context), discussed in Chapters 8 and 9 respectively.

7.1.1 Synthesising the initial theoretical model (*Impact Value Chain*) from literature

The initial IVC theoretical model, described in Section 3.8 and shown below in Figure 59 was based on four underpinning theoretical models, including: (1) the Theory of Change (Weiss, 1995 and Stein and Valters, 2012); (2) Creating Shared Value (Porter, 1985, 2011); (3) infrastructure systems approach (Hall et al., 2016, 2017; Thacker and Hall, 2018 and Thacker, et al., 2019); and (4) the Triple Bottom Line (Elkington, 1994, 2013, 2018 and Griggs et al., 2013). The last of these, the TBL, provided the link to SDGs through a more holistic "systems approach" to address infrastructure sustainability in the SDG context. The IVC provides a new holistic method to potentially improve sustainability on projects and programmes by guiding decision makers in their investment choices through confidence that they link to specific SDG targets.



Figure 59: The Impact Value Chain (IVC) theoretical model, adapted from ICAS/IIRC's 'The Sustainable Development Goals, integrated thinking and the integrated report' (Adams, 2017).

7.1.2 Prototype Approach

For this research study the definition of a prototype is taken from a civil engineering study (Kim, 2019) into methodologies for developing early-stage solutions in a project design process. The study defines 'prototype' as both an 'original model of something that serves as a basis for other things' and an 'early model that can be examined through tests to find a design solution" (Dam and Siang, 2017). Importantly, the second description places emphasis on the designer's assuming-evolving-solving process in the design processes. The 'process involving the test-refinement-completion of designs using prototypes is called 'prototyping' (Ulrich and Eppinger, 1995, 2004). Importantly, it not just about product design, it 'is also, and perhaps more importantly, a phase in a critical process' (Lauff et al., 2018). Considering these viewpoints, this study adopts an adapted version of Kim's definition (2019): 'The MISI prototype is an early sample IVC model to test the concept and process of measuring SDG impacts at portfolio, megaproject and project levels, to act as a basis for further MISI learning'.

7.1.3 Triangulation Approach

As discussed in Chapter 4, triangulation has been used extensively in qualitative and social research. The thinking behind triangulation was introduced by Norman Denzin (1970) in the 1970s when he developed the proposals by Webb et al. (1966) and before them, Campbell and Fiske (1959). Denzin suggests (1970) it is a method that can improve the validity (extent to which a study accurately reflects or evaluates the concept or ideas being investigated) and credibility (trustworthiness and how believable a study is). Thus, the method design included an Explanatory Sequential Design (Creswell, 2015; Merriam and Grenier, 2019), followed by a triangulation of data to inform the development of the prototype, as shown below in Figure 60.



Figure 60: The Triangulation process to build the IVC Prototype.

7.2 TRIANGULATION OF RESULTS

The structuring of the triangulation evidence is aligned to the four concept areas that were derived from the outputs of the literature review and the theoretical chapter, shown in Section 3.8. These are:

- Concept 1 The Theory of Change's Impact-Value Chain (IVC)
- Concept 2 Delivery of Projects within an Organisational Structure
- Concept 3 Infrastructure investments as 'system of systems'
- Concept 4 Delivering Impact measured against the TBL/SDGs

7.2.1 Concept 1 - The Theory of Change's Impact-Value Chain (IVC)

7.2.1.1 Learnings from literature and theoretical review (Chapters 2 and 3):

• The first concept aligns the Theory of Change (ToC) with the Triple Bottom Line (TBL) as the fundamental basis of the theoretical IVC model. The ToC has its roots in the development of the Logical Framework Approach (Baccarini, 1999), also known as the 'LogFrame' or Goal Oriented Project Planning (GOPP) and Objectives Oriented Project Planning (OOPP) methods, in the development sector. This has been extensively tested, especially across international development projects, and is a basis for further prototype development.

7.2.1.2 Learnings from survey data (Chapter 5):

- The engineers surveyed (Chap 5, Finding #C3) highlighted the need for simple tools to measure project success using the SDGs, that should guide the design of the prototype.
- The survey noted (Chap 5, Finding #C2) that 'definition of success' and 'excessive focus on outputs instead of outcomes' (Chap 5, Finding #O2) were two of the five top challenges facing practical application of MISI. Thus, the IVC prototype should be developed with an emphasis of defining long-term goals and then map backward to identify necessary preconditions. In this way it should demonstrate the process of change by outlining causal linkages in an initiative, i.e., its shorter-term, intermediate, and longer-term outcomes. The IVC should enable the identified changes to be mapped as the 'outcomes pathway'.

7.2.1.3 Learnings from interviews (Chapters 6):

• There was almost unanimous conviction that the "ends" of achieving the desired "outcomes" was good for business (Chapter 6, Finding #4) and this should be messaged when seeking to get organisations to engage with the IVC prototype. This was supported by participants being optimistic that their organisation would achieve the broader outcomes by making SDG measurement more usable, consistent and verifiable across the construction sector, with increasing balance to their investment decisions across environment, economic and societal factors (Chapter 6, Finding #1).

- CEO's mentioned that a MISI tool needs to ensure strategic alignment, which aligns with use of the 'ends, ways, means' model using the Theory of Change.
- Whilst recognising the imperative of ensuring short-term commercial success when delivering project engagements for clients, all CEOs recognised the importance of defining outcomes and benefits from the start, which assumes an explicit need to plan for longer-term impacts. The IVC model should align with the existing benefits realisation processes.
- A number of the interview participants noted the importance of building a coalition of the 'willing' and that the alignment would be enhanced by using the IVC's ToC and TBL to demonstrate progress on the achievement of outcomes. In this way, evidence of success confirms the causal linkages and indicates that the initiative is effective in achieving its outcomes and SDG impacts.

7.2.2 Concept 2 - Delivery of Projects within an Organisational Structure

7.2.2.1 Learnings from literature and theoretical review (Chapters 2 and 3):

- The IVC prototype needed to further align (and be tested against) the project management development process, referred to as the project life cycle (Morris, 2013), which is typically based on a number of iterative and normative stages, such as: plan, design, deliver, operate/maintain, and decommission. The prototype should have utility at all stages, but especially at the initiation stage to align stakeholders and inform the development of the business case. Consideration should be given to testing the prototype against projects at different stages of design-delivery.
- The OPM model (Muller et al., 2019) provides useful insights to the organisational context and project context. The IVC prototype therefore needs to be flexible for the needs from portfolio, megaproject and project levels.
- The OPM model also notes the project lifecycle imperatives and thus the prototype needs to be adaptable to different stages of the project. For example, the early definition stage will not have full details of the activities, but the prototype should align to the Green Book Five Case Model so that it can be used across the public sector at the investment appraisal stage.

7.2.2.2 Learnings from survey data (Chapter 5):

- The survey results (Chap 5, Finding #M2) highlighted the need to enhance business skills of engineers to better define benefits and impacts. In this regard, it was suggested by a number of respondents that embedding business skills learning within core engineering educational programmes would help provide opportunities for meaningful improvements in the measurement of SDG performance on projects. This was also linked to the survey feedback that there was a need to move from excessive focus on outputs towards more emphasis being placed on outputs-outcomes. The prototype should therefore also include the need for iteratively building knowledge within IVC teams to ensure sufficient foundation busines skills. This would include ensuring a minimum level of knowledge on SDGs as well as, for example, the capability to accurately define and measure benefits. This learning from the survey data is subsequently embedded within the SDGiPro illustrated in outline at Figure 61.
- The survey responses also showed that there is significant room for improvement on availability of 'fit for purpose' engineering tools and methodologies to measure SDGs. These results highlight the need for a new simple tool, which should be a design principle of the prototype.
- The survey also noted that measurement of SDG performance should accommodate the requirements at different organisational levels, namely portfolio, programme and project levels. The prototype should accommodate the flexibility of working across these hierarchies.

7.2.2.3 Learnings from interviews (Chapters 6):

 Most interviewees referred to the myriad of existing sustainability reporting frameworks available and that not all of these provided value to their organisations. Many referred to the inter and intra organisational boundaries, where each organisation has its own sustainability reporting requirements as part of an annual reporting cycle. The testing of the prototype should therefore not be viewed as "yet another" reporting mechanism that is loaded onto already stressed project managers, but instead, be seen as a way of providing global credibility and clarity of purpose when using the IVC to measure SDG impacts at organisational level. The tests in the following two chapters explicitly sought to determine the opportunity of using the IVC at the organisational-portfolio level.

A further prototype design criterion that emerged from the interviews, to enable the mechanism for measuring SDG impacts to achieve the outcomes, was the ability to find a golden thread from enterprise portfolio level to project level (Chap 6, Finding #10). This was most clearly explained by the participants that were most developed in their SDG measurement processes (2, 3, 11 and 20) but also included others who were actively developing SDG processes (8, 9, 14, 19, 27, 28 and 36). This could be done by choosing the Global Reporting Initiative as one framework to test for the 'Golden Thread' in Test 1, and secondly, to select an established sustainability champion at an organisational level for Test 2. However, in regard to testing the prototype, there was a lack of evidence given by interview participants on their ability to achieve the golden thread of SDG measurement from project to portfolio level (Chap 6, Finding #10) because, often, it was not available at any credible depth or backed up by verifiable evidence. It was therefore proposed that this was an area to be tested in Test 1 (Chap 8) to assess whether aspirations to achieve this linkage were realistic.

7.2.3 Concept 3 – Infrastructure investments as 'system of systems'

7.2.3.1 Learnings from literature and theoretical review (Chapters 2 and 3):

• The prototype should emphasise the concept of a system of systems in recognition that infrastructure projects in the built environment are more than the 'sum of their parts'. This can be achieved by highlighting the right-hand side of the ToC value chain that focuses on outcomes and impacts. In this way, the prototype could support organisations to plan in a more cohesive and systems-led way, which ultimately seeks to better manage the inherent inter and intra organisational complexities and boundary management issues.

7.2.3.2 Learnings from survey data (Chapter 5):

 The survey results (Chap 5, Finding #O2) indicated that there was strong belief that the choice of a career in engineering, was determined by a desire to provide a service to society. This was tested with relative weightings of priority goals
 – commercial or for broader societal or environmental goals, as defined by Creating Shared Value. The millennials were particularly strong in their desire to link engineering projects to SDGs. Thus, the prototype should enable a clearer alignment with delivering a construction 'system' that can be linked to broader TBL thematics, including using the SDG as icons to define that wider purpose, since they have strong resonance with millennials.

7.2.3.3 Learnings from interviews (Chapters 6):

- Concept 3 challenges the traditional understanding of infrastructure as standalone physical assets and the interviewees frequently (>50%) referred to the wider goals of the assets, and that this required a new way of thinking. Of note were three megaproject CEOs, who had harnessed a systems approach to delivery, who all stated they were champions of linking their projects to SDGs. The prototype could therefore add value to existing system integration tasks.
- Again, the megaproject leaders were consistent in their view that there was a need to develop a MISI approach that focuses on the identification of interrelationships between components (i.e. sub-systems) of a system. The immediate design and testing of the prototype should include the steps to better understand the sub-systems from the start, such as breaking down the carbon issues across stakeholders and especially across the supply chain. In this way, systems mapping can align some of the prototype IVC modelling. The prototype should be evolved in future research to understand how this can happen.
- The interviews highlighted the broad range of National Economic Infrastructure systems that they were involved with, such as: Water, Energy, Transport, Waste, and Telecommunications. The prototype should be flexible to work across all categories as well as the functions which the infrastructure system enables, such as: across healthcare services, transport services, and education services.

7.2.4 Concept 4 – Delivering SDG Impact measured against the TBL/SDGs

7.2.4.1 Learnings from literature and theoretical review (Chapters 2 and 3):

• The fourth concept aligns the TBL with SDGs and strengthens the composite IVC value chain. The IVC prototype should therefore provide a means 'to determine stakeholder's preferences and the trade-offs they choose to make

given their scarce resources, or the value the marketplaces on an item' (Porter, 1985).

• There was evidence from the literature review that the definitions of impact, outcomes and benefits are often confused. This was further confirmed in the survey and interviews. The IVC prototype should therefore explicitly define the different parts of the model and why a deeper understanding is necessary to ensure consistent use of the IVC tool. This supports the earlier recommendation to develop the SDGiPro alongside the IVC to help project teams have a sequential series of workshops to build knowledge across these areas.

7.2.4.2 Learnings from survey data (Chapter 5):

• The survey summary results noted that existing sustainability measurement at organisational and project levels is well established, and therefore, SDG measurement should be aligned to existing successful approaches, not created as an 'add-on' i.e., the organisational level will likely have different SDG imperatives and reporting requirements, such as using the GRI, from the project level, which might have limited capability and capacity to track too many targets and indicators. The prototype should therefore be seen to build on existing approaches, not as a replacement to existing sustainability and benefits management processes.

7.2.4.3 Learnings from interviews (Chapters 6):

- The results showed that interview participants have the appetite and resolve to employ SDG measurement at business and project levels (Finding #2) in order to achieve outcomes that benefit people, the planet and profit. This should be included as part of the communications to entice organisations to trial the IVC prototype.
- Many executives mentioned the need to articulate their contribution to society with improved focus on the ultimate societal SDG impact. For example, in a flooding project context, one executive explained that any MISI prototype should explicitly seek to understand their flood project systems' service to society by determining the causal link to the SDG impacts and use this as a coherent way to assess sustainability across the TBL of economic, social and

environmental impact areas as well as SDG impacts. The interviewee stated that the prototype would thus be able to engage stakeholders in a more meaningful way that would likely improve positive engagement as well as raising contributory funds from them (such as local councils) when they were better informed of the societal value-add.

• As part of the interview discussions on strengths and weaknesses, the participants identified a number of contextual issues that affected the likely success of the prototype mechanism likelihood of achieving the desired outcomes. These "context" issues included: leadership (Chap 6, Finding #3): outcome-output definition (Finding #4); knowledge (Finding #5); and change management (Finding #8) capabilities. The latter would be a key consideration to test the IVC prototype's SDGiPro methodology and would need to be addressed if agreement from trial organisations was to be achieved.

7.3 ITERATION OF THE IVC BY INCLUDING A METHODOLOGY FOR ADOPTION

Whilst the triangulation of empirical evidence, shown above, has strengthened understanding of what the IVC design should incorporate, there was also a large amount of useful evidence that addressed the 'employability' of the IVC. As such, there was an obvious gap that needed filling – the design of a complementary methodology for using the IVC prototype model. Two areas of design focus emerged: the design of a simple framework that would allow project and portfolio teams to develop, in a workshop environment, a causal path using the IVC, to define the 'ends, ways and means' of their project's SDG impacts; and secondly, a broader methodology titled the 'SDG Impacts on Projects', shortened to: SDGiPro© that would support the operational teams through a series of iterative steps, to build their knowledge on SDGs, definition of project success and other prerequisite capability areas (as emerged from the findings discussed above). These are discussed below.

7.3.1 Iteration of the IVC by simplification and 'baseline' model

Based on the feedback from the surveys and interviews, the prototype must be simple to use and ensure a clear causal link across the Theory of Change, showing a logical progression from inputs-activities-outputs-outcomes-impacts. Therefore, the IVC was developed further by designing a simple framework that would allow project and portfolio teams to develop, in a workshop environment, a causal path using the IVC, to define the 'ends, ways and means' of their project's SDG impacts

In practice, the TBL can be mapped against the five stages of the IVC as shown in Figure 29. The examples shown indicate that there are clear 'Theory of Change' patterns that build through the iterative stages and this can be linked directly to project and organisational level understanding of sustainability reporting.

	(a) Input	(b) Activity	(c) Output	(d) Outcome	(e) Impact
Economy	Finance/investment, insurance, risk contingency allocations, WLC analysis, stable government and noncorrupt financial context.	Job creation; income; wages; source, move and assemble materials; build iteratively through defined activities, such as early earthworks, and local and wider supply chain activity	Project completion to time/cost/scope— bridge, building, road, etc.; income; profit; taxes from in- project business and net present value provides strong RoI against whole life costs.	Economic growth enabled by completed assets as a system, more resilience, wealth creation, ownership, increased future investment and additional job creation.	SDGs 8, 9, 10 and 12.
Social	People, social networks, cultural and technical knowledge, and listening and working with stakeholders.	Collaborative innovation, health and wellbeing, stakeholder engagement, skills and learning, working conditions, production activity and user engagement.	Asset's social utility, meeting stakeholders' objectives, individual and group learning, and reinforced community stakeholder groups.	Infrastructure enabled change across health, education, etc., e.g., reduced mortality; gender equality; social equity; justice and post-project knowledge sharing.	SDGs 1, 2, 3, 4, 5, 7 and 11.
Environm ent	Raw materials, land take, water, light, clean air, energy, planned land use and ecology ecosystem valuation assessment.	GHG emissions; pollution; noise and air quality and works' effects pre and during production, e.g., waste management, nitrogen, carbon dioxide and acidification levels.	Managed effects on completion of asset; replanted trees, etc.; improved local area; no net loss on eco system footprint and short-term environmental targets met.	Restored/improved biodiversity and natural balance, e.g., increased long-term positive effect on environment through improved sustainability.	SDGs 6, 13, 14 and 15.

Table 23: IVC table illustrating golden thread mapping of the TBL in a 'baseline' model.

7.3.1.1 Development of the SDGiPro Methodology

Secondly, to support implementation of the IVC, a practical model was developed. This methodology, titled the '*SDG Impacts on Projects*', shortened to SDGiPro©, was aimed to support the project teams through a series of iterative steps,

that would enable them to build their knowledge on SDGs, definition of project success and other prerequisite capability areas (as emerged from the findings discussed above). The outline description of the methodology, with three design principles, that has been fully developed separately (not fully described in this thesis for reasons of brevity) into a workable methodology, is discussed below.

- SDGiPro Design Principle 1: Provide a sequential stepping-stone approach for accessing the MISI terminology and theories. SDGiPro is a framework that provides the guidance needed to initiate systemic change in the planning, design and delivery of sustainable and resilient infrastructure. SDGiPro is a decisionmaking guide, not a set of prescriptive measures. SDGiPro provides a gateway to access the language of the United Nations SDGs. SDGiPro enables project teams to adapt the organisational SDG intent into relevant local SDG metrics for all types and sizes of projects to help users assess and measure the extent to which their project contributes to conditions of SDGs across the full range of social, economic, and environmental indicators.
- SDGiPro Design Principle 2: Designed to enable the IVC to align with existing processes. Fundamentally, SDGiPro is about supporting higher performance through more sustainable choices in infrastructure development. The methodology provides a flexible system of SDG criteria and performance objectives, aligned to existing business case and benefits management processes as well as existing sustainability tools such as CEEQUAL. This aids decision makers and helps project teams identify sustainable approaches during planning, design, and construction that will carry forward throughout the project's operations and maintenance and end-of-life phases. Using SDGiPro as a guidance tool, owners, communities, designers, contractors, and other stakeholders are able to collaborate to make more informed decisions about the sustainability of infrastructure.
- SDGiPro Design Principle 3: Adaptable for portfolio, megaprojects and projects. Community infrastructure development is subject to the resource constraints of multiple departments and agencies, each with different schedules, agendas, mandates, budget cycles, and funding sources. SDGiPro assesses not only individual project performance, but how well the infrastructure project contributes to the efficiency and long-term sustainability of the communities it serves. In this

way, SDGiPro not only asks, "Are we doing the project right?" but also, "Are we doing the right project?"

The design principles led to a methodology with five proposed steps that have emanated from the triangulation of empirical evidence. The framework is a proposed way to initiate the "right project" in the "right way" and with increased clarity of "Ends, Ways and Means" aligned to SDGs.



Figure 61: The proposed 5-step SDGiPro methodology.

The SDGiPro methodology included the development of a detailed set of workshops that are summarised below:



Figure 62: The SDGiPro iterative workshops.

7.4 CHAPTER CONCLUSIONS

This chapter has summarised research into the development of a prototype from the triangulation of data from Stage One Exploratory Design. This has led to the IVC being jointly developed with a methodology, the SDGiPro©, that had been identified from the survey and interviews as a necessary extension to the IVC to support its implementation in 'active' projects. This evolution of the MISI IVC tool meant that the early IVC prototype had been developed to a stage that was ready for testing in Test 1 (SDG 'Golden Thread') and Test 2 (in a water utility company context), discussed in Chapters 8 and 9 respectively. It should be noted that 'test' is actually a 'demonstration', since full validation testing is required prior to building a digitised version of the IVC that would have broader utility across the sector.

The further strengthening of Proposition 4 is shown in the table below.

Confidence in	P1:	P2:	P3:	P4:
Evidence for	Engineer's view of	Organisational	'Golden Thread'	Prototype
Propositions	current MISI	view of MISI	global - to - local	Development
Theoretical	Not yet	Not yet	Very Low	Very Low
Analysis	-	-	-	
Survey Analysis	Moderate	Very Low	Very Low	Very Low
Interviews	High	Moderate	Very Low	Moderate
Analysis	_		-	
Test 1: Golden	Not yet	Not yet	Not yet	Not yet
Thread	-	-	-	
Test 2: Case	Not yet	Not yet	Not yet	n/a
Study	-	-	-	

Table 24: Development of stability of Propositions' Results (changes since previous Chap in red)

(Note 1: The assessment of the quality of propositions' evidence is based on a qualitative formula designed by the Clinical Grading Working Group's British Medical Journal's study (GRADE, 2004). Full description and justification for this approach is shown in Table 8.)

Chapter 8: Test 1: Is there a 'Golden Thread' from global SDG, through the Organisational Layer (portfolio), to local project level?

8.1 OUTLINE OF CHAPTER

This chapter discusses the first case study test, known as the 'Golden Thread' test. First, in Section 8.2, it places the investigation into the context of the broader aims of the Stage Two Detailed Investigation. In Section 8.3 it describes the context for the test and follows this by describing the 'Golden Thread' test's methodology in Section 8.4. The results are shared in Section 8.5 and this leads to the discussion of the implications that inform the research objectives and propositions.

8.2 STAGE TWO – DETAILED INVESTIGATION

The detailed research design was described in Chapter 4. This determined that the second stage of the investigation sought to build on Stage One by firstly triangulating the data with analysis from the literature review, the survey and the interviews, and secondly, to build a prototype model that could be tested. Through the iterative steps of Stage Two, shown in Figure 63 below, the purpose was to address a gap in knowledge through a series of test events that enabled the gradual deepening of understanding and thereby, to address the research question as well as the propositions.



Figure 63: Detailed Investigation (DI) Flow Chart.

8.2.1 Context for the Test – Identifying a Golden Thread

In the search for a way to link 'bottom-up' project delivery and the 'top-down' Global Goals, empirical evidence was needed to substantiate the third proposition⁹. The intent was to identify a 'golden thread' between best practice sustainability reporting frameworks at the project level, with those at organisational level. In doing so, it aimed to establish whether there is sufficient linkage to embed SDG impact targets at the design stage of an infrastructure project, thereby providing a more robust investment appraisal at the project design phase, which defines project success more widely across the 'Triple Bottom Line' (TBL) of economic, social and environmental outcomes and associated SDG impact.

8.3 METHODOLOGY FOR GOLDEN THREAD TEST APPROACH

In the search for a 'golden thread' between ground-level project delivery and the SDGs (to address the Research Objectives RO6 and RO7¹⁰ and answer the Propositions 3 & 4¹¹), two sub-tests were conducted (shown in Figure 64). Sub-Test 1.1 analysed whether the project-level sustainability reporting tool CEEQUAL (BRE, 2019), mapped to the TBL. Sub-Test 1.2 explored whether CEEQUAL could be mapped to SDG global level goals. Sub-Test 1.3 explored whether the organisationallevel GRI approach (GRI, 2019) could be mapped to project-level (CEEQUAL) and this was followed by Test 1.4 that assessed linkage to SDG global level goals. If these tests proved positive, then there would be evidence to support the measurement of SDG at organisational and project levels. The logic-based flow of the tests is shown in Figure 64.

⁹ Proposition 3: Current sustainability measurement at organisational and project levels can be used to demonstrate a 'golden thread' from 'global' SDG to 'local' portfolio and project levels.

¹⁰ Research Objectives: RO6: To test whether a 'golden thread' of SDG measurement could be identified from global to local levels. RO7: To test whether the prototype could be validated with a case study organisation.

¹¹ Proposition 4: A MISI prototype can provide a plausible, testable and achievable logic chain for defining project and portfolio SDG impacts.

Chapter 8: Test 1: Is there a 'Golden Thread' from global SDG, through the Organisational Layer (portfolio), to local project level?



Figure 64: The IVC Testing Research Design.

Another view of the conceptual link that Test 1 was seeking to explore, across the TBL (economic, social, environmental) and the OPM model (project to portfolio) to national and global levels, is shown in Figure 65 below. The diagram shows a conceptual pathway of how an SDG measured at project level, might inform measurement at the portfolio, national and global levels.



Figure 65: Conceptual map of potential 'golden thread' from local to global levels.

Chapter 8: Test 1: Is there a 'Golden Thread' from global SDG, through the Organisational Layer (portfolio), to local project level?

8.3.1 Selection of representative project and organisation level methods

To establish a potential golden thread from global to local, the study required two leading sustainability reporting methods, with international and national credibility. This led to the selection of CEEQUAL (which was compared with other global project measuring tools) and the Global Reporting Initiative's (GRI) global standard for organisational sustainability measurement. This stage of the research was conducted with the collaboration of both GRI and the Building Research Establishment (BRE), which is UK's leading centre of building science. BRE, as the owners of CEEQUAL, gave full access to their systems and standards to enable completion of the detailed text and process analysis of both standards in comparison to the SDG targets and indicators. A summary of some of the leading sustainability reporting frameworks from this evaluation of suitable tools, at organisational and project levels, is shown in Table 25 with a brief analysis of their explicit or implicit alignment with SDG measurement. It does not purport to provide a full in-depth comparison or discussion of the relative merits.

Table	25:	Summary	of	leading	infrastructure	sustainability	reporting	tools/methods	at
organi	sation	al and proje	ect le	evels.					

Tools and Methods	Relevance for the Research
<u>1. Organisational level tools and</u> <u>methods.</u> Global Reporting Initiative (GRI, 2019), UN Global Compact (2019), Carbon Disclosure Project (Matisoff et al., 2013), GHG Protocol (Barrett et al., 2013), OECD guidelines (Barkemeyer et al., 2014) and integrated reporting (De Villiers et al., 2014).	Based on analysis of the industry leading sustainability reporting frameworks (Corporate Reporting Dialogue, 2019, and PwC SDG Reporting Challenge, 2018), GRI was shown to be the most frequently used by leading companies. Indeed, of the world's largest 250 corporations, 92% report on their sustainability performance and 74% of these use GRI's standards to do so, with 23,000 corporate sustainability reports currently in the GRI database (Global Reporting Initiative, 2019). For example, it was used by 6671 organisations in 2017 (GRI database, 2019) and 75% of Fortune 250 companies (KPMG, 2017) across 91 countries.
	Whilst the UN Global Compact has the

Whilst the UN Global Compact has the "SDG Compass" methodology to support organisations to measure SDG impacts at

subnational level, it remains at a high level and does not include any accepted standards for measurement of sub-national criteria. The case study expands on the challenge of trying to use the national level targets at organisational and project levels.

2. Project level tools and methods.

Thirteen sustainability assessment methods were examined, including the following: CEEQUAL (UK & Ireland Projects/International Projects) (BRE, 2019), BREEAM (BRE, 2019), Halstar (Pearce et al., 2012); SPeAR (McGregor and Roberts, 2003), ASPIRE (Siew et al., 2013), ISO14001 (ISO, 2019), OHSAS 45001 (ISO, 2019), Jacobs Value (Gasparatos, 2010), LEED (Awadh, 2017), ENVISION rating system by ISI and Harvard University (Shivakumar et al., 2014), IS rating scheme by the Infrastructure Sustainability Council of Australia (ISCA, 2019), infrastructure voluntary evaluation sustainability tool (INVEST) (Clevenger et al., 2013), SuRe® Standard for Sustainable and Resilient Infrastructure (Butler et al., 2014), sustainable transportation appraisal rating system framework (STARS) (Sakamoto, 2014), IFC Performance Standards on Environmental and Social Sustainability, and World Bank Environmental and Social Framework.

The project-level sustainability frameworks were assessed against their ability to measure SDGs. Most of these were developed before the SDGs were agreed at the UN by the 193 states in 2015 and thus have no formal linkage to SDG measurement. Some, such as CEEQUAL, have started to link to both SDGs and to the GRI to establish a golden thread from project level to organisational level to national-global levels.

However, although this research has confirmed there is the potential for the golden thread from project to global goals, there is only sporadic evidence of projects and organisations having achieved this requirement.

Therefore, this confirms the knowledge gap and explains why the case study discussed in the next chapter (the choice of Anglian Water was motivated by their award of the UK's national prize in 2017 as "Sustainability Company of the Year").

8.3.2 Identifying the sustainability tools as the 'Reference Class' for analysis.

The focus of the investigation was on the detailed analysis of existing sustainability reporting methods across two of the hierarchy levels, i.e. at the project and organisational levels. Whilst there are literally hundreds of sustainability methods used globally, from simple spreadsheet-based approaches to enterprise wide, cloudbased systems, the selection of the two methods was based on meeting four criteria: (1) extent of uptake based on the percentage of use; (2) recognition by reporting authorities, including having government endorsement; (3) currency, with the latest updates reflecting 2018-2019 changes in legal and advisory frameworks; and, (4) accessibility of data sets to enable detailed analysis. Based on these criteria CEEQUAL (BRE, 2019) was identified as the leading international sustainability reporting method for infrastructure at the project level. It also identified the Global Reporting Initiative's (GRI) Standard (2019) as the most frequently used reporting tool at the organisational level. Indeed, from the world's largest 250 corporations, 92% report on their sustainability performance and 74% of these use GRI's Standards to do so, with 23,000 corporate sustainability reports currently in the GRI database (Global Reporting Initiative, 2019). Both of these methods are described in more detail below.

8.3.3 Project-level selection of sustainability assessment technique

Thirteen sustainability assessment methods were examined, including the following: CEEQUAL (BRE, 2019); BREEAM (BRE, 2019); Halstar (Pearce et al., 2012); SPeAR (McGregor and Roberts, 2003); ASPIRE (Siew et al, 2013); ISO14001 (ISO, 2019); OHSAS 45001 (ISO, 2019); Jacobs Value (Gasparatos, 2010); LEED (Awadh, 2017); ENVISION Rating system by ISI and Harvard University (Shivakumar et al., 2014); IS Rating Scheme by Infrastructure Sustainability Council of Australia (ISCA, 2019); Infrastructure Voluntary Evaluation Sustainability Tool (INVEST) (Clevenger et al., 2013); SuRe® Standard for Sustainable and Resilient Infrastructure (Butler et al, 2014); and, Sustainable Transportation Appraisal Rating System framework (STARS) (Sakamoto, 2014). These frameworks were assessed against the selection criteria set out above and CEEQUAL scored the highest and was adopted within the research. CEEQUAL was the first evidence-based sustainability assessment, rating and awards scheme for civil engineering. It is less 'stick' and more 'carrot' to support a positive learning environment through structured discussions and performance management of sustainability issues. The CEEQUAL method provides a

rigorous and comprehensive sustainability assessment and rating approach that supports clients, designers and contractors to improve the specification, design and construction of infrastructure.

8.3.4 Organisational-level selection of sustainability assessment technique

Seven sustainability approaches were considered at the organisational level: Global Reporting Initiative (GRI, 2019); UN Global Compact (2019); Carbon Disclosure Project (Matisoff et al., 2013); GHG Protocol (Barrett et al., 2013); OECD Guidelines (Barkemeyer et al., 2014); Integrated Reporting (De Villiers et al., 2014). Based on the selection criteria and analysis by the industry leaders (Corporate Reporting Dialogue, 2019; PwC SDG Reporting Challenge, 2018), GRI scored highest amongst the global methods analysed, particularly on acceptance and recognition criteria. For example, it was used by 6,671 organisations in 2017 (GRI database, 2019) and 75% of Fortune 250 companies (KPMG, 2017) across 91 countries.

8.3.5 Selection of methods for each of the tests

Two tests were developed to address the research proposition. These required a variety of analytical methods, which are discussed below. The methods chosen reflected the different nature of the two sustainability reporting tools. Both are voluntary, allow selective use of areas that are self-assessed as relevant to the project/business and have an embedded management process that encourages dialogue with stakeholders. Most importantly, they both champion the fundamental principles of effective governance (OECD, 2011) of accountability, responsibility, transparency and fairness (Muller, 2017). However, despite these similarities, there are some fundamental differences, which are shown in Table 26.

Feature	CEEQUAL	GRI Standards
Coverage	Project level.	Organisational level.
Sectors	• Infrastructure / built environment across public, private and NGOs.	• All sectors across public, private and NGO.
Accountability	• The project director takes accountability for the report and its management.	• Report usually authorised by the Corporate Board.
Responsibility	• Voluntary.	• Voluntary.
Assessed	• Verification and rating issued.	• Self-assessed, with option of external assurance - although only 31 (1.1%) of the 2,902 reports uploaded to-date in 2018 and analysed on the GRI database, described their

		external assurance as 'Reasonably High'. No rating given on reports.
Transparency	• Detail kept private but award rating made public unless the client opts out of sharing data.	• Public.
Fairness	• A tightly controlled structure with assessment of evidence provides a balanced rating award.	• The GRI standard is widely used although only a small % use the full report, and very few (31 out of 2,902 in 2018) have a high level of external assurance.
Measurement against TBL	• Implicit (embedded within criteria focused on project team delivery).	• Explicitly structured on the three core areas of: GRI 200 Economic; GRI 300 Environmental; GRI 400 Social.
Link to SDG	• No current linkage.	 No current linkage, although GRI part of UN Global Compact (UN-Business leaders' group for SDGs) to build connections e.g. SDG Compass has a methodology to do so.
Updates	• New version launched in June 2019.	• New GRI Standards launched in July 2018.

8.3.6 Does CEEQUAL map to GRI across the IVC thematic areas?

The first technique applied was the use of a high-level matrix mapping technique that compared the CEEQUAL Categories with GRI Materiality Topics. The second method used was a text mining/analysis technique to identify intertextual patterns (Foucault, 1973) of significance. Both of these methods used the IVC framework to structure and prioritise the topics of value for analysis.

8.3.7 High Level Analytical Matrix Mapping of linkage to TBL

The method for building high-level associations between CEEQUAL Categories with GRI Materiality Topics was a simplified version of the 'ecosystem service matrix' (Jacobs et al., 2015; Burkhard et al., 2012). This approach builds a tabular format to test strength of linkages across two dimensions and then subsequently uses expert groups to test the strength of the connection points. This part of the test was limited to input from a 'reference group' convened and hosted by the Institution of Civil Engineers (including BRE, ICE, UNOPS and UCL collaborators – See Appendix 6 for the write-up by the ICE Knowledge Manager of one of these events). The aim of the review group was to construct an initial composite measure, such as identifying key indicator words and primary 'hot spots' across the samples that could be used in the second phase of Test One.

8.3.8 Detailed Text Mining-Analysis to establish IVC links between CEEQUAL and GRI

The chosen method for detailed analysis was Text Mining-Analysis. With the advances of software solutions, Text Mining is used as a methodology for social scientists to support text analysis because it offers the ability to manage and quantify huge amounts of data in a very short time. It is used across academic disciplines such as economics (Levenberg at al., 2014), political science (Grimmer and Stewart, 2013) and sociology (Mische, 2014). The specific technique used for this study was Named Entity Recognition which provides a statistical technique to capture key 'indicator' words as part of the content analysis (Krippendorff, 2018). This requires a coding frame that was built on the IVC four core concepts (see Section 3.4). An advanced technique of comparing key words between texts was first defined by the philosopher and historian Foucault (1973) who identified the intertextual patterns that can determine answers to social science questions. In order to identify intertextual patterns, text mining requires a hierarchy model, or 'tree map' that in this case used the IVC framework to link nodes of key information, with sub-nodes and specific words that are associated with the four IVC concepts. For example, the first concept is based on the Theory of Change that has a linear progression linking inputs, through activities and outputs, to outcomes and impacts. These are shown in the top part of the relationship chart, with the inclusion of 'benefits' and 'value' as additional words of high interest.



Figure 66: Tree map linking IVC four concepts to the Key Indicator Words via nodes and subnodes.

The tree map in Figure 66 illustrates 6 primary nodes, the 13 sub nodes and 42 Key Indicator Words. The analysis of the words was enabled by a specialist software tool, NVivo, which is a qualitative data analysis software package that enables rapid analysis of large quantities of data. The tool was used to provide detailed text analysis of the prioritised Key Indicator Words as shown in Figure 66, across the two publications in Table 27.

Table 27: Selected Manuals for analysis: CEEQUAL and GRI.

Methodology Manual	Title	Pages	Words
CEEQUAL	CEEQUAL V5.2 Technical Assessment Manual	148	77,698
Global Reporting Initiative	Consolidated Set of GRI Sustainability Reporting Standards 2018	542	152,797

Using these techniques, it was anticipated that the research in Test One would provide evidence as to whether the Analytical Model (embedded in Figure 64), using the IVC concepts, enabled a way to find a 'golden thread' from project to organisational levels. Test Two was aimed at providing the means to extend the linkage all the way through to the SDG Impacts.

8.4 RESULTS AND DISCUSSION

In the search for a 'golden thread' between bottom-up project delivery and the strategic level of the Global Goals, four related theoretical models were used. The results are captured below.

8.4.1 Test 1.1. Does CEEQUAL map to GRI across the IVC thematics?

8.4.1.1 Part One of Test 1.1

Through the use of the high-level Analytical Matrix Mapping, it was confirmed that there are verifiable linkages between the CEEQUAL Categories with GRI Materiality Topics. This approach builds a tabular structure (Figure 67) that is captured in bar chart format (Figure 68) to show the level of connectivity across the three TBL areas of Economic, Social, and Environment.

The data in Figure 68 shows that the CEEQUAL Categories (Y axis) has strong correlation with GRI Standards' (X axis) thematic topics of Management (GRI 101), Environment (GRI 300) and to a lesser degree, there is reasonably strong mapping in 40% of the GRI materiality topics in Economic (GRI 200) and Social (GRI 400) areas, as shown in Figure 68.



Figure 67: High-level Analytical Matrix Mapping showing linkages between CEEQUAL Categories (y-axis) with GRI Materiality Topics (x-axis).

The chart shown in Figure 68 below, can be interpreted as: Y-axis shows the number of occurrences in CEEQUAL's 246 questions; On X-axis, the left (blue) indices are GRI 200 Economic Material Topics, middle (green) = GRI 300 Environmental, right (brown) = GRI 400 Social.



Figure 68: Bar Chart showing the instances of 'hot spots' where alignment is identified.

The results identified the following key findings:

- The three areas of TBL do link across from CEEQUAL to GRI, although they are only implicit in CEEQUAL, whereas for GRI, the labelling is explicit.
- There are sufficient linkages to give confidence of a credible basis to assume that project level sustainability reporting using CEEQUAL, could be grouped under similar TBL categories to GRI, which would help organisations align sustainability reporting. It also provides the first half of the 'golden thread'.
- The evidence is subjective (since it is based on the review group's views) and needs further development to further strengthen the stability of the findings. This was done in part 2 of this test, using text analysis techniques.

8.4.1.2 Part Two of Test 1 (eg 1.2). Detailed Text Mining-Analysis to establish IVC links between CEEQUAL and GRI

The chosen method for detailed analysis was Text Mining-Analysis, using the qualitative analysis NVivo software tool (Bazeley and Jackson, 2013). The test analysed Key Indicator Words that relate to IVC's four concepts across CEEQUAL and GRI.

								GRI			CEEQUAL						
	f1	Deine and Nicola	f2	Cub Norda	f3	Key Indicator	# of	% for	f2.1	f3.1	# of	% for	f2.2	f3.2			
	%	Primary Node	%	Sub-Node	%	Words	occurrences	document	%	%	occurrences	document	%	%			
						Value	100	0.01			96	0.2					
				Ends' Impact-		Impacts	976	0.97		0.07	267	0.55		0.24			
		Impact-Value		Value Chain		Benefits	86	0.09		0.27	75	0.15		0.24			
		Chain	0.14		0.18	Outcomes	11	0.01	0.20		30	0.06	0.16				
		(IVC Concept 1)		Ways & Means'		Outputs	10	0.01			16	0.03					
				Impact-Value	0.1	Activities	361	0.36		0.13	79	0.16		0.07			
				Chain		Inputs	31	0.03			13	0.03					
						Economic	633	0.63			159	0.33					
		Impact-Value		TBL	0.53	Environmental	546	0.54		0.58	395	0.8		0.48			
		TBL / SDG	0.37			Social	578	0.57	0.43		147	0.31	0.31				
		(IVC Concept 4)		Sustainable	0.21	SDG	0	0		0.20	0	0		0.15			
				Development	0.21	Sustainability	557	0.56		0.20	140	0.29		0.15			
				Hiorarchy	0.7	Project	81	0.08		0.14	1186	2.44		1.26			
				Theratchy	0.7	Organisation	204	0.2		0.14	37	0.07		1.20			
		Impact Value		Employment	0 17	Employment	408	0.4		0.21	13	0.03		0.04			
	Impact-value	Structures (IVC Concept 2)	Structures	0.27	Linpioyment	0.17	Safety	331	0.21	0.22	0.51	25	0.05	052	0.04		
			0.37	Stakeholders	0.24	Client	0	0	0.22	0.22	289	0.38	0.52				
						Supplier / contracto	443	0.44			101	0.2		0.26			
						stakeholders	279	0.28		0.22	32	0.07		0.20			
Imnact-	mnact					communities	160	0.16			183	0.38					
Value	03					Water	580	0.58			236	0.49		r i			
Framework	0.5			Thematic		Energy	189	0.19			131	0.27					
	Iework	Impact-Value		Topics	0.23	Health	390	0.39		0.24	34	0.07		0.22			
		Reporting		Topics		Transport	55	0.05			121	0.25					
		Thematics	0.20			ICT	0	0	0.21		0	0	0.20				
		(IVC Concept 3)				Emissions	393	0.29			60	0.12		ſ			
		(,		Footprint	0.18	Effluent/discharge/	219	0.23		0.18	209	0.43		0.18			
						carbon	13	0.01		0.10	72	0.15		0.120			
						GHG	187	0.19			12	0.02					
						Disclosure	1786	1.78		Ĩ	0	0		ſ			
				Capture of		evidence	0	0			638	1.31					
		Impact-Value		responses and	0.52	methodology / proc	114	0.11		0.45	137	0.28		0.60			
		Management		data		Assessment	267	0.27			607	1.25					
		process	0.53			achievement	66	0.07	0.23		82	0.17	0.83				
		(IVC Concept 5)		0		score	0	0			808	1.66		[
				Quanitrying	0.53	verifiers	0	0		0.01	60	0.12		1.06			
				relative success		monitoring	15	0.01			120	0.25					
						award	9	0.01			46	0.09		-			
		Castan (Commercial	0.17	legal	125	0.15		0.22	/2	0.15		0.13			
		Sector /	0.29			contract	284	0.28	0.12		56	0.1	0.47	17			
		commercial		Sector	0.41	Construction / Infra	27	0.03		0.02	678	1.39		0.81			
									Engineering	0	0			112	0.23		

Table 28: Data Analysis using NVivo: Nodal-Word linkages.

The table above is interpreted as follows: Column description: f1 = the average % of the 42 Key Indicator Words usage in the combined documents of CEEQUAL and GRI Standards; f2 = the nodal average %; f3 = the sub-node %; f2.1 and f3.1 are the GRI average % use of each key word within the nodes and sub-nodes; f2.2 and f3.2 are the equivalent for the CEEQUAL document.

It should be noted that the percentage figures in the two columns in Table 28, that are titled '% for document', represent how many times the key indicator word appeared in the relevant document as a percentage of the total words (only counting the words of 3 and above letters). It was a coincidence that the GRI total words came close to 100,000 words, thereby giving a metric correlation. For example, the key word 'impact', which is part of the 'Ends IVC' sub-node group, had 976 appearances in the GRI document which neatly represents 0.97%, (representing nearly 1 in every 100 words of 3 letters and above, therefore highly relevant), and 267 (0.55%) in the CEEQUAL document, representing about 1 in 200 words. The summary of the table is shown below:



Figure 69: Data captured from NVivo analytical tool showing strength of connections across the 4 concepts in IVC from project level to organisational level (full data in Annex). For description of the columns, see Table 28 title.

The results shown in Table 29 below and Figure 69 above, illustrate the percentage of occurrences of each key indicator word across the documents which has allowed results to be interpreted and a possible link from project-to-organisational level sustainability reporting to be assessed. Using the example given above on the analysis of the 'impact' key word, it implies that there is more emphasis on the post-project impacts in the GRI, but caution should be applied to linear linguistic comparisons because there are subtleties that need to be considered, (noting that a key

issue influencing the findings is that CEEQUAL is largely project orientated and that GRI is organisational focused) such as:

- CEEQUAL does not explicitly refer to economic issues as frequently as GRI but implicitly considers economic benefits from approaching sustainability from an efficiency and effectiveness perspective.
- CEEQUAL has less use of the word 'social' but places more of an emphasis on social issues through reference to stakeholders and communities. As a result, these should be seen as synonymous.
- CEEQUAL uses language specific to the engineering and infrastructure sector, whereas GRI uses generic language due to it being for all sectors.
- CEEQUAL is more detailed in its language, reflecting the tactical nature of its projects' activities and outputs. It is apparent that CEEQUAL does not use the language of outcomes and benefits, but instead, partly covers for this by use of 'impact' but used in a different sense to the IVC definition.

The main findings from the analysis are captured in Table 29, with the corresponding findings indicated in bold **'F**', which are summarised in the table below: Table 29: Key data results from the NVivo text analysis (See Appendix 10 for full data).

Nodes	CEEQUAL	GRI
Impact Value Chain (Concept 1)	 CEEQUAL uses 'impact' but at a level of half the frequency of GRI. It tended to use 'value' and 'benefits' more, perhaps as compensation. Both rarely used 'outcomes' that suggests the Theory of Change and global programme management terms are not well known or widely used. 	 The GRI had the strongest alignment to Theory of Change terminology, especially 'Impacts' (0.97%) – i.e. almost 1 in every 100 words. GRI rarely uses 'value' or 'outcomes', both at less than 0.001%. (F1)
TBL (IVC Concept 4)	• CEEQUAL had fewer references to 'economic' factors (0.33% vs. 0.63%) but has implicit economic criteria embedded in the efficiency of the management processes that address the sustainability questions. Both CEEQUAL and GRI had no reference to 'SDG' (0%). (F2, F3)	 GRI had stronger reference to the two of the core areas of TBL ('Econ', 0.63%; 'social', 0.57%). GRI had stronger reference to 'sustainability' (0.56%)
Structures (IVC Concept 2)	 CEEQUAL has an explicit focus on the 'project' level (2.5%) but an equal focus on stakeholder engagement. It has greater focus on 'communities' (x2) and a main focus on the client – in effect, CEEQUAL is about the value chain working better. The high use of 'communities' could have been aligned with 'social' in the TBL/Concept 4 – they are synonymous. 	• GRI has an explicit focus on the 'organisational' level and a greater focus on 'safety' (x4 of CEEQUAL, which recognises there are other tools covering safety at project level) and 'employment' (x10).

Reporting (IVC Concept 3)	• Both levels give equal priority to thematic reporting across 'water/energy' etc.	 Both have equal focus on footprint areas ('GHG', 'emissions', 'discharge' at 0.18%). 'Carbon' is rarely used by both.
Management processes – e.g. this relates to the process of CEEQUAL and not what is being assessed	 CEEQUAL has a significantly greater interest in 'score', 'verify', 'monitor', 'award', 'assessment'. This indicates the strong focus on verifiable evidence. In effect, this gives it teeth, albeit, in a low reputational risk way – data remains confidential. Also, 'achievement' sits across a number of nodes because it also aligns with 'outputs and outcomes' of the IVC in the first node. The focus of CEEQUAL assessment is split between internal governance and external verification. 	 Both have a similar level of emphasis on the capture of 'response' data. GRI uses the term 'disclosure' as primary term. The reports are loaded onto the GRI website, but the strength of reporting varies significantly, which is not easily identified on the website. Whereas, for CEEQUAL, the assessment is about encouraging verification so that they are having the right discussions on the right issues, early enough to impact sustainability. Thus, CEEQUAL is proactive, GRI is more reflective in approach. (F4)
Sector specific / commercial	• The focus on 'infrastructure' and 'construction' was reflected in the key word usage (1.4 in every 100 used).	 Very low reference to specific sectors since GRI is for all sectors. Similar use of 'legal' but more use of 'contract'. (F5)

8.4.2 Emerging issues from Test 1.1 (Parts 1&2):

The research appears to indicate that there is supporting evidence of a golden thread, across all of the TBL lines, as shown in Figure 69. The data in the tree map highlights that on the left originating side, there is an average of 0.3% use of the 42 key indicator words (see Figure 66) across the two core documents. The diagram (Figure 69) shows the quantitative data that indicates six main similarities and differences between the two methodologies, which are as follows: 1) There are specific areas of verifiable linkages between CEEQUAL Categories with GRI Materiality Topics, as well as gaps - the linkages suggest a verifiable golden thread; 2) CEEQUAL's project-level sustainability reporting places more emphasis on environmental issues and social issues; 3) Economic issues are addressed at half the frequency at project level than at organisational level, which suggests that other economic tools, often related to the business cases, are being used at project level and also, that economic criteria are implicitly embedded in the efficiency of the management processes that address the sustainability questions; 4) The 'SDG' key indicator word is not used which is partly explained because SDGs are a relatively new concept and sustainability reporting frameworks have been developed over many years and take years to change, but this potentially delays the ability of making explicit linkages from projects through to SDG targets; 5) The CEEQUAL reporting approach has a significant focus on assessment and verification of evidence to encourage the client/contractor/designer to have the right sustainability discussions on the right issues, early enough to impact the efficiency and effectiveness of the project's

sustainability footprint. Thus, CEEQUAL is proactive. GRI is more reflective in approach, capturing sustainability achievements and actions against the TBL themes in their annual reports; 6) Both are intended to be voluntary and rely on the 'carrot' of highlighting good performers, instead of the 'stick' of reputational or fiscal penalties.

8.4.3 Findings from Test 1.1 (Parts 1&2 – see Table 29):

- Different tools are needed for different project and organisational levels. A suite of tools enables the optimal performance level of sustainability measurement specific to both the project level and organisational level. However, a golden thread runs through all levels, based on the TBL, which provides a route from tactical level project delivery to strategic SDG impacts.
- While recognising that the two approaches are focused at different levels, there is an opportunity to strengthen SDG coherence in future versions by increasing use of IVC terminology, especially the terms of: 'outcomes' and 'impact'. This could be supported by the ICE providing learning and development (L&D) education of the IVC theoretical and practical usage, perhaps aligned with the Enterprise-view of Project 13 (ICE, 2018). Both encourage a value and outcomes related view of investment appraisal and benefits realisation.
- The linkage between project-organisation sustainability reporting can be increased by explicitly labelling project level thematics areas by TBL headings. Given that most users do not have recognition of the TBL terms, an overlay of explicit 'signposting' to the TBL could be applied and supported by further L&D.
- SDGs in both project level and organisational level reports need to be explicitly referenced.
- Economic TBL-IVC issues at project level need to be explicitly increased, so that TBL parameters are considered holistically across economic, social and environmental related topics. This could include a mechanism to cost social and environmental impact/value so that economics aspects more explicitly drive the TBL sustainability decision-making process.
- With strengthened requirements for reporting at government and industry levels, the collection of reporting data at project level should be centralised and shared, in order to allow knowledge sharing and increase efforts to improve results.

Project level reporting increases the linkage to economic targets to emphasise the overlapping areas of influence across all three TBL. This is of greater significance to the second area, post-project. By doing this, there will be increased recognition by senior managers of their interconnectivity. TBLs are currently reported in silos at project level and this loses understanding of potential positive and negative impacts of the investments. For example, increased use of TBL valuation tools, from the start through to project completion, would strengthen investment decisions-making and analysis of lessons learned.

8.4.4 Test 1.2. Does CEEQUAL map to SDGs?

The second sub-test explored whether CEEQUAL could be mapped to SDG global goals. The outputs from the full matrix mapping tool are held electronically by the researcher (available on request), but are shown at a high-level below in Figure 70. The pie chart indicates a strong focus (50%) on environmental issues, with approximately a half of the questions spread across the economic (19%) and social (34%) TBL related areas.



Figure 70: CEEQUAL's relative focus on TBL and across SDGs (full matrix analysis shown in Appendices).

The bar chart illustrates the relative connectivity (i.e. touch points) across the individual SDGs which is further illustrated in the systems mapping diagram shown below in Figure 71.



Figure 71: Systems mapping of connections between CEEQUAL and SDGs.

The results of the analysis from Test 2 are as follows: Three of the SDGs (9, 12 and 15) have strong connectivity (where a linear, evidence-based, linkage can be identified that could provide an objective level of 'attribution') to CEEQUAL; eight of the SDGs (3, 4, 6, 7, 8, 11, 13, 14) have an indirect connection (where a linkage is identified at a 'contribution' level which is without an evidence-base to objectively substantiate the link) and six have low or no connection (1, 2, 5, 10, 16, 17). This provides insight to the prioritisation process at a project design stage as to which SDGs are used to assess SDG impact. There is confidence that a link can be made from project level tactical activities and outputs to the more strategic level outcomes and impacts of SDGs.

Examples of these three categories are as follows:

Strong connection identified: Target 6.1 for SDG6, with 169 targets at https://sustainabledevelopment.un.org/sdgs), By 2030, 'achieve universal and equitable access to safe and affordable drinking water for all', that can be captured by CEEQUAL under question 3.5.4, which relates to the number of people with access to safely managed drinking water. Attributes could for

instance be the increased number of local communities who have access to clean water.

- Indirect connection identified: Target 7.3, '*By 2030, double the global rate of improvement in energy efficiency*', that linked to CEEQUAL question 8.4.1-4, 8.5.1-2, but where there is no attribution metrics to justify this linkage.
- No connection identified: Target 8.1, 'Sustain per capita economic growth in accordance with national circumstances and, in particular, at least 7 per cent gross domestic product growth per annum in the least developed countries', which is not relevant to project level measurement and no metrics identify contribution to the improvements.

8.4.5 Limitations of the research

For this stage of the doctoral research, it is recognised that the analysis can only be considered as early exploratory research without definitive conclusions. However, it is used as a way of supporting the evidence gathering to respond to Research Objectives 6 and 7 as well as answering Propositions 3 and 4 (listed in Section 8.3). The specific limitations of the approach were as follows: the matrix mapping was only completed by the lead researcher and verified at a high level by a BRE Director and two senior academics (who supported this stage of the research study) and should be more widely tested to strengthen the findings; and, the text analysis technique provides only limited indications. Consequently, these are not conclusive findings because the terminology is nuanced and specific to the contextual purpose of the methodology in relation to its organisational level; and finally; the SDG targets analysed are specifically designed for national level measurement and as such, are not easily cascaded to project or organisational level, thus reducing the strength of linkage between them.

8.4.6 Contribution to answering the Research Question, further research and potential applications

The aforementioned limitations suggest that this first Test in Stage Two of the overall MISI study has not provided definitive findings. However, it has helped provided evidence to respond to Research Objectives 6 and 7 as well as answering Propositions 3 and 4 (listed in Section 8.3). It has also established priorities for future research design. In this context, further research and potential applications include:
- Continue research into improving the understanding of the linkage between project level success, organisational level success and the global SDG goals.
- Develop understanding further on how to embed SDG impact targets at the design stage of an infrastructure project, thereby providing a more robust investment appraisal at the project design phase. This will help define project success more widely across the 'Triple Bottom Line' (TBL) of economic, social and environmental outcomes as well as associated impact.
- Build an agreed, common, accessible and adaptable database of indicators and a corresponding criteria framework that can be used to select measurements at the project level that are aligned with specific SDG targets and indicators.
- Conduct a case study investigation to build more detailed qualitative and quantitative data, which the findings of the exploratory research can be tested against. This feeds initially into Test 2, the testing of the IVC with Anglian Water in Chapter 9, but also looks further ahead to the future research conducted with the Environment Agency and Tideway megaproject.

8.5 CONCLUSIONS

This chapter summarises research into the existence of a 'golden thread' between sustainability reporting at the tactical delivery-level of projects and the strategic-level outcomes and impacts of the UN's 17 Sustainable Development Goals. The research selected the CEEQUAL reporting methodology at project level and the GRI methodology at organisational level since both approaches had the best attributes of accessibility, wide usage, currency and credibility. The results from this 'Test 1 of a Golden Thread' research study indicate that the golden thread can be evidenced across the TBL themes of economic, social and environmental thematic areas, at both project and organisational levels. It also showed that there was confidence that tactical-level sustainability tools on projects can be widened to include SDG linkages. This has particular value to stakeholders when assessing both the project delivery phase of related TBL success definition, as well as the second phase (i.e. post-project), of the wider project outcomes and SDG impacts. Given the findings from the research the Proposition 3, Current sustainability measurement at organisational and project levels can be used to demonstrate a 'golden thread' from 'global' SDG to 'local' portfolio and project levels, was supported, albeit with the stated limitations and according to the defined spectrum of high-to-low connections across the CEEQUAL to GRI linkages and the CEEQUAL to SDG linkages.

The next chapter uses a case study of a UK water utility company, Anglian Water, to demonstrate how the IVC process model can integrate the "triple bottom line" (Elkington, 1994, 2013, 2018, and Griggs et al., 2013) to ensure balanced definition of success across economic, environmental and social thematics. The emphasis is switched from "doing projects right" to "doing the right projects"; both are important, but the latter is critical. This is an explicit part of the IVC model, ensuring that short-term project success measures are balanced with post-project longer term outcomes and SDG strategic impact, which many (Morris, 2013, and Cooke-Davies, 2002 and 2007) have suggested are improved definitions of project success.

Confidence in	P1:	P2:	P3:	P4:
Evidence for	Engineer's view of	Organisational	'Golden Thread'	Prototype
Propositions	current MISI	view of MISI	global - to - local	Development
Theoretical	Not yet	Not yet	Very Low	Very Low
Analysis				
Survey Analysis	Moderate	Very Low	Very Low	Very Low
Interviews	High	Moderate	Very Low	Very Low
Analysis	_			
Test 1: Golden	High	Moderate	Moderate	Moderate
Thread				
Test 2: Case	Not yet	Not yet	Not yet	Not yet
Study	-	-	-	

Table 30: Development (4th) of stability of Propositions (changes since previous Chap in red)

(Note 1: The assessment of the quality of propositions' evidence is based on a qualitative formula designed by the Clinical Grading Working Group's British Medical Journal's study (GRADE, 2004). Full description and justification for this approach is shown in Table 8.)

Chapter 9: Test 2: Case Study of IVC in Water Utility context

9.1 OUTLINE OF CHAPTER

This chapter outlines Test 2, the case study using the Impact Value Chain (IVC) in a water utility organisation. In Section 9.2 it introduces the context for the case by describing the identification of the water utility company, Anglian Water. It follows this in Sections 9.3 and 9.4 with the methodology and research design using the C-M-O mechanism, which includes the use of desk-level analysis of publicly available information and interviews with three leaders from the company to assess the potential practical use of the IVC. Following this, Section 9.5 outlines the findings stemming from the analysis to answer Proposition 4: *A MISI prototype can provide a plausible, testable and achievable logic chain for defining project and portfolio SDG impacts*.

9.1.1 Context for the Case Study – Using critical success factors (CSF)

This final part of the empirical testing builds on the previous development in Chapter 7 of an innovative infrastructure model, called the "Infrastructure SDG Impact-Value Chain" (IVC) to link local-level project delivery with global-level SDG impacts. It uses a case study of a water utility company to demonstrate how the IVC business model can integrate the 'triple bottom line' to ensure the balanced definition of success across economic, environmental and social thematic areas. The case study includes the selection of longer-term outcomes and strategic SDG impacts, which, it is suggested, are improved definitions of project success.

9.2 METHODOLOGY FOR CASE STUDY APPROACH

9.2.1 Using the Realist Evaluation Methodology to Structure the Research

As discussed in Chapters 3 and 4, the research study adopted the critical realism perspective of ideological philosophers, such as Bhaskar (1978), to inform the choice of the realist evaluation's context-mechanism-outcome (C-M-O) configuration (Pawson and Tilley, 2004) because it provides a strong framework for analysing engineers' perceptions of the context of SDG measurement as well as the potential outcome on redefining investment decisions to achieve broader SDG impacts.

The aim of this stage of the research was to demonstrate that, by using the IVC, the new model approach could be used to link an organisation's strategic priorities with project level delivery using the SDGs. The validation exercise of the research was to review a leading organisation's sustainability model, that already used the SDGs at goal (10 of the 17 were used) and target (35 of the 169 were used) levels, to assess whether the IVC would add clarity and structure to cascade the strategic goals down, through a portfolio office, to the project level. The assessment also considered the wider stakeholders' views and what the application of the SDG framework had on their perception of value and practical usability. The choice of the organisation, Anglian Water, was based on it being one of the UK's largest water utility companies. It is amongst the UK's leading sustainability and sustainable development reporting pioneers (with early use of SDG targets) and was the winner of Business in the Community's (BITC) Responsible Business of the Year Award in 2017. This recognised Anglian Water's ambitions, laid out in its "Love Every Drop" (of water) vision, which aimed to create a resilient environment that allowed sustainable growth and the ability to cope with the pressures of climate change.

The data for the case study was accessed by interviewing (1.5 h) a senior boardlevel member of the Anglian Water executive who, at the time, was the Director for Asset Management (DirAM). A second interview was held with the Head of Anglian Water's Sustainability Management, as a further source of data and information. The latter worked alongside senior board-level members which added to his credibility as a data reference point to understand the emerging IVC. The DirAM was also the chair of the UK government's Green Construction Board's (2015) Infrastructure Working Group and has been a major sponsor and champion of the sustainable development programme across Anglian Water, as well as the infrastructure sector more generally, for the past 10 years. The DirAM provided publicly available documents (i.e., as a form of secondary research) to support the insights into the company's pioneering work in sustainable development. A third interview was conducted with the head of the organisation's supplier interface Joint Venture entity (the '@One Alliance'). This research was triangulated by evaluation of other forms of secondary data on the company, including information from the company's website as well as related corporate documents and technical reports on the company's approach to sustainable development in order to verify the data's validity. Formal agreement for the review

and the publication of the findings was agreed by the company in writing by DirAM and Anglian Water's Director of Brand and Communications.

9.3 RESULTS AND FINDINGS

9.3.1 Case Study Investigation: Anglian Water - Organisational Focus on Sustainable Development

The Anglian Water approach to sustainability and the SDGs is explained in their Annual Integrated Report (Anglian Water, 2018a). The report includes a description of their impact-value objectives (performance against outcomes) assessment, which correlates with the triple bottom line of the economic, social and environmental thematics. In summary, Anglian Water (AW) describe their TBL priorities as follows (Table 31).

Table 51. Anglian water's performance against outcomes	Table 31:	Anglian	Water's	performance	against	outcomes.
--	-----------	---------	---------	-------------	---------	-----------

Anglian Water Outcomes	Objectives
1. Smart business. Innovating by exploring new ways to operate more sustainably and helping customers, business partners and employees to embrace our Love Every Drop strategy.	 i. Resilient business. ii. Investing for tomorrow. iii. Fair charges, fair returns. iv. Our people: healthier, happier, safer.
2. Smart communities. Collaborating and engaging with customers, colleagues and business partners, and inspiring them to take positive steps towards achieving our vision for a sustainable future.	i. Positive impact on communities.ii. Safe, clean water.iii. Delighted customers.
3. Smart environment. Transforming behaviours by playing a leading role in reshaping how society values and uses water and reducing our combined impact on the world around us.	i. A smaller footprint.ii. Flourishing environment.iii. Supply meets demand.

These are shown below in the images from the *Annual Report* (Anglian Water, 2018a, pages, 24-25, 29) (Figure 72).



Figure 72: Anglian Water alignment of purpose-outcomes and SDGs (Anglian Water, 2018a). [Permission to re-use all graphics agreed by AW].

The following analysis of the case study is structured along the C-M-O 'Variables Framework'. The data is shown in the form of key quotes from the interviewees, supported by data gathered from open-source documents.

9.3.2 Context (#1): Strong Leadership. What is the Role of Leadership to Champion the SDG Impacts across the TBL?

Consistent with the survey results (Mansell, 2018), Anglian Water place a high priority on leadership to galvanise commitment to their corporate-level sustainability objectives. They achieve this through consistent and strong communications, both graphically, such as through their "Purpose Wheel" (Figure 73), and by the high-profile championing of their sustainable development approach by their board and executive.



Figure 73: Anglian Water's Purpose Wheel (Anglian Water, 2018a), aligned to the triple bottom line.

DirAM, a Director and Executive Board member at Anglian Water, observed (note: in future, all quotes from the interview are labelled as "DirAM" followed by the quotation) (a repeated quote from earlier): "*Leadership is the most important critical success factor, both internally and externally, to align and galvanise our employees, our communities and the supply chain. It was about getting us all to be more collaborative in finding novel, innovative ways of delivering sustainable solutions... It is about the leaders capturing the hearts and minds of the stakeholders to champion changed behaviours to achieve big, bold strategic outcomes.*" In his view, it played an important part in Anglian Water becoming a sustainable development leader across the sector. DirAM: "there are a number of reasons why we won Business in the Community's (BITC) Responsible Business of the Year Award in 2017—but a key part was that our CEO brought a very specific challenge back to the business having been inspired by a 'Seeing is Believing' visit, organised by BITC, to an area near the Olympic Park in London. The visit looked at how businesses were able to create opportunities and skills for those living in areas of high deprivation and low social mobility. The CEO's response was: 'how can we do something on a similar scale, in the region we serve, to make a real difference?'. This led to our hugely successful programme in Wisbech and helped us develop an approach that we have subsequently used on project work in Nepal alongside Water Aid."

(Note: The Wisbech project, was a forerunner of the Lahan project in Nepal. Lahan was the first WaterAid project with significant engagement from the utilities' supply chain and became a beacon to demonstrate how such projects can be driven across Nepal and beyond.) The quote also reinforces Porter's theory of creating shared value (1985, 2011) because, in this example, there are tangible benefits for the business to be seen to be actively "putting back" into society.

He also notes the moral values that are implicit in the choice of making sustainable development a core business priority for Anglian Water. DirAM: "a vital part of leadership is doing the right thing, just because it is the right thing to do, not because of a box-ticking exercise". DirAM expands this to state the following: "Our leadership was engaging the supply chain proactively to collaboratively change the way we thought about, and did, our business... We wanted the approach to become part of the way we jointly became leaders in delivering our businesses successfully... We wanted to establish meaningful change across the supply chain, and we recognised that, to do this, we had to develop long-term relationships; hence, we contracted on a five-, plus five-year basis. This built longevity into our thinking and allowed true innovation to develop solutions to the bigger sustainable development issues across the environment—driving efficiency and effectiveness."

This was not necessarily an approach that was either quick or easy and it needed a tough commitment from the leadership, DirAM: "*It is 50% belief and 50% belligerence when you start something like this; that is, holding yourself and others to account. That is what I mean by belligerence. In other words, 'seeing it through'.*" The core principles of governance (OECD, 2015) of accountability, responsibility and transparency were also noted; DirAM: "a key part of the leadership is the ownership of the sustainable development strategy. It is also about accountability and having the resources to deliver the solution. That is why the 'Infrastructure Clients' are the single most important stakeholders in addressing sustainable development. If they 'own' and champion the solution, then the supply chain will follow... hence, leadership and procurement are the biggest elements of the recent Green Construction Board's 'Three Years On Report—Reducing Carbon Reduces Cost' report" (Green Construction Board, 2015).

9.3.3 Outcomes (#2): Clarity of IVC Project Success Definition. Do Businesses Have a Clear Understanding of the Need to Separate Definitions of Success between "In-Project" Inputs/Activities/Outputs and "Post-Project" Outcomes and Impact?

In the Anglian Water Integrated Report 2018, (Anglian Water, 2018a, p.8), the CEO says: "We are continuing to plan and to invest in protecting customers and the environment. This year saw the publication of our draft Water Resources Management Plan, which sets out how we propose to balance supply and demand in a fast-growing region over the next 25 years and to protect customers from severe water restrictions in a future drought." The Annual Report highlights that Anglian Water explicitly assesses both the short-to-medium term economic factors that their investors value as well as the longer term strategic sustainable development impacts that are more aligned to SDG targets.

DirAM explains how Anglian Water used the overall "Love Every Drop" banner campaign to balance long-term and short-term priorities: "In 2015 we refreshed our 'Love Every Drop' goals and aligned them with the Outcomes Wheel shown in the Annual Report. So, we thought long and hard about not just the goals that we created but how that fit with a set of longer-term outcomes in our region and what that would look like in terms of implementation. This was our way of meaningfully connecting the strategy with outcomes that our stakeholders recognised."

It was also noted that Anglian Water uses simple and accessible language to explain their "Purpose Wheel" and its linkage to outcomes-impacts. This aligns with the IVC model and indicates a viable way of thinking "big and long" whilst managing the activities and outputs on a short-term basis to track progress.

9.3.4 Mechanism (#3): Prioritising SDG Goals Aligned to Strategic Vision. Do Businesses Have a Clearly Defined Strategy that Can Guide the Prioritisation of SDG Goals?

The Anglian Water approach aligns closely with the IVC model, since it also uses an "Ends, Ways, Means" logic similar to the Theory of Change concept (Weiss, 1995 and Stein and Valters, 2012). DirAM: "you must start with the end in mind, even if you have not got a detailed routemap to deliver at every stage of the journey. Part of the mantra is to set big audacious goals and then adopt an attitude of 'I have started so I will finish' and, by the way, you never actually finish, because the end goal is moving; it is like you achieve one peak but realise it is a false horizon, and so you continue your climb to the next summit."

As well as the ten prioritised goals, Anglian Water had also prioritised 35 targets that are most easily measured at project level, shown below (Figure 74).





The value of having clarity of the strategic ends is noted, albeit with a caution that the identification of targets for tracking performance must not become a "box-ticking" exercise that distorts clarity of outcomes; DirAM: "*if you actually begin with the end in mind of the outcome you are seeking and how you wire your DNA to achieve*

that, you are far more likely to achieve those outcomes, and in so doing the boxes get ticked. But if you predicate your thinking with thoughts about just filling the boxes, you have constrained yourself."

Therefore, to overcome the box-ticking mentality, DirAM explained their approach: "Anglian Water thought long and hard about its position in the region and how we contributed strategically as a major player in the region and we created the concept of "Love Every Drop" and, in essence, our own SDGs to align our strategy with local outcomes... We used the "Love Every Drop" goals to identify ambitious aspirations, which meant that our business had to think longer term."

9.3.5 Mechanism (#4): Prioritisation of (a limited) Number of SDG Targets Relevant to the Infrastructure Project

The chart shown in Figure 74 illustrates the 35 targets selected by Anglian Water, which at first sight is impressive, but the interview identified that it is challenging to move beyond the rhetoric of great sounding qualitative statements. Therefore, it is important to agree and publish hard quantitative targets that the success of the organisation can be assessed against; DirAM: "... so we nailed our colours to the mast and started reporting against those. One of them was to take 50% of the carbon out of the assets we build by 2015. It was the one that had a specific date on and a specific quantity, and I deliberately did that because I believed it and I was belligerent enough to drive it.... That is the one that, perhaps, out of all sustainability targets and goals, Anglian Water had the greatest recognition from and probably reflects the greatest change programme that has gone on across the whole of the supply chain."

9.3.6 Mechanism (#5): Aligned Business Priorities/Integrate the Targets across the TBL. How Are the Project Success Criteria Balanced across the Triple Bottom Line - What Trade-Offs Are Made?

A representation of the linkage of the Anglian Water three TBL thematic outcomes (Elkington, 1994, 2013, 2018), aligned to their ten prioritised SDG goals, is shown below .



Figure 75: Anglian Water has three business priorities that are balanced across the triple bottom line.

In the Anglian Water integrated report of 2018 (AW, 2018a, p.9), the CEO, Peter Simpson, says: "Since becoming Responsible Business of the Year, we have been working hard to show others how sustainability makes good business sense". This quote emphasizes the Anglian Water experience that aligns with the creating shared value (Porter, 1985 and 2011). It implies that the TBL (Griggs et al., 2013) can be balanced—a strategy that focuses on the environment and society, which can equally achieve economic success. When in harmony, real growth is delivered to the benefit of all, as shared by DirAM: "For example, our approach to 'product lifecycle management' was learned from the aeronautical and automotive industry from 2004–2005 and this meant that we looked at the whole life costs, which not only ensured we were more outcomes focused, but, by the way, improved our productivity by 3% each year, year on year, highlighting that good sustainable development also made good business sense".

9.3.7 Mechanism (#6): Reporting and Communication. What is the Best Way to Share Data on SDG Progress, Internally and Externally?

It has already been noted that Anglian Water had a policy of thinking long-term, explaining their sustainable development approach in accessible language and also the need to uphold strong governance principles of accountability and transparency (OECD, 2015). This has led to a strong ethic of being held accountable for delivering meaningful change, including publishing their strategic objectives in quantifiable

terms (such as the carbon figures noted in the paragraph above) as well as, equally importantly, the results; DirAM: "learning from the likes of Marks and Spencer's Plan A, we realised you had better publish your sustainability plans and outcome targets so that you are kept honest in the process—there is very little point nailing your colours to the mast and then not living to the high expectations... so the message was that we must commit to do the things that matter to us. That is what gets people excited, because it really matters. We are tough on ourselves on reporting what happens, and this allows us to measure what impact we are having so that we can measure the benefit."

The theme of honesty and allowing stakeholders to hold the executive and board to account is a powerful lesson that also relates to measuring SDG impacts at project level; DirAM: "But the point about turning your ambitious goals into reality, to avoid superficial statements, is that it is all recorded—it is published annually, which is an important part of defining where you are going. Driving towards it with no 'U' turns when some tough decisions have to be made. It is obvious that you have to make loads of tough decisions rather than duck them, and then recording your progress in an open and visible way helps keep you honest in that process."

A cautionary note about communication was that the messaging should be kept simple and accessible; DirAM: "We found that our campaign and collaborative working with partners had created a different conversation with different language. Ultimately, accessible language on meaningful outcomes is what people can buy into and this is what creates the momentum of changed behaviours... Through engagement and innovative solutions addressing the big problems, Wisbech is an example of working with the community to achieve meaningful long-term changes."

9.4 OVERVIEW ANALYSIS OF ANGLIAN WATER'S PROJECTS SET AGAINST THE IVC FRAMEWORK

The research of MISI at Anglian Water was only envisaged as an initial demonstration of the utility of the IVC approach, rather than any kind of full-blown test. It provided insights that informed the full test that was subsequently completed with the Environment Agency and Thames Tideway megaproject from December 2019 to April 2020. Therefore, the discussion below should be viewed as providing 'guidance' not 'confirmation' of the IVC's potential use by portfolios and projects.

The reference to Anglian Water's Wisbech project in the previous quote provides a holistic test against the six critical success factors and a useful way to cap the case study analysis. Launched in January 2013 as part of Anglian Water's "Wisbech 2020" vision, the Wisbech project was chosen as part of this case study because data on its delivery are open source on the internet. It was delivered by Anglian Water with its partners as part of their commitment to make a long-term impact on the market town of Wisbech for more than the five years that the initial project covered. Located just 40 miles from Cambridge, UK, Wisbech faced many socio-economic challenges but also had potential for significant growth and development. The vision proposed a new garden town with 10,000 homes, bringing transport, education and health benefits to the town and surrounding region. By using this project as an example, Anglian Water wanted to assess whether a broad programme of social, economic and environmental change to improve the local communities' lives could be linked to the SDGs using the IVC.

The table below mirrors the formatting of the IVC table (Table 32) and has been updated with data from the Wisbech project (Anglian Water, 2018b). The simple steps to achieve the Wisbech-adapted IVC included: reading and analysis of the publicly available documentation of the Wisbech project, identification of key data across the IVC framework, cross-checking across authors to assess the credibility of interpretation and sharing the final table with Anglian Water to ensure the consistency and accuracy of project data. This provides an assessment as to whether projects could have both the "in-project" successes measured as well as the "post-project" outcomes and SDG impacts as defined in the Theory of Change (Weiss, 1995, and Stein and Valters, 2012). It was evident that it is easier to define quantifiable success criteria for the inputs-activity-outputs during the in-project phase because they are tangible and delivered as core delivery performance measures, such as time, cost and scope/quality. On the other hand, the outcome and impacts are typically delivered after the completion of the project and are more diffuse. Thus, the example from the Wisbech project shown below is not conclusive but gives indications that the IVC provides a useful framework to engage stakeholders on what project success looks like during and post-delivery. It should be noted that the Wisbech project is an outreach community programme inspired by HRH The Prince of Wales' "Seeing is Believing" initiative, which seeks to find ways to support marginalised communities. The SDGs therefore offer a framework to address the more diffuse outcomes and impacts that might not typically have been defined and measured using traditional project measurement approaches.

	Input	Activity	Output	Outcome	Impact
Economy	Seconded a Senior Operational Manager to Wisbech in 2013; agreed support from other supply chain partners to become involved in the project; this allowed the cost, expertise and effort to be shared across a broad range of partners.	Worked jointly with the local Fenland District Council to develop a longer term strategy beyond their existing 2020 Vision, which was thought to be too short-term to encompass the 'big, hairy, audacious' strategic goals that could achieve transformational change; building a business case for the 'Garden Town' that would attract investment and large transport infrastructure improvements.	Championing apprenticeships and training scheme with 20 trained and employed year on year; turn the community centre from a £30k pa loss making entity to a vital community hub fuelling future economic success; a confirmed lease and implemented the creation of the 'Jobs Fair' and the 'Jobs Café'; The campaigning body for getting rail back – now in the County Transport Plan.	Bills, affordability and profits to stimulate and sustain the local economy, especially those on lower incomes (bills have only increased by 10% since 1990). Viability of future rail and integrated transport system attracting more regional investment and raising local people's aspirations; Market Town proposal, with planning for over 10,000 new homes, providing 'scale of growth' confidence.	SDGs 8, 9, 10, 12
Social	Started by listening – to understand the local issues from the local community's perspective; Brought together senior leaders from 'The @One Alliance'; creating a collaborative multi- stakeholder approach; focused on building long-term sustainable relationships with the local community.	Collaborative innovation with the local community in open and honest talks; health & wellbeing; stakeholder engagement; skills and learning; working conditions; production activity; user engagement; keeping the local community at the heart of the project plans and delivery; worked with the College of West Anglia to train more mechanical and electrical engineers; designed and ran new courses; providing IT support from partners to raise	Providing a community centre (refurbishment of the Queen Mary Centre) that is the hub of employment opportunities; active STEM subjects engagement with schools; specifically focus efforts on helping those not in employment, education or training; untapped unused human resource; organised the BITC 'Big Connect' event align business connectors from across UK; a second phase for the Queen Mary community Centre to include	Achieving 'Business in the Community' outcomes such as regeneration; Building on the 'Seeing is Believing' community initiatives; Understanding the value of long- term thinking; Providing safe, clean and reliable water; Improve the town/regions standing as the 6 th worst ranked town on social mobility index in UK; addressing the life expectancy that was 3 years less than in Cambridge.	SDGs 1,2,3,4,5,7,11

		aspirations of unemployed.	theatres and a music teaching centre.		
Environment	Raw materials; land take; water; light; clean air; energy; planned land use; ecology ecosystem valuation assessment.	Management plans for the flood risk, building resilience into engineering designs; using innovative modelling techniques developed by the Dutch government.	A commitment to protecting and restoring our wealth of wetland habitats. make a difference to rare and common species, be they in wet grasslands, open water, fens, or mires.	Build resilience to cope with future challenges. Protecting the environment, we live in; Through its Flourishing Environment Fund, helps environmental organisations deliver real benefits for nature	SDGs 6,13,14,15

Table 32: Applying Anglian Water's Wisbech project initiative to the IVC grid with mapping of the TBL with the five stages of the IVC.

9.4.1 Policy Implications Derived from Analysis of Anglian Water's Use of the IVC Framework

There are a number of policy implications that emanate from the analysis of the Anglian Water case study. These are listed at both the organisational and project levels and involve multiple stakeholders, including clients, investors, suppliers and communities, who all benefit from the use of the derived models proposed in this study.

9.4.2 Organisational policy implications:

There is evidence that businesses identify value in the adoption of global SDG performance measurement at the local level. This is consistent with the theory of creating shared value (Porter, 1985, 2011) that identified a greater benefit to businesses than CSR, that is often perceived as being an add-on and not core to its success. The complexity of the global-national measurement framework makes measurement at subnational level challenging. The need for simplicity is important and examples of success, such as this case study, are helpful in galvanizing others to follow and share lessons learned. This is important for users of the models because the case study makes clear that some organisations are employing the language of SDG measurement but without a formalised methodology to do so. This makes it difficult to replicate because the ad-hoc nature of the measuring methodology used by Anglian Water does not easily support cross-sector comparisons using a common framework that would have facilitated further knowledge sharing and delivery improvements.

The SDG measurement approach can align with existing approaches to sustainability measurement. This offers efficiency of processes and systems if they can be linked. The case study gives confidence that existing reporting approaches to sustainability, such as CEEQUAL, are complementary to the proposed SDG measuring methodology. This highlights that the IVC can be adapted, such as by using language that "makes sense" to the local stakeholders and does not alienate existing project delivery teams who would not want an additional large reporting system mandated. The opportunity to align existing sustainability reporting metrics to SDG targets offers a valuable line of future research, going beyond what was achieved in Chapter 8.

There is evidence that businesses that already have a strong track record in sustainability measurement can readily adapt to the language and approach of using SDGs. Anglian Water had recently been awarded the UK's Sustainability Company of the Year, which meant that the case study interviews, and review of their documentation were conducted with a highly mature organisation that had a welldeveloped plan for delivering sustainable impacts. They also had a strong leadership team to champion the trialling of the SDG measurement approach. The bigger question remains how successful the less mature (in terms of SDG measurement journey) performing companies might be at addressing the complexities of SDG measurement. Again, this is an area for further study since that is where the majority of benefit might come from, by developing an approach that is easily replicated across the sector.

The 'Variables Framework' issues, such as leadership, are important. Strong leadership that is meaningfully engaged in championing the use of SDG measurement will be more likely to deliver tangible evidence of SDG impacts. This becomes a critical point as the strategic nature of organisational change has to be driven from the top (Kotter, 2012). There was recognition by the Anglian Water executive that, in reality, this meant that leaders at all levels were needed as champions, which, for SDG measurement, needed to be aligned with success stories that would make sense to the target audience written in their language and justifying "why" followed by explaining clearly "how".

9.4.3 Project level policy implications:

The effective use of SDG measurement at project level needs buy-in from both internal and external stakeholders. The engagement of suppliers is critical to ensure common focus on identifying what SDG success looks like and to work collaboratively to seek innovative solutions to deliver meaningful SDG success.

There are a number of mechanistic issues that become critical to SDG measurement success. These include: prioritising relevant targets and indicators (do not select too many); seeking to understand how the few selected goals and targets can have a simple indicator framework that allows the capture of reliable evidence; and ensuring that reporting and communicating is open, honest and timely, sharing both good news and bad news. There is also a need to continually learn and evolve and so build a better framework that achieves a more balanced investment decision across the TBL of people, profit and planet (Elkington, 1994, 2018, and Griggs et al., 2013).

9.5 CONCLUSIONS FROM THE TEST 2 CASE STUDY

The central investigation in the case study of Anglian Water was to test and validate whether the new infrastructure business model, called the "Infrastructure SDG Impact-Value Chain" (IVC), could link local-level project and organisational delivery with global-level strategic SDG impacts. The study used the "golden thread" of the TBL thematic areas (namely economic, social and environmental) (investigated in Chapter 8) to interrogate whether one of the UK's leading water utility companies, Anglian Water, was already delivering strategic sustainable development solutions that could be mapped to SDG targets. Although the research was conducted in the UK, the findings have potential broader applicability to other countries since it was observed that the SDGs are by nature a framework to be adopted by all countries and hence there is an international commonality of purpose in this regard. This is a valuable area of future research that could potentially engage with a number of construction firms with global footprints to compare the differences and similarities of measuring SDGs across and within different regional areas. For example, UNOPS (2018) research indicates that there are many contextual global issues that affect the use and measurement of SDGs but, while noting the differences, they suggest that all issues should have a consistent framework to enable cross-cutting comparisons.

The results of the case study investigation have indicated that there is a verifiable link across the IVC of activities-inputs-outputs during the "in-project" phase, connecting to the "post-project" outcomes and SDG impacts. A number of Anglian Water's projects were mapped to this schematic (although, for brevity, only one, Wisbech, is reproduced in this thesis) and this gave confidence that the approach could have wider applicability. Therefore, the results led to a proposed methodology for project leaders to use as a way of strategically aligning stakeholders on a common definition of success, linking tactical "in-project" success of outputs with the more strategic outcomes and SDG impacts "post-project". The methodology would ideally be used during the design phase of the projects. The emphasis is switched from "doing projects right" to "doing the right projects". It includes the selection of longer-term outcomes and strategic SDG impacts, which, it is suggested, offer improved definitions of project success.

The Test 2 research case study has focused on a single case study in the UK and cannot, as stated earlier, be automatically extended to the entire water industry, either nationally or internationally. The methodology adopted, however, has potential to be used to evaluate multiple projects across different industry sectors. In this way, the results can thus provide insights for further research across the water industry and also potentially across other infrastructure sectors and geographical regions.

The final iteration of the maturity of evidence underpinning the response to the Propositions is shown below:

Confidence in	P1:	P2:	P3:	P4:
Evidence for	Engineer's view of	Organisational	'Golden Thread'	Prototype
Propositions	current MISI	view of MISI	global – to – local	Development
Theoretical	Not yet	Not yet	Very Low	Very Low
Analysis				
Survey Analysis	Moderate	Very Low	Very Low	Very Low
Interviews	High	Moderate	Very Low	Very Low
Analysis				
Test 1: Golden	High	Moderate	Moderate	Moderate
Thread				
Test 2: Case	High	Moderate-High	High	High
Study		-		

Table 33: Development (5th) of stability of Propositions' Results (changes from this Chap in red)

(Note 1: The assessment of the quality of propositions' evidence is based on a qualitative formula designed by the Clinical Grading Working Group's British Medical Journal's study (GRADE, 2004). Full description in shown in Table 8.)

In the next Chapter the thesis provides a holistic view across the full doctoral research study to describe the summary findings and implications for science and practitioners. It also describes the follow-on research approach, and emerging results, where the IVC has been extensively tested with collaborative partners including the Environment Agency, at portfolio and project levels, and with Thames Tideway at the megaproject level.

10.1 OUTLINE OF CHAPTER

This chapter concludes the thesis. A summary of the research study is provided in Section 10.2 with a description of the high-level conclusions. Following the discussion of limitations in Section 10.3, the chapter provides recommendations in Section 10.4, including the implications of this research on the adaption of the Theory of Change for the future adoption of the proposed MISI approach using the IVC model and SDGiPro methodology. In Sections 10.5 and 10.6 it addresses potential value to both the scientific and practitioner communities and shares the emergent results from the collaboration with the Environment Agency, Thames Tideway and the UK Government's Infrastructure and Projects Authority (with UCL, ICE, BRE and UN Global Compact – all of which have their letters of impact support included as Appendices 17-23). Final thoughts from the researcher using reflexivity, provide concluding remarks to close the thesis.

10.2 SUMMARY OF THE RESEARCH STUDY RESULTS

The research was based on two main stages. The first stage was led by a systematic literature review that informed the research objectives, shared in Chapter 1, underpinning the research question. This helped to understand the contextual situation in which MISI mechanisms are used, and through empirical research, assess whether the current outcomes could be improved by the development of a theory-led prototype model that is workable at portfolio and project levels. The intent was that the IVC mechanism would provide a practical way to link from the local projects, through the organisational portfolio construct, to global SDG goals. The method chosen comprised a Sequential Explanatory Design (Creswell, 2017) using mixed methods that involved a survey of 350 engineers (due to late completion of 25, only 325 were included in the analysis) to derive quantitative data (Mansell, et al., 2020b) along with interviews with 40 CEOs and corporate Heads of Sustainability to capture qualitative data (Mansell, et al., 2020c). The second stage involved the development

of a prototype that was examined¹² through further exploratory investigations at two levels: (1) Test 1: is there a Golden Thread from global SDGs, through the organisational-level, down to project level SDG impact measurement (Mansell, et al., 2019a); (2) Test 2: does the prototype model, the *Impact Value Chain*, have coherence in the analysis of a brief case study investigation that, through application of the main findings from the empirical stage, evaluated the scope to measure SDG performance for infrastructure projects at a Water Utility Company (Anglian Water) (Mansell, et al., 2020d).

The Logic Map shown below in Figure 76 illustrates the development of the Propositions and the C-M-O Variables Framework. The red boxes and red connecting lines illustrate how this measurement framework was built and how it flows. It highlights the connecting loop, that starts from the SLR, connects through a series of evidence gathering and development stages, and returns in the Concluding Chapter to answer the Research Question, Objectives and Propositions originally established in Chapters 1 and 2.



Figure 76. Logic Map for Thesis Findings.

¹² **Note**. The 'tests' are more aligned to a demonstration. Eg. The research of MISI at Anglian Water was only envisaged as an initial demonstration of the utility of the IVC approach, rather than any kind of full-blown test. It provided insights that informed the full test that was subsequently completed with the Environment Agency and Thames Tideway megaproject from December 2019 to April 2020. Therefore, 'tests' should be viewed as providing 'guidance' not 'confirmation' of the IVC's utility.

Each chapter, with its unique research focus, captured the main findings from its research at the end of each stage, and linked this sequentially to the next steps of the investigation. In this way, the evidence is cumulatively built. This final chapter provides summary findings from each chapter in a holistic review against the overall research question. This is structured against four areas, listed below:

- **Results Area #1: Research Objectives** results are captured in Section 10.2.1 and in Table 34;
- Results Area #2: C-M-O Variables Framework results are shown in Section 10.2.2;
- **Results Area #3: Propositions'** cumulative evidence supporting the assessment of its results are described in Section 10.2.3; and
- **Results Area #4: Recommendations** are discussed in Section 10.2.4.

10.2.1 Summary Results against the Research Objectives

Based on the **research question**: 'How can global SDG goals be used to define and measure infrastructure projects' SDG impact at organisational and local project levels?', the table below summarises where the research investigation sought to achieve the **research objectives** and whether, in the view of the researcher, they were met. The more detailed evidence to support the analysis of research findings, is captured in Sections 10.2.2 to 10.2.4. This sub-section, shown in Table 34 below, seeks to reflect on the original Research Objectives, where they were evaluated and whether they were achieved.

Research Objectives	Summary of where	Was the Research Objective
	investigation completed	met?
RO1: To understand the <u>existing knowledge</u> (in theory and practice) on how organisations and projects measure infrastructure projects' SDG Impact.	 Literature Review (Ch 2); Theory (Ch 3); Survey (Ch 5); Interviews (Ch 6) 	High confidence that the objective was met by ensuring wide networking with academic and practitioners to test validity of research findings eg through 12 peer-reviewed publications in international journals &
		conferences
RO2: To understand the <u>context</u> (the 'variables') of the current use of mechanisms to measure	 Literature Review (Ch 2); Theory (Ch 3); Survey (Ch 5); Interviews (Ch 6) 	High confidence that the objective was met due to extensive publications of research findings

Table 34: Research (Objectives Summa	ry.
----------------------	------------------	-----

infrastructure projects' SDG Impact. RO3: To assess the <u>current</u> <u>mechanisms</u> used for the Measurement of Infrastructure projects' SDG Impact (MISI) at portfolio and project levels.	 Theory (Ch 3); Survey (Ch 5); Interviews (Ch 6) 	as well as coherence in model development and trials. High confidence that the objective was met from the extensive feedback through surveys and interviews.
RO4: To understand the <u>perception of individual</u> <u>engineers and organisations'</u> relative perception of the <u>outcomes</u> of the current use of mechanisms to measure infrastructure projects' SDG Impact	Survey (Ch 5);Interviews (Ch 6)	As above
RO5: To use the theory-led study to inform the <u>development of a prototype</u> model to improve the measurement of infrastructure projects' SDG Impact	 Development of the Prototype (Ch 7); Conclusions and Recommendations (Ch 10) 	High confidence that the objective was met because the model have been tested and feedback from practitioners provided that it can demonstratively support organisations MISI efforts.
RO6: To <u>test whether a</u> <u>'golden thread'</u> of SDG measurement could be identified from global to local levels.	• Case Study (Ch 8) Test 1: GRI & CEEQUAL	High confidence that the objective was met through the findings from the Chapter 8 demonstration.
RO7: To <u>test whether the</u> <u>prototype could be validated</u> with a case study organisation.	Case Study (Ch 9) Test 2: Water utility company	High confidence that the objective was met through the findings from the Chapter 9 demonstration.
RO8: To <u>build a framework</u> for further development, for researchers and practitioners to utilise, driving improved investment decisions across planet, profit and people outcome criteria, aligned to SDG impacts.	 Theory (Ch 3); Triangulation of Results and Development of a Prototype Conclusions and Recommendations (Ch 10) 	High confidence that the objective was met by having an evidence- based development of the theories of ToC and TBL to underpin the IVC. Evidence from the practitioners that are using the research outputs also confirms success as shown in Appendices 17-23.

10.2.2 Summary Results against the C-M-O Variables Framework

The MISI study sought to answer whether the existing UN SDG goals are adequate for defining success at project level in the infrastructure sector. The conceptual development was based on literature research across the Triple Bottom Line and the Theory of Change. This enabled the development of a theoretical model, the Impact Value Chain (IVC), that provided a mechanism that could be tested across three levels, including the business-portfolio, the mega-project and the project levels, which has recently been completed at the Environment Agency and Tideway Tunnel's mega-project. Although the results are not included in detail in this thesis, the organisations have agreed for their letters of impact support from this research project to be included in Appendices 17-23.

The resultant view against the C-M-O Variables Framework after the Test 1 and 2 (note caveat in Footnote 9 above), is shown below, based on critical analysis of the Tests' results. As discussed in Chapter 8, the 'Harvey Balls' are commonly used in comparison tables to illustrate the degree to which a particular item meets a particular criterion, and users of them, such as Groenland (2016), suggest that they discourage readers of a report from '*attributing quantitative meanings to qualitative data*'. In this case, the Harvey Balls indicate that there is reasonable strength of findings for seven of the nine contextual issues, two of the three mechanism issues and for both of the two outcome issues. In effect, C6, C7 and M3 would need further research to strengthen the findings.



Figure 77. View 3 – Maturity of C-M-O evidence across variables post-Tests 1 & 2.

A summary of the results from the investigation into the C-M-O Variables Framework, shown below in Table 35, indicates that the MISI study has added new knowledge to address a gap in the current understanding. The confidence rating for the derived stability of the findings is shown in column 'b', that is based on the evidence listed in column 'c'. This leads to the conclusion that the framework is robust and provides a solid basis for further MISI development, in that the C-M-O has allowed the logical experimentation of causal chains, both horizontally as well as vertically, across portfolio-megaproject-project boundaries. The analysis of the Variables Framework has also indicated that some of the variables will require further investigation. Suggestions for the nature of the future work is captured in column 'd'.

Table 35: Summary Results of C-M-O Variables Framework.

derived Context 1: High Leadership backing.	 <u>Context (conditions-variant</u> <u>Issues & Variables' & C</u> Leadership was identified contextual issue in the m SDG impacts at organisat levels. Because of the leaders reat transformation which limits and the second seco	ables) DPM (Muller et al, 2019) as a dominant neasurement of ational and projectThe development of a new theoretical lens in Chap 7, the IVC, should
derived Context 1: High Leadership backing.	 m SLR 'Issues & Variables' & C Leadership was identifie contextual issue in the m SDG impacts at organisa levels. Because of the leaders ra transformation which lip 	DPM (Muller et al, 2019)d as a dominant neasurement of ational and projectThe development of a new theoretical lens in Chap 7, the IVC, should
Context 1: High Leadership backing.	 Leadership was identified contextual issue in the m SDG impacts at organisat levels. Because of the leaders real transformation, which lip 	td as a dominant neasurement of ational and projectThe development of a new theoretical lens in Chap 7, the IVC, should
Content 2	 adopting the SDG lens to and project success, the strong direction with vis strongly referenced as or factors in the survey and The SDGs have similar to sustainability, with an adaptive systems that pla demand on leaders. Leaders need the skillset the complexities of achie business success of profi increasing demands on or environmental and socia The visionary leaders with harness the CSV mindse empower and align their the people, profit and pla TBL, thinking and acting systems' models that see solutions to the SDG characteristics 	es at the core of o measure business importance of ible leaders was itical success the interviews. complexity patterns lequal need for ace an extraordinary ts that can balance eving the economic itability with the to-balancing with l objectives. Ill be the ones that t and then can organisations with anet thematics of g within 'systems of ek innovative allenges. (be supported by tools and processes that enable the practical application of the proposed IVC model. Potential for a re-run of the original survey conducted in June 2018, to test emerging positions. The Environment Agency and Thames Tideway megaproject are currently undergoing their own trials, championed by their leaders.
Context 2: Mod Strategic High Alignment.	 Recognition of the need and impact performance ensure strategic plans ar effectively and efficientl corporate and stakehold were agreed as a good w with a common lexicon. CEO's mentioned that a ensure strategic alignme 	for using benefits measurement to e deliveredNeeds 'live' testing with organisations at the portfolio, megaproject and project levels to ensure practical, plausible, achievable use

		•	The results showed that interview participants have placed a high priority in strategic alignment of project goals with SDGs and also have the resolve to employ SDG measurement at business and project levels (Finding #2) in order to achieve outcomes that benefit people, the planet and profit. The survey results (Chap 5, Finding #M2) highlighted the need to enhance business skills of engineers to better define benefits and impacts that can more effectively align with organisational strategic goals.	
Context 3: Knowledge of SDG & success definition	High	•	The survey showed that engineers agreed that too often, projects define success by traditional outputs using the so-called 'iron triangle' of time/cost/scope (and quality) dimensions to deliver on the SDG goals. Instead, the majority of survey respondents agreed that the engineering and project management communities need to place a greater emphasis on the achievement of long-term outcomes aligned to SDGs and adopt a correspondingly broader definition of success. The survey results also indicated the need to build capability and capacity amongst engineers. The interviews reinforced the point above, by noting the importance of defining success through outcomes and not just through the traditional outputs of time, cost and scope (and quality). The broadening of skillsets to include business skills of innovation and definition of broader TBL success will play a more dominant part of the education and learning syllabus in future. In this regard, it was suggested by a number of respondents that embedding business skills learning within core engineering educational programmes would help provide opportunities for meaningful improvements in the measurement of SDG performance on projects.	The development of research into what changes to current education curriculum could be progressed and ideally with standards bodies such at the APM and ICE.
Context C4: Complexity of organisational levels: (1) Business- Portfolio; (2) Mega-project; (3) project	High	•	The OPM model (Muller et al., 2019) provides useful insights to the organisational context and project context. The IVC prototype therefore needs to be flexible for the needs from portfolio, megaproject and project levels. Although the survey questions did not specifically address the separate levels of organisations and projects, a number of the interviewees mentioned the need for a	There is an opportunity to continue the research with the Environment Agency and Thames Tideway to ensure a portfolio, megaproject and project

		 coherent cascade of goals from portfolio to project level and a knowledge innovation return loop that shares measurement of benefits realisation (aligned to SDG). That existing sustainability measurement at organisational and project levels is well established, and therefore, SDG measurement should be aligned to existing successful approaches, not created as an 'add-on' (i.e., the organisational level will likely have different SDG imperatives and reporting requirements, such as using the GRI, from the project level, which might have limited capability and capacity to track too many targets and indicators.) This becomes important in the development of a prototype since the measurement of SDGs at the enterprise level (the portfolio) will likely be different from that at programme and project levels. This requires further research. 	perspectives are incorporated into the future development of the IVC.
C5: Tools & process maturity	Moderate- High	 The engineers surveyed (Chap 5, Finding #C3) highlighted the need for simple tools to measure project success using the SDGs, that should guide the design of the prototype. It was frequently noted that there was significant room for improvement on availability of 'fit for purpose' engineering tools and methodologies to measure SDGs. These results highlight the need for a new simple tool (such as a set of KPIs linked to the SDG indicators). There is an opportunity to allow the engineering community to align projects' SDG reporting with the growing trend of using global standards to report sustainability. The literature review also noted that there is a growing body of evidence (Allen et al., 2019; Bali Swain and Yang-Wallentin, 2020; Jones and Comfort, 2020) that suggests that the complexity of the 17 Goals and the 169 targets needs to be simplified and a reduced selection prioritised for measurement. 	There is a need to maintain the pace of MISI development, with a strong focus on practical implementation. This aligns with the UK government's ambition to achieve Net Zero by 2050, which is one area that be tracked with meaningful data. There is potential to align with COP26 in Glasgow in Nov 2021 if MISI gains further traction with the Cabinet Office and HMT (BBC, 2021).
C6: Type of Organisation - Public or private sector	Low	• There was little coverage of the difference between public and private sectors. A few of the CEOs did refer to the challenges of MISI, where the client was not demanding SDG measurement as part of the procurement process.	There is an opportunity to continue the research with the Environment Agency and Thames Tideway to compare the

C7: Lifecycle	Low	• The OPM model also notes the project	differences and similarities of employing a MISI approach in the private and public sectors. There is a future
phase		 Intervention of the project of the project intervention of the project intervention of the project. For example, the early definition stage will not have full details of the activities, but the prototype should align to the Green Book Five Case Model (5CM) so that it can be used across the public sector at the investment appraisal stage. Both the Environment Agency and Thames Tideway MISI trials have tested the IVC on megaprojects that are c50% through the delivery phase as well as some that have been completed. It is anticipated by the Environment Agency that they will be ready to use the IVC by April 2021 on the full portfolio of projects, including new initiatives. 	 proposition for new research that might seek to verify whether there is greater utility of MISI at the investment stage of a project. MISI to continue to support the HMT/CO team to update the Green Book 5CM by making it 'greener'.
C8: Business Case & Benefits analysis	High	 The survey showed that many engineers agreed that the choice of SDG goals and targets should be primarily selected on the basis of business profitability. This is counter to CSV and TBL. As a result, the longer-term value of making investment decisions based on broader TBL principles could be weakened. It is therefore proposed that the next research stage, to inform the development of the prototype, investigates how the TBL could be integrated with the measurement of SDGs. As noted above (C7), this should be done in partnership with the HMT and Cabinet Office's Green Book update team. Whilst recognising the imperative of ensuring short-term commercial success when delivering project engagements for clients, all CEOs recognised the importance of defining outcomes and benefits from the start, which assumes an explicit need to plan for longer-term impacts. The IVC model should align with the existing benefits realisation processes. 	The development of a new theoretical lens in Chap 7, the IVC, should be supported by tools and processes that enable the practical application of the proposed IVC model. Potential for a re-run of the original survey conducted in June 2018, to test emerging positions.
C9: Organisational transformation tempo	Moderate- High	• A number of the interviewees noted the importance of building a coalition of the 'willing' and that the alignment would be enhanced by using the IVC's ToC and TBL to demonstrate progress on the achievement	Further research into the challenges of building a new measurement approach into

		of outcomes. In this way, evidence of success confirms the causal linkages and indicates that the initiative is effective in achieving its outcomes and SDG impacts.	existing processes. The Environment Agency have agreed to lead.
		(underlying) Mechanism	
(derived fr	om Theory	Chap 2 and tested in Stage 1 and using prototype	e in Stage 2)
M1: Using the IVC & the SDG goals and targets to understand the 'ends' of the project – eg definition of success.	High	 There was almost unanimous conviction that the "ends" of achieving the desired "outcomes" was good for business (Chapter 6, Finding #4) and this should be communicated when seeking to get organisations to engage with the IVC prototype. The survey noted (Chap 5, Finding #C2) that 'definition of success' and 'excessive focus on outputs instead of outcomes' (Chap 5, Finding #O2) were two of the five top challenges facing practical application of MISI. The model should harness the core concepts of the Theory of Change and the Logic Model, with their focus on outcomes measurement, including the analysis of causal linkages, engagement of stakeholders and strategic design with the 'ends' being the starting point for a right to left causal mapping. Measurement of SDG performance should accommodate the perspective of Creating Shared Value (CSV) (i.e., seeking solutions that are good for business in the short and longer term through balance of profit–planet–people objectives). This was supported by participants being optimistic that their organisation would achieve the broader outcomes by making SDG measurement more usable, consistent and verifiable across the construction sector, with increasing balance to their investment decisions across environment, economic and societal factors (Chapter 6, Finding #1). 	Needs 'live' testing with organisations at the portfolio, megaproject and project levels. Increased emphasis on Climate Change and Net Zero by global players. Eg. in January 2021, 110 states had legally binding commitments, including China and led by UK in 2017 – (BBC, 2021)
M2: Adoption	High	• The survey highlighted a clear preference	This indicates an
of the right 'ways' (MREL processes, systems etc)	-	 for measuring just five SDG goals (there was a 50% reduction in the preferences for the next two SDG goals). This indicated a long tail of ten further goals that did not appear to resonate as much with participants. The findings from the survey on the top five SDGs and the complexity noted in the literature review is consistent with the 	important area for further research to assess how this simplification can be achieved at organisational and project levels.

		advice given by the UN Global Compact in their proposed methodology (GRI, 2015).	
M3: Application of adequate 'means' (resources, time, scope) to achieve the ends.	Moderate- Low	 Measurement of SDG performance should be viewed from a systemic perspective and thereby move beyond the traditional 'iron triangle' view of projects in the short term (i.e., according to schedule, budget, scope and quality performance) and additionally, take account of longer-term project outcomes and impacts. The megaproject leaders that were interviewed were consistent in their view that there was a need to develop a MISI approach that focuses on the identification of interrelationships between components (i.e. sub-systems) of a system. The immediate design and testing of the prototype should include the steps to better understand the sub-systems from the start, such as breaking down the carbon issues across stakeholders and especially across the supply chain. The fourth concept aligns the TBL with SDGs and strengthens the composite IVC value chain. The IVC prototype should therefore provide a means 'to determine stakeholder's preferences and the trade-offs they choose to make given their scarce resources, or the value the marketplaces on an item' (Porter, 1985). 	The development of a new theoretical lens in Chap 7, the IVC, should be supported by tools and processes that enable the practical application of the proposed IVC model. Potential for a re-run of the original survey conducted in June 2018, to test emerging positions.
	<u> </u>	(Expected) Outcome	
(derive	ed from Theo	ory Chap 2 and tested using prototype in Stage 2	, Chaps 8 & 9)
O1: A 'golden	Moderate-	• Evidenced primarily from the Test 1 in	There is a need
thread' has been reinforced from 'global to local level' that can link project impacts to SDGs.	High	 Chapter 7. The conclusion was that there is strong evidence of a golden thread from SDG to GRI, and through the organisational level, down to projects. Initially, the study identified a lack of evidence given by interview participants on their ability to achieve the golden thread of SDG measurement from project to portfolio level (Chap 6, Finding #10) because, often, it was not available at any credible depth or backed up by verifiable evidence. This was examined further in Test 1 (Chap 7) to test whether aspirations to achieve this linkage are realistic. A further prototype design criterion that emerged from the interviews, to enable the mechanism for measuring SDG impacts to achieve the outcomes, was the ability to find a golden thread from enterprise 	to test and track the actual measurement of SDG targets. The Environment Agency are currently undergoing trials to assess the validity of this approach.

		portfolio level to project level (Chap 6, Finding #10).	
O2: The IVC logic chain is usable by organisations, potentially at portfolio, megaproject and project levels (it is plausible, testable and achievable)	High	 Evidenced primarily from Test 2 in Chapter 8. The Anglian Water case study indicated the IVC's value and relevance to existing organisations that wanted to harness SDGs to be more aligned with global challenges and methods to respond to them. Measurement of SDG performance should accommodate the perspective of the Triple Bottom Line (i.e., social, environmental and economic performance). This will drive a broader definition of project sustainability that includes the three pillars (i.e., social, environmental and economic performance). It provides simplicity and structure for the analysis in regard to selecting and measuring SDGs. Measurement of SDG performance should accommodate the required different organisational levels, namely portfolio, programme and project levels. 	As above

The evidence collected from the investigation of the variables has informed the response to the Propositions, that are answered below.

10.2.3 Summary Results against the Propositions Framework

The final assessment of the maturity of evidence against the Propositions is shown in Table 36, below. This indicates that, as expected in a qualitative-based investigation, the empirical research has not given statistical certainty. However, it has, through the tracking of evidence maturity through the investigation stages, provided sufficient evidence to give confidence to the findings discussed at the end of each Chapter and in summary, below in Table 36. This summation is in-line with most qualitative led mixed research methods that can only give a view of the world, since as the Critical Realist philosophers (eg Bhaskar, 1999) might suggest, there is only a limited view of the real and actual worlds that empirical investigations uncover. Thus, the MISI study has supported the evaluation of the transitive domain through observation of events and people's interpretation of them. This leads to the summation of propositions' maturity of evidence shown below:

Confidence in	P1:	P2:	P3:	P4:
Evidence for	Engineer's view of	Organisational	'Golden Thread'	Prototype
Propositions	current MISI	view of MISI	global - to - local	Development
Theoretical	Not yet	Not yet	Very Low	Very Low
Analysis		-	-	-
Survey Analysis	Moderate	Very Low	Very Low	Very Low
Interviews	High	Moderate	Very Low	Very Low
Analysis	_		-	-
Test 1: Golden	High	Moderate	Moderate	Moderate
Thread	_			
Test 2: Case	High	Moderate-High	High	High
Study				

Table 36: Summary table of the stability of Propositions

The grading assessment of the quality of propositions' evidence used the qualitative formula designed by the Clinical Grading Working Group's British Medical Journal's study (GRADE, 2004), based on a combination of quality and consistency. Using their definitions, (shown in full below Table 8), the evidence from the study has provided high confidence for Propositions 1, 3 and 4. This means that further research is very unlikely to change the confidence in the estimate of effect. Proposition 2 was slightly less strongly evidenced, and it is suggested that further research is likely to have an important impact on the confidence in the estimate of effect and may change the estimate. In addition to the maturation of evidence to respond to the propositions, shown above in Table 36, the summary of the thesis results is shown below:

10.2.3.1 Proposition 1: Individual engineers are 'supportive but frustrated' in methods to measure global SDG goals at project level.

- The survey provided empirical evidence that engineers (over 90% of the 325 surveyed) were very strongly supportive of measuring SDGs on projects. However, they noted a gap between what is wanted-needed and what is available, i.e. there is insufficient leadership; tools; processes; knowledge; and pace of change to support SDG adoption.
- There were extensive reasons offered to explain the gap, and these were captured in the C-M-O Variables Framework discussed in Table 35 above.
- It was frequently noted that there was significant room for improvement on availability of 'fit for purpose' engineering tools and methodologies to measure SDGs. The engineers surveyed (Chap 5, Finding #C3) highlighted the need for

simple tools to measure project success using the SDGs, that played a major part in the design of the prototype. This was realised by the development of a new theoretical lens in Chap 7, the IVC and the SDGiPro, that should be supported by tools and processes that enable the practical application of the proposed IVC model.

- The literature review also noted that there is a growing body of evidence (Allen et al., 2019; Bali Swain and Yang-Wallentin, 2020; Jones and Comfort, 2020) that suggests that the complexity of the 17 Goals and the 169 targets needs to be simplified and a reduced selection prioritised for measurement.
- The first proposition is therefore supported by the research evidence.

10.2.3.2 Proposition 2: Organisations' leaders champion SDG measurement without having evidence-based methods to verify claimed impact at portfolio and project levels.

- The interviews provided evidence of a 'gap' between organisations' MISI 'ambitions' and 'reality' by accepting that there is a mismatch between what companies 'say' they are doing and what they are 'actually' doing (this is sometimes characterised as 'greenwash'). There was evidence of a growing commitment to adopt the SDGs in the Anglian Water case study and the interviews supported this view by demonstrating that engineers are embracing the shift to SDG adoption, but are finding considerable contextual challenges to achieve MISI efficiently and effectively.
- The interviews confirmed that there is a wide variance in MISI knowledge and perceived value.
- There was almost unanimous conviction that the "ends" of achieving the desired "outcomes" was good for business (Chapter 6, Finding #4). Whilst the leaders recognised the challenges noted above, they wanted to move beyond 'greenwash'. They recognised that the new, and rapidly evolving, legal norms driven by the 'Net Zero in 2050' laws, that there was reputational and economic value in being seen to lead in their MISI commitments. This should be messaged when seeking to encourage organisations to engage with the IVC prototype.

- The survey noted (Chap 5, Finding #C2) that 'definition of success' and 'excessive focus on outputs instead of outcomes' (Chap 5, Finding #O2) were two of the five top challenges facing practical application of MISI.
- As a result of the findings from CEO's interviews, the prototype was developed further with greater focus on outcomes measurement, including the analysis of causal linkages, engagement of stakeholders and strategic design with the 'ends' being the starting point for a right to left causal mapping.
- It was noted that measurement of SDG performance should accommodate the perspective of Creating Shared Value (CSV) (i.e., seeking solutions that are good for business in the short and longer term through balance of profit–planet–people objectives). This was supported by participants being optimistic that their organisation would achieve the broader outcomes by making SDG measurement more usable, consistent and verifiable across the construction sector, with increasing balance to their investment decisions across environment, economic and societal factors (Chapter 6, Finding #1).
- The second proposition is therefore supported by the research evidence.

10.2.3.3 Proposition 3: Current sustainability measurement at organisational and project levels can be used to demonstrate a 'golden thread' from 'global' SDG to 'local' portfolio and project levels.

- Initially, the study identified a lack of evidence given by interview participants on their ability to achieve the golden thread of SDG measurement from project to portfolio level (Chap 6, Finding #10) because, often, it was not available at any credible depth or backed up by verifiable evidence.
- A further prototype design criterion that emerged from the interviews, to enable the mechanism for measuring SDG impacts to achieve the outcomes, was the ability to find a golden thread from enterprise portfolio level to project level (Chap 6, Finding #10).
- The evidence was subsequently derived from Test 1, as described in Chapter 7. The conclusion was that there is strong evidence of a golden thread from SDG global goals (by testing the GRI framework linkages to CEEQUAL and to SDGs) and through the organisational level, down to projects.

- There is a need to test and track the actual measurement of SDG targets. The Environment Agency are currently undergoing trials to assess the validity of this approach
- The third proposition is therefore supported by the research evidence.

10.2.3.4 Proposition 4: A MISI prototype can provide a plausible, testable and achievable logic chain for defining project and portfolio SDG impacts.

- This was evidenced primarily from Test 2 in Chapter 8. The Anglian Water case study demonstrated the IVC's value and relevance to existing organisations that wanted to harness SDGs to be more aligned with global challenges and methods to respond to them.
- The findings from the case study suggested that measurement of SDG performance can accommodate the perspective of the Triple Bottom Line (i.e., social, environmental and economic performance). This supports a broader definition of project sustainability that includes the three pillars (i.e., social, environmental and economic performance). It is also important to provide simplicity and structure for the analysis in regard to selecting and measuring SDGs.
- The case study indicated that measurement of SDG performance can accommodate the required different organisational levels, namely portfolio, programme and project levels.
- The evidence from the case study indicates there are high MISI performers, such as Anglian Water, at an early-adopter stage, that have potential to drive change across the sector.
- The fourth proposition is therefore supported by the research evidence.

The overall results indicate that SDG measurement practices are embraced in principle but are problematic in practice and that rarely does action match rhetoric. While the research was completed in the UK, the findings have broader applicability to other countries since most construction firms have extensive global business footprints.
Researchers can use the findings to extend the current understanding of measuring outcomes and impact at project-portfolio levels, and, for practitioners, the study provides insights into the contextual preconditions necessary to achieve the intended outcomes of adopting a mechanism for the measurement of SDGs. The national relevance of this research is corroborated by the project partners, whose letters confirming the derived Impact to their organisations of this work, are included at Appendices 17-23: Institution of Civil Engineers; the Buildings Research Establishment; the Cabinet Office's Infrastructure and Projects Authority; the Environment Agency; and, the UN Global Compact. These letters of support confirm that the research study has provided meaningful insights into how infrastructure investment can be focused to increase impact across SDGs with broader benefits in the longer-term achieved. It also demonstrates how the benefit of the MISI approach is aligned to Creating Shared Value that drives more successful and sustainable businesses with increased contributions across the TBL impacts of profit, people and planet.

Civil engineering practitioners are encouraged to reflect on the findings from this research and consider how sustainability can be incorporated throughout the project lifecycle – from the design to construction, operation and disposal stages. As described herein, infrastructure investment and the corresponding projects represent a major opportunity for the construction sector to establish sustainable building practices in the industry that reduce environmental impacts and help construction enterprises to remain competitive. Moreover, this research has attempted to tackle the inherent complexity associated with the SDG framework and supporting indicators as well as the challenge of how to measure performance against such goals for infrastructure projects. In this context, the civil engineering community is well placed to contribute to further developments in the field through applying the findings from both theoretical and empirical research to improve the measurement of SDGs and drive sustainability across the sector.

10.3 LIMITATIONS

The research is limited in time and space. As regards time, the study has not been longitudinal, as illustrated in Saunders' research Onion Ring in Figure 31 (Saunders, et al., 2009), because it has only investigated an event at a single point in time. Future research could include an assessment of whether the findings from the survey and interviews of this MISI study have evolved in light of increasing international awareness, such as at the COP26 global conference hosted in Glasgow in November 2021 (BBC, 2021). In terms of 'space' the study has had limited resources in terms of research bandwidth, and whilst the collaborative partnership with ICE, UCL, BRE, UN Global Compact, IPA, the Environment Agency and Tames Tideway megaproject have broadened and deepened the research, it has necessarily been constrained by the doctoral researcher's finite capacity.

There are inevitably limitations of the research study. Although a significant level of empirical analysis has been undertaken through the survey of 325 practitioners and interviews with 40 senior leaders along with the initial case study investigations, there is nevertheless scope for more detailed case study research to be carried out. In this regard, further case studies would build more detailed qualitative and quantitative data that the findings of the research study can be evaluated against. In this way, the proposed research will likely provide more meaningful insights into how infrastructure investment can be better focused and lessons that increase impact across SDGs will be applied more effectively. This is important because infrastructure projects have always been an essential underpinning aspect for society, but today's global business context gives new weight to infrastructure's importance, and this approach of measuring SDG impact at the project level provides a golden thread to link the projects' delivery outcomes with national and global SDG targets. Conversely, if projects do not widen the definition of success to incorporate SDG impacts, they will fail to accommodate the unique enabling role of engineering and infrastructure, inadvertently weakening the resilience and wellbeing of both business and society.

10.4 RECOMMENDATIONS

There are theoretical implications for the development of an adapted Theory of Change framework for future research into this embryonic study area. Most importantly, given the immediacy and Grand Challenge of achieving the SDG 2030 goals, practical application is significant since the Impact Value Chain has flexibility to be used at both project and portfolio levels (as described by the Environment Agency in their Impact Statement shown at Appendix 18), thereby linking tactical delivery on projects to organisational SDG impacts at the portfolio level and potentially leads to improved investment decisions with increased likelihood of success in achieving the SDG 2030 targets, involving broader benefits in regard to people, profit and planet considerations. The summary recommendations can be briefly stated across four themes as, shown below in Figure 78. Three of the recommendations are shown aligned to the project/portfolio lifecycle and business case stages, the fourth recommendation is to further the research into MISI and its IVC and SDGiPro methodology.



Figure 78. Lifecycle view of MISI Recommendations.

10.4.1 Recommendation 1 - The MISI (IVC and SDGiPro) approach should be used to support the Investment Decision stage of infrastructure projects and their strategic portfolios.

MISI should be used to:

- Inform the design and choice of the project by better understanding the broader, longer-term outcomes and impacts.
- Widen discussion on benefits and disbenefits for portfolios, thereby seeking stakeholder alignment with project's aims.
- Embed IVC framework in the options appraisal stage by using the adapted ToC theory to enable an SDG causal chain and a TBL trade-off.

The IVC and SDGiPro should be used during the problem consideration and design phase (through the appropriate use of the adapted causal logic chains of the ToC and TBL theories) to increase the likelihood that stakeholders will have clearly specified the initiative's intended outcomes, the activities that need to be implemented in order to achieve those outcomes, and the contextual factors that are likely to influence them. In this way, the early use of the IVC at the concept stage of project design helps show how activities will lead to interim and longer-term outcomes and SDG impacts. It also identifies the contextual conditions (as developed in the MISI 'Variables Framework') that may affect them. This helps strengthen the business case for attributing subsequent change in these outcomes (from initial levels) to the delivery of project activities and outputs.

This recommendation has been actively taken forward with the UK Government's Infrastructure and Projects Authority who have leveraged the MISI work by employing the researcher to propose edits to the new Green Book's (2018) updates to include SDG measurement. Therefore, this recommendation can be viewed in the context of extending the field of ToC/TBL into project sustainability and indeed, the thematical framework of the Green Book's Five Case Model (5CM)¹³. It is anticipated that by influencing the new edits to the 5CM, the MISI research will achieve significant indirect influence on investment decisions.

10.4.2 Recommendation 2 - The IVC should be used to facilitate the measurement and data collection (monitoring, reporting and evaluation) process.

MISI should be used to:

- Shape the project and portfolio MRE (monitoring, reporting and evaluation) process of SDG-benefits/disbenefits.
- Validate SDG contributions and separate direct attribution from indirect contribution.

The adaption of the ToC and TBL theories provides the building blocks of any effective evaluation, but they are especially useful for mid-course feedback to managers and for developing a knowledge base about how and why SDG

¹³ The Green Book's Five Case Model is the core document and approach for the justification of taxpayers' money, through a defined business case protocol.

measurement can be applied. For example, the IVC (and SDGiPro) asked that participants be as clear as possible about not only the ultimate outcomes and SDG impacts they hope to achieve but also the avenues through which they expect to achieve them (Weiss, 1995). An evaluation based on a ToC and TBL, therefore, identifies what to measure, ultimate and interim outcomes, and the implementation of activities intended to achieve these outcomes. It also helps to guide choices about when and how to measure those elements. The unique insight of this research is that MISI, through alignment of ToC and TBL in a single framework, provides a new way to find a causal pathway to SDG impacts, which allows a clearer measurement framework with a vocabulary that enables stakeholders to align around short and longer term TBL-SDG outcomes and impacts.

10.4.3 Recommendation 3 - The MISI approach should be used to support knowledge building through the learning and sharing from MISI good practice.

MISI should be used to:

- Complete the learning stage of the MREL for the benefit of further research and also for practitioners.
- Build confidence in leaders' selection of 'stretch' SDG targets that they might not achieve, such as ambitious Net Zero targets.
- ICE should bring together industry leaders (at all levels, not just CEOs) to champion the adoption of realistic, practical and verifiable MISI.

MISI could contribute to the learning stage of the MREL process by providing the link from local to regional, national and global goals. In doing so, the lessons have potential to have wider utility than the local project and could be leveraged for greater benefits. For example, lessons from MISI on a water project in East Anglia (UK) could have relevant lessons for an international development project in East Africa.

MISI lessons sharing should leverage this research by using the ICE as a trusted organisation to build a coalition of willing leaders (CEOs and key individuals across the age and gender spectrum) to share their MISI experiences so that others may benefit, even if they underperform in achieving their aspired SDG targets.

MISI could also be used by the ICE to build engineers' better understanding of the MISI business knowledge embedded in the IVC (ToC, TBL and CSV) and the definition of SDG outcome success. This could be considered as a potential broadening of a civil engineer's (and other related professions) professional qualifications.

10.4.4 Recommendation 4 - The MISI research should be used to guide future research with potential implementation of an enhanced knowledge and learning protocol for infrastructure stakeholders.

MISI should be used to:

- Establish a basis for further research that builds on early empirical evidence from working with the Environment Agency and Tideway Ltd.
- Build a MISI indicator framework, based on MISI learnings, that enables project leaders to select which of the targets and indicators are relevant for their size of project.
- Seek funding for the establishment of a MISI Centre of Excellence that can sustain the learning.
- Consider cross-disciplinary approaches, such as with Artificial Intelligence, to find ways that can leverage large quantities of data to better inform the learning cycle.

The study provides a foundation to guide future research on project and portfolio capabilities based on the MISI Variables Framework established through this study. This should include the development of an indicators framework, derived from the existing UN SDG targets and indicators as well as those used at the portfolio level (by the GRI) and project level (such as CEEQUAL), thereby building on the SDG golden thread established in Chapter 8.

The study has implications for the operations field, broader project studies, leadership/governance studies, international development studies, supply chain management and many other areas. Each of these have overlaps and perhaps even more opportunity lies with multi-disciplinary studies, such as seeking to work with the education sector to share ideas on how to develop knowledge and feed it into professional learning courses, such as suggested by the ICE, who have proposed leveraging the MISI research to enhance their professional learning and development courses (as discussed in Recommendation 3).

It is also proposed that further research uses the IVC model and the SDGiPro methodology to develop an improved understanding of the organisational context within which the definition and measurement of infrastructure project success is made. It could be used to examine the leadership and governance theories and relationships that underpin the overall analysis of project success definition and measurement. For example, recent studies (Müller, 2017) have provided clarity on how project governance, that shapes the reporting, directing and management of projects, is best understood by scrutinizing the overlapping influences of corporate governance at the organisational level and the separate, but related, governance at project level that has its own customs, rules and approaches for reporting project success.

10.5 MISI IMPACT AND IMPLICATIONS FOR FUTURE USE

The research produced twelve peer-reviewed papers including being published in seven internationally recognised academic journals, such as: *Sustainability* (2 articles), *Administrative Sciences*, and the *Proceedings of the Institution of Civil Engineers–Engineering Sustainability*. The MISI research outputs have been taken forward by the government and industry partners, specifically the Environment Agency and Thames Tideway Project, working together to establish this new approach for measuring sustainability on infrastructure projects.

10.5.1 Future Work – MISI with Environment Agency and Thames Tideway

A useful derivative of the MISI research has been the collaborative partnership to test the prototype model and its approach across the Environment Agency's full portfolio of projects and also, the Thames Tideway Tunnel megaproject. Although this detailed case study was an extension to the MISI research captured in this thesis, the ongoing and future Environment Agency and Tideway work has provided a platform for further investigations to highlight the utility and applicability of the IVC.

However, the 4-month in-depth case study of using the IVC in the 'real world', that was entirely led and designed by the doctoral researcher, has provided further insights into the applicability, usefulness, coherence and utility by these leading government and private sector organisations. The support letters proving the veracity of this MISI doctoral research impact is evidenced by senior leaders' letters in Appendices 17-23. The practical application is significant since, with improved linkage of tactical delivery to strategic SDG impacts, improved investment decisions

will be made, and systemic level lessons can be applied to increase the likelihood of success in achieving the SDG 2030 targets.

10.5.2 Future Work – Learning from MISI research approach

The next stage of the research is to develop the infrastructure SDG measurement methodology proposed in Figure 61 into a fully defined methodology that is adaptable to the scale of the project and also its position in the project-programme-portfolio hierarchy. Thus, the model could be tested in active projects. The existing case studies could be broadened to include both developing and developed countries, and this might focus on a single asset type across the national economic infrastructure categories of energy, waste, water, transport and ICT.

The future research could close the current gaps of the existing propositions (as discussed above in sections 10.2.2.1-10.2.2.4) by application of a greater emphasis on 'engaged' research design, as introduced in Chapter 5 (Table 10) and now updated in Table 37 below.

	Intensive $\leftarrow \rightarrow$ Extensive (a)								
	What is the mechanism?	How do context & mechanism typically behave x OPM levels?	How do context & mechanism historically interact?	What are the context variables that impact outcomes?					
Detached (b)	Case-study (Future MISI)	Comparative case- study (Future MISI study)	Institutional / historical analysis	Surveys					
Engaged (c)	Action research (Future MISI)	Intensive realist literature evaluations	Barefoot research (Future MISI)	Extensive realist evaluation					

Table 37: Critical Realist Research Designs (amended from Vincent and O'Mahoney, 2018; Ackroyd and Karlsson 2014)

Based on this critical realist framework, the proposed future research design methods could be developed as follows:

• **Case-study**. The most frequent form of CR research that enables MISI to test the exploration of using the IVC prototype to assess whether it is practical, plausible and achievable. The next case study would seek to align the application of the IVC with existing business case design (Green Book, 2018) and also the existing protocols of benefits realisation.

- Organisational behavioural impacts. The discussion in Chapter 5 introduced the work of Peter Morris (2017) and Frank Geels (2007). Both apply systems thinking to seek understanding of complex relationships between operating models of strategy, processes, systems and people. Further study could harness these insights to consider organisational change models such as Kotter (2012) to explore the interaction of stakeholders on the IVC model. The research could align with the Institution of Civil Engineers' current and future topics of '*Transformative Collaborations*' (as part of Project 13) and '*Shaping Net Zero*' (the 2021 President's primary knowledge campaign) that have strong parallels with the MISI area of study.
- Action research. The summary results suggest this as a proposed new research phase of the MISI study, with the Environment Agency and Thames Tideway megaproject. This method of research was discounted in Chapter 4, Section 4.3.2, due to a lack of time, resources and buy-in from potential organisations to develop the approach with. However, in the final 12 months of the doctoral research a strong collaboration with the Environment Agency has opened the way for an Action Research based approach in future. This would allow the training of the employees to jointly work with the lead researcher to develop the follow-on 'trials stage'. This is currently the case at the Environment Agency, where the Chair of the organisation has sanctioned the Portfolio Design team to take forward the learnings from the earlier MISI research work. The CEO of the Cabinet Office's Infrastructure and Projects Authority has explicitly backed this approach to widen the use of SDGs across the full government portfolio of projects.
- **Comparative case-study**. The proponents of this approach such as Kirkpatrick et al. (2005) and Delbridge (1998) suggest this as a suitable way to explore how similar mechanisms operate in different contexts. So, for example, this might be chosen as the proposed approach for future research, beyond this thesis write-up, as a way to more comprehensively test the IVC across the OPM levels of portfolio and project levels with the Environment Agency and the megaproject level with Thames Tideway.
- **Barefoot research**. Lindqvist (1979) describes how organisations can develop their own interpretations by use of training and encouraging employees of the

case study organisation to do their own research. This is the proposed approach to be taken with the collaborative MISI partnership between the Environment Agency and Thames Tideway megaproject, whereby they can be trained to develop their own solutions based on the MISI research findings from this thesis.

10.6 PERSONAL REFLECTIONS

This has been a long journey. It has been a privilege to conduct the research with such a broad range of senior collaborative partners. It has allowed the opportunity to harness 'concrete experience', as proposed by Kolb's (1981, 1984) Experiential Learning Cycle theory, to help inform my approach across the other three stages: Reflective Observation of the New Experience; Abstract Conceptualisation; and, Active Experimentation. Through this process, which is actually a continual series of smaller loops, each producing insights, set-backs, connections and experiences, the learning iterations have successfully increased the knowledge in the MISI area. The challenge is to now identify how I can contribute to the wider issues of influencing government policy to increase TBL and SDG impact at project and portfolio levels. This has started by the invitation in October and November 2020 from the Cabinet Office to use the PhD MISI knowledge to suggest 'green' sustainable development updates to the family of government investment decision guidance books, known as the 'Green Books'. Ultimately, this will ensure the MISI research achieves the greatest impact from its insights. Future infrastructure investment decisions in UK and abroad could proactively use the IVC to help broaden their definition of success to include the TBL of profit, planet and people to drive a more sustainable future for us all, whilst supporting the achievement of the UN's 2030 SDG ambitions.

As Antonio Guterres, the UN Secretary General, says in October 2020:

"As Member States recognized at the SDG Summit held last September, global efforts to date have been insufficient to deliver the change we need, jeopardizing the Agenda's promise to current and future generations. Now, due to COVID-19, an unprecedented health, economic and social crisis is threatening lives and livelihoods, making the achievement of Goals even more challenging' (UN, 2020). He suggests our immediate and concerted global efforts in 2021, and specifically COP26 hosted by UK in November 2021 in Glasgow, is of paramount importance. He concludes:

'We are at a make or break moment" (BBC, 2021).

References

Aarseth, W., Ahola, T., Aaltonen, K., Økland, A., & Andersen, B. (2017). Project sustainability strategies: A systematic literature review. International Journal of Project Management, 35(6), 1071-1083 accessed on 4 March 2020 at https://ntnuopen.ntnu.no/ntnuxmlui/bitstream/handle/11250/2445998/Project+sustainability+strategies-

+A+systematic+literature+review.pdf?sequence=2

Adams, C. A (2017). The Sustainable Development Goals, integrated thinking and the integrated report, published by the International Integrated Reporting Council and the International Chartered Accountants of Scotland, accessed at http://tsss.ca/wp-content/uploads/2017/11/SDGs-and-theintegrated-report full17.pdf on 6 June 2020.

Adshead, Daniel, Scott Thacker, Lena I. Fuldauer, and Jim W. Hall. (2019). Delivering on the Sustainable Development Goals through long-term infrastructure planning. Global Environmental Change 59: 101975.

Ainger, C. M., & Fenner, R. A. (2014). Sustainable infrastructure: principles into practice. London, UK: ICE publishing.

Albuquerque, Rui, Yrjö Koskinen, and Chendi Zhang. (2019). Corporate social responsibility and firm risk: Theory and empirical evidence. Management Science 65: 4451-69.

Allen, C., G. Metternicht, and T. Wiedmann. (2016). National pathways to the Sustainable Development Goals (SDGs): A comparative review of scenario modelling tools. Environmental Science & Policy 66: 199–207.

Allen, C., G. Metternicht, and T. Wiedmann. (2019). Prioritising SDG targets: assessing baselines, gaps and interlinkages. Sustainability Science 14: 421-38.

Anglian Water. Responsible Business. Annual Integrated Report 2018 (2018). Available online: https://www.anglianwater.co.uk/siteassets/household/about-us/annual-intergrated-report-2018.pdf (accessed on 6 June 2020).

Anglian Water. New Models for Collaborative Working, (2018). A Guide to Community Wisbech—An Anglian Water Regeneration in Perspective. Available online: https://www.bitc.org.uk/sites/default/files/bitc lr wisbech v1.compressed.pdf (accessed on 6 June 2020).

Anwar, B., Xiao, Z., Akter, S., & Rehman, R. U. (2017). Sustainable Urbanization and Development Goals Strategy through Public-Private Partnerships in a South-Asian Metropolis. Sustainability, 9(11), 1940.

Arif, M.; Egbu, C.; Haleem, A.; Kulonda, D.; Khalfan, M, (2009). State of green construction in India: Drivers and challenges. J. Eng. Des. Technol. 2009, 7, 223-234.

Armanios, D. E., C. E. Eesley, J. Li, and K. M. Eisenhardt (2017). How entrepreneurs leverage institutional intermediaries in emerging economies to acquire public resources. Strategic Management Journal 38: 1373–90.

Astbury, B.; Leeuw, F.L., (2010). Unpacking black boxes: Mechanisms and theory building in evaluation. Am. J. Evaluat. 2010, 31, 363-381.

Association of Project Management. (2012) APM's Body of Knowledge 6th Edition, Definitions. Interfaces, 12. https://www.apm.org.uk/media/1605/final-proof-bok-6-definitions.pdf (Accessed 20 March 2020).

Association for Project Management, UK. (2019). APM Body of Knowledge. Princes Risborough: Buckinghamshire. ISBN 978-1-903494-83-7.

Astbury, B. and Leeuw, F.L., (2010). Unpacking black boxes: mechanisms and theory building in evaluation. American journal of evaluation, 31(3), pp.363-381.

Atkinson, R. (1999). Project management: cost, time and quality, two best guesses and a phenomenon. It's time to accept other success criteria, International Journal of Project Management 17, 337-342. doi:10.1016/s0263-7863(98)00069-6.

Aust, V., Morais, A. I., & Pinto, I. (2020). How does foreign direct investment contribute to Sustainable Development Goals? Evidence from African countries. Journal of Cleaner Production, 245, 118823.

Awadh, O., (2017). Sustainability and green building rating systems: LEED, BREEAM, GSAS and estidama critical analysis. *J. Build. Eng.* 2017, *11*, 25–29.

Baccarini, D. (1999). The logical framework method for defining project success. *Project Management Journal*, 30(4), pp.25-32.

Baird, C. H. (2015). Myths, exaggerations and uncomfortable truths: The real story behind Millennials in the workplace. The IBM Institute for Business Value. Available online: https://www.ibm.com/downloads/cas/Q3ZVGRLP (accessed on 7 August 2020).

Bali Swain, R., and F. Yang-Wallentin. (2020). Achieving sustainable development goals: Predicaments and strategies. *International Journal of Sustainable Development & World Ecology* 27: 96–106.

Banihashemi, S., Hosseini, M. R., Golizadeh, H., & Sankaran, S. (2017). Critical success factors (CSFs) for integration of sustainability into construction project management practices in developing countries. *International Journal of Project Management*, *35*(6), pp. 1103-1119.

Barkemeyer, R., Holt, D., Preuss, L. and Tsang, S (2014). What happened to the 'development'

in sustainable development? Business guidelines two decades after Brundtland. Sustainable development, 22(1), pp.15-32.

Barrett, J., Peters, G., Wiedmann, T., Scott, K., Lenzen, M., Roelich, K. and Le Quéré, C (2013). *Consumption-based GHG emission accounting: a UK case study*. Climate Policy, 13(4), pp.451-470.

Baumeister, R. F., & Leary, M. R. (1997). Writing narrative literature reviews. *Review of general psychology*, 1(3), 311-320.

Bazeley, P. and Jackson, K. eds. (2013). *Qualitative data analysis with NVivo*. Sage Publications Limited.

BBC. (2021). *Why 2021 could be turning point for tackling climate change* Accessed on 1 January 2021 on the BBC digital website https://apple.news/AkBHByxVZSSCXaGiuClYkhA.

Bebbington, A. (1999). Capitals and capabilities: a framework for analyzing peasant viability, rural livelihoods and poverty. *World development*, 27(12), 2021-2044.

Bénabou, R., and J. Tirole. (2010). Individual and corporate social responsibility. *Economica* 77: 1–19.

Berge, M. S. (2017). Telecare–where, when, why and for whom does it work? A realist evaluation of a Norwegian project. *Journal of Rehabilitation and Assistive Technologies Engineering* 4: 2055668317693737.

Bergeron, Dave A., and Isabelle Gaboury. (2020). Challenges related to the analytical process in realist evaluation and latest developments on the use of NVivo from a realist perspective. *International Journal of Social Research Methodology* 23: 355–65.

Berk, M.L.; Schur, C.L.; Cantor, J.C. (1995). Ability to Obtain Health Care: Recent Estimates from the Robert Wood Johnson Foundation National Access to Care Survey. *Heal. Aff.* 1995, *14*, 139–146, doi:10.1377/hlthaff.14.3.139.

Bhaskar, Roy, (2013). A Realist Theory of Science. London and New York: Routledge.

Bhattacharya, A., J. Oppenheim, and N. Stern. (2015). Driving Sustainable Development Through better Infrastructure: Key Elements of a Transformation Program. Brookings Global Working Paper Series, Washington DC, USA. Available online: https://g24.org/wp-content/uploads/2016/02/Driving-Sustainable-Development-Through-Better-Infrastructure-Key-Elements of a Transformation Program Phottachary Oppenheim Stern July 2015 rdf (accessed

Elements-of-a-Transformation-Program-Bhattacharya-Oppenheim-Stern-July-2015.pdf (accessed on 10 July 2020).

Bhatti, W., & Zaheer, A. (2014). The role of intellectual capital in creating and adding value to organizational performance: A conceptual analysis. *The Electronic Journal of Knowledge Management*, *12*(3), pp. 187-194.

Bielenberg, A., M. Kerlin, J. Oppenheim, and M. Roberts. (2016). Financing change: How to mobilize private-sector financing for sustainable infrastructure. *McKinsey Center for Business and Environment*.

Bloom, J.D. and Hinrichs, C.C. (2011) Moving local food through conventional food system infrastructure: Value chain framework comparisons and insights. *Renewable Agriculture and Food Systems*, 26(1), pp.13–23.

Bonini, S. and Emerson, J. (2005) *Maximizing blended value–Building beyond the blended value map to sustainable investing, philanthropy and organizations*. http://community-wealth.org (accessed on 2 April 2020).

Bosch-Badia, M. T., Montllor-Serrats, J., & Tarrazon-Rodon, M. A. (2018). Sustainability and ethics in the process of price determination in financial markets: A conceptual analysis. *Sustainability*, *10*(5), 1638.

Boswell, James F., David R. Kraus, Scott D. Miller, and Michael J. Lambert. (2015). Implementing routine outcome monitoring in clinical practice: Benefits, challenges, and solutions. *Psychotherapy Research 25*: 6–19.

Brady, T., Davies, A., & Gann, D. M. (2005). Creating value by delivering integrated solutions. *International Journal of Project Management*, 23(5), 360-365.

Brundtland, G.H., Khalid, M. and Agnelli, S. (1987) *Our Common Future: Report of the World Commission on Environment and Development*; Oxford: Oxford University Press. **8(9).**

Bugg-Levine, A. and Emerson, J. (2011) Impact investing: Transforming how we make money while making a difference. *Innovations: Technology, Governance, Globalization*, 6(3), pp.9-18.

Building Research Establishment (BRE). (2019). Available online: https://www.breeam.com/ and https://www.breeam.com/engage/research-and-development/consultation-engagement/ceequal-v6-consultation/ (accessed on 26 August 2020).

Building Research Establishment (2020), *The Built Environment and Future Sustainability - The relationship between BREEAM and the Sustainable Development Goals*,

https://files.bregroup.com/breeam/sdg/BREEAM_SDB_A4%20_BRE_115430_0720_web.pdf.

Burkhard, B., Kroll, F., Nedkov, S. and Müller, F (2012). *Mapping ecosystem service supply, demand and budgets*. Ecological indicators, 21, pp.17-29.

Butler, D., Farmani, R., Fu, G., Ward, S., Diao, K. and Astaraie-Imani, M (2014). A new approach to urban water management: Safe and sure. Procedia Engineering, 89, pp.347-354.

Campbell, K. J. (1998). Once burned, twice cautious: explaining the Weinberger-Powell doctrine. *Armed forces & society*, *24*(3), 357-374.

Carvalho, M. M., & Rabechini Jr, R. (2017). Can project sustainability management impact project success? An empirical study applying a contingent approach. *International Journal of Project Management*, *35*(6), pp. 1120-1132.

Cassell, C. & Johnson, P. (2006). 'Action research: Explaining the diversity.' *Human Relations*, 59:6, 783-814.

Chen HT, Rossi PH. (1983). Evaluating with sense: The theory-driven approach. *Evaluation Review*. 1983, 7, 283–302, doi: 10.1177/0193841X8300700301

Chen HT, Chen HT. (2005). *Practical Program Evaluation: Assessing and Improving planning, Implementation, and Effectiveness.* Sage: Thousand Oaks, CA, USA, 2005.

Cheng, S., Li, Z., Uddin, S. M. N., Mang, H. P., Zhou, X., Zhang, J., ... & Zhang, L. (2018). Toilet revolution in China. *Journal of environmental management*, *216*, 347-356.

Ciegis, R., Ramanauskiene, J. and Martinkus, B. (2009) The concept of sustainable development and its use for sustainability scenarios. *Engineering Economics*, 62(2).

Clark, P. (2012). Organizations in Action: Competition Between Contexts. Taylor & Francis. Clausewitz, C. (1982). On war (Vol. 20). Penguin UK.

Clevenger, C.M., Ozbek, M.E. and Simpson, S (2013). April. Review of sustainability rating systems used for infrastructure projects. In *49th ASC Annual International Conference Proceedings* (pp. 10-13). Associated Schools for Construction.

Cooke-Davies, T. (2007). The "real" success factors on programmes, *International Journal of Programme Management*, 20, 3, pp. 185–190. In Morris and Pinto, j. (eds) in *The Wiley Guide to Managing Projects*, Wiley: Hoboken, NJ, pp106-108.

Costanza, R., Fioramonti, L. and Kubiszewski, I., (2016). The UN sustainable development goals and the dynamics of well-being. *Frontiers in Ecology and the Environment*, 14(2), pp.59-59. Costanza, R., & Patten, B. (1995). Defining and predicting sustainability. *Ecological Economics* 15(3), 193–196.

Cooke-Davies, Terry. (2007). The real success factors on programmes. *International Journal of Programme Management* 20: 185–90.

Corporate Reporting Dialogue (CRD). (2019). SDGs and the Future of Corporate Reporting. Available online: http://integratedreporting.org/wp-content/uploads/2019/02/The-Sustainable-Development-Goals-and-the-future-of-corporate-reporting-1.pdf (accessed on 19 March 2019). Crawford, L. (2005). Senior management perceptions of project management competence. *Int. J. Proj. Manag.* 2005, *23*, 7–16, doi:10.1016/j.ijproman.2004.06.005.

Creswell, J. W., Plano Clark, V. L., Gutmann, M. L., & Hanson, W. E. (2003). Advanced mixed methods research designs. *Handbook of mixed methods in social and behavioral research*, 209, 240.

Creswell, J. W., and Creswell, J., David. (2017). *Research Design: Qualitative, Quantitative, and Mixed Methods Approaches*. Thousand Oaks: Sage Publications.

Creswell, J.W.; Clark, V.L.P. (2017). *Designing and Conducting Mixed Methods Research*; Sage Publications: Thousand Oaks, CA, USA.

Crosthwaite, C., Jolly, L., Brodie, L., Kavanagh, L., Buys, L. and Turner, J., (2012). Curriculum design and higher order skills: challenging assumptions. In Proceedings of the 5th International Conference: Innovation, Practice and Research in Engineering Education (EE 2012). Loughborough University.

Cummins, D. D., Lubart, T., Alksnis, O., & Rist, R. (1991). Conditional reasoning and causation. *Memory & Cognition*, 19(3), 274-282.

da Silva, L., Domingos, P., Prietto, M., Pavan Korf, E., (2019). Sustainability Indicators For Urban Solid waste Management IN Large And Medium-Sized Worldwide Cities Journal of Cleaner Production (IF 6.395) Pub Date : 2019-07-26, DOI: 10.1016/j.jclepro.2019.117802.

Davies, A., & Brady, T. (2000). Organisational capabilities and learning in complex product systems: towards repeatable solutions. *Research policy*, 29(7-8), 931-953.

Davies, A. (2004). Moving base into high-value integrated solutions: a value stream approach. *Industrial and corporate change*, *13*(5), 727-756.

Davies, A., & Hobday, M. (2005). *The business of projects: managing innovation in complex products and systems*. Cambridge University Press.

Davies, A., Gann, D., & Douglas, T. (2009). Innovation in megaprojects: systems integration at London Heathrow Terminal 5. *California management review*, *51*(2), 101-125.

Davies, A., & Brady, T. (2016). Explicating the dynamics of project capabilities. *International Journal of Project Management*, 34(2), 314-327.

Davies, A., Manning, S., & Söderlund, J. (2018). When neighboring disciplines fail to learn from each other: The case of innovation and project management research. *Research Policy*, 47(5), 965-979.

Dean, K., Trillo, C., & Bichard, E. (2017). Assessing the Value of Housing Schemes through Sustainable Return on Investment: A Path towards Sustainability-Led Evaluations? *Sustainability*, 9(12), 2264.

Delbridge, R. (1998). *Life on the Line in Contemporary Manufacturing: The Workplace Experience of Lean Production and the "Japanese" Model*. Oxford University Press.

Deloitte. (2018a). The Business Case for Inclusive Growth. Available online: https://www2.deloitte.com/content/dam/Deloitte/global/Documents/About-Deloitte/gx-abt-wef-business-case-inclusive-growth-global%20report.pdf (accessed on 7 August 2020).

Deloitte. (2018b). The Deloitte Millennial Survey 2018—Millennials' Confidence in Business, Loyalty to Employers Deteriorate. Available online: https://www2.deloitte.com/content/dam/Deloitte/global/Documents/About-Deloitte/gx-2018-

millennial-survey-report.pdf (accessed on 7 August 2020).

Denicol, J., Davies, A., & Krystallis, I. (2020). What Are the Causes and Cures of Poor Megaproject Performance? A Systematic Literature Review and Research Agenda. *Project Management Journal*, 8756972819896113.

De Villiers, C.; Rinaldi, L.; (2014). Unerman, J. Integrated reporting: Insights, gaps and an agenda for future research. *Account. Audit. Account. J.* 2014, *27*, 1042–1067.

De Vries S. R. de Wit,W. C. A.; Folke, C.; Gerten, D.; Heinke, J.; Mace, G. M.; Persson, L. M. Ramanathan, V. Reyers, B. Sorlin, S. (2015). "Planetary boundaries: Guiding human development on a changing planet". *Science*. 347 (6223).

Diaz-Sarachaga, J. M., Jato-Espino, D., Alsulami, B., & Castro-Fresno, D. (2016). Evaluation of existing sustainable infrastructure rating systems for their application in developing countries. *Ecological indicators*, *71*, 491-502.

Diekhoff, George. (1992). *Statistics for the Social and Behavioral Sciences*. Univariante, Bivariate, Multivariante. No. HA29. D46 1992. London: William C Brown Pub; ISBN-10: 0697285146; ISBN-13: 978-0697285140.

Ding, G.K.; Shen, L. (2010). Assessing sustainability performance of built projects: a building process approach. *Int. J. Sustain. Dev.* 2010, *13*, 267, doi:10.1504/ijsd.2010.037558.

Ding, X., Zhou, C., Mauerhofer, V., Zhong, W., & Li, G. (2019). From environmental soundness to sustainable development: Improving applicability of payment for ecosystem services scheme for diverting regional sustainability transition in developing countries. *Sustainability*, *11*(2), 361.

Di Vaio, A., & Varriale, L. (2018). Management innovation for environmental sustainability in seaports: Managerial accounting instruments and training for competitive green ports beyond the regulations. *Sustainability*, 10(3), 783.

Donohue, I., Hillebrand, H., Montoya, J. M., Petchey, O. L., Pimm, S. L., Fowler, M. S., ... & O'Connor, N. E. (2016). Navigating the complexity of ecological stability. *Ecology Letters*, 19(9), 1172-1185.

Doran, G. T. (1981). There's a S.M.A.R.T. way to write management's goals and objectives. *Management Review*. AMA Forum. 70 (11): 35–36.

Drucker, P. F. (1995). *People and performance: The best of Peter Drucker on management*. Routledge.

Drucker, P. F. (2006). Classic Drucker: essential wisdom of Peter Drucker from the pages of Harvard Business Review. Harvard Business Press.

Drucker, P. F. (2020). The essential drucker. Routledge.

Dudwick, N.; Kuehnast, K.; Jones, V.N.; Woolcock, M. (2006). *Analyzing Social Capital in Context. A Guide to Using Qualitative Methods and Data*; World Bank Institute: Washington, DC, USA, 2006; pp. 1–46

Dushenko, M., Bjorbæk, C. T., & Steger-Jensen, K. (2018). Application of a Sustainability Model for Assessing the Relocation of a Container Terminal: A Case Study of Kristiansand Port. *Sustainability*, *11*(1), 1-18.

Easterby-Smith, M.; Thorpe R.; Lowe, A. (2002). *Management Research: An Introduction*, 2nd ed; Sage Publications: London, UK, 2002, pp. 342.

Easton, Geoff. (2010). Critical realism in case study research. *Industrial Marketing Management* 39: 118–28.

Eccles, Robert G., and Michael P. Krzus. (2010). *One Report: Integrated Reporting for a Sustainable Strategy*. Hoboken: John Wiley & Sons.

The Economist (2020). The new normal – Covid-19 is here to stay. The world is working out how to live with it. *Economist International*, 4 July 2020 Edition.

Edum-Fotwe, F.T.; Price, A.D.F. (2009). A social ontology for appraising sustainability of construction projects and developments. *Int. J. Proj. Manag.* 2009, *27*, 313–322.

Elkington, J. (1994). Towards the sustainable corporation: Win-win-win business strategies for sustainable development. *California management review*, *36*(2), 90-100.

Elkington, J. (2013). Enter the triple bottom line. In *The triple bottom line* (pp. 23-38). Routledge.

Elkington, J. (2018). 25 Years Ago, I Coined the Phrase "Triple Bottom Line." Here's Why It's Time to Rethink It, Harvard Business Review, 25 June 2018. Available online:

https://hbr.org/2018/06/25-years-ago-i-coined-the-phrase-triple-bottom-line-heres-why-imgiving-up-on-it (accessed on 6 June 2020)

Emas, R. (2015). The concept of sustainable development: definition and defining principles. *Brief for GSDR*, 2015. Available online:

https://sustainabledevelopment.un.org/content/documents/5839GSDR%202015_SD_concept_def initon_rev.pdf (accessed on 10 July 2020).

Emerson, Jed, Jay Wachowicz, and Suzi Chun. (2000). Social return on investment: Exploring aspects of value creation in the non-profit sector. *The Box Set: Social Purpose Enterprises and Venture Philanthropy in the New Millennium* 2: 130–73.

Engert, S., Rauter, R., & Baumgartner, R. J. (2016). Exploring the integration of corporate sustainability into strategic management: a literature review. *Journal of cleaner production*, *112*, 2833-2850.

Executive Office of the President, Office of Science and Technology Policy. (1989). "The

Farinosi, F., Giupponi, C., Reynaud, A., Ceccherini, G., Carmona-Moreno, C., De Roo, A., ... & Bidoglio, G. (2018). An innovative approach to the assessment of hydro-political risk: A spatially explicit, data driven indicator of hydro-political issues. *Global environmental change*, *52*, 286-313.

Federal High-Performance Computing Program" Sept. 1989, pp. 49–50: Appendix A Summary.

Fenner, R. A., Ainger, C. M., Cruickshank, H. J., & Guthrie, P. M. (2006, December). Widening engineering horizons: addressing the complexity of sustainable development. In *Proceedings of the Institution of Civil Engineers-Engineering Sustainability* (Vol. 159, No. 4, pp. 145-154). Thomas Telford Ltd.

Flyvbjerg, B. (2003). Delusions of Success: Comment on Dan Lovallo and Daniel Kahneman. *Harvard Business Review,* December Issue, pp. 121-122. USA.

Flyvbjerg, B. Bruzelius, N. Rothengatter, W. (2003). *Megaprojects and Risk – An anatomy of Ambition*. Cambridge Uni Press, 1st print in 2003; 7th print in 2009.

Flyvbjerg, B. (2006). Five misunderstandings about case-study research. *Qualitative inquiry*, *12*(2), 219-245.

Forum for the Future. (2018). Our Net Positive Approach. Available online: https://www.forumforthefuture.org/net-positive (accessed on 2 April 2020).

Foucault, M (1973). *The Order of Things: An Archaeology of the Human Sciences*. 1966. New York: Vintage.

Freeman, M and Beale, P. (1992). Measuring project success, *Project Management Journal* 1, 8–17.

Frels, R.K.; Onwuegbuzie, A.J. (2013). Administering quantitative instruments with qualitative interviews: A mixed research approach. J. Couns. Dev. 2013, 91, 184–194.

Friedman, V. J. & Rogers, T. (2009). 'There is nothing so theoretical as good action research.' *Action Research*, 7:1, 31-47.

Fukuda-Parr, S., and D. McNeill. (2019). Knowledge and Politics in Setting and Measuring the SDG s: Introduction to Special Issue. *Global Policy* 10: 5–15.

Fuldauer, L. I., Ives, M. C., Adshead, D., Thacker, S., & Hall, J. W. (2019). Participatory planning of the future of waste management in small island developing states to deliver on the Sustainable Development Goals. *Journal of cleaner production*, *223*, 147-162.

Fullan, M. (2005). Leadership & Sustainability: System Thinkers in Action. Thousand Oaks: Corwin Press.

Galli, Alessandro, Gordana Đurović, Laurel Hanscom, and Jelena Knežević. (2018). Think globally, act locally: Implementing the sustainable development goals in Montenegro. *Environmental Science & Policy* 84: 159–69.

Garavan, T.; McGuire, D. (2010). Human resource development and society: Human resource development's role in embedding corporate social responsibility, sustainability, and ethics in organizations. *Adv. Dev. Hum. Resour.* 2010, *12*, 487–507.

Gasparatos, A (2010). Embedded value systems in sustainability assessment tools and their implications. J. Environ. Manag. 2010, 91, 1613–1622.

Geels, F. W. (2002). Technological transitions as evolutionary reconfiguration processes: a multi-level perspective and a case-study. *Research policy*, *31*(8-9), 1257-1274.

Geels, F. W. (2004). From sectoral systems of innovation to socio-technical systems: Insights about dynamics and change from sociology and institutional theory. *Research policy*, *33*(6-7), 897-920.

Geels, F. W., & Schot, J. (2007). Typology of sociotechnical transition pathways. *Research policy*, *36*(3), 399-417.

Gemuenden, Hans. (2016). Project Governance and Sustainability — Two Major Themes in Project Management Research and Practice. Project Management Journal. 47. 3-6.

Giffen, J (2009). The Challenges of Monitoring and Evaluating Programmes. *M&E Paper 5*, INTRAC, December 2009. Accessed on 12 November 2020 at

https://www.intrac.org/wpcms/wp-content/uploads/2016/06/Monitoring-and-Evaluation-Series-Outcomes-Outputs-and-Impact-7.pdf.

Okali, C; Sumberg, J; and Farrington, J. (1994). Farmer Participatory Research: Rhetoric and Reality. IT Publications, London

Glavic, P., & Lukman, R. (2007). Review of sustainability terms and their definitions. Journal of Cleaner Production, 15, 1875–1885.

Global Compact, (2018). United Nations. Reporting on SDGs. Making Global Goals Local Business. Twitter 2018. Available online: https://twitter.com/globalcompact (accessed on 20 August 2019).

Global Infrastructure Hub. Infrastructure Investment need in the Compact with African countries. (2019). Available online:

 $https://outlook.gihub.org/?utm_source=GIHub+Homepage \& utm_medium=Project+$

tile&utm_campaign=Outlook+GIHub+Tile (accessed on 6 June 2020).

Global Infrastructure Outlook. (2019). Infrastructure Investment need in the Compact with African countries, available at:

https://outlook.gihub.org/?utm_source=GIHub+Homepage&utm_medium=Project+tile&utm_ca mpaign=Outlook+GIHub+Tile and accessed on 23 July 2019.

Global Reporting Initiative. (2016). Carrots and Sticks, Global trends in sustainability reporting regulation and policy'. Available online: https://www.carrotsandsticks.net/wp-content/uploads/2016/05/Carrots-Sticks-2016.pdf (accessed on 10 July 2020).

Global Reporting Initiative. (2019). Available online: https://www.globalreporting.org/Pages/default.aspx (accessed on 10 July 2020).

Global Task Force of Local and Regional Governments. (2020). *Towards The Localization of the SDGs – How to accelerate transformative actions in the aftermath of the COVID-19 outbreak*. 4th Report to the 2020 HLPF. Accessed on 23 November 2020 at https://www.global-taskforce.org/sites/default/files/2020-

07/Towards%20the%20Localization%20of%20the%20SDGs.pdf.

Goel, A., Ganesh, L. S., & Kaur, A. (2019). Sustainability integration in the management of construction projects: A morphological analysis of over two decades' research literature. *Journal of Cleaner Production*, 236, 117676.

Goubran, S. (2019). On the role of construction in achieving the SDGs. *Journal of Sustainability Research*, *1*(2), e190020, doi.org/10.20900/jsr20190020.

GRADE Working Group. (2004). Grading quality of evidence and strength of recommendations. *British Medical Journal*, 328(7454), 1490.

Green Construction Board. (2015). *Three Years on Report—Reducing Carbon Reduces Cost*; Leeds City Council: Leeds, UK.

Greenhalgh, Trisha, Charlotte Humphrey, Jane Hughes, Fraser Macfarlane, Ceri Butler, and Ray Pawson. (2009). How do you modernize a health service? A realist evaluation of whole-scale transformation in London. *The Milbank Quarterly* 87: 391–416.

GRI, U., WBCSD (2015). SDG Compass-The guide for business action on the SDGs. SDG Compass. Accessed on 11 August 2020 at https://sdgcompass.org/.

Griggs, D., Stafford-Smith, M., Gaffney, O., Rockström, J., Öhman, M. C., Shyamsundar, P., ... & Noble, I. (2013). Policy: Sustainable development goals for people and planet. Nature, 495(7441), 305.

Grimmer, J. and Stewart, B.M (2013). *Text as data: The promise and pitfalls of automatic content analysis methods for political texts*. Political analysis, 21(3), pp.267-297.

Groves, R. M., Fowler Jr, F. J., Couper, M. P., Lepkowski, J. M., Singer, E., & Tourangeau, R. (2011). *Survey methodology* (Vol. 561). John Wiley & Sons.

Hak, T., S. Janoušková, and B. Moldan. (2016). Sustainable Development Goals: A need for relevant indicators. *Ecological Indicators* 60: 565–73.

Hakkinen, T.; Belloni, K. Barriers and drivers for sustainable building. *Build. Res. Inf.* 2011, *39*, 239–255.

Hall, J. W. Tran, M. Hickford, A. J. Nicholls, R. J. (2016). The Future of National Infrastructure: A System of Systems Approach. Cambridge University Press, Cambridge.

Hall, R. P., Ranganathan, S., & GC, R. K. (2017). A general micro-level modeling approach to analyzing interconnected SDGs: achieving SDG 6 and more through multiple-use water services (MUS). *Sustainability*, *9*(2), 314.

Handel, M. I. (Ed.). (1986). *Clausewitz and modern strategy* (Vol. 9, No. 1). Psychology Press. Hartshorn, J., Maher, M., Crooks, J., Stahl, R., & Bond, Z. (2005). Creative destruction: Building toward sustainability. *Canadian Journal of Civil Engineering*, 32, 170–180.

Hayward, R., J. Lee, J. Keeble, R. McNamara, C. Hall, S. Cruse, P. Gupta, and E. Robinson. (2013). The UN global compact-accenture CEO study on sustainability 2013. UN Global Compact *Reports* 5: 1–60.

Hayward, K. (2019). UK utility activity through the SDG lens, *The Source - Magazine of the International Water Association*, https://www.thesourcemagazine.org/uk-utility-activity-through-the-sdg-lens/.

Head, B. W., & Alford, J. (2015). Wicked problems: Implications for public policy and management. *Administration & Society*, 47(6), 711-739.

Heravi, G., Fathi, M., & Faeghi, S. (2017). Multi-criteria group decision-making method for optimal selection of sustainable industrial building options focused on petrochemical projects. *Journal of Cleaner Production*, *142*, 2999-3013.

HM Treasury. The Green Book—Central Government Guidance on Appraisal and Evaluation. (2013). Available online:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/

attachment_data/file/685903/The_Green_Book.pdf (accessed on 6 June 2020).

Holden, E., K. Linnerud, and D. Banister. (2017). The imperatives of sustainable development. *Sustainable Development* 25: 213–26.

Holling, C. S. (2001). Understanding the complexity of economic, ecological, and social systems. *Ecosystems*, 4(5), 390-405.

Horrocks, Ivan, and Leslie Budd. (2015). Into the void: A realist evaluation of the eGovernment for You (EGOV4U) project. *Evaluation* 21: 47–64.

Howe, Neil, and William Strauss. (1991). *Generations: The History of America's Future, 1584 to 2069*. New York: William Morrow & Company, p. 538.

Hubbard, G. (2009). Measuring organizational performance: beyond the triple bottom line. *Business Strategy and the Environment* 18: 177–91.

Hwang, B. G., & Ng, W. J. (2013). Project management knowledge and skills for green construction: Overcoming challenges. *International Journal of Project Management*, 31(2), pp. 272-284.

Infrastructure and Projects Authority (2017). Transforming Infrastructure Performance. https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file /664920/transforming_infrastructure_performance_web.pdf (Accessed 15 March 2020).

Infrastructure and Projects Authority. (2020). The 2020 Annual Report on the Government Major Projects Portfolio. Available online: https://www.gov.uk/government/publications/infrastructure-and-projects-authority-annual-report-2020 (accessed on 6 August 2020).

Infrastructure Sustainability Council of Australia (ISCA) (2019). Available online: https://www.isca.org.au/is_ratings (accessed on 24 March 2019).

Institution of Civil Engineers. (2018). *Project 13* Available online: http://www.p13.org.uk/wp-content/uploads/2018/06/P13-Blueprint-Web.pdf (accessed on 2 April 2020). *Blueprint and Commercial handbook*. Institution of Civil Engineers. May 2018, London.

Institution of Civil Engineers, (2019) https://www.ice.org.uk/news-and-insight/latest-ice-news/gec-what-next (Accessed 14 March 2020).

Inter-Agency and Expert Group on SDG Indicators (IAEG-SDGs). (2017). Resolution Adopted by the General Assembly on Work of the Statistical Commission Pertaining to the 2030 Agenda for Sustainable Development (A/RES/71/313). Available online: https://undocs.org/A/RES/71/313 (accessed on 2 April 2019).

Inter-American Development Bank, IDB. What is Sustainable Infrastructure? A Framework to Guide Sustainability Across the Project Cycle. (2018). Available online: https://publications.iadb.org/publications/english/document/What is Sustainable Infrastructure

A Framework_to_Guide_Sustainability_Across_the_Project_Cycle.pdf (accessed on 10 July 2020).

International Fund for Agricultural Development, (2002). A Guide for Project M&E. Managing for Impact in Rural Development.

http://www.depocen.org/ckfinder/userfiles/files/publications/Books/M%26E%20Manual_Eng.pd f (Accessed 1 March 2020).

International Organisation for Standardization (ISO). ISO 14000 Family—Environmental Management. 2019. Available online: https://www.iso.org/iso-14001-environmental-

management.html and https://www.iso.org/iso-45001-occupational-health-and-safety.html (accessed on 21 March 2019).

IPCC. (2018). Global Warming of 1.5 °C, an IPCC Special Report on the Impacts of Global Warming of 1.5 °C Above Pre-Industrial Levels and Related Global Greenhouse Gas Emission Pathways, in the Context of Strengthening the Global Response to the Threat of Climate Change, Sustainable Development, and Efforts to Eradicate Poverty. Geneva: IPCC.

Jacobs, S., Burkhard, B., Van Daele, T., Staes, J. and Schneiders, A (2015). *The Matrix Reloaded: a review of expert knowledge use for mapping ecosystem services*. Ecological Modelling, 295, pp.21-30.

Jama, A. A., & Mourad, K. A. (2019). Water Services Sustainability: Institutional Arrangements and Shared Responsibilities. *Sustainability*, 11(3), 916.

Jarvis, C. B., MacKenzie, S. B., & Podsakoff, P. M. (2003). A critical review of construct indicators and measurement model misspecification in marketing and consumer research. *Journal of consumer research*, *30*(2), 199-218.

Jenner, S. (2016). *Transforming government and public services: realising benefits through project portfolio management.* CRC Press.

Jones, P., and D. Comfort. (2020). A commentary on the localisation of the sustainable development goals. *Journal of Public Affairs* 20: e1943.

Joyce, A., and R. L. Paquin. (2016). The triple layered business model canvas: A tool to design more sustainable business models. *Journal of Cleaner Production* 135: 1474–86.

Kallio, H.; Pietilä, A.M.; Johnson, M.; Kangasniemi, M. (2016). Systematic methodological review: Developing a framework for a qualitative semi-structured interview guide. *J. Adv. Nurs.* 2016, *72*, 2954–2965.

Kaplan, R. S., and D. P. Norton. (1996). *Using the Balanced Scorecard as a Strategic Management System*. Boston: Harvard Business School.

Kapos, V., A. Balmford, R. Aveling, P. Bubb, P. Carey, A. Entwistle, and M. Walpole. (2009). *Outcomes, not implementation, predict conservation success*. Oryx 43: 336-342.

Kashan, A. J., & Mohannak, K. (2014). A conceptual analysis of strategic capability development within product innovation projects. *Prometheus*, *32*(2), pp. 161-180.

Kazi, M., Blom, B., Moren, S., Perdal, A.-L. & Rostila, I. (2002). 'Realist evaluation for practice in Sweden, Finland and Britain.' *Journal of Social Work Research and Evaluation*, 3:2, 171-86. Kazi, M. A. F. (2003). *Realist Evaluation in Practice: Health and Social Work*. SAGE Publications.

Keeys, L. A., & Huemann, M. (2017). Project benefits co-creation: Shaping sustainable development benefits. *International Journal of Project Management*, *35*(6), 1196-1212. Kempster, S. & Perry, K. (2014). 'Critical Realism and Grounded Theory.' In P. Edwards, J. Kibert, C. J. (2013). Sustainable construction: Green building design and delivery, 3rd ed. Hoboken, NJ: Wiley.

Kim, D. Y. (2019). A Design Methodology Using Prototyping Based on the Digital-Physical Models in the Architectural Design Process. *Sustainability*, *11*(16), 4416.

Kirkpatrick, I., Ackroyd, S. & Walker, R. (2005). *The New Managerialism and Public Service Professions: Change in Health, Social Services and Housing.* Palgrave Macmillan.

Klaufus, C., Van Lindert, P., Van Noorloos, F., & Steel, G. (2017). All-inclusiveness versus exclusion: Urban project development in Latin America and Africa. *Sustainability*, *9*(11), 2038.

Klopp, J. M., & Petretta, D. L. (2017). The urban sustainable development goal: Indicators, complexity and the politics of measuring cities. *Cities*, 63, 92-97.

Kolb, D. A. (1976). The Learning Style Inventory: Technical Manual. Boston, MA: McBer.

Kolb, D.A. (1981). Learning styles and disciplinary differences, in: A.W. Chickering (Ed.) *The Modern American College* (pp. 232–255). San Francisco, LA: Jossey-Bass.

Kolb, D. A. (1984). *Experiential learning: Experience as the source of learning and development* (Vol. 1). Englewood Cliffs, NJ: Prentice-Hall.

Kolb, D. A., & Fry, R. (1975). Toward an applied theory of experiential learning. In C. Cooper (Ed.), *Studies of group process* (pp. 33–57). New York: Wiley.

Kolb, D. A., Rubin, I. M., & McIntyre, J. M. (1984). Organizational psychology: readings on human behavior in organizations. Englewood Cliffs, NJ: Prentice-Hall.

Kotter, J.P. Leading Change; Harvard Business Press: Brighton, MA, USA, 2012.

Kramer, M. R. (2007). Strategy and Society: The Link Between Competitive Advantage and Corporate Social Responsibility. *Harvard Business Review*, Volume December.

Krippendorff, K (2018). Content analysis: An introduction to its methodology. First published 2004. Sage publications.

KPMG, Blasco, J.L. and King, A. (2017). *The road ahead: the KPMG survey of corporate responsibility reporting 2017*. KPMG International, Zurich. https://assets.kpmg/content/dam/kpmg/xx/pdf/2017/10/executive-summary-the-kpmg-survey-of-corporate-responsibility-reporting-2017.pdf (Accessed 11 March 2020).

Labuschagne, C., & Brent, A. C. (2005). Sustainable project life cycle management: the need to integrate life cycles in the manufacturing sector. *International Journal of Project Management*, 23(2), pp. 159-168.

Lavagnon, A. Ika. (2009). Project success as a topic in project management journals. *Project Management Journal*, 40(4).

Leal Filho, Walter. (2009). Professionals' Perspectives of Corporate Social Responsibility. Springer-Verlag Berlin Heidelberg.

Lee, A. (2008). "How Are Doctoral Students Supervised? Concepts of Research Supervision." *Studies in Higher Education* 33 (3): 267–281. doi:10.1080/03075070802049202.

Lenth, Russell V. (2001). Some practical guidelines for effective sample size determination. *The American Statistician* 55: 187–93.

Liberati, A., Altman, D. G., Tetzlaff, J., Mulrow, C., Gøtzsche, P. C., Ioannidis, J. P., ... & Moher, D. (2009). The PRISMA statement for reporting systematic reviews and meta-analyses of studies that evaluate health care interventions: explanation and elaboration. *Annals of internal medicine*, *151*(4), W-65.

Lim, S. S., K. Allen, Z. A. Bhutta, L. Dandona, M. H. Forouzanfar, N. Fullman, P. W. Gething, E. M. Goldberg, S. I. Hay, M. Holmberg, and et al. (2016). Measuring the health-related Sustainable Development Goals in 188 countries: A baseline analysis from the Global Burden of Disease Study 2015. *The Lancet* 388: 1813–50.

Lindqvist, S. (1979). 'Dig where you stand.' Oral History, 24-30.

Linsley, Paul, David Howard, and Sara Owen. (2015). The construction of context-mechanismsoutcomes in realistic evaluation. *Nurse Researcher* 22: 28–34.

Liu, L. Y., Zheng, B. H., & Bedra, K. B. (2018). Quantitative analysis of carbon emissions for new town planning based on the system dynamics approach. *Sustainable Cities and Society*, *42*, 538-546.

Lucas, P. L., Hilderink, H. B., Janssen, P. H., Samir, K. C., van Vuuren, D. P., & Niessen, L. (2019). Future impacts of environmental factors on achieving the SDG target on child mortality—A synergistic assessment. *Global Environmental Change*, *57*, 101925.

Lundquist, K. J., & Trippl, M. (2013). Distance, proximity and types of cross-border innovation systems: A conceptual analysis. *Regional Studies*, 47(3), pp. 450-460.

Makino, T., Noda, K., Keokhamphui, K., Hamada, H., Oki, K., & Oki, T. (2016). The Effects of Five Forms of Capital on Thought Processes Underlying Water Consumption Behavior in Suburban Vientiane. *Sustainability*, *8*(6), 538.

Malhi, Y., Aragao, L.E.O., Metcalfe, D.B., Paiva, R., Quesada, C.A., Almeida, S., Anderson, L., Brando, P., Chambers, J.Q., Da Costa, A.C. and Hutyra, L.R. (2009). *Comprehensive assessment of carbon productivity, allocation and storage in three Amazonian forests*. Global Change Biology, 15(5), pp.1255-1274.

Malhi, Yadvinder, Luiz Eduardo O. C. Aragão, Daniel B. Metcalfe, Romilda Paiva, Carlos A. Quesada, Samuel Almeida, Liana Anderson, Paulo Brando, Jeffrey Q. Chambers, Antonio C. L. Da Costa, and et al. (2009). Comprehensive assessment of carbon productivity, allocation and storage in three Amazonian forests. *Global Change Biology* 15: 1255–74.

Mansell, P. Quantitative Survey Analysis: *What Engineers and CEOs Currently Think about Sustainability and the SDGs.* (2018). Available online: https://www.researchgate.net/publication/327602328 Engineers perception of value of SDGs and the current ability to measure projects' SDG impact (accessed on 6 June 2020).

Mansell, P., Philbin, S.P., Plodowski, A. (2019) Why Project Management is Critical to Achieving the SDGs, and How This Can be Achieved. Available online: https://pmcongress2019.org/ (accessed on 6 June 2020).

Mansell, P., Philbin, S.P., Konstantinou, E. (2019) 'Call to Arms': Using the Creating Shared Value Business Governance Paradigm to Deliver Projects' Business-Society Impact Against the UN SDG 2030 Targets. *Proceedings of the EURAM (European Academy of Management) 2019 Conference*, Lisbon, Portugal. Available online: http://www.euram-online.org/annual-conference-2019.html. (accessed on 6 June 2020).

Mansell, P., Philbin, S.P., and Boyd, T. (2019). Infrastructure Projects' Impact on Sustainable Development – Case Study of a Water-Utility Company. In the *Proceedings of 7th IPMA Research Conference and the 14th International OTMC Conference*. 4-7 September 2019. Zagreb, Croatia, pp 570-592.

Mansell, P., Philbin, S.P., Broyd, T., Nicholson, I. (2020). Assessing the Impact of Infrastructure Projects on Global Sustainable Development Goals, *Proceedings of the Institution of Civil Engineers: Engineering Sustainability*, doi.org/10.1680/jensu.19.00044. Thomas Telford Ltd.: London, UK, 2019; pp. 1–17.

Mansell, P., Philbin, S.P. (2020). Measuring Sustainable Development Goal Targets on Infrastructure Projects. *J Modern Project Manag.* 2020, *8* Available online: https://openresearch.lsbu.ac.uk/item/8q171 (accessed on 6 June 2020)

Mansell, P., Philbin, S.P., Broyd, T. (2020). Development of a New Business Model to Measure Organizational and Project-Level SDG Impact—Case Study of a Water Utility Company. *Sustainability*. 2020, *12*, 6413.

Mansell, P., Philbin, S.P., Konstantinou, E. (2020a). Redefining the Use of Sustainable Development Goals at the Organisation and Project Levels—A Survey of Engineers. *Administrative Sciences*. 2020, *10*, 55, doi:10.3390/admsci10030055.

Mansell, P.M., Philbin, S.P., Konstantinou, E. (2020b). Delivering UN Sustainable Development Goals' Impact on Infrastructure Projects: An Empirical Study of Senior Executives in the UK Construction Sector. *Sustainability*, 12(19), 7998.

Mansell, P.M., Philbin, S.P., Van Rooyen, D., Sabini, L. (2020c). 'A Systematic Literature Review of Infrastructure Projects - Assessment Through SDG Targets: Towards a Comprehensive Framework', Submitted to *EPOC (Engineering Project Organization Society)* on 2 June 2020 - Paper accepted for presentation on 23 Oct 2020.

Marchal, B., van Belle, S., van Olmen, J., Hoerée, T. & Kegels, G. (2012). 'Is realist evaluation keeping its promise? A review of published empirical studies in the field of health systems research.' Evaluation, 18:2, 192-212.

Marnewick, C. (2016). Benefits of information system projects: The tale of two countries. *International Journal of Project Management*, 34(4), 748-760.

Márquez, A. J. C., Cassettari Filho, P. C., Rutkowski, E. W., & de Lima Isaac, R. (2019). Landfill mining as a strategic tool towards global sustainable development. *Journal of Cleaner Production*, *226*, 1102-1115.

Martens, M.L.; Carvalho, M. (2016a). The challenge of introducing sustainability into project management function: multiple-case studies. *J. Clean. Prod.* 2016, *117*, 29–40, doi:10.1016/j.jclepro.2015.12.039.

Martens, M. L., & Carvalho, M. M. (2016b). Sustainability and Success Variables in the Project Management Context: An Expert Panel. *Proj. Manag. J.* 2016, *47*, 24–43, doi:10.1177/875697281604700603.

Martens, M. L., & Carvalho, M. M. (2016c). Key factors of sustainability in project management context: A survey exploring the project managers' perspective. International Journal of Project Management. Volume 35, Issue 6, 2017, Pages 1084-1102.

May, T. ed., (2002). Qualitative research in action. Sage.

Matisoff, D.C., Noonan, D.S. and O'Brien, J.J (2013). *Convergence in environmental reporting: assessing the Carbon Disclosure Project*. Business Strategy and the Environment, 22(5), pp.285-305.

McGregor, I.M. and Roberts, C (2003). Using the SPeARTM assessment tool in sustainable master planning. In Proceedings of US Green Building Conference, Pittsburgh, PA, USA. US Green Building Council, Pittsburgh, PA, USA.

McKinsey Global Survey result. (2011). *The business of sustainability*. Available at https://www.mckinsey.com/~/media/McKinsey/dotcom/client_service/Sustainability/PDFs/McK %20on%20SRP/SRP_11_Biz%20sustainability.ashx and accessed on 12 October 2018.

McNeilly, M. (2003). *Sun Tzu and the art of modern warfare*. Oxford University Press on Demand.

McNeilly, M., & McNeilly, M. R. (2012). Sun Tzu and the art of business: Six strategic principles for managers. OUP USA.

Mebratu, D. (1998). Sustainability and sustainable development: historical and conceptual review. Environmental impact assessment review, 18(6), pp.493-520.

Menhas, R., Mahmood, S., Tanchangya, P., Safdar, M. N., & Hussain, S. (2019). Sustainable Development under Belt and Road Initiative: A Case Study of China-Pakistan Economic Corridor's Socio-Economic Impact on Pakistan. *Sustainability*, *11*(21), 6143.

Merriam, S.B.; Grenier, R.S. *Qualitative Research in Practice: Examples for Discussion and Analysis.* John Wiley & Sons: Hoboken, NJ, USA, 2019.

Merry, Sally Engle. (2019). The Sustainable Development Goals Confront the Infrastructure of Measurement. *Global Policy* 10: 146–48.

Mertz, D.R. (2005). *Grand Challenges: A Strategic Plan for Bridge Engineering*, NCHRP Project 20-07/Task 199, Transportation Research Board, Washington, DC.

Metcalf, Louise, and Sue Benn. (2013). Leadership for sustainability: An evolution of leadership ability. *Journal of Business Ethics* 112: 369–84.

Micheli, Pietro, and Jean-Francois Manzoni. (2010). Strategic performance measurement: Benefits, limitations and paradoxes. *Long Range Planning* 43: 465–76.

Millar, R. and Hall, K. (2013). Social return on investment (SROI) and performance measurement: The opportunities and barriers for social enterprises in health and social care. *Public Management Review*, 15(6), pp.923-941.

Mingers, J. (2006). *Realising Systems Thinking: Knowledge and Action in Management Science*. Springer US.

Mische, A (2014). *Measuring futures in action: Projective grammars in the Rio+20 debates*. Theory and Society. 43. 437-464. 10.1007/s11186-014-9226-3.

Monteiro, N. B. R., Moita Neto, J. M., & da Silva, E. A. (2018). Bibliometric study of the crushed stone mining sector. *Mineral Processing and Extractive Metallurgy Review*, *39*(4), 235-249.

Moore, J. E., Mascarenhas, A., Bain, J., & Straus, S. E. (2017). Developing a comprehensive definition of sustainability. Implementation Science, 12(1), 110.

Morgan, J. & Olsen, W. (2008). 'Defining Objectivity in Realist Terms: Objectivity as a Second-Order 'Bridging'Concept Part II: Bridging to Praxis.' *Journal of Critical Realism*, 7:1, 107-32. Morris, P. (2013). Reconstructing Programme Management. Chichester: John Wiley & Sons. UK.

Morris, P. (2017). Climate Change and what the project management profession should be doing about it. Association for Project Management. Available at

https://www.apm.org.uk/media/7496/climate-change-report.pdf and accessed on 11 May 2018.

Morse, S. 2013. *Indices and Indicators in Development: An Unhealthy Obsession with Numbers*. London: Routledge.

Müller, R.; Jugdev, K. (2012). Critical success factors in projects. *Int. J. Manag. Proj. Bus.* 2012, *5*, 757–775.

Müller, R. (2017). Governance and Governmentality for Projects – Enablers, Practices, and Consequences. Routeledge. New York and London.

Müller, R., Drouin, N., & Sankaran, S. (2019). Modelling Organizational Project Management. *Project Management Journal*, *50*(4), 499–513. https://doi.org/10.1177/8756972819847876.

Munyasya, B. M., & Chileshe, N. (2018). Towards sustainable infrastructure development: drivers, barriers, strategies, and coping mechanisms. *Sustainability*, *10*(12), 4341.

Mutch, A. (2007). 'Reflexivity and the institutional entrepreneur: a historical exploration.' *Organization Studies*, 28, 1123-40.

Mutch, A. (2013). 'Sociomateriality—Taking the wrong turning?' *Information and organization*, 23:1, 28-40.

Mutch, A. (2014). 'History and Documents in Critical Realism.' In P. Edwards, J. O'Mahoney & S. Vincent (Eds.) *Studying Organisations Using Critical Realism: A Practical Guide*: 223-40. Oxford: Oxford University Press.

Nardi, P. M. 2015. Doing Survey Research. London: Routledge.

National Audit Office. 2005. *Improving Public Services through better Construction*. Report by the Comptroller and Auditor General HC 364-I Session 2004–2005. London: TSO.

National Infrastructure Commission (2018). *National Infrastructure Assessment*. Accessed report. https://www.nic.org.uk/wp-content/uploads/CCS001_CCS0618917350-001_NIC-

NIA_Accessible.pdf (Accessed 24 March 2020)

Nerini, F.F., Tomei, J., To, L.S., Bisaga, I., Parikh, P., Black, M., Borrion, A., Spataru, C., Broto, V.C., Anandarajah, G. and Milligan, B., (2018). Mapping synergies and trade-offs between energy and the Sustainable Development Goals. *Nature Energy*, 3(1), pp.10-15.

New Climate Economy. (2016). *The Global Commission on the Economy and Climate*. Washington: World Resources Institute. Available online: https://www.deutsches-klima-konsortium.de/fileadmin/user_upload/pdfs/Briefings/Morgan_12_Nov_15.pdf (accessed on 10 July 2020).

Ochieng, E.G.; Price, A.D.F.; Moore, D. (2013). *Management of Global Construction Projects*; Palgrave Macmillan's Global Academic: Hampshire, UK.

Office of National Statistics (ONS). 2019. Available online: https://sustainabledevelopment-uk.github.io/reporting-status/ (accessed on 6 June 2020).

Office of Science and Technology Policy (OSTP) (2013). 21st Century Grand Challenges. Accessed at https://obamawhitehouse.archives.gov/tout/ostp-grand-challenges (Accessed 19 March 2020).

Okali, C; Sumberg, J; and Farrington, J. (1994). *Farmer Participatory Research: Rhetoric and Reality*. IT Publications, London.

Økland, A. (2015). Gap analysis for incorporating sustainability in project management. Procedia Computer Science, 64, 103-109.

Omenn, Gilbert S. (2006). Grand challenges and great opportunities in science, technology, and public policy. Science, 314.5806. 1696-1704.

Organisations for Economic Co-Operation and Development (OECD) (2007). *What is Social Capital?* Paris. Accessed at https://www.oecd.org/insights/37966934.pdf on 20 March 2019. Organisation for Economic Co-Operation and Development (OECD), (2010). *Glossary of Key*

Terms in Evaluations and Results Based Management. OECD, 2002, re-printed in 2010. Organisation for Economic Co-Operation and Development (OECD), (2015). G20/OECD Principles of Corporate Governance. Paris. 2015. Available online: https://www.oecdilibrary.org/governance/g20-oecd-principles-of-corporate-governance-2015 9789264236882-en

(accessed on 20 March 2019). Organisation for Economic Co-Operation and Development (OECD), (2019). Measuring Distance to the SDG Targets 2019: An Assessment of Where OECD Countries Stand. Available at https://read.oecd-ilibrary.org/development/measuring-distance-to-the-sdg-targets-2019 a8caf3fa-en#page1 (accessed on 15 July 2019).

Oxford University School of Geography & Environment (2014). Step by Step Guide to Monitoring and Evaluation.

https://www.geog.ox.ac.uk/research/technologies/projects/mesc/guide-to-monitoring-andevaluation-v1-march2014.pdf Accessed on 4 March 2020).

Patel, Zarina, Saskia Greyling, David Simon, Helen Arfvidsson, Nishendra Moodley, Natasha Primo, and Carol Wright. (2017). Local responses to global sustainability agendas: Learning from experimenting with the urban sustainable development goal in Cape Town. *Sustainability Science* 12: 785–97.

Patton, M.Q. (1988). Reports on Topic Areas: The Evaluator's Responsibility for Utilization. *Am. J. Evaluation* 1988, *9*, 5–24, doi:10.1177/109821408800900201.

Patton MQ. (1996). A world larger than formative and summative. Am. J. Evaluation. 1996, 17, 131–44.

Pawson, R., Tilley, N. (1997). Tilley, N. *Realistic evaluation*. Sage publications: Thousand Oaks, CA, USA, 1997.

Pawson, R., and N. Tilley. (1997). An introduction to scientific realist evaluation. *Evaluation for the 21st century: A handbook* 405–18.

Pawson, R., Tilley, N. (2001). Realistic evaluation bloodlines. Am. J. Evaluation 2001, 22, 317–324

Pawson, R., T. Greenhalgh, G. Harvey, and K. Walshe. (2005). Realist review-a new method of systematic review designed for complex policy interventions. *Journal of Health Services Research* & *Policy* 10 (Suppl. 1): 21–34.

Pearce, O.J., Murry, N.J. and Broyd, T.W (2012). June. *Halstar: systems engineering for sustainable development*. In Proceedings of the Institution of Civil Engineers-Engineering Sustainability (Vol. 165, No. 2, pp. 129-140). Thomas Telford Ltd.

Pengue, W., Muller, A., Sukhdev, P. and Gemmill, H. (2018). A synthesis of the results and recommendations of TEEB for Agriculture and Food's Scientific and Economic Foundations Report. <u>http://teebweb.org/agrifood/wp-content/uploads/2018/Synthesis_report_highres.pdf</u> (Accessed 19 March 2020).

Perrini, F., and A. Tencati. (2006). Sustainability and stakeholder management: The need for new corporate performance evaluation and reporting systems. *Business Strategy and the Environment* 15: 296–308.

Peters, L.D., Pressey, A.D., Vanharanta, M. and Johnston, W.J., (2013). Constructivism and critical realism as alternative approaches to the study of business networks: Convergences and divergences in theory and in research practice. *Industrial Marketing Management*, 42(3), pp.336-346.

Pinkerton WJ, Pinkerton WJ. Project Management: Achieving Project Bottom-line Success. McGraw-Hill Education: New York, NY, USA, 2003.

Pinto, J. K. and Slevin, D. P. (1987). Critical factors in successful programme implementation, *IEEE Transactions on Engineering Management*, 34, 1, pp. 22-27.

PIMCO Investment Management (2019). *Corporate Reporting on SDGs, Mapping a Sustainable Future*. <u>https://www.pimco.co.uk/en-gb/insights/viewpoints/2018/12/corporate-reporting-on-the-un-sdgs-mapping-a-sustainable-future</u> (Accessed 20 March 2020).

PMI Standards Committee, and Project Management Institute. (1996). A Guide to the Project Management Body of Knowledge. Project Management Institute: Newtown Square, PA, USA.

Poirier, E., D. Forgues, and S. Staub-French. (2016). Collaboration through innovation: Implications for expertise in the AEC sector. *Construction Management and Economics* 34: 769–89.

Porter, M.E.(1985). *Creating and Sustaining Superior Performance. Competitive Advantage*; Free Press: New York, NY, USA, 1985; p. 167.

Porter, M. E. and Kramer, M. R. (2011). The Big Idea: Creating Shared Value, Rethinking Capitalism. *Harvard Business Review* Volume January-February 89(1-2).

Prasad, R., C. Dovrolis, M. Murray, and K. C. Claffy. (2003). Bandwidth estimation: Metrics, measurement techniques, and tools. *IEEE Network* 17: 27–35.

Preston, F. (2012). A global redesign?: Shaping the circular economy. London: Chatham House.

Price Waterhouse Coopers, (2016). *Make it Your Business: Engaging with the Sustainable Development Goals.* London: PwC.

Price Waterhouse Coopers, Scott, L., McGill, A. (2018). *From promise to reality: Does business really care about the SDGs?* London: PwC. <u>https://www.pwc.com/gx/en/sustainability/SDG/sdg-reporting-2018.pdf</u> (Accessed 11 March 2020).

Proctor, Enola, Hiie Silmere, Ramesh Raghavan, Peter Hovmand, Greg Aarons, Alicia Bunger, Richard Griffey, and Melissa Hensley. (2011). Outcomes for implementation research: Conceptual distinctions, measurement challenges, and research agenda. *Administration and Policy in Mental Health and Mental Health Services Research* 38: 65–76.

Rainey, H.J., Pollard, E.H., Dutson, G., Ekstrom, J.M., Livingstone, S.R., Temple, H.J. and Pilgrim, J.D. (2015). A review of corporate goals of No Net Loss and Net Positive Impact on biodiversity. *Oryx*, 49(2), pp.232-238.

Rea, L. M., & Parker, R. A. (2014). *Designing and conducting survey research: A comprehensive guide*. John Wiley & Sons.

Reffat, R. (2004). Sustainable construction in developing countries. In Proceedings of the First Architectural International Conference, Cairo University, Cairo, Egypt, 24–26 February 2004 Roche, C (1999). Impact Assessment for Development Agencies. Oxford: Oxfam/NOVIB.

Rockström, J., Steffen, W., Noone, K., Persson, Å., Chapin, F. S., Lambin, E. F., ... & Nykvist, B. (2009). A safe operating space for humanity. *nature*, *461*(7263), 472-475.

Rockström, J (2016). Wedding Cake (Donut) view of the economy and social embedded as parts of the biosphere (Stockholm Resilience Centre) accessed at <u>https://www.stockholmresilience.org/research/research-news/2016-06-14-how-food-connects-all-the-sdgs.html on 20 March 2019</u> on 20 March 2019.

Rossi PH, Lipsey MW, Henry GT. (2018). *Evaluation: A Systematic Approach*. Sage publications: Thousand Oaks, CA, USA

Rubin, A. and Babbie, E.R. (2016). *Empowerment series: Research methods for social work*. Cengage Learning.

Rubin, D.B. (2005). Causal Inference Using Potential Outcomes. J. Am. Stat. Assoc. 2005, 100, 322–331, doi:10.1198/016214504000001880.

Ruddin, L. P. (2006). You can generalize stupid! Social scientists, Bent Flyvbjerg, and case study methodology. *Qualitative inquiry*, *12*(4), 797-812.

Sabini, L., Muzio, D., & Alderman, N. (2017). *Integrating Sustainability into Project Management Practice: The Perspective of Professional Institutions. IRNOP*, April, 2017. https://doi.org/10.5130/pmrp.irnop2017.5661 (Accessed 4 October 2020).

Sabini, L., Muzio, D., & Alderman, N. (2019). 25 years of sustainable projects. What we know and what the literature says. *International Journal of Project Management*, 37(6), 820-838. <u>https://doi.org/10.1016/J.IJPROMAN.2019.05.002</u> (Accessed 4 October 2020).

Sachs, J., Schmidt-Traub, G., Kroll, C., Durand-Delacre, D., & Teksoz, K. (2016). SDG index & dashboards: A global report. Bertelsmann Stiftung.

Sachs, J.; Woo, W.T.; Yoshino, N.; Taghizadeh-Hesary, F (2019). Importance of green finance for achieving sustainable development goals and energy security. In *Handbook of Green Finance: Energy Security and Sustainable Development*, Springer: Singapore, 2019; pp. 3–12.

Sachs, J., Schmidt-Traub, G., Kroll, C., Lafortune, G., Fuller, G., Woelm, F. (2020). *The Sustainable Development Goals and COVID-19. Sustainable Development Report 2020*. Cambridge University Press. Accessed on 11 November 2020 at https://www.sdgindex.org/.

Sakamoto, K (2014). *Toward a sustainability appraisal framework for transport*. Accessed at <u>https://think-asia.org/bitstream/handle/11540/1417/sdwp-031.pdf?sequence=1</u> on 24 March 2019. Sakhrani, V., Chinowsky, P, S., and Taylor, J. (2016). Grand Challenges. The Engineering

Project Organization Journal (August 2017) 7, 1 The Engineering Project Organization Journal Engineering Project Organization Society.

Sánchez, M. A. (2015). Integrating sustainability issues into project management. *Journal of Cleaner Production*, *96*, pp. 319-330.

Sawaf, A and Gabrielle, R. (2014). *Sacred Commerce: A Blueprint for a New Humanity* (2nd Edition). EQ Enterprises. pp. 24–28.

Sayer, Andrew. (2004). Why critical realism? Andrew Sayer. Critical Realist Applications in Organisation and Management Studies 11: 23–42.

Sayer, R. A. 1992. *Method in Social Science: A Realist Approach*. East Sussex: Psychology Press. Saunders, M., Lewis, P., & Thornhill, A. (2009). *Research methods for business students*. Pearson education.

Schaltegger, Stefan, and Roger Burritt. (2000). Contemporary Environmental Accounting: Issues, Concepts and Practice. Sheffield: Greenleaf Publishing, p. 111.

Scheyvens, Regina, Glenn Banks, and Emma Hughes. (2016). The private sector and the SDGs: The need to move beyond 'business as usual'. *Sustainable Development* 24: 371–82.

Schwanitz, V. J., Wierling, A., & Shah, P. (2017). Assessing the impact of renewable energy on regional sustainability—A comparative study of Sogn og Fjordane (Norway) and Okinawa (Japan). *Sustainability*, *9*(11), 1969.

Scott, L.; McGill, A. From Promise to Reality: Does Business Really Care about the SDGs? London: PwC. (2018). Available online: https://www.pwc.com/gx/en/sustainability/SDG/sdg-reporting-2018.pdf (accessed on 10 July 2020).

Seinfeld, J. H., and Pandis, S. N. (2016). Atmospheric chemistry and physics: from air pollution to climate change. John Wiley & Sons.

Sergi, B.S.; Popkova, E.G.; Borzenko, K.V.; Przhedetskaya, N.V. (2019). Public–Private Partnerships as a Mechanism of Financing Sustainable Development. In *Financing Sustainable Development*; Palgrave Macmillan: Cham, Switzerland, 2019; pp. 313–339.

Shen, L., Tam, V. W. Y., Tam, L., & Ji, Y. (2010). Project feasibility study: The key to successful implementation of sustainable and socially responsible construction management practice. Journal of Cleaner Production, 18(3), 254–259.

Shivakumar, S., Pedersen, T., Wilkins, S. and Schuster, S (2014). *Envision: A Measure of Infrastructure Sustainability*. In Pipelines 2014: From Underground to the Forefront of Innovation and Sustainability (pp. 2249-2256).

Siew, R., Balatbat, M. and Carmichael, D (2013). *A review of building/infrastructure sustainability reporting tools (SRTs). Smart and Sustainable Built Environment*, 2(2), pp.106-139.

Silvius, G., Schipper, R., Planko, J., van der Brink, J., & Köhler, A. (2012). Sustainability in project management. Surrey, Gower Publishing Limited.

Silvius, A. G., & Schipper, R. P. (2014). Sustainability in project management: A literature review and impact analysis. *Social Business*, 4(1), 63-96.

Silvius, A.G., Kampinga, M., Paniagua, S. and Mooi, H., (2017). Considering sustainability in project management decision making; An investigation using Q-methodology. *International Journal of Project Management*, 35(6), pp.1133-1150.

Steffen, Will, Katherine Richardson, Johan Rockström, Sarah E. Cornell, Ingo Fetzer, Elena M. Bennett, Reinette Biggs, Stephen R. Carpenter, Wim De Vries, Cynthia A. De Wit, et al. (2015). Planetary boundaries: Guiding human development on a changing planet. *Science* 347: 1259855. Singleton R. A., and B. C. Straits. 2010. *Approaches to Social Research*. New York: Oxford University Press.

Sosik, John J., and Dongil Jung. (2018). Full Range Leadership Development: Pathways for People, Profit, and Planet. London and New Tork: Routledge.

Stein, D., and C. Valters. (2012). Understanding theory of change in international development. Available online: http://www.theoryofchange.org/wpcontent/uploads/toco_library/pdf/UNDERSTANDINGTHEORYOFChangeSteinValtersPN.pdf (accessed on 7 August 2020).

Sperling, J., Romero-Lankao, P., & Beig, G. (2016). Exploring citizen infrastructure and environmental priorities in Mumbai, India. *Environmental Science & Policy*, *60*, 19-27.

Subedi, D. (2016). Explanatory sequential mixed method design as the third research community of knowledge claim. *Am J Educ Res*, 4(7), 570-577.

Suess, Erwin. (1980). Particulate organic carbon flux in the oceans—Surface productivity and oxygen utilization. *Nature* 288: 260.

Švajlenka, J.; Kozlovská, M.; Pošiváková, T. (2018). Analysis of Selected Building Constructions Used in Industrial Construction in Terms of Sustainability Benefits. *Sustainability* 2018, *10*, 4394. Švajlenka, J.; Kozlovská, M. (2018). Perception of User Criteria in the Context of Sustainability of Modern Methods of Construction Based on Wood. *Sustainability* 2018, *10*, 116.

Sverdrup, H.; Rosen, K. (1998). Long-term base cation mass balances for Swedish forests and the concept of sustainability. *For. Ecol. Manag.* 1998, *110*, 221–236.

Swain, Ranjula Bali. (2018). A Critical Analysis of the Sustainable Development Goals. In *Handbook of Sustainability Science and Research*. Edited by Leal Filho W. World Sustainability Series. Cham: Springer.

Tansey, Oisín. (2007). Process tracing and elite interviewing: A case for non-probability sampling. *PS: Political Science and Politics* 40: 765–72.

Teferi, Z. A., & Newman, P. (2018). Slum upgrading: Can the 1.5° C carbon reduction work with SDGs in these settlements? *Urban Planning*, 3(2).

Terrapon-Pfaff, J., Ortiz, W., Dienst, C., & Gröne, M. C. (2018). Energising the WEF nexus to enhance sustainable development at local level. *Journal of environmental management*, 223, 409-416.

Thacker, S. Hall, J. (2018). Engineering for Sustainable Development. Infrastructure Transition Research Consortium (ITRC), University of Oxford.

Thacker, S., Hall, J.W., Adshead, D., O'Regan, N., Rozenberg, J., Hallegatte, S., Fay, M., Harvey, M., Meller, H., and Watkins, G. (2019). Infrastructure for sustainable development. Nature Sustainability.

The Global Commission on The Economy and Climate. Washington DC: World ResourcesInstitute.(2014).Availableonline:https://www.deutsches-klima-konsortium.de/fileadmin/userupload/pdfs/Briefings/

Morgan 12 Nov 15.pdf (accessed on 10 July 2020).

Themistocleous, G., & Wearne, S. H. (2000). Project management topic coverage in journals. *International Journal of Project Management*, 18, 7–11.

Thiry, M. (2004). Value Management. In Morris P and Pinto J (Eds), Wiley Guide to Managing Projects, Wiley. Holbroken, NJ.

Tilley, Nick. (2016). EMMIE and engineering: What works as evidence to improve decisions? *Evaluation* 22: 304–22.

Tilt, C. A. (2007). *Corporate Responsibility Accounting and Accountants*. Idowu, Samuel O.; Tukey, John W (1977). *Exploratory Data Analysis*. Addison-Wesley. <u>ISBN 0-201-07616-0</u>. <u>OCLC 3058187</u>.

United Nations. (2015). *Transforming Our World: The 2030 agenda for sustainable development,* Resolution Adopted by the General Assembly. United Nations, New York. 2015. Torraco, R. J. (2005). Writing integrative literature reviews: Guidelines and examples. *Human resource development review, 4*(3), 356-367.

Tranfield, D., Denyer, D., & Smart, P. (2003). Towards a methodology for developing evidenceinformed management knowledge by means of systematic review. *British journal of management*, 14(3), 207-222.

Ulrich, K. T., Eppinger, S. D., & Alvarez, R. V. M. (2004). *Diseño y desarrollo de productos: enfoque multidisciplinario*. McGraw-Hill.

United Nations. (2015). Transforming Our World: The 2030 agenda for sustainable development, Resolution Adopted by the General Assembly. United Nations, New York. 2015.

United Nations Development Programme (UNDP) and Infrastructure Transition Research Consortium (ITRC). (2009). Handbook on Planning, Monitoring and Evaluating for Development. <u>http://web.undp.org/evaluation/handbook/documents/english/pme-handbook.pdf</u> (Accessed 12 March 2020).

United Nations Intergovernmental Panel on Climate Change (IPCC). (2018). Revised January 2019. *IPCC special report in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty*. Accessed at https://www.ipcc.ch/site/assets/uploads/sites/2/2018/07/SR15_SPM_version_stand_alone_LR.pd f accessed on 19 March 2019

United Nations Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services (2019). *Report of the Plenary of the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services on the work of its seventh session*. Available at https://www.ipbes.net/system/tdf/ipbes_7_10_add-1-

advance 0.pdf?file=1&type=node&id=35245 (accessed on 15 July 2019).

United Nations. (2018). *The Sustainable Development Goals Report 2018*. Accessed online: https://unstats.un.org/sdgs/files/report/2018/TheSustainableDevelopmentGoalsReport2018-EN.pdf (accessed on 15 July 2019).

United Nations Office for Project Services (UNOPS). (2017). Importance of Infrastructure for Development. UNOPS, Copenhagen, Denmark.

United Nations Office for Project Services (UNOPS). (2018). *Infrastructure: Underpinning Sustainable Development*. UNOPS, Copenhagen, Denmark. <u>https://www.itrc.org.uk/wp-content/PDFs/ITRC-UNOPS-Infrastructure_Underpining_Sustainable%20Development.pdf</u> (Accessed 19 March 2020).

United Nations. (2019). *World Population Prospects*. Available online: http://esa.un.org/unpd/wpp/Publications/Files/Key_Findings_WPP_2015.pdf (accessed on 6 June 2020).

US Public Interest Research Group. 2(016). Available online: https://uspirg.org/reports/usp/millennials-motion (accessed on 2 April 2020).

Vaske, J. J. (2019). *Survey research and analysis*. Sagamore-Venture. 1807 North Federal Drive, Urbana, IL 61801.

Vincent, S., & O'Mahoney, J. (2018). Critical realism and qualitative research: An introductory overview. *The Sage handbook of qualitative business and management research methods: History and traditions*, 201-216.

Weiss, C.H. (1983). The stakeholder approach to evaluation: Origins and promise. *New Dir. Program Evaluation* 1983, *1983*, 3–14, doi:10.1002/ev.1322.

Weiss, C.H. (1995). Nothing as practical as good theory: Exploring theory-based evaluation for comprehensive community initiatives for children and families. In *New Approaches to Evaluating Community Initiatives: Concepts, Methods, and Contexts*; The Aspen Institute: Washington, DC, USA, 1995, Volume 1, pp. 65–92.

Weiss, C.H. (1998a). *Evaluation: Methods for Studying Programs and Policies*, 2nd ed; Prentice Hall: Upper Saddle River, NJ, USA, 1998.

Weiss, C.H. (1998b). Have we learned anything new about the use of evaluation? Am. J. Evaluation 1998, 19, 21-33.

Weiss, C.H. (2018). Theory-Based Evaluation: Theories of Change for Poverty Reduction Programs. *Eval. Poverty Reduct.* 2018, 103–112, doi:10.4324/9781351325325-16.

Whelan, T. Fink, C. (2016). The Comprehensive Business Case for Sustainability.

Wong, K. K. (2013). Partial least squares structural equation modelling (PLS-SEM) techniques using SmartPLS. *Marketing Bulletin*, 24(1), 1-32.

Yamey, B. S. (1949). Scientific bookkeeping and the rise of capitalism. *The Economic History Review*, 1(2-3), 99-113.

Zachariadis, Markos, Susan Scott, and Michael Barrett. (2013). Methodological implications of critical realism for mixed-methods research. *MIS Quarterly* 2013: 855–79.

Zahra, Shaker A., Harry J. Sapienza, and Per Davidsson. (2006). Entrepreneurship and dynamic capabilities: A review, model and research agenda. *Journal of Management Studies* 43: 917–55.

Zhang, Q., Liu, S., Wang, T., Dai, X., Baninla, Y., Nakatani, J., & Moriguchi, Y. (2019). Urbanization impacts on greenhouse gas (GHG) emissions of the water infrastructure in China: Trade-offs among sustainable development goals (SDGs). *Journal of Cleaner Production*, 232, 474-486.

Zuofa, T., & Ochieng, E. (2016). Sustainability in construction project delivery: A study of experienced project managers in Nigeria. *Project Management Journal*, 47(6), 44-55.

Appendices

Appendix 1 - Table of SLR data across the MISI Research Thematics

The main data points from the SLR analysis have been collated into the table below that summarises the key thematics across the seven codified areas. This provides a high-level overview of the focus areas of the selected dataset.

Combined Search Keywords	Authors	Article Title	Type of Contribution	Primary Research Design	Primary Geographical Focus	Infrastructure Sector Focus	Industry / Narrowed Sector	Level	SDGs Mentioned
SDG & Project	Hall R.P., Ranganathan S., Raj Kumar G.C.	A general micro-level modelling approach to analyzing interconnected SDGs: Achieving SDG 6 and more through multiple-use water services (MUS)	Framework proposition	Multiple case study	Mozambique	Water	Rural water services	Project; Organisational	SDG 6
SDG & Project	Klaufus C., van Lindert P., van Noorloos F., Steel G.	All-inclusiveness versus exclusion: Urban project development in Latin America and Africa	Framework proposition	Multiple case study	Latin America & Africa	Urban development	N/A	Local; Sectoral	SDG 11
SDG & Project	Farinosi F., Giupponi C., Reynaud A., Ceccherini G., Carmona- Moreno C., De Roo A., Gonzalez- Sanchez D., Bidoglio G.	An innovative approach to the assessment of hydro- political risk: A spatially explicit, data driven indicator of hydro-political issues	Framework proposition	Empirical analysis	Global	Water	Trans- boundary resources	Global	SDG 6.5.2

Table 38: Consolidated table of data across the seven MISI Thematics

Combined Search Keywords	Authors	Article Title	Type of Contribution	Primary Research Design	Primary Geographical Focus	Infrastructure Sector Focus	Industry / Narrowed Sector	Level	SDGs Mentioned
SDG & Project	Dean K., Trillo C., Bichard E.	Assessing the value of housing schemes through sustainable return on investment: A path towards sustainability-led evaluations?	Framework testing	Multiple case study	United Kingdom	Housing	Social housing	Project	SDG 11
SDG & Project	Goel A., Ganesh L.S., Kaur A.	Deductive content analysis of research on sustainable construction in India: current progress and future directions	Framework proposition	Systematic literature review	India with developing economy reach	Not specified	Construction	Industry	General
SDG & Project	Heravi G., Fathi M., Faeghi S.	Evaluation of sustainability indicators of industrial buildings focused on petrochemical projects	Theoretical	Empirical analysis	Iran	Buildings	Industrial (Petro- chemical)	Industry	General
SDG & Project	Ding X., Zhou C., Mauerhofer V., Zhong W., Li G.	From environmental soundness to sustainable development: Improving applicability of payment for ecosystem services scheme for diverting regional sustainability transition in developing countries	Framework proposition	Single case study	China	Water	Water transfer	Regional; National	General
SDG & Project	Lucas P.L., Hilderink H.B.M., Janssen P.H.M., KC S., van Vuuren D.P., Niessen L.	Future impacts of environmental factors on achieving the SDG target on child mortality—A synergistic assessment	Framework testing	Empirical analysis	Global	Health, Water, Energy Nexus	N/A	Global	General
SDG & Project	Ishikawa Y., Murata M., Kawaguchi T.	Globally applicable water quality simulation model for river basin chemical risk assessment	Framework testing	Single case study	Japan	Water & Sanitation	River basins; Water quality	Organisational	General

Combined Search Keywords	Authors	Article Title	Type of Contribution	Primary Research Design	Primary Geographical Focus	Infrastructure Sector Focus	Industry / Narrowed Sector	Level	SDGs Mentioned
SDG & Project	Wang W., Chen J., Liu Q., Guo Z.	Green project planning with realistic multi-objective consideration in developing sustainable port	Framework proposition	Single case study	China	Port	N/A	Sectoral	General, leaning towards environme ntal
SDG & Project	Zhang L., Chu Z., He Q., Zhai P.	Investigating the constraints to building information modelling (BIM) applications for sustainable building projects: A case of China	Exploratory	Empirical analysis	China	Buildings	Sustainable buildings	Local; Sectoral	General
SDG & Project	Calderón Márquez A.J., Cassettari Filho P.C., Rutkowski E.W., de Lima Isaac R.	Landfill mining as a strategic tool towards global sustainable development	Exploratory	Multiple case study	North America, Europe, Asia and the Middle East	Waste management	Solid Waste	Regional; Global	General
SDG & Project & Infrastructure	Dushenko M., Bjorbaek C.T., Steger- Jensen K.	Application of a sustainability model for assessing the relocation of a container terminal: A case study of kristians and port	Framework testing	Single case study	Scandinavia	Port	N/A	Industry	General
SDG & Project & Infrastructure	Terrapon- Pfaff J., Ortiz W., Dienst C., Gröne M C.	Energising the WEF nexus to enhance sustainable development at local level	Framework testing	Empirical analysis	Global south	Water, Energy, Food Nexus	N/A	Project	General
SDG & Project & Infrastructure	Menhas R., Mahmood S., Tanchangya P., Safdar M.N., Hussain S.	Sustainable development under Belt and Road Initiative: A case study of China-Pakistan Economic Corridor's socio-economic impact on Pakistan	Theoretical	Single case study	Pakistan	Multi-sector	N/A	Regional	General

Combined Search Keywords	Authors	Article Title	Type of Contribution	Primary Research Design	Primary Geographical Focus	Infrastructure Sector Focus	Industry / Narrowed Sector	Level	SDGs Mentioned
SDG & Project & Infrastructure	Anwar B., Xiao Z., Akter S., Rehman R U.	Sustainable urbanization and development goals strategy through public-private partnerships in a South-Asian metropolis	Theoretical	Multiple case study	South Asia	Urban development	Mega-cities; PPPs	Regional	SDG 11
SDG & Infrastructure	Schwanitz V.J., Wierling A., Shah P.	Assessing the impact of renewable energy on regional sustainability-A comparative study of Sogn og Fjordane (Norway) and Okinawa (Japan)	Framework proposition	Multiple case study	Norway; Japan	Energy	Renewable energy	Regional	General
SDG & Infrastructure	Diaz- Sarachaga J.M., Jato- Espino D., Castro-Fresno D.	Evaluation of LEED for neighbourhood development and envision rating frameworks for their implementation in poorer countries	Framework testing	Multiple case study	Developing countries	Urban development	N/A	Project	General
SDG & Infrastructure	Sperling J., Romero- Lankao P., Beig G.	Exploring citizen Infrastructure and environmental priorities in Mumbai, India	Theoretical	Empirical analysis	India	Urban development	N/A	Local	General, leaning towards SDG 11
SDG & Infrastructure	Aust V., Morais A.I., Pinto I.	How does foreign direct investment contribute to Sustainable Development Goals? Evidence from African countries	Theoretical	Empirical analysis	Africa (44 countries)	Multi-sector	N/A	National; Regional	General, leaning towards SDG 13
SDG & Infrastructure	Wei X., Xu H., Zhang B., Li J.	Infrastructure operation efficiency and influential factors in developing countries: Evidence from China	Exploratory	Multiple case study	China	Roads	Expressways; Toll road operations	Industry	General
SDG & Infrastructure	Fuldauer L.I., Ives M.C., Adshead D.,	Participatory planning of the future of waste management in small island developing states to deliver on the	Framework proposition	Single case study	Curacao	Waste management	N/A	National	General

Combined Search Keywords	Authors	Article Title	Type of Contribution	Primary Research Design	Primary Geographical Focus	Infrastructure Sector Focus	Industry / Narrowed Sector	Level	SDGs Mentioned
	Thacker S., Hall J.W.	Sustainable Development Goals		U					
SDG & Infrastructure	Zheng B., Bedra K.B.	Recent sustainability performance in China: Strength-weakness analysis and ranking of provincial cities	Framework proposition	Empirical analysis	China	Urban development	N/A	Local	General
SDG & Infrastructure	Di Vaio A., Varriale L.	SDGs and airport sustainable performance: Evidence from Italy on organisational, accounting and reporting practices through financial and non-financial disclosure	Framework testing	Multiple case study	Italy	Airports	N/A	Organisational; Industry	SDG 11, 17
SDG & Infrastructure	Teferi Z.A., Newman P.	Slum regeneration and sustainability: Applying the Extended Metabolism Model and the SDGs	Framework testing	Single case study	Ethiopia	Urban development	Slums and settlements	Local	SDG 1
SDG & Infrastructure	da Silva L., Marques Prietto P.D., Pavan Korf E.	Sustainability indicators for urban solid waste management in large and medium-sized worldwide cities	Framework testing	Multiple case study	Brazil	Waste management	Solid waste	Local	General
SDG & Infrastructure	Monteiro N.B.R., da Silva E.A., Moita Neto J.M.	Sustainable development goals in mining	Exploratory	Multiple case study	Brazil	Mining	N/A	Industry	SDG 1, 2, 5, 8, 13
SDG & Infrastructure	Makino T., Noda K., Keokhamphui K., Hamada H., Oki K., Oki T.	The effects of five forms of capital on thought processes underlying water consumption behaviour in suburban vientiane	Framework testing	Empirical analysis	Vientiane, Lao	Water	Water supply	Local	General
SDG & Infrastructure	Cheng S., Li Z., Uddin	Toilet revolution in China	Exploratory	Literature review	China	Sanitation	N/A	National	General, leaning

Combined Search Keywords	Authors	Article Title	Type of Contribution	Primary Research Design	Primary Geographical Focus	Infrastructure Sector Focus	Industry / Narrowed Sector	Level	SDGs Mentioned
	S.M.N., Mang HP., Zhou X., Zhang J., Zheng L., Zhang L.								towards SDG 6
SDG & Infrastructure	Munyasya B.M., Chileshe N.	Towards Sustainable Infrastructure Development: Drivers, barriers, strategies, and coping mechanisms	Theoretical	Empirical analysis	Australia	Not specified	Construction	Industry; Sectoral	General
SDG & Infrastructure	Zhang Q., Liu S., Wang T., Dai X., Baninla Y., Nakatani J., Moriguchi Y.	Urbanization impacts on greenhouse gas (GHG) emissions of the water Infrastructure in China: Trade-offs among sustainable development goals (SDGs)	Framework testing	Empirical analysis	China	Water, Energy Nexus	N/A	Local; Regional	SDG 6, 11, 13
SDG & Infrastructure	Jama A.A., Mourad K.A.	Water services sustainability: Institutional arrangements and shared responsibilities	Theoretical	Empirical analysis	Somalia	Water	Water supply	Sectoral; National	SDG 6.1, emphasisi ng its link to many others
Appendix 2 – SLR Analysis of Data with Derived Emerging Propositions

The nodal structure shown at Figure 15 is expanded below in Table 39 and illustrates the evolving propositions emanating from the SLR findings. This gives insight into the research issues in column 'b' and the context variables in column 'c', with emerging propositions (high, medium, low) shown in column 'e'. These issues and variables were used to shape the evolving evidence-based framework for structuring the MISI research.

Authors (a)	MISI Research Issue (b)	Context Variables (c)	Research Issues and thematics extracted from the selected most relevant (based on keyword occurrence articles) (d)	Emerging Proposition Strength (e) (High, Med, Low)
Hall R.P., Ranganathan S., Raj Kumar G.C.	Cascading from Global to the local level	Policy to Implementation	• The challenge now facing development organisations and governments is how to operationalise this interconnected set of goals and targets through effective projects and programmes.	High: core to the global-to- local research
		Micro-level OPM modelling approach	The research indicates a micro-level modelling approach that can quantitatively assess the impacts associated with rural water interventions that are tailored to specific communities.	Med: consider use of a water- based case study
			• The multilevel modelling framework provides a generalisable template that can be used in multiple sectors	High: aim for multi sector value
Dean K., Trillo C., Bichard E.	Definition and evaluation of Project Success	Inclusion of socio-economic assessment	• Current evaluative methods that support decision making on social housing interventions fail to capture all of the socio-environmental value contained in the UN SDG 11.	High: balance of TBL – eg include social
			• The paper addresses the issue by demonstrating how Sustainable Return on Investment can successfully describe and analyse a range of externalities related to the sustainable value generated by social housing regeneration schemes.	High: aim for prototype that informs investment decisions
		Inclusion of environmental and socio- economic assessment	• The findings show that, historically, the environmental and social value of regeneration schemes have been largely disregarded because of a gap in the evaluation methods, and that there is room for significant improvement for future evaluation exercises.	High: balance of TBL – eg include social, env and econ

Table 39: MISI SLR Research Findings (issues & variables) from the top 13 articles, leading to emerging propositions.

			-	
Goel A., Ganesh L.S., Kaur A.	Sustainable Construction	Lack of relevant research for project levels	• There is a lack of studies that synthesize and critically evaluate the available literature to provide an overview of the current state of sustainable construction (SC) research in India and provide directions for future research	Low: not core to research
		Lack of relevant research for project levels	• Current SC research endeavours are predominantly oriented towards the macro- industry level, the environmental dimension and the internal stakeholders.	Med: incl. micro project level
		Outputs versus outcomes success	• Additionally, more emphasis has been provided on the final project deliverable compared to the project processes.	High: include Theory of Change
		Research approach	Overall, this study makes three specific contributions [of which the first two are]: i) the current thrust areas of SC research in India have been identified while pointing out the imbalance in this academic pursuit; ii) a deductive content analysis framework has been developed that provides a generic template for conducting similar SLRs in the context of other countries	Med: informs design of the prototype and testing propositions
Zhang L., Chu Z., He Q., Zhai P.	Application of BIM technology for sustainability	Challenges to delivery of sustainability	Conducted a questionnaire survey with 389 respondents to investigate the applications of BIM technology in sustainable building projects. The results showed that there were four main constraining factors: "Public participation", "technology application", "economic cost", and "application management" "public participation" was particularly important.	High: assists the design of the first proposition to test engineer's views
		Opportunities for delivery of sustainability	The study offers practical and managerial implications based on the findings for local government and the private sector	Med: value of MISI application
Dushenko M., Bjorbaek C.T., Steger- Jensen K.	Practical application of Theoretical sustainability models	Limitations of theoretical models	• When documenting a sustainable design of port projects, decision-makers use theoretical sustainability models to conceptualize features of a sustainable society. However, a major challenge for the decision-makers was that the sustainability assessment results did not show, as expected, the same results as those of three existing theoretical sustainability models.	High : need for practical models – informs the testing of propositions for practicality
			• The benchmark results indicate a disparity between the importance of what sustainability models describe and what is important in practice.	High: as above
Terrapon- Pfaff J., Ortiz W., Dienst C., Gröne MC.	Cascading from Global to the local level	Policy to Implementation	• Until now, the focus of WEF [water- energy-food] nexus discussions and applications has mainly been on national or global levels, macro-level drivers, material flows and large Infrastructure developments. This overlooks the fact that major nexus challenges are faced at local level.	High: links to the global-to- local thematic

			•	
		Micro-level OPM modelling approach	• The study identifies the complex links which exist between sustainable energy projects and the food and water sectors and highlights that these needs are currently not systematically integrated into project design or project evaluation.	High: as above
		Outputs versus outcomes success	• A more systematic approach, integrating the water and food pillars into energy planning at local level in the global south, is recommended to avoid trade-offs and enhance the development outcomes and impacts of energy projects.	High: include Theory of Change in proposition structure
Schwanitz V.J., Wierling A., Shah P.	Definition and evaluation of Project Success	Policy to Implementation	• Apply a range of assessment methods and study their usefulness as tools to identify trade- offs and to compare the sustainability performance. We calculate cross-sectoral footprints, self-sufficiency ratios and perform a simplified Energy-Water-Food nexus analysis.	Med: as above
		Outputs versus outcomes success	• We recommend a general upgrade to indicators and visualization methods that look beyond averages and a fostering of infrastructure for data on sustainable development based on harmonized international protocols.	Low: possibly include Theory of Change in proposition structure
		Research approach	• We warn against rankings of countries or regions based on benchmarks that are neither theory-driven nor location-specific.	Low: not applicable to study
Diaz- Sarachaga J.M., Jato- Espino D., Castro- Fresno D.	Definition and evaluation of Project Success	Policy to Implementation	• Green rating systems have been launched during the last decades to facilitate the assessment of sustainable development in terms of building and infrastructure, including the evaluation of sustainable urban development through the study of communities. The absence of metrics in the New Urban Agenda led to relate its commitments to the SDGs, which revealed that the prerequisites and credits included in LEED ND and Envision mainly focused on managerial and environmental aspects and disregarded the economic and social dimensions. Consequently, the premises under which LEED ND and Envision were developed must be updated and complemented with the two latest guidelines recently adopted by the United Nations in the field of urban and sustainable development.	High: aim for testing use of existing sustainability measurement at project and organisational levels
Sperling J., Romero- Lankao P., Beig G.	Cascading from Global to the local level	Policy to Implementation	• With growing discussion and tensions surrounding the new urban sustainable development goal, announced by the UN in late September 2015, and a new global urban agenda document to be agreed upon at 'Habitat III', issues on whether sustainable urbanization priorities should be set at the international, national or local level remain controversial.	High: core to the global-to- local research

		Micro-level OPM modelling approach	• As such, this study aims to first understand determinants of and variations in local priorities across one city, with implications discussed for local-to-global urban sustainability.	High: core to the global-to- local research
Aust V., Morais A.I., Pinto I.	Investment priorities in SDG	Policy to Implementation	• The public and the private sectors play fundamental roles in mobilizing capital to achieve the 2030 Agenda of Sustainable Development. In particular, developing countries can benefit from foreign direct investment (FDI) as a source of external financing in the private sector. This study aims to investigate whether FDI contributes to the achievement of Sustainable Development Goals (SDGs) in Africa. We analyse a sample of 44 African countries regarding their SDG scores and apply a multivariate analysis and an ordered profit model.	Low: not core to study but of interest for future research topics and themes
		Opportunities for delivery of sustainability	• Our results indicate that the presence of foreign investors positively influences SDG scores.	Low : not applicable to this study
		Challenges to delivery of sustainability	However, although FDI has a positive impact in areas such as basic infrastructure, clean water, sanitation, and renewable energy, some adverse environmental consequences may occur for host countries. In fact, the relationship between FDI and the probability of achieving SDG13 (Climate action) is negative.	Low : as above
		Policy to Implementation	• This study contributes to the literature on sustainable development and can be useful for decision-makers in developing investment plans to support the achievement of SDGs.	High: aim for prototype that informs investment decisions
		Policy to Implementation	• Furthermore, we provide evidence of a positive influence of FDI on the SDGs, which might encourage further investments in Africa.	Low: not applicable to this study
Di Vaio A., Varriale L.	Investment priorities in SDG	Policy to Implementation	• This article investigates the 17 Sustainable Development Goals (SDGs) Agenda introduced by the United Nations in 2015 outlining if and which organisational, accounting and reporting practices are adopted to sustainable performance. Specifically, adopting the sustainability disclosure framework, we analyse how firms within the airport industry achieve the SDGs 11 and 17 showing how the initiatives are developed and implemented.	High : aim for prototype that informs investment decisions
		Research approach	• The article conducts a qualitative study through the reading and processing of financial statements and non-financial reports (sustainability and social reporting) of seven major strategic airport infrastructures in Italy to outline the initiatives implemented for meeting the SDGs.	Med: include in Theory of Change proposition structure

		Opportunities for delivery of sustainability	This article outlines the need to create conditions for developing and better implementing the accounting and reporting practices, like the SBSC (Sustainable Balanced Scorecard), as well as adequate organisational architectures and educational training and management programs for achieving the SDGs goals within firms.	Med: as above
da Silva L., Marques Prietto P.D., Pavan Korf E.	Definition and evaluation of Project Success	Policy to Implementation	• This work aimed to select a relevant set of sustainability indicators to analyse municipal solid waste management (MSWM) in large and medium-sized worldwide cities and to apply these findings in three municipalities located in the state of Rio Grande do Sul, in southern Brazil.	Med: include in Theory of Change proposition structure
		Policy to Implementation	• The result was the selection of a set of 49 indicators for application in a case study. It was only possible to measure 11 indicators with the information publicly available for the three Brazilian cities studied, demonstrating the fragility of information regarding sustainability issues.	Low: as above
		Challenges to delivery of sustainability	• Also, data related to social issues and natural and energy resources were insufficient for indicators to be measured. The analysis revealed difficulties regarding the availability of information in database	Med: include in Theory of Change proposition structure
Munyasya B.M., Chileshe N.	Drivers of Sustainable Infrastructure	Opportunities for delivery of sustainability	• While there is a plethora of studies around sustainable infrastructure, there are limited studies undertaken on the influencing drivers and barriers particularly within the South Australian construction industry.	High: include in proposition on SWOT of engineers and CEOs
		Challenges to delivery of sustainability	• "Lack of steering mechanism", "multi- disciplinary nature of the word "sustainability", and "lack of cooperation and networking" were the critical barriers.	Low: included in SLR

Appendix 3 - Survey questions and selection of the type of question and metric to align with analysis requirements for measuring engineers' views on projects' SDG impact

Q#	Theme	Question	Measuring Value		Aeasuring Value		
		·	Y / No	1 to 6	text	List	option
	Opening Page	With ICE and GEC logos. Thank the respondent for their time. Explain the purpose of the survey. Who is it for? How long will it take (10 mins)? What will happen with the data? Who is the POC at ICE.					
а		Are you completing this survey as an individual or for an engineering firm?					allow single choice for either individual or as engineering firm
1a	Engineering Organisation - General Data on your organisation						
1.1		Q: What is the name of your organisation.					complete text
1.2		Q: In which country is your organisation based?					text for country and list for continets
1.3		Q: Name of person completing Survey for firm					complete text
1.4		Q: Your role/grade					complete text
1.5		Q: Contact details (email)					complete text
1.6		Q: Number of Engineers in your firm					options include 1-49; 50-99, 100 +etc
1.7		Q: Number of current engineering projects underway (from design through to completion)					options include 1-20; 20-49; 50-99, 100 +etc
1.8		Q: In which countries do you deliver projects?					just home country; 2-5 countries; 6-20 countries; 20+ countries
1.9		Q: Who is the company's SDG or CSR lead?					complete text
2	Engineering Organisation - SDG Data on your organisation						
2.1	organisation	C. How do you plan to assess your impact on the SDGs? 1 We have no intention to assess our impact on the SDGs 2 We plan to assess our impact on the SDGs but have not thought through how 3 We plan to assess our impact on some of the SDGs and indicators relevant to our business 4 We plan to assess our impact on all the SDGs and indicators relevant to our business 5 We plan to assess our impact on all 17 SDGs and indicators 6 Don't know					select one of the six choices
2.2		Q. We fully understand the SDG priorities of the governments in our key markets and countries of operation.					1=strongly agree; 2=agree; 3= neither agree or disagree; 4=disagree; 5=strongly disagree 6=n/a
2.3		Q. Our company has fully defined the tools that will help it to assess its impact against the SDGs.					1=strongly agree; 2=agree; 3= neither agree or disagree; 4=disagree; 5=strongly disagree 6=n/a
2.4		Q. We can comprehensively report to governments and other key stakeholders on how our company is contributing to SDGs.					1=strongly agree; 2=agree; 3= neither agree or disagree; 4=disagree; 5=strongly disagree 6=n/a
1b	General SDG Data - as a member of the engineering community						
1.1b		Q: What is the name of your organisation?					complete text
1.2b		Q: In which country is your organisation based?					text for country and list for continets
1.3b		Q: Contact details (email) - optional					complete text
1.4b		Q: Years of experience as qualified engineer					options incl eg 1-5; 5-10; 10-15; 15+; n/a
1.5b		Q: Are you a millenial (born 1980-2000)?					select Y or N; this allows to filter data to assess any difference between them and others
1.6b		Q. As an engineer I want to know more about UN's Sustainable Development Goals, and what we are doing to measure our impact against them.					1=strongly agree; 2=agree; 3= neither agree or disagree; 4=disagree; 5=strongly disagree 6=n/a
3	SDG-Engineering specific Questions						
3.1		Q. From the list below, please rank the five SDGs where you believe engineers have the greatest impact and opportunity.					list of all 17 Goals with titles
3.2		Q. Having read the UN's Sustainable Development Goals, do you agree it is important that engineering business' sign up to these goals?					1=strongly agree; 2=agree; 3= neither agree or disagree; 4=disagree; 5=strongly disagree 6=n/a
3.3		Q. There is strong evidence that we have a 'fit for purpose' SDG measuring approach to track our contribution from our projects.					1=strongly agree; 2=agree; 3= neither agree or disagree; 4=disagree; 5=strongly disagree 6=n/a
3.4		Q. Commercial realities dictate that you should cherry pick the best SDGs for your business instead of the best ones for the planet.					1=strongly agree; 2=agree; 3= neither agree or disagree; 4=disagree; 5=strongly disagree 6=n/a
4	GEC related Questions: Engineering Community – Sharing Best Practice	The Global Engineering Congress is being hosted at the ICE from 22-28 Oct 18 (please see details at: Global Engineering Congress Info					Global Engineering Congress Info
4.1		Q. I/we are planning to actively engage with the GEC discussions and support plans to agree and implement a global engineering response roadmap to the SDGs?					1=strongly agree; 2=agree; 3= neither agree or disagree; 4=disagree; 5=strongly disagree 6=n/a
4.2		Q. As an engineer, I/we support the Global Engineering Congress' objective to unite the engineering community to agree and mobilise a response roadmap to the UNSDG?					1=strongly agree; 2=agree; 3= neither agree or disagree; 4=disagree; 5=strongly disagree 6=n/a
4.3		Q. We should look to engineering associations and standards bodies for advice, support and guidance on measuring project contribution to specific SDGs.					1=strongly agree; 2=agree; 3= neither agree or disagree; 4=disagree; 5=strongly disagree 6=n/a

Appendix 4 - Data capture from survey for question "Select the six SDGs that you believe that engineers have the greatest impact and opportunity".



Appendix 5 - Data capture from the survey's Chi-Square Tests (with continuity correction, likelihood ratio, and linear-by-linear association).

Are you a millennial (born between 1980-2000)? * q0023_0001 Crosstabulation						
			q0023_0001			
			Agree	Strongly Agree	Total	
Are you a millennial (born	Yes	Count	21	2	23	
between 1980-2000)?		Expected Count	17.7	5.3	23.0	
	No	Count	22	11	33	
		Expected Count	25.3	7.7	33.0	
Total		Count	43	13	56	
		Expected Count	43.0	13.0	56.0	
Chi-Square Tests						

	Value	df	Asymp. Sig. (2-sided)	Exact Sig. (2- sided)	Exact Sig. (1 sided)		
Pearson Chi-Square	4.615 ^a	1	.032				
Continuity Correction ^b	3.337	1	.068				
Likelihood Ratio	5.087	1	.024				
Fisher's Exact Test				.052	.03		
Linear-by-Linear Association	4.533	1	.033				
N of Valid Cases	56						

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 5.34. b. Computed only for a 2x2 table

Are you a millennial (born between 1980-2000)? * g0024_0001 Crosstabulation

			q002	4_0001	
			Agree	Strongly Agree	Total
Are you a millennial (born	Yes	Count	24	9	65
between 1980-2000)?		Expected Count	21.4	6.2	65.0
	No	Count	28	6	93
		Expected Count	30.6	8.8	93.0
Total		Count	52	15	158
		Expected Count	52.0	15.0	158.0

Chi-Square Tests

	Value	df	Asymp. Sig. (2-sided)		
Pearson Chi-Square	7.150 ^a	4	.128		
Likelihood Ratio	7.372	4	.117		
Linear-by-Linear Association	3.358	1	.067		
N of Valid Cases	158				

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 6.17.

Are you a millennial (born between 1980-2000)? * q0024_0001 Crosstabulation

Т

			Total
Are you a millennial (born between 1980-2000)?	Yes	Count	65
		Expected Count	65.0
	No	Count	95
		Expected Count	95.0
Total		Count	160
		Expected Count	100.0

Chi-Square Tests

	Value	df	Asymp. Sig. (2-sided)
Pearson Chi-Square	8.564 ^a	5	.128
Likelihood Ratio	9.474	5	.092
Linear-by-Linear Association	4.358	1	.037
N of Valid Cases	160		

a. 2 cells (16.7%) have expected count less than 5. The minimum expected count is .81.

Are you a millennial (born between 1980-2000)? * q0022_0001 Crosstabulation

			q00	22_0001	
			Agree	Strongly Agree	Total
Are you a millennial (born	Yes	Count	23	39	62
between 1980-2000)?		Expected Count	27.7	34.3	62.0
	No	Count	39	38	77
		Expected Count	34.3	42.7	77.0
Total		Count	62	77	139
		Expected Count	62.0	77.0	139.0

Chi-Square Tests

	Value	df	Asymp. Sig. (2-sided)	Exact Sig. (2- sided)	Exact Sig. (1- sided)
Pearson Chi-Square	2.553 ^a	1	.110		
Continuity Correction ^b	2.034	1	.154		
Likelihood Ratio	2.567	1	.109		
Fisher's Exact Test				.125	.077
Linear-by-Linear Association	2.535	1	.111		
N of Valid Cases	139				

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 27.65 b. Computed only for a 2x2 table

Question 2. We conducted SPSS analysis to ascertain Chi-Square between the two age groups (millennial and non-millennial) and their responses. There were 159 answers that gave a p-value (Pearson Chi-Square) of 0.136. This was not fully statistically significant but indicates a viable trend that might justify further analysis. However, when the optionality of questions was condensed, combining the agree and strongly agree as well as then separately combining the disagree and strongly disagree, then the results became more statistically significant at p-value of 0.110.

Question 3: There was initially a p-value of 0.001, suggesting that the non-millennials were not having any markedly different opinions on the answers to this question. However, when further analysis was conducted by combining agree and strongly agree, the p-value was 0.032, indicating that the Millennials had similar numbers agreeing but a much higher proportion of millennials were strongly agreeing. It is difficult to interpret what this categorically means but it could indicate that there is likely to be a stronger viewpoint from a generation that prefer to give higher ratings for an issue that has such catastrophic impacts if it is not dealt with effectively.

Appendix 6 - Blog of ICE workshop (between survey and interview stages to validate results) led by Paul Mansell with practitioners & academics at the Institution of Civil Engineers – The 'Control Group'

Written by the Head of Knowledge at the ICE – Elodie Huiban, published 1 May 2019:

Measuring SDG impact across Infrastructure projects Workshop – 25th April 2019 @ ICE

If the global engineering congress enthused more than 3500 engineers worldwide last October, the engineering community is still asking itself: **how do we translate high level goals to project context?** And since what can't be measured can't be managed it feels like the most important task on our to-do list post-congress.

ICE dedicated some time to this question at a workshop with a cross section of practitioners, academics and other thought leaders on **Measuring SDG Impact across Infrastructure Projects**.

Why?

The construction industry has a major role in achieving measurable impact against the Sustainable Development Goals 2030 targets. The estimated USD \$94 trillion (Global Infrastructure Hub, 2017 & 2018) of investment in infrastructure projects that is required globally by 2040, represents a massive opportunity to stimulate economic prosperity, reduce poverty and raise standards in health, education and gender equality.

However, the linking of infrastructure project success to SDG targets is problematic as a recent Institution of Civil Engineers' survey led by Paul Mansell in 2018 demonstrated: whilst the appetite for SDG reporting at project level is very strong (87%), especially by millennials, only a third of the 325 respondents assessed current tools as 'fit for purpose'.

What?

ICE brought a group of like-minded partners together to share the latest research on sustainability measurements methods at organisational and project level. We did a lot of listening in the morning and utilised those methods to inform practical exercises throughout the afternoon.

1. Nathu Puri Institute for Engineering and Enterprise / London South Bank University

Paul Mansell, a Doctoral Researcher who is working closely with ICE to transform the way engineers engage with SDGs presented his latest research focused on finding methods that will help engineers and engineering firms to demonstrate SDG impact across infrastructure projects >> Sharing empirical evidence of a 'golden thread' from project level SDG reporting to organisational level

As Paul has demonstrated throughout his research linking infrastructure project sustainability performance to Sustainable Development Goals (SDG) targets is problematic. Through his doctoral research, a new Infrastructure Project Transformation Process Model is being proposed, called the 'Infrastructure SDG Impact-Value Chain' (IVC) to link tactical-level project delivery with global-level strategic SDG impacts. He demonstrated the viability of this method using a water utility company case study to showcase how the model can integrate the 'Triple Bottom Line' to ensure balanced definition of success across economic, environmental and social thematic areas. The proposed methodology aimed at project leaders enables stakeholders' alignment on a common definition of project success during the design phase. It includes selection of longer-term outcomes and strategic SDG impacts.

Having assessed several sustainability measurement tools and methodologies including the UN Global Compact Compass; GRI and CEEQUAL, Paul outlined that only 9% of the 232 SDG indicators are highly relevant to engineering projects, which poses the crucial question what set of specific

infrastructure indicators (not SDG indicators) can project leaders choose from that are relevant for infrastructure projects?

Closely related is the Inclusive growth agenda, which has the combined purpose of ensuring increased prosperity with greater equity and improved wellbeing and living standards. Richard suggested that the focus of all project leaders should be to find monetary value to improving peoples' lives by understanding which project options deliver greater net benefit.

Engineering projects must move away from the traditional cost benefit model and put environmental and social bottom line on the same pedestal as economical drivers. Most engineering projects impacts might not be financial. A Social value analysis allows for the total value created by projects to be measured in a common currency using techniques endorsed by HM Treasury and the OECD.

Throughout the day, the energy in the room was high, valuable conversations took place and the participants discussed their understanding of the interconnections between SDGs and appreciated the sheer complexity of choosing between 17 SDGs, 169 targets or 232 indicators at project level.

How?

After lunch, participants were asked to validate and test some of the research presented in the morning whilst assessing the feasibility of doing such exercise with project teams.

Paul presented us with a series of cards representing the SDGs and another set representing the 169 targets so we could choose a set of primary and secondary SDGs to prioritise for the development of an infrastructure project that is seeking to provide safe drinking water to a village in Zambia. Once we prioritized our SDGs we drew lines between the SDGs to build a nexus map of interactions and analyzed our network, by quantitative scoring to identify relationship strengths between SDGs.

SDG PRIORITIES

The 3 groups had different approaches and chose different SDGs as priorities, whilst all groups included SDG6 Clean water and sanitation and SDG3 Good health and wellbeing (as the 2 most obvious ones) the other SDGs were different from one group to another, choosing SDG 13 Climate action over SDG8 Decent work and economic growth or SDG2 no hunger – the links between our project and those SDGs were not as clear and when we delved into the targets, it got even more complicated...

We felt empowered by discovering more about each SDG we had chosen to prioritise for the project and yet challenged by the urge to want to know more about each target to ensure the project will achieve its priority SDG and be measured against its indicators.

At that point most of us were overwhelmed by the nature of the 232 indicators which sits underneath the 17 SDGs and its 169 targets.

Another realisation occurred - no one from a project team will go through all of the above in great detail - it's too complex, time consuming and indicators are inappropriately applicable to a project or possible to easily integrate as part of measurement tools.

It highlights again the need for the engineering industry to agree on how SDG targets/indicators should be translated from high level goal to project context – and ICE is forming a working group on Measuring, Monitoring and Reporting impact against SDGs across infrastructure projects to further develop methods to answer this apparent need.



SDG SYSTEM MAP OF INTERDEPENDENCIES

Finally, each group built a generic Infrastructure SDG system map of interdependencies between the SDG chosen as priorities which could be used at the project design stage to effectively assess pre and post-project success as defined within the IVC model presented and developed by Paul and piloted throughout the afternoon.



NEXT STEPS

To conclude we assessed our achievements throughout the day and discussed next steps.

- 1. Set further discussions and workshops on translating high level goals to project context to find out how:
 - *a.* To standardise and consistently use which metrics to report on and how do you collect *it*?
 - b. Create a central repository of all projects above 50 million pounds which will demonstrate SDGs alignment and enable progress on the below
 - *c.* Identify benefits of building SDGs into your project = to incentivise and build a strong business case
 - d. Test bed a pilot repository (Starting with assessing DFID global infrastructure programme open data information) which could lead to an open data platform of SDG realisation on projects globally
 - e. Use this workshop/game approach to further educate the workforce
- 2. How could these methods inform/influence the selection of projects and empower project teams to create the winning business case (this is the major gap identified in the industry at the moment)

Appendix 7 – List of Interviewees and	l company background data
---------------------------------------	---------------------------

	Role			0	rganisation					Interview data		Awareness \	's capability	Transcri	ption Project		
Participant ID	Role	Management Cat	Organisation		Size	Size Category (>50; 25- 50;10- 25;<10;oth			Geography of Business		Interview length	Words transcribed	joint or Individual	Three level of SDG focus awareness Vs application	Level of Self- assessed Capability / Maturity 1-5	Transcr ipted	Nvivo coding completed
	Past President ICE; Chair of Construction & Infrastructure	Board	UCL BSC&PM	n/b	(FTE)	er) other	Share value	Global / National	Category	Interview date	45	4.752	Individual			-	
	Policy;			multidaciplinary consultancy; one of the	16,000	10-25k		150 countries	Other	1.30pm 3 July				n/a 3 = aware and doing	n/a 2 - early processes in		
2	Global Sustainability Leader	Senior Executive	MettMac	world. About 16,000 projects per year Design, Engineering consulting,	48,000	25-50k	Revenue	offices in 50	Global	11am 4 July	55	5,892	Individual	ł	place	yer	Yes
3	UK Director of Sustainability	Head of Sustainability	WSP	Environmental consulting, Planning, Professional services UK multidisciplinary consultancy operating	2.000 staff	~10k	C\$6.4 Bn in 2016	countries Capita Group has	Global	3pm 4 July	61	7,177	Individual	it aware and boing	2 - using sustainable process	yes	yes
	azo	CEO	Capita Symonds	In the building design, civil engineering, environment, management and transport sectors, part of the Capita Group.	with contractors of total of c4500			4,500 staff in 50 offices across UK and Ireland	Regional	10am 10 July	42	5,508	Individual	2 = aware and not doing it	1 - basic level at definition stage	yes	Yes
5	œo	CEO	BAM Nutral	A construction and civil engineering company HO is Camberley, UK. Involved in a portfolio of road, rail, nuclear, and other major projects workholde. Subsidiary of the Dutch Royal BAM Group.	3,100 in UK at 120 sites of 25,000 in Group	<10k	10574 m turnover in 2017				s	4,923	Joint	3 = aware and doing it	2 - early processes in place	yes	yas
6	People & Culture Director	Senior Executive	BAM Nuttal	ditto	ditto	-10k	ditto	ditto	Regional	9am 1 August 9am 1 August	53		Joint	3 = aware and doing it	2 - early processes in place	yes	yes
				AECOM is an American multi-national engineering firm. #164 in Fortune 500 in 2018. (exercise on all exercise of the	1,600 in 50 countris	>50k	\$21 Bn in 2018	Global									
7	Chief Exec Environment & Ground Engineering	CED	AECOM	and its recalling of an aspects of the environmental consultance which is contaminated land to environmental access, social impact assessment, noise and air quality of the whole gamat Crossral is the name of the rail construction	business of 87000	<10k			Global	1pm 10 July	40	4,222	Individual	1 = unaware and not doing it	0 - processes not developed	ун	hat
•	æ	CEO	Crossral	project and of the limited company, wholly owned by TR, thirt was formed to carry out construction works Crossnal, reformade an another than the second of the second of the hold bill service with an earth-and to come into bill service with an earth-and to come on the bill service with an earth-and to come of the bill service with an earth-and to come of the bill service with an earth-and to come of the bill service with an earth-and the bill service with an earth-and to come of the bill service with and to come of the	55,000 jobs				National	10am 23 July	42	5,324	individual	2 = aware and not doing it	0 - processes not developed	har	<i>be</i>
9	Head of Sustainability	Head of Sustainability	Anglan Water	Anglan Water is a water company that operates in the East of England. Anglan Water is regulated under the United	4,000	-10k	Revenue in 2015 was £1244 M	UK and Netherlands			36	3,105	Individual	3 = aware and doing it	2 - early processes in place	yes	Yes
10	Director of Strategy & Investment	Senior Government or UN Policy Director	Environment Agency	Kingdom Weler Industry Act 1921. The Environmet Agency (EA) as non- departmental public body, established in 1926 and sponsored by the United Kingdom government's Department for Environment, Food and Rural Affrian (CEPRA), with responsibilities making to the production and enhancement the servicement in England. "We have	11,200	<10k	8 directorates	England	National	9am 10 July	52	5,643	Individual	2 = aware and not doing it	1 - basic level at definition stage	har	yas
				Anglan Water is a water company that operates in the East of England, Anglan	4,000	<10k	Revenue in 2015 was	UK and Netherlanda	National	4pm 4 July				2 - many and ching	2 . and a more set in		
"	Director of Asset Management	Senior Executive	Anglian Water	Water is regulated under the United Kingdom Water Industry Act 1991. Heathrow exception is a second to the	76,000	-10k	£1244 M	UK	National	9am 10 July	36	3,207	Individual	it it	place	yer	Yes
12	Expansion Programme Director	CEO	Heathrow	autional interest, which will help deliver new domestic routes and connect more of the	at LHR. Increase to	" mark			Ì		25	3,458	Individual	3 = aware and doing it	2 - early processes in place	yes	yes
13	Technical Director	Senior Exercition	Crossmil	world. see above	supported	+10k	<u> </u>	uk	National	4pm 1 August	£9,	5,435	Individual	2 = aware and not	0 - processes not	V	
				What the igone Alliance does for Anglian Water. The igone Alliance is a	55,000 jobs n/s	×10k	n'a	uk	National	10am 23 July		-,		doing it	developed		
14	Managing Director of Partnership	CED	Anglian	collaborative, virtual joint venture formed to deliver the large part of the Anglian Water capital investment programmes.	D				National	1pm 13 August	52	5,231	Individual	it it	definition stage	yes	Yes
15	Sustainability Director	Head of Sustainability	Mace	privately owned company that does not have shareholder pressure to perform in a certain way.	5,042 (2017	-10k	Revenue £2,037 (2017)	uk	Global	Jom 10 July	54		Joint	2 = aware and not doing it	1 - basic level at definition stage	yes	Yes
16	Gp FD	Board	Mace	Mace Group Ltd, commonly known as Mace, is a global consultancy and construction firm bandwartened in London.	5,042 (2017	10k	Revenue £2,037 (2017)	uk	-	Term 40 July	54	5,838	Joint	3 = aware and doing it	1 - basic level at definition stage	yes	Yes
17	Partner for Sustainability	Senior Executive	Deloitte	Defotte, is a multinational professional services network. Delotte is one of the "Big Four" accounting organizations and the language professional services network in the world by revenue and	286,200 (2018) global	HSOk	Revenue US\$ 43.2 Bn (2018)	Global		Jun to stay	21	2,731	Individual	3 = aware and doing it	2 - early processes in place	yes	yas
18	Infrastructure & Materials	Senior Government or UN Policy Director	BEIS	numeer or protessionas. The Department for Business, Energy and Industrial Strategy (BEIS) is a department of the government of the United Kingdom, which was created on 14 July 2016		other	Annual Budget £13.8 Bn (2016-17)		Global	12.00 12 July	36	3,452	Individual	n/a	n/a	yes	yas
19	Global Sustainable Development Leader	Head of Sustainability	Алар	Anap (officially Anap Group Limited) is a multinational professional services firm headquartered in London which provides engineering, design, planning, project management and consulting services for al aspects of the built environment. It is a	13,840 (2018)	10-25k	Revenue £1.55 Bn (2018)	90 offices in 35 countries. Arup has participated in projects in over 160 countries.			75	9,146	Individual	3 = aware and doing it	3 - using sustainable process	yes	yes
20	Design Delivery Manager, HS2	Head of Scotainability	WSP	Trust owned company. Design, Engineering consulting, Environmental consulting, Planning,	48,000	10-25k	Revenue CS5.4 Bn in	offices in 50 countries	Global	Spm 10 July	55	4903	bilideal	3 = aware and doing	2 - using sustainable		Late
	Curzon Street Station General Counsel, Exec Board	Barrel	Collined To:	Professional services Galiford Try pic is a British construction and bruss-building company registered	5,485 (2018	<10k	2016 Revenue 12 932 (2018)		Global	11am 16 July			inited.	it 2 = aware and not	process 1 - basic level at		
22	member Head of Communications	Head of Sustainability	Galiford Try	in Uxbridge, London. Founded 1905 ditto	ditto	-10k	ditto	ditto	Regional	2pm 12 July	45	4,069	Joint	doing it 2 = aware and not	definition stage 1 - basic level at	ym	yes
	Head of Environmental	Manual of Prostationability	Laine Officiality	Laing O'Rourke is a multinational construction company	12,796 (2017/18)	10-25k	£2,928.9 million		regional	2pm 12 July		4.303	hereitert	doing it 2 = aware and not	0 - processes not		
	Sustainability	The of Summerly	Lang Orloake	headquartered in Darfford, UK. It was founded in 1978 by Ray O'Rourke. It is the bestell musicipal and the second sectoral second Pinsent Masons LLP is an international law	2,500	<10k	(2017/18) £432.1 million	25 offices in UK,	Global	11am 17 July				doing it	developed	ju.	yes
24	Head of Infrastructure	Senior Executive	Pinsent Masons	Firm which apecialises in the energy, infrastructure, financial services, real estate and advanced manufacturing and technology sectors. The firm ranks among the top hundred law firms in the world by Matienel California.	22.600	10.385	(2016/17)	400 partners; legal team of c1,800 and over 2,500 staff	Global	11am 13 July	39	3,289	Individual	2 = aware and not doing it	0 - processes not developed	yes	yes
25	Director, Electricity Transmission Owner	Senior Executive	National Grid	British multinational electricity and gas utility company headquartered in Warelck, UK Its originate articities are in the UK	(2018)	10 aux	£15.25 Bn (2018)				43	5,027	Individual	3 = aware and doing it	1 - basic level at definition stage	yes	yes
	Head of IPM Strategic Initiations			and NE US. founded 1990. UNOPS implements more than \$1 billion	600 open	-10k	400 to 500	operating in more	National	12.30 14 August							
26	Infrastructure and Project Management Group	Senior Government or UN Policy Director	UNOPS	worth of peace and security, humanitarian and development projects for its partners every year, The £1.2 billion Prosperity Fund operating	projects at the moment n/a	other	milion a year	than 80 countries.	Global	9am 2 July	28	2,458	Individual	2 = aware and not doing it	1 - basic level at definition stage	Aur	Yes
27	Acting Head, Prosperity Fund, Joint Funds Unit	Senior Government or UN Policy Director	FCO - Prosperity Fund	until 2023, across: investment in infrastructure and human capital;innovation and technology; increasing trade; financial	1						47	4,147	Individual	3 = aware and doing it	2 - early processes in place	yes	Yes
25	Leader Infrastructure Design Gp	Senior Executive	Апр	and economic reform; and ease of doing business. Anup (officially Anup Geoup Limited) is a multinational professional services firm headquartered in London which provides engineering, design, planning, project management and consulting services for al meaning of the hold moderament. It is no	13,840 (2018)	10-25k	Revenue £1.55 Bn (2018)	90 offices in 35 countries. Anup has participated in projects in over 160 countries.	National	12.00 18 July, FCD	ж	3,125	istividual	3 = aware and doing it	2 - early processes put in place as trial	yes	Yes
29	Environmental Sustainability	Head of Sustainability	National Grid	Trust owned company. In the B National Grid pic is a British multitastional electricity and gas utility company headquartered in Warwick,	22,600 (2018)	10-25k	Revenue £15.25 Bn (2018)		Gilobal	5.30pm 23 July	45	5,743	Individual	2 = aware and not	1 - basic level at definition character	yes	Yes
				and NE US. founded 1990. The Infrastructure and Projects	n/a	other	nia	nia	National	2pm 17 July					and a stage		_
30	Infrastructure Leadership Council; chair the government construction board; green construction board, as part of leadership council;	Senior Government or UN Policy Director	Infrastructure and Projects Authority, Cabinet Office, UK Government	Authority (IPA) is the United Kingdom government's centre of expertise for infrastructure and major projects. The IPA sits at the heart of government, reporting to the Cabinet Office and HM Treasury. The core learns include experts in							65	7,052	Individual	2 = aware and not doing it	0 - processes not developed	yes	yes
31	Vice President, Strategy and Partnership, GEC sneaker	Senior Executive	Engineers Canada	mmastructure, project delivery and project	n/a	+10k	n'a	n/a	Other	4pm 6 July 9am EST/2pm BST 14 Aug	59		Joint	n/a	nla	yes	Yes
32	Vice President, Strategy and Partnership, GEC speaker	Senior Executive	Engineers Canada	5%	n/a	~10k	n'a	nia	National	9am EST/2pm BST 14 Aug	59	7,422	Joint	n/a	n/a	yes	Yes
33	Head of Sustainability	Head of Sustainability	United Utilities	Kingdomis larger lated water company, kingdomis larger lated water company, was founded in 1925 as a result of the merger of North Wealt Water and NORMEE The group manages the regulated water and waste water creations: In North Wealt England, which includes Cumbris, Cheshire, Greater Marchester, Lancashire and Merseysido, which have a combined population of		in a SK	n.,, 30.4 milio n (2018)		Global	10am 23 July	kost	lost	individual	2 = aware and not doing it	1 - basic level at definition stage	lost	hae
34	Head of Sustainability	Head of Sustainability	Network Rail	Network Rail is the owner (via its aubsidiary Network Rail Infrastructure Ltd, which was known as Railizack pic before 2002 and infrastructure manager of most o the railway network in Great Britain.	37,000	10-25k	05.2 billion (2013)		Nation	12.30am 23. kdv	44	307	joint	2 = aware and not doing it	1 - basic level at definition stage	yes	yas
35	Group Director of Safety, Technical & Engineering	Board	Network Rail	ditto Network Rail is an arm's length public body of the Department for Transport with no shareholders, which	ditto	10-25k	ditto				44		joint	2 = aware and not doing it	1 - basic level at definition stage	yes	Yes
				reinvests its income in the naiways. The Building Research Establishment (RBP) is a contra of b.	n/a	other	nia	nia	National	1pm 23 July				2 - aware and not doing P	1 - basic level at		
36	Director - CEEQUAL & Infrastructure	Serior Executive	BRE Group	Inclunces in the Carlied Strongeneration on caldBMS characteristics in the Carlied Strongeneration of the Strongeneration in the Strongeneration of the Strongeneration in the as former LK government matinnal laboration research, advice, training, testing, carling and standards for for both public and private sectororganisations in the LK and abroads.					Global	2am 17 Aug	65	5,480	individual			yes	Yes
37	Head of Environment and Sustainability	Head of Sustainability	Taylor Woodrow	raylor Woodrow was one of the largest housebuilding and general construction companies in Britain.	8,132 (2005)	+10k	£3,572.1 million (2006)		Global	10AM 14 August	57	4,498	Jaint	3 = aware and doing it	1 - basic level at definition stage	har	Yes
38	Head of Environment and Sustainability	Head of Sustainability	Taylor Woodrow Institution of Engineer*	ditto n'a	ditto n/a	<10k other	ditto n/a	ditto n/a	Global	10AM 14 August Written responses	57		Joint	3 = aware and doing it n/a	1 - basic level at definition stage n/a	yes	Yes
40	President	CEO	Pakistan COIT, Spain (Telecom***)	nia	n/a	other	n'a	nia	National	13/7	written	741	individual	nia	nia	n/a yes	n/a yes
					1	1	1			t responses	47	153,013					

Appendix 8 – Nodal Incidence of participants Per Node and Frequency of Statements Per Node



PARTICIPANTS' (N=40) FREQUENCY OF CAPTURED STATEMENTS (N=598) AND NUMBER OF NODES THAT STATEMENTS WERE ALIGNED WITH (N=317)



Participant #3	Participant #29	Particip	ant #19	Particip	ant #8	Partici	ipant #26	Partici	pant //11
Participant #2	Participant #10 Participant #20		Participant	112 1	Participant I	23 P	Participant #7	Part	icipant #31
Participant #2	Participant #5		Participant	#13	Particip	ant II9	Participant	Pa	rticipant
Participant #27			Participant	116	Particip	ant #16	i Partici	pant	Participa
Participant/20			Participant	//15	Particip	ant //24	Partie	Parti	Par
	Participant #25		Participant	114	Particip	ant //36	; Partici Partici	pant	

Appendix 9 - NVivo Codebook of Node Descriptions and Occurrences of Codified Data Collection

Name	Description	Files	References
1. What role and lev	vel in the Organisation was the Participant?		
Board Level	The Participant was currently at Board level, or, had recently retired from a similar level role.	6	
CEO	The Participant was had a title of CE, CEO or MD. Their business area was a stand-alone unit that they had P&L accountability for.	7	
Head of Sustainability	They had the primary responsibility for coordinating and managing the company's sustainability approach. None of them had SDG in their job title, although some had SDG in their job description.	11	
Senior Executive	They had an executive role for a discreet business area. Typically had the word Director in their job title.	11	
Senior Government or UN Policy Director	At Director or Deputy Director level in Government departments, or, International Organisation.	5	
2. Outcome – how o	lid the organisation identify with SDG outcomes?		
Terminology of SDG - Sustainability - CSR - CSV	After the salutations and clarification of the ethics clearances (eg confirmed that the form had been signed and returned) the interview started with a discussion on their company's use of terminology across sustainability, Sustainable Development, SDG and Corporate Social Responsibility (CSR).	31	81
3. Mechanism – hoy	w important and advanced was the organisation's SDG measurement?		
Theme 1. Perceptions & Awareness of SDG Measurement	The third area of discussion was for the Participant to self-assess their company's 'Awareness & Application' of SDG measurement and also their assessed level of 'SDG measurement Process Maturity'.	16	28
Code 1.1 - Company's awareness & importance	Self-assess company's 'Awareness & Application' of SDG measurement	16	23
Import. Low.	This banding indicated that the company placed low importance on SDG measurement. They were unaware and didn't apply them.	10	10
Import. High- High	This banding indicated that the company recognised the importance of SDG measurement, and had a high level of engagement with SDGs.	4	4
Import. High- Low	This banding indicated that the company recognised the importance of SDG measurement, but there was low level of engagement with SDGs.	7	9
Code 1.2 - Company's level of SDG Measurement Maturity	Company's self-assessed level of 'SDG measurement Process Maturity'.	13	21
Process - High	This banding indicated that the company had a higher than average maturity of SDG measurement processes: either 2 = early processes in place; 3 = sustainable SDG processes.	3	4
Process - Low	This banding indicated that the company had a higher than average maturity of SDG measurement processes: either $0 = no$ SDG processess; $1 = currently$ defining processes.	4	4
4. Context – what is	ssues effect (SWOT) the success of SDG measurement?		
Theme 2. Key Challenges & Opportunities	The analysis of the contextual issues that affected companies' ability to measure SDG impacts successfully were captured using a SWOT (strength, weakness, opportunity, threat) approach. There were 5 primary areas of nodes that were derived from the preceding survey of	1	1

Name	Description	Files	References
	325 engineers.		
Code 2.1 Knowledge	The knowledge codes had been identified by the preceding survey of 325 engineers as the single largest factor that negatively impacted SDG measurement at organisational and project levels.	5	9
2.1.1 Outcome Vs Output	Within Code 2.1, the first sub-code was the challenge of differentiating between outputs and outcomes. Too few knew how to do this well and as a result, the wrong 'targets-indicators' were being measured to define success.	13	30
2.1.2 skills	Skills covered a number of areas including: the skills to be able to define success definitions; business skills to be able to build performance frameworks; sustainability/SDG skills that helped understand the SDG framework and how they relate at sub global-national levels, at organisational and project levels.	9	10
2.1.3 starting projects	The preceding survey had not highlighted the importance of 'starting projects well'. This node was added during the interviews since it was often referred to as the 'key investment decision point' and its linkage to SDG impacts.	7	11
Code 2.2 - Leadership & Strategy	The Leadership and Strategy codes had been identified by the preceding survey of 325 engineers as the second highest challenge in SDG measurement at organisational and project levels.	23	63
2.2.1 Stakeholders and customers	The interviews highlighted the importance of understanding the stakeholders. The word analysis categorises a number of stakeholders that were most frequently referred to by the participants, such as: government, communities, investors, users, suppliers etc.	11	21
2.2.2 Planning horizons	There was frequent reference to the need to think longer term. SDGs have targets and goals for 2030, but the references were to moving from short term RoI that were driven by economics only, to a longer term more balanced approach across Environment and society issues. Hence the SDG were viewed as a good way to achieve this ambition.	10	13
Code 2.3 - Tools, Processes & Techniques	The research longer term aim was to assess whether a new approach, by developing existing theory, could be designed to measure SDG with greater success. This led to common reference to the use of tools, systems and methodologies, although at this stage, the interviews did not seek detailed analysis on any specific aspect.	10	18
2.3.1 What to measure	There were many references to what is measured; The selection of targets becomes critical in a business environment that is already awash with data collection. Is it quantitative data or qualitative? What is the balance between too little data collection and too much?	26	82
Code 2.4 - Other, eg Culture	The single most important area for ensuring SDG measurement success is having a successful change programme that ensures a practical approach is made to work for the 'users' with added value of what they are doing.	12	27
2.4.1 Millennials - gender	The preceding survey included a number of specific questions relating to the interests-perceptions-values of millennials. Some consistent and interesting considerations, especially in senior executives recognising the demands of millennials in this area and the need to adjust their approach in consideration of the recruitment and retention HR issues.	14	18
Code 2.5 - Cost	Many participants referred to the realities of ensuring that at the bottom line, businesses remained profitable.	2	2
Code 2a - Opportunities for improvement	This node was a catch-all for the positive suggestions for improving SDG measurement in their organisations.	17	45
Code 2b - Weaknesses & Challenges	This node was a catch-all for the the challenges for improving SDG measurement in their organisations.	10	19

Appendix 10 - Data analysis using NVivo: nodal-word linkages

Category	С-М-О	Sub-Category	Word	Count	f1 Weighted %	f2 %	f3 %
			a vata in a bility	077	0.710/		
		Sustainability	sustainability	377	0.71%	0.71%	
			sustainable	154	0.71%		
Sustainability;	_	Sustainable	development	142	0.23%	0.56%	
Sustainable	Context	Development	aoroiopinoni	296	0.56%		2.24%
Development; SDG			sdgs	328	0.62%		
		SDGs	sdg	188	0.35%	0.97%	
				516	0.97%		
			social	//	0.15%		
		TBL - Social	society	49	0.09%	0.35%	
			CSI	185	0.35%		
Triple Bottom Line	-		environment	79	0.05%		
(TBL)	Outcome	TBL -	environmental	52	0.10%	0.25%	0.68%
		Environment		131	0.25%		
			economic	20	0.04%		
		TBL - Economic	Prosperity	19	0.04%	0.08%	
				39	0.08%		
			resources	20	0.04%		
			cost	57	0.11%		
		Input to Activity -	COSIS	17	0.03%		
		'iron triangle' of	officiency	10	0.03%	0.33%	
		cost-time-scope	money	10	0.03%		
			commercial	10	0.03%		
			spend	17	0.03%		
Theory of Change		Activity to Output	initiatives	27	0.05%		
incory or onlango,		(time, cost,	activities	15	0.03%	0.12%	
(causal logic chain	Mechanism	scope)	outputs	19	0.04%		1.26%
from inpts to		Outcome and	outcomes	60	0.11%		
impacts)		benefits	outcome	34	0.06%		
		delivered	benefits	23	0.04%	0 17%	
		derived from using the project's	benefit	19	0.04%	0.1776	
		measurable improvement resulting	impact	219	0.41%		
		from an outcome that is perceived as an arkentage by one or more	impacts	19	0.04%		
		longer term	value	101	0.19%		
		goals - values at	end	73	0.14%	0.64%	
		end of project	yoais	165	0.31%		
				902	1./9%		
			leadership	83	0.16%		
		Laadar-bir -	leaders	30	0.06%		
		Leadership as	leading	20	0.04%	0.29%	
		uleme	leads	15	0.03%		
				148	0.29%		
			strategy	75	0.14%		
		Strategy	strategic	23	0.04%	0.18%	
1.4.4	Ocarta i			98	0.18%		0.000
Leadership	Context	Loodorphin rate	Ceo	26	0.05%	0 1 20/	0.80%
		Leadership role	executive	37	0.07%	0.12%	
			align	03	0.12%		
		Strategic	governance	25	0.05%		
		planning	alignment	18	0.03%		
		competency	direction	18	0.03%	0.21%	
			vision	28	0.05%		
				114	0.21%		

Sub-Category			Word	Count	f1 Weighted %	f2 %	f3 %
			learning	30	0.06%		
		Learning &	training	22	0.04%	0.15%	
		Education	education	29	0.05%		
Knowledge	Context		maturity	30	0.06%	· · · · ·	0.26%
		Experience	experience	29	0.05%	0.11%	
			experience	140	0.00%		
				140	0.2070		
			target	31	0.06%		
			targets	208	0.39%		
			objectives	26	0.05%		
			indicators	76	0.14%		
			measure	142	0.27%		
		Targets and	measuring	72	0.14%	1.36%	
		measuring	measuring	72	0.14%		
			measurement	54	0.10%		
			measured	17	0.03%		
Performance	Mechanism		measures	21	0.04%		1.90%
Management				719	1 36%		
			management	83	0.16%		
		Performance	performance	54	0.10%		
		management -	metrics	46	0.09%	0.43%	
		guuantitative	obiectives	26	0.05%		
		1	quantitative	18	0.03%		
		Qualitative	contribution	40	0.08%		
		contribution	qualitative	17	0.03%	0.11%	
			284 0.38%				
		lools	tools	32	0.06%	0.06%	
Tools, Systems;	M	_	processes	0.05%		0.0.00	
Processess	Mechanism	Process	process	93	0.18%	0.23%	0.34%
		Systems	system	26	0.05%	0.05%	
			-	177	0.34%		
			change	129	0.24%		
		Change Culture	culture	23	0.04%	0.28%	
				152	0.28%		
			honest	22	0.04%		
		Behavioure	collaborative	14	0.03%	0 10%	
		Denaviours	collective	14	0.03%	0.1076	
				50	0.10%		
			opportunity	54	0.10%		
Change	Contout	Innovativa	opportunities	20	0.04%		0.700/
Management	Context	Commitment to	commitment	42	0.08%	0.24%	0.78%
		change	innovation	32	0.06%	0.54%	
		change	investment	32	0.06%		
				180	0.34%		
					0.000/		
		O	communication	14	0.03%		
		Communication			0.000/	0.06%	
		S	communications	14	0.03%		
				28	0.06%		
			project	278	0.52%		
		Project Level	projects	185	0.35%	12.61%	
				6751	12.61%		
Project;			program	65	0.12%		
Programme;	Context	Programme	program	65	0.12%		13.01%
Portfolio		Level	programme	31	0.06%	0.35%	
			programs	27	0.05%		
		D (())		188	0.35%		
		Portiolio Level	portiolio	26	0.05%	0.05%	L
			dobal	107	0.040/		
		Global	yiuuai world	127	0.24%	0.44%	
Geography - levele	Context	National	national	100	0.20%	-	0.63%
	00.110/11		local	38	0.12 /0	0.19%	0.0070
				333	0.63%		
1	1			000	0.00 /0		

Appendix 11 – participants' Data Charts of Frequency of Nodes and References



Participant Charts of Frequency of Nodes & ReferenceParticipant Charts of Frequency of Nodes & Reference(17-24 of highest frequency range)(25-32 of highest frequency range)





Appendix 13 - Interview semi-structured approach – Outline of Questions

Draf	t Questions for 1-2-1	<u>I Interviews</u> for PhD Research into 'Measuring Impact o	f
<u>Cligi</u>			I
_		They leafer we write a	2
1	Opening	> Thanks for your time > Have you read and signed the consent form?	2 mins
		> Can I confirm that we have the full 45 mins for the interview?	
2	Role	Your current role	3 mins
3	Personal	Your personal commitment to SDG's	5 mins
	commitment to		
	SDGs		
4	Company	Your company's current approach to measuring SDG impact at	5 mins
-	commitment to	project level:	
	SDGs & level of	> Do you have a SDG lead? Is that the same role/person as your	
	moturity	CSR lead?	
	maturity	> Which category are you in, as regards measuring projects' SDG	
		Impact?:	
		> Aware and not doing it	
		 > Unaware and not doing it 	
		> If you are aware and dong it, what maturity level are you at	
		(using P3M3 level descriptors)?	
		> Level 1 - Awareness of Process - informal, no standard	
		process	
		> Level 2 - Repeatable Process - limited consistency and	
		coordination	
		> Level 3 - Defined Process - centrally controlled processes	
		VITICIT Call De Callored > Level 4 - Managed Process - management metrics canability	
		assessment & quality management	
		> Level 5 - Optimized Process - continuous process	
		improvement & optimised processes	
5	Future SDG	SWOT discussion on the challenges and opportunities of doing	25 mins
	Impact	this better.	
	Development		
5.1		Strength: of current SDG impact measurement on engineering	
0.1		projects?	
5.2		Weakness: of current SDG impact measurement on engineering	
		projects?	
5.3		Opportunities for future SDG impact measurement on	
		engineering projects?	
5.4		projects?	
6	GEC engagement	Are you planning to attend the Global Engineering Congress from 22-26 Oct 182	2 mins
7	Closing	> Any final points that we haven't covered that you feel are	3 mins
/	CIOSING	important to this subject area?	
		> Thanks for your time and I will ensure that the copy of the ICE	
		report and the academic papers are passed to you personally	

ON SOUTH BANK UCI **Participation Information Form** Project title: 'Measuring Impact of Engineering Projects to achieve the Sustainable Development Goals' Lead researcher: Paul Mansell Contact Details: Email: paulrmansell@gmail.com Mobile number: 07795302258 Dear participant, Please read through the following information sheet before deciding to be a participant in this study. The investigation is completely voluntary giving you the option to choose whether or not you wish to participate. Aims of research: The purpose of this study is to provide a more in-depth understanding of UN Sustainable Development Goals, specifically SDG 13 on climate change, and why there is a gap between the high-level SDG political rhetoric and what is actually happening on engineering projects to measure their SDG impact against the 2030 targets. The research will be closely aligned with the engineering community to develop tools and approaches that can enable the SDG impacts to be measured effectively and thus to ensure maximum SDG value from the critical infrastructure's

Appendix 14 - Participant Information Form

What you will be required to do as a participant of this study:

investment beyond the traditional success definition of time, cost and quality.

During the study you will be asked to engage in a semi structured interview with the PhD research investigator, Paul Mansell. The interview will last approximately 45 minutes. As a participant you will be questioned about: your current role; your personal commitment to SDG's; your company's current approach to measuring SDG impact at project level; and a discussion on the challenges and opportunities of doing this better.

Pre Interview all participants will receive a copy of the questions to become familiar with some of the themes that will be discussed during the interview.

1

What type of participants am I seeking?

Having conducted a survey jointly with the ICE with over 300 respondents, I am now seeking the opportunity to engage with CEOs and their SDG/CSR leads. Ideally this would enable two interviews with each organisation, since the views of the leadership will be critical to gauge the viability of responding to this issue effectively.

What are the risks of participating in this study?

All the information given in the interview will <u>remain_anonymous</u>, unless specific quotes are to be published in which case the research lead and ICE would ensure that clearance for the quotes were formally obtained prior to inclusion in the publication.

What are the benefits to the participant involved in the study?

The results will provide insight into how the engineering firms can engage more effectively with SDG, specifically finding ways to measure SDG impact on their projects. There will be one 'glossy' report produced by the ICE in August, and one academic paper in September. Both will be inputs to the Global Engineering Congress from 22-26 October.

What will happen to the information collected?

All the information will be held securely and will only be accessible to the researcher. The ICE and University will hold the information for up to seven years, after this the data will be destroyed.

What's next?

If you have any questions regarding the study, please do not hesitate to contact me. If you would like to take part in this investigation you will need to fill out a 'Participation consent form' which will be provided by the research investigator.

Many thanks,

Paul Mansell – PhD Student sponsored by the Nathu Purri Institute

Email: paulrmansell@gmail.com Mobile number: 07795302258

Supervisor: Professor Rao Bhamidimarri, School of Engineering, Nathu Purri Institute; 07548 030346

2

Full matrix analysis of CEEQUAL's Categories and 246 questions' relative focus on TBL and across the 17 SDGs. Each shaded square represents a linkage.







Appendix 16 – Anglian Water's mapping of SDG targets to the 27 Projects.

Anglian	Water Priority SD Goals and Targets	WRE	Wisbech 2020	Dron CO2	Every Drop	Counts - Shop UEA Centre for Motor Studion	thc - 1	Building a	tbc - 2	Community Education	Catchment Management	Business in the Community	Biodiversity	Natural Capital	Keep it Clear	Water Efficiency	WaterAid	WaterAid Nepal	MK: Smart	Waterparks	#h2gr0wth	Green Construction	PR19 Consult	Safe & Well	LIFE	Pollution Watch	Greater Peterborough	The College WestAnglia	Totals
SMART	MAMI INITIES								-																				
Similari	SDG 6: Clean Water & Sanitation																												9
	6.3 - Improve water quality by reducing pollution halving the																												
	proportion of untreated wastewater and substantially										1				1														2
	6.4 - Increase water-use efficiency across all sectors and		1	-			-	-																					
	substantially reduce numbers of people suffering from water	1														1													2
	scarcity.																												
	6.5 - Implement integrated water resource management at all	1					1																						2
	6.6 - Protect and restore water-related ecosystems.			_	_	_		_	_	_				1			1	1								_			3
	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,													_			_	-											
	6.b - Support and strengthen the participation of local																												0
	communities in improving water and sanitation management.		_	-		_	_		_	_																			-
	SDG 11: Sustainable Cities & Communities		-	-	_	_	_	_	-																				9
	11.3 - Enhance sustainable urbanisation and capacity for sustainable human settlement planning and management		1		1	1													1										3
	11.4 - Strengthen efforts to protect and safeguard the world's			1					_		1	1																	1
	decrease the direct economic losses caused by disasters.						1	1																					2
	11.7 - Provide access to safe, inclusive and accessible, green																			1									1
	11.a - Support positive economic, social and environmental																				1								1
	strengthening national and regional development planning																												
	11.b - Increase the number of settlements adopting and																					1							
	implementing integrated policies and plans towards resource																												1
	efficiency, mitigation and adaptation to climate change,																				-								-
	resilience to disasters.			-	_	_	_	_	-													1							-
	SDG 12: Responsible Consumption & production			-					-																				
	of natural resources.				1	1													1										2
	12.4 - Achieve the environmentally sound management of											1																	
	chemicals and all wastes throughout their lifecycle.										1												_						1
	12.5 - Substantially reduce waste generation through																					1							1
	prevention, reduction, recycling and reuse.			-	_	_	-	_	-	_																			
	and to integrate sustainability information into their reporting cycles.											1																	1
	12.8 - Ensure that people everywhere have the relevant information and awareness for sustainable development and lifestyles in harmony with nature.																						1						1
SMART BL	siness		_	_	_		_		_	_																			-
	2.4 - Reduce by one third promoture mortality from pon-		-	-					-			<u> </u>	1																- 5
	communicable diseases and promote mental halvah nd well- being.											1																	1
	3.6 - Haive the number go global deaths and injuries from road traffic accidents																							1	1				2
	3.9 - Substantially reduce the number of deaths and illnesses																												
	from hazardous chemicals and air, water and soil pollution										1															1			2
	SDG 4: Quality Education			-			-	-																			1		3
	4.4 - Substantially increase the number of youth and adults																												1
	who have relevant skills, including technical and vocational skills, for employment.																										1	1	2
	4.7 - Ensure that all learners acquire knowledge and skills																												1
	needed to promote sustainable development, including,									1																			1
	among others, through education for sustainable development																												
	SDG 8: Decent Work & Economic Growth																												8
]							
	8.3 - Promote development-oriented policies that support																												
	productive activities, decent job creation, entrepreneurship, creativity and innovation, and encourage formalization and	1																			1								2
	growth of micro-,small- and medium-sized enterprises,																												
	including through access to financial services.		-		_	_		_	-	-												-				-			
	8.4 - Improve resource efficiency in consumption and production and endeavour to decouple economic growth from		1																		1								2
	environmental degradation.																												_
	8.5 - By 2030, achieve full and productive employment and																												
	decent work fro all women and men, including for young		1							1																			2
	equal value																												
	8.6 - By 2020, substantially reduce the proportion of youths not		1	E						1																			2
	in employment, education or training.	-	-	_		_	-	-	-		-															-			+



Table 40: Anglian Water's mapping of SDG targets to the 27 Projects. The y-axis shows the 35 SDG targets selected at Anglian Water corporate level; the x-axis shows the 27 projects that they are allocated to by Anglian Water (AW).

The representation, shown in Table 40, of mapping Anglian Water's top 27 projects to their prioritised SDG targets shows that all projects had at least one target to measure success against, while one project had 10 targets to map success against. This mapping by Anglian Water highlights that only a few targets can realistically be measured at project level. It also suggests that, if the targets are measured across a portfolio of projects and programmes, then a composite SDG impact measurement could be made. This would provide useful insights to support investment appraisals that seek to better understand the strategic impacts of investments and their broader TBL's return on investment.



<u>....</u>

-

.

<u>. . .</u>

-------- -. SDG (DG 1

.

*

•

* * * * * * * * * * *

Anglian Water's Prioritised SDGs - Frequency against CEEQUAL Category Themes

SDG 9 DG 13

DG 1

DG 1:

SDG

SDG 4 SDG 8

DG 14

Appendix 17 – MISI Impact Statement from Government & Industry Partners

(Note: all the letters of support for MISI Impact have been formally sanctioned to be included in this thesis by the named individuals)

The MISI research outputs, designed and led by the doctoral researcher, have been taken forward by the government and industry partners, specifically the Environment Agency and Thames Tideway Project, working together to establish this new approach for measuring sustainability on infrastructure projects.

Impact at the Environment Agency. "Following on from the MISI Project, we will be able to embed the knowledge that was generated by the project to support the measurement of SDG performance across our new portfolio of Environment Agency projects to be launched on 1st April 2021, which will total £5.2billion and include around 15 major projects in excess of £50million. Moreover, the approach to SDG measurement developed in the MISI Project will directly help us deliver our new sustainability strategy (known as eMissiion2030) and this will have huge positive impact that directly contributes to: saving lives; protecting hospitals, schools and homes; and regulating environmental impacts; as well as providing value for money for the UK Government". Quote from the Deputy Director Allocation & National Programme Management, Environment Agency (See Appendix 18).

Impact at Tideway. The Thames Tideway Tunnel is one of the largest infrastructure projects being delivered in Europe with a budget of £4.9billion and with a primary purpose to reduce sewage overflows into the River Thames.

"The development of the processes, tools and insights from the MISI Project can directly be utilised across the Thames Tideway Tunnel to ensure we are adequately capturing SDG performance for this major infrastructure project".

"Tideway aligned the financing of the project to the company's sustainability commitments and issued £1.8 billion of sustainable debt so far. We developed a sustainable finance communication strategy to attract and retain investors and aligned the reporting with the relevant SDGs. To this end, the work we have done on the MISI Project helped enhance the way we communicate our work on the SDGs, both in our Annual Report and in our Sustainable Finance Report. This has been well received by various stakeholders". Quotes from the Group Treasurer of Tideway (See Appendix 19).

Impact with wider stakeholders. Other quotes about the research impact were:

"The outputs of the MISI Project, which builds on earlier research from LSBU, has made it clear to us that significant progress on measuring the SDG performance of infrastructure projects has been secured". Quote from the Executive Director, UN Global Compact Network UK (See Appendix 20).

"At the IPA we recognised at an early stage that the MISI Project had significant potential to help all projects with their contribution to the SDGs". Quote from the Director Project Delivery, **Infrastructure and Projects Authority** (See Appendix 21).

"The outputs of the MISI Project, which builds on earlier research from LSBU, has made it clear to us and our community of ca. 100,000 members across 150 plus countries that major progress on measuring the SDG performance of infrastructure projects has been made in a short timeframe". Quote from the Director General and Secretary, **Institution of Civil Engineers** (See Appendix 22).

The impact of the SDG focused research on infrastructure projects at LSBU has also been recognised in an article in "*The Source - Magazine of the International Water Association*" (Hayward, 2019) and more recently in a publication by BRE on the BREEAM scheme, which is a key sustainability assessment tool used in the construction and built environment sector (**Building Research Establishment**, 2020) (See Appendix 23).

Appendix 18 – MISI Impact Statement of this research from the Environment Agency.

Letter of support from the Deputy Director Allocation & National Programme Management, Environment Agency, 24th September 2020 (available upon request):



The progress of the MISI Project and underpinning research by LSBU and partners has been rapid and has delivered real value for us and will likely be greatly valued by the wider sector. For example, we have been working closely with the LSBU lead researcher, Paul Mansell, and the UK Cabinet Office and HM Treasury's Infrastructure Project Authority to develop our learning in a way that we can roll-out the research more widely across UK's full infrastructure portfolio. It is excellent to see the team's hard work and dedication to translate the findings from the doctoral research at LSBU into knowledge and benefits arising from the MISI Project.

Yours sincerely

JMRWSSM

John Russon Deputy Director Allocation & National Programme Management Mobile: 07741 007218 john.russon@environment-agency.gov.uk

Appendix 19 – MISI Impact Statement of this research from the Thames tideway project.

Letter of support from the Group Treasurer of Tideway, 27th October 2020:



Tideway

We have aligned our commitments to the UN Sustainable Development Goals (SDGs), identifying the main SDGs to which Tideway makes a direct contribution to. Tideway will make a long term direct contribution to SDG 6 Clean Water and SDG 11 Sustainable Cities. During construction, Tideway is making a significant contribution to eight other SDGs – some of these will have a lasting impact and/or will be handed over to other organisations.

It is evident to me that major progress has been made by the MISI Project that builds on the earlier research conducted by LSBU. As Tideway's project was over 50% complete at the time of engaging with the MISI Project, the analysis undertaken by Tideway's Sustainability and Treasury teams focused on mapping Tideway's experience to MISI's methodology and focused on matters that could be taken forward by Tideway. The development of the processes, tools and insights from the MISI Project can directly be utilised across the Thames Tideway Tunnel to ensure we are adequately capturing SDG performance for this major infrastructure project. Further to developing the Impact Value Change analysis for Tideway, we extended the mapping of our legacy commitments to the SDGs at target level and developed further our SDG narrative.

Tideway aligned the financing of the project to the company's sustainability commitments and issued £1.8 billion of sustainable debt so far. We developed a sustainable finance communication strategy to attract and retain investors and aligned the reporting with the relevant SDGs. To this end, the work we have done on the MISI Project helped enhance the way we communicate our work on the SDGs, both in our Annual Report and in our Sustainable Finance Report. This has been well received by various stakeholders.

Following on from the MISI Project, we have been able to embed the learning and knowledge from the project and original research from LSBU in order to improve our understanding of how we can effectively measure SDG performance across the Tideway project. As a result, we have continued discussions with the Environment Agency and the Infrastructure and Projects Authority, sharing our experience and legacy work. The MISI Project and the underpinning research from LSBU represents a leap forward in how sustainability can be measured across infrastructure mega projects such as the Thames Tideway Tunnel.

Finally, I would like to pass on our thanks for the diligence and dedication by Paul Mansell for his doctoral research studies at LSBU and for delivering the MISI Project earlier in 2020.

Yours sincerely

Ines Faden da Silva Group Treasurer Bazalgette Tunnel Ltd, trading as Tideway

2

Appendix 20 – MISI Impact Statement of this research from the UN Global Compact Network UK.

Letter of support from the Executive Director, UN Global Compact Network UK, 2nd September 2020:



Appendix 21 – MISI Impact Statement of this research from the Infrastructure and Projects Authority, UK Government.

Letter of support from the Director Project Delivery, Infrastructure and Projects Authority, UK Government, 15th October 2020:



Appendix 22 – MISI Impact Statement of this research from the Institution of Civil Engineers.

Letter of support from the Director General and Secretary, Institution of Civil Engineers, 1st September 2020:



Appendix 23 – MISI Impact Statement of this research from the Building Research Establishment (BRE).

Letter of support from the Strategic Relationship Lead BRE Global and Director CEEQUAL, Building Research Establishment, 19th October 2020:

	BRE Global Bucknalls Lane Waftord, Herts WD25 9XX T: +44 (0)333 321 8811 E: <u>BREGlobalEnquiries@bregroup.com</u> W: <u>www.bregroup.com/breglobal</u> Dang G. Gingge D. Dbillyin	bre
	Director Nathu Puri Institute for Engineering and Enterprise London South Bank University 103 Borough Road London SE1 0AA	
–	Date: 19 October 2020 Our Ref.: Your Ref.:	
	Dear Prof. Philbin,	
	Impact of the MISI (Measurement of Infrastructure SDG Impact) Project	
	This letter is to provide feedback on the MISI (Measurement of Infrastructure SDG Impact has been undertaken by London South Bank University (LSBU) and collaborative partners BRE, Institution of Civil Engineers, University College London, and various industry partne Environment AgencyThe BRE Trust provided some funding to this project to enable LSB student, Paul Mansell, to extend his research to cover a specific review of the SDGs again CEEQUAL sustainability rating and certification scheme for infrastructure.) Project, which s, including the ers including the U's PhD hst our
	This review was important because we have been increasingly aware, as the construction required to measure their performance with the SDG goals, of the need to ensure our fam sustainability rating schemes (CEEQUAL, BREEAM and Home Quality Mark) are aligned framework of SDG goals and indicators.	sector is being ily of to the
	In order to adopt the goals in the construction industry, direct and indirect solutions must be Understanding a layered approach to tackling the SDGs is key to understanding the role of business can carry out in order to play their part. As the SDGs are country level goals, dire the specific targets and indicators can be difficult. Understanding the indirect relationship, business has with the goals can be just as important in facilitating the national or global ad the SDGs.	be considered. each individual ect correlation to a commercial chievement of
	Paul, along with researchers at BRE, University College London and London South Bank able to identify a link between best-practice sustainability-reporting frameworks at the proj those at the organisational level (such as GRI). This enabled the alignment of such frame SDG's allowing SDG impact targets to be embedded into the design and construction of a project. The work carried out has been published by the Institute of Civil Engineers: 'Asse of infrastructure projects on global sustainable development goals.' Mansell, P., et. al. 201 Proceedings of the Institution of Civil Engineers-Engineering Sustainability (Vol. 173, No. Thomas Telford Ltd.	University, was ect level and works to the in infrastructure ssing the impact (9, November. In 4, pp. 196-212). 2590X with VAT Registration
	Number GB 689 9499 27 BRE Global is a trading name of the Company, using brands LPCB, BREEAM, BRE Academy, CEEQUAL and BRE Global Certification TP024 Rev 1.0 © BRE Global Limited 2019	
Date: 19 October 2020 BRE Global Ref.:

In addition since the completion of this research, BRE have adopted the approach Paul used on CEEQUAL to analyse the whole family of BREEAM sustainability schemes and have now published a mapping document, to help industry available here https://www.breeam.com/sdg/

Yours sincerely

lan Nicholson

Ian Nicholson Strategic Relationship Lead - Infrastructure For and on behalf of BRE Global Limited Telephone: +44 (0)1923 664409 E-mail: <u>ian.nicholson@bregroup.com</u>

Page 2 of 2

The LSBU Research Degree Code of Practice (2018), states the thesis should contain: abstract; statement of research objectives; references; info on any publications produced as part of the research, "either included as part of the thesis or placed in a pocket at the end of the thesis" (Research Degree Code of Practice, p.24). The list of 12 publications below includes the current list of publications (incl 1 in review process).

Journal Papers

[1]. Mansell, P., Philbin, S. P., Broyd, T., & Nicholson, I. (2020). Assessing the impact of infrastructure projects on global sustainable development goals. *Proceedings of the Institution of Civil Engineers–Engineering Sustainability*, 173(4), 196-212, https://doi.org/10.1680/jensu.19.00044.

[2]. Mansell, P., Philbin, S. P., & Konstantinou, E. (2020). Redefining the Use of Sustainable Development Goals at the Organisation and Project Levels—A Survey of Engineers. *Administrative Sciences*, 10(3), 55, <u>https://doi.org/10.3390/admsci10030055</u>.

[3]. Mansell, P., Philbin, S. P., & Konstantinou, E. (2020). Delivering UN Sustainable Development Goals' Impact on Infrastructure Projects: An Empirical Study of Senior Executives in the UK Construction Sector. *Sustainability*, *12*(19), 7998, https://doi.org/10.3390/su12197998.

[4]. Mansell, P., Philbin, S. P., & Broyd, T. (2020). Development of a New Business Model to Measure Organizational and Project-Level SDG Impact—Case Study of a Water Utility Company. *Sustainability*, *12*(16), 6413, <u>https://doi.org/10.3390/su12166413</u>

[5]. Mansell, P., Philbin, S.P. (2020). Measuring Sustainable Development Goal Targets on Infrastructure Projects. *Journal of Modern Project Management*: <u>https://openresearch.lsbu.ac.uk/item/8q171</u>

[6]. Mansell, PM, Philbin, SP, Broyd, T, Nicholson - 'Measuring Infrastructure Projects' Impact on UN SDG Global Goals: Development of an SDG Impact-Value Chain for the Infrastructure Sector based on the Triple Bottom Line', *International Journal of Sustainability Society (IJSSOC)*. Submission code: IJSSOC-291855. Paper Accepted.

Paper Under Review

[7]. Mansell, P.M., Philbin, S.P., Van Rooyen, D., Sabini, L. (2020c). 'A Systematic Literature Review of Infrastructure Projects - Assessment Through SDG Targets: Towards a Comprehensive Framework', *Engineering Project Organization Journal* invited for inclusion post conference; in review process.

Conference Papers (Double Blind Reviewed) – published in proceedings

[8]. Mansell, P. Quantitative Survey Analysis: What Engineers and CEOs Currently Think about Sustainability and the SDGs. (2018). ICE Global Engineering Conference (reviewed by ICE editorial publishing team and the GEC Council prior to acceptance) https://www.researchgate.net/publication/327602328_Engineers_perception_of_valu e_of_SDGs_and_the_current_ability_to_measure_projects'_SDG_impact

[9]. Mansell, P., Philbin, S.P., Plodowski, A. (2019) Why Project Management is Critical to Achieving the SDGs, and How This Can be Achieved. Project Management Conference 2019, Delft University. <u>https://pmcongress2019.org/</u>

[10]. Mansell, P., Philbin, S.P., Konstantinou, E. (2019) 'Call to Arms': Using the Creating Shared Value Business Governance Paradigm to Deliver Projects' Business-Society Impact Against the UN SDG 2030 Targets. Proceedings of the EURAM (European Academy of Management) 2019 Conference, Lisbon, Portugal. http://www.euram-online.org/annual-conference-2019.html

[11]. Mansell, P., Philbin, S.P., and Boyd, T. (2019). Infrastructure Projects' Impact on Sustainable Development. In the Proceedings of 7th IPMA Research Conference and the 14th International OTMC Conference. 4-7 September 2019. Zagreb, Croatia, pp 570-592.

[12]. Mansell, P.M., Philbin, S.P., Van Rooyen, D., Sabini, L. (2020c). 'A Systematic Literature Review of Infrastructure Projects - Assessment Through SDG Targets: Towards a Comprehensive Framework', Paper accepted and presented to EPOC (Engineering Project Organization Society) on 2 June 2020.